



Monte Carlo Based Techniques for Quantum Magnets with Long-Range Interactions

Patrick Adelhardt [†], Jan A. Koziol [†], Anja Langheld [†] and Kai P. Schmidt ^{*}

Department of Physics, Friedrich-Alexander Universität Erlangen-Nürnberg (FAU), 91058 Erlangen, Germany; patrick.adelhardt@fau.de (P.A.); jan.koziol@fau.de (J.A.K.); anja.langheld@fau.de (A.L.)

* Correspondence: kai.phillip.schmidt@fau.de ⁺ These authors contributed equally to this work.

Abstract: Long-range interactions are relevant for a large variety of quantum systems in quantum optics and condensed matter physics. In particular, the control of quantum-optical platforms promises to gain deep insights into quantum-critical properties induced by the long-range nature of interactions. From a theoretical perspective, long-range interactions are notoriously complicated to treat. Here, we give an overview of recent advancements to investigate quantum magnets with long-range interactions focusing on two techniques based on Monte Carlo integration. First, the method of perturbative continuous unitary transformations where classical Monte Carlo integration is applied within the embedding scheme of white graphs. This linked-cluster expansion allows extracting highorder series expansions of energies and observables in the thermodynamic limit. Second, stochastic series expansion quantum Monte Carlo integration enables calculations on large finite systems. Finitesize scaling can then be used to determine the physical properties of the infinite system. In recent years, both techniques have been applied successfully to one- and two-dimensional quantum magnets involving long-range Ising, XY, and Heisenberg interactions on various bipartite and non-bipartite lattices. Here, we summarise the obtained quantum-critical properties including critical exponents for all these systems in a coherent way. Further, we review how long-range interactions are used to study quantum phase transitions above the upper critical dimension and the scaling techniques to extract these quantum critical properties from the numerical calculations.

Keywords: quantum spin systems; long-range interactions; Ising interactions; XY interactions; Heisenberg interactions; Monte Carlo; series expansion; perturbative continuous unitary transformation; stochastic series expansion; quantum phase transitions; critical exponents; quantum simulation

1. Introduction

Since the advent of the theoretical description of classical and quantum phase transitions (QPTs), long-range interactions between degrees of freedom challenged the established concepts and propelled the development of new ideas in the field [1–5]. It is remarkable that, only a few years after the introduction of the renormalisation group (RG) theory by K.G. Wilson in 1971 as a tool to study phase transitions and as an explanation for universality classes [6–11], it was used to investigate ordering phase transitions with long-range interactions. These studies found that the criticality depends on the decay strength of the interaction [1–3]. It then took two decades to develop numerical Monte Carlo (MC) tools capable of simulating basic magnetic long-range models with thermal phase transitions following the behaviour predicted by the RG theory [12,13]. The results of these simulations sparked a renewed interest in finite-size scaling above the upper critical dimension [12,14–19] since "hyperscaling is violated" [13] for long-range interactions that decay slowly enough. In this regime, the treatment of dangerous irrelevant variables (DIVs) in the scaling forms is required to extract critical exponents from finite systems.

Meanwhile, a similar historic development took place regarding the study of QPTs under the influence of long-range interactions. By virtue of pioneering RG studies [20,21],



Citation: Adelhardt , P.; Koziol, J.A.; Langheld, A.; Schmidt, K.P. Monte Carlo Based Techniques for Quantum Magnets with Long-Range Interactions. *Entropy* **2024**, *26*, 401. https://doi.org/10.3390/e26050401

Academic Editors: Bertrand Berche and Yurij Holovatch

Received: 5 March 2024 Revised: 17 April 2024 Accepted: 19 April 2024 Published: 1 May 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the numerical investigation of long-range interacting magnetic systems has been triggered [22–38]. In particular, Monte Carlo-based techniques became a popular tool to gain quantitative insight into these long-range interacting quantum magnets [22,25,26,29–40]. On the one hand, this includes high-order series expansion techniques, where classical Monte Carlo integration is applied for the graph embedding scheme, allowing extracting energies and observables in the thermodynamic limit [25,29]. On the other hand, there is stochastic series expansion quantum Monte Carlo [39], which enables calculations on large finite systems. To determine the physical properties of the infinite system, finite-size scaling is performed with the results of these computations. Inspired by the recent developments for classical phase transitions [15–19,41], a theory for finite-size scaling above the upper critical dimension for QPTs was introduced [32,34].

When investigating algebraically decaying long-range interactions $\sim r^{-(d+\sigma)}$ with the distance r and the dimension d of the system, there are two distinct regimes: one for $\sigma \leq 0$ (strong long-range interaction) and another one for $\sigma > 0$ (weak long-range interaction) [5,42–45]. In the case of strong long-range interactions, common definitions of internal energy and entropy in the thermodynamic limit are not applicable and standard thermodynamics breaks down [5,42–45]. We will not focus on this regime in this review. For details specific to strong long-range interactions, we refer to other review articles such as Refs. [5,42–45]. For the sake of this work, we restrict the discussion to weak long-range interaction or competing antiferromagnetic strong long-range interactions, for which an extensive ground-state energy can be defined without rescaling of the coupling constant [5].

The interest in quantum magnets with long-range interactions is further fuelled by the relevance of these models in state-of-the-art quantum–optical platforms [5,46–87]. To realise long-range interacting quantum lattice models with a tunable algebraic decay exponent, one can use trapped ions, which are coupled off-resonantly to motional degrees of freedom [5,81–85,88]. Another possibility is to couple trapped neutral atoms to photonic modes of a cavity [5,86,87]. Alternatively, one can realise long-range interactions decaying with a fixed algebraic decay exponent of six or three using Rydberg atom quantum simulators [46–55] or ultracold dipolar quantum atomic or molecular gases in optical lattices [56–73]. Note that, in many of the above-listed cases, it is possible to map the long-range interacting atomic degrees of freedom onto quantum spin models [5,52,89]. Therefore, they can be exploited as analogue quantum simulators for long-range interacting quantum magnets, and the relevance of the theoretical concepts transcends the boundary between the fields.

From the perspective of condensed matter physics, there are multiple materials with relevant long-range interactions [90–106]. The compound LiHoF₄ in an external field realises an Ising magnet in a transverse magnetic field [102–105]. A recent experiment with the two-dimensional Heisenberg ferromagnet Fe₃GeTe₂ demonstrates that phase transitions and continuous symmetry breaking can be implemented by circumventing the Hohenberg–Mermin–Wagner theorem with long-range interactions [106]. This material is in the recently discovered material class of 2D magnetic van der Waals systems [107,108]. Further, dipolar interactions play a crucial role in the spin ice state in the frustrated magnetic pyrochlore materials Ho₂Ti₂O₇ and Dy₂Ti₂O₇ [90–101].

In this review, we are interested in physical systems described by quantum spin models, where the magnetic degrees of freedom are located on the sites of a lattice. We concentrate on the following three paradigmatic types of magnetic interactions between lattice sites: first, Ising interactions, where the magnetic interaction is oriented only in the direction of one quantisation axis; second, XY interactions with a U(1)-symmetric magnetic interaction invariant under planar rotations; and third, Heisenberg interactions with a SU(2)-symmetric magnetic interaction invariant under rotations in 3D spin space. In the microscopic models of interest, a competition between magnetic ordering and trivial product states, external fields, or quasi-long-range order leads to QPTs.

In this context, the primary research pursuit revolves around how the properties of the QPT depend on the long-range interaction. The upper critical dimension of a QPT in

magnetic models with non-competing algebraically decaying long-range interactions is known to depend on the decay exponent of the interaction for a small enough exponent, and decreases as the decay exponent decreases [20,21]. If the dimension of a system is equal to or exceeds the upper critical dimension, the QPT displays mean-field critical behaviour. At the same time, standard finite-size scaling, as well as standard hyperscaling relations are no longer applicable. Therefore, these systems are primary workhorse models to study finite-size scaling above the upper critical dimension. In this case, the numerical simulation of these systems is crucial in order to gauge novel theoretical developments. Further, QPTs in systems with competing long-range interactions do not tend to depend on the long-range nature of the interaction [23–26,29,30,32]. In several cases, long-range interactions then lead to the emergence of ground states and QPTs, which are not present in the corresponding short-range interacting models [27,30,54,55,109–112].

In this review, we are mainly interested in the description and discussion of two Monte Carlo-based numerical techniques, which were successfully used to study the lowenergy physics of long-range interacting quantum magnets, in particular with respect to the quantitative investigation of QPTs [22,25,29–32,34–38,40]. The success of Monte Carlo techniques in this field is due to the occurrence of high-dimensional sums and integrals that commonly arise in the formulation of many-particle statistics. In contrast to many deterministic integration techniques, for which the standard error scales exponentially with the dimension of the underlying integral, the standard error of an integral calculated with Monte Carlo integration does not scale with the dimension of the underlying integral. We further chose to review this topic due to our personal involvement with the application and development of these methods [25,29–32,34,35]. On the one hand, we explain in detail how classical Monte Carlo integration can enhance the capabilities of linked-cluster expansions (LCEs) with the pCUT+MC approach (a combination of the perturbative unitary transform approach (pCUT) and MC embedding). On the other hand, we describe how stochastic series expansion (SSE) quantum Monte Carlo (QMC) integration is used to directly sample the thermodynamic properties of suitable long-range quantum magnets on finite systems.

This review is structured as follows. In Section 2, we review the basic concept of a QPT in a condensed way, focusing on the details relevant for this review. We define the quantum-critical exponents and the relations between them in Section 2.1. Here, we also have the first encounter with the generalised hyperscaling relation, which is also valid above the upper critical dimension where conventional hyperscaling breaks down. As the SSE QMC method discussed in this review is a finite-system simulation, we discuss the conventional finite-size scaling below the upper critical dimension in Section 2.2 and the peculiarities of finite-size scaling above the upper critical dimension in Section 2.3. In Section 3, we summarise the basic concepts of Markov chain Monte Carlo integration: Monte Carlo sampling, Markovian random walks, stationary distributions, the detailed balance condition, and the Metropolis-Hastings algorithm. We continue by introducing the series-expansion Monte Carlo embedding method pCUT+MC in Section 4. We start with the basic concepts of a graph expansion in Section 4.1 and introduce the perturbative method of our choice, the perturbative continuous unitary transformation method, in Section 4.2. We introduce the theoretical concepts for setting up a linked-cluster expansion as a full graph decomposition in Section 4.3 and, subsequently, discuss how to practically calculate perturbative contributions in Sections 4.4 and 4.5. We prepare the discussion of the white graph decomposition in Section 4.6 with an interlude on the relevant graph theory in Sections 4.6.1 and 4.6.2 and the important concept of white graphs in Section 4.6.3. Further, in Section 4.7, we discuss the embedding problem for the white graph contributions. Starting from the nearest-neighbour embedding problem in Section 4.7.1, we generalise it to the long-range case in Section 4.7.2 and then introduce a classical Monte Carlo algorithm to calculate the resulting high-dimensional sums in Section 4.7.3. This is followed by some technical aspects on series extrapolations in Section 4.8 and a summary of the entire workflow in Section 4.9. In the next section, the topic changes towards the review of the SSE QMC method, which is an approach to simulate thermodynamic properties of suitable

quantum many-body systems on finite systems at a finite temperature. First, we discuss the general concepts of the method in Section 5. We review the algorithm to simulate arbitrary transverse-field Ising models introduced by A. Sandvik [39] in Section 5.1. We then review an algorithm used to simulate non-frustrated Heisenberg models in Section 5.2. After the introduction to the algorithms, we summarise techniques on how to measure common observables in the SSE QMC scheme in Section 5.3. Since the SSE QMC method is a finite-temperature method, we discuss how to rigorously use this scheme to perform simulations at effective zero temperature in Section 5.4. We conclude this section with a brief summary of path integral Monte Carlo techniques used for systems with long-range interactions (see Section 5.5). To maintain the balance between algorithmic aspects and their physical relevance, we summarise several theoretical and numerical results for quantum phase transitions in basics long-range interacting quantum spin models, for which the discussed Monte Carlo-based techniques provided significant results. First, we discuss longrange interacting transverse-field Ising models in Section 6. For ferromagnetic interactions, this model displays three regimes of universality: a long-range mean-field regime for slowly decaying long-range interactions, an intermediate long-range non-trivial regime, and a regime of short-range universality for strong decaying long-range interactions. We discuss the theoretical origins of this behaviour in Section 6.1.1 and numerical results for quantum critical exponents in Section 6.1.2. Since this model is a prime example to study scaling above the upper critical dimension in the long-range mean-field regime, we emphasise these aspects in Section 6.1.3. Further, we discuss the antiferromagnetic long-range transverse-field Ising model on bipartite lattices in Section 6.2 and on nonbipartite lattices in Section 6.3. The next obvious step is to change the symmetry of the magnetic interactions. Therefore, we turn to long-range interacting XY models in Section 7 and Heisenberg models in Section 8. We discuss the long-range interacting transversefield XY chain in Section 7 starting with the U(1)-symmetric isotropic case in Section 7.1, followed by the anisotropic case for ferromagnetic (see Section 7.2) and antiferromagnetic (see Section 7.3) interactions, which display similar behaviour to the long-range transversefield Ising model on the chain discussed in Section 6. We conclude the discussion of the results with unfrustrated long-range Heisenberg models in Section 8. We focus on the staggered antiferromagnetic long-range Heisenberg square lattice bilayer model in Section 8.1 followed by long-range Heisenberg ladders in Section 8.2 and the long-range Heisenberg chain in Section 8.3. We conclude in Section 9 with a brief summary and with some comments on the next possible steps in the field.

2. Quantum Phase Transitions

This review is part of the Special Issue with the topic "Violations of Hyperscaling in Phase Transitions and Critical Phenomena". In this work, we summarise investigations of low-dimensional quantum magnets with long-range interactions targeting, in particular, quantum phase transitions (QPTs) above the upper critical dimension, where the naive hyperscaling relation is no longer applicable. In this section, we recapitulate the relevant aspects of QPTs needed to discuss the results of the Monte Carlo-based numerical approaches. First, we give a general introduction to QPTs. After that, we discuss in detail the definition of critical exponents and the relations among them in Section 2.1, as well as the scaling below (see Section 2.2) and above (see Section 2.3) the upper critical dimension.

Any non-analytic point of the ground-state energy of an infinite quantum system as a function of a tuning parameter λ is identified with a QPT [113]. This tuning parameter can, for instance, be a magnetic field or pressure, but not the temperature. Quantum phase transitions are a concept of zero temperature as there are no thermal fluctuations and all excited states are suppressed infinitely strong such that the system remains in its ground state. There are two scenarios for how a non-analytic point in the ground-state energy can emerge [113]: First is an actual (sharp) level crossing between the ground-state energy and another energy level. Second, the non-analytic point can be considered as a limiting case of an avoided level crossing. Historically, phase transitions are classified by the lowest order derivative of the free energy that is discontinuous [113,114]. Therefore, a first-order phase transition is discontinuous in the order parameter (first derivative) and a second-order phase transition is discontinuous in the response functions (second derivative). Since, in second-order phase transitions, the order parameter is still continuous across the phase transition, we use the term "continuous phase transition" as an equivalent for "second-order phase transition".

In this review, we are interested in second-order QPTs, which fall into the second scenario. At a second-order QPT, the relevant elementary excitations condense into a novel ground state, while the characteristic length and time scales diverge. Apart from topological QPTs involving a long-range entangled topological phase, such continuous transitions are described by the concept of spontaneous symmetry breaking. On one side of the QPT, the ground state obeys a symmetry of the Hamiltonian, while on the other side, this symmetry is broken in the ground state and a ground-state degeneracy arises.

Following the idea of the quantum-to-classical mapping [113,115], *d*-dimensional quantum systems can be mapped in the vicinity of a second-order QPT to models of statistical mechanics with a classical (thermal) second-order phase transition in d + 1 dimensions. In many cases, the models obtained from a quantum-to-classical mapping are rather artificial [113]. However, such mappings often allow categorising QPTs in terms of universality classes and associated critical exponents by the non-analytic behaviour of the classical counterparts [10,11,113,116,117]. The mapping further illustrates that the renormalisation group (RG) theory is also applicable to describe QPTs [10,11,113,116,117].

In the RG theory, each QPT belongs to a non-trivial fixed point of the RG transformation [10,11], whereas a trivial fixed point would, for instance, be a fully ordered state with maximal correlation or a fully disordered state with no correlation at all. Critical exponents are connected to the RG flow in the immediate vicinity of these non-trivial fixed points [10,11,113]. The concept of universality classes arises from the fact that different microscopic Hamiltonians can have a quantum critical point that is attracted by the same non-trivial fixed point under successive application of the RG transformation [10,11]. Due to this, the QPTs in these models have the same critical exponents.

Another remarkable result of the RG theory is the scaling of observables in the vicinity of phase transitions. Historically, the theory of scaling at phase transitions was heuristically introduced before the RG approach [118–124]. The latter provided the theoretical foundation for the scaling hypothesis [6,7]. The main statement of the scaling theory is that the non-analytic contributions to the free energy and correlation functions are mathematically described by generalised homogeneous functions (GHFs) [124]. A function with *n* variables $f(x_1, x_2, \ldots, x_n)$ is called a GHF, if there exist $a_1, a_2, \ldots, a_n \in \mathbb{R}$ with at least one being non-zero and $a_f \in \mathbb{R}$ such that, for $b \in \mathbb{R}_+$,

$$f(b^{a_1}x_1, b^{a_2}x_2, \dots, b^{a_n}x_n) = b^{a_f}f(x_1, x_2, \dots, x_n).$$
⁽¹⁾

The exponents $a_1, a_2, ..., a_n$ are the scaling powers of the variables, and a_f is the scaling power of the function f itself. An in-depth summary of the mathematical properties of GHFs can be found in Appendix B. The most important properties of GHFs are that their derivatives, Legendre transforms, and Fourier transforms are also GHFs. As we will outline in Section 2.2, the theory of finite-size scaling is formulated in terms of GHFs and relates the non-analytic behaviour at QPTs in the thermodynamic limit with the scaling of observables for different finite system sizes. In this, the variables x_i are related to physical parameters like the temperature T, control parameter λ , symmetry-breaking field H, and also irrelevant, more abstract parameters that parameterise the microscopic details of the model like the lattice spacing. Later in this section, we will define irrelevant variables in the context of the RG and GHFs.

Another aspect relevant for this work is that quantum fluctuations are the driving factor with QPTs [113]. In general, fluctuations are more important in low dimensions [117]. The universality class of QPTs for a certain symmetry breaking depends on the dimensionality of the system.

An important aspect regarding this review is the so-called upper critical dimension $d_{\rm uc}$. The upper critical dimension is defined as a dimensional boundary such that, for systems with dimension $d \ge d_{uc}$, the critical exponents are those obtained from mean-field considerations. The upper critical dimension is of particular importance for QPTs in systems with non-competing long-range interactions. For sufficiently small decay exponents of an algebraically decaying long-range interaction $1/r^{d+\sigma}$, the upper critical dimension starts to decrease as a function of the decay exponent σ [20,32,34]. In the limiting case of a completely flat decay ($\sigma = -d$) of the long-range interaction resulting in an all-to-all coupling, the model is intrinsically of the mean-field type and mean-field considerations become exact. For a certain value of the decay exponent, the upper critical dimension becomes equal to the fixed spatial dimension, and for decay exponents below this value, the dimension of the system is above the upper critical dimension of the transition [20,32,34]. This makes long-range interacting systems an ideal test bed for studying phase transitions above the upper critical dimension in low-dimensional systems. In particular, long-range interactions can make the upper critical dimension accessible in real-world experiments as the upper critical dimension of short-range models is usually not below three.

Although phase transitions above the upper critical dimension display mean-field criticality, they are still a matter worth studying, since naive scaling theory describing the behaviour of finite systems close to a phase transition (see Section 2.2) is no longer applicable [16,19,125]. Moreover, the naive versions of some relations between critical exponents, as discussed in Section 2.1, do not hold any longer [15,16]. The reason for this issue are the dangerous irrelevant variables (DIVs) in the RG framework [126–128]. During the application of the RG transformation, the original Hamiltonian is successively mapped to other Hamiltonians, which can have infinitely many couplings. All these couplings, in principle, enter the GHFs. In practice, all but a finite number of these couplings are set to zero since their scaling powers are negative, which means they flow to zero under renormalisation. These couplings are, therefore, called irrelevant. This approach of setting irrelevant couplings to zero can be used to derive the finite-size scaling behaviour as described in Section 2.2. However, above the upper critical dimension, this approach breaks down because it is only possible to set irrelevant variables to zero if the GHF does not have a singularity in this limit [126]. Above the upper critical dimension, such singularities in irrelevant parameters exist, which makes them DIVs [127]. We explain the effect of DIVs on scaling in Section 2.3.

2.1. Critical Exponents in the Thermodynamic Limit

As outlined above, a second-order QPT comes with a singularity in the free energy density. In fact, also, other observables experience singular behaviour at the critical point in the form of power-law singularities. For instance, the order parameter m as a function of the control parameter λ behaves as

$$m(\lambda \to \lambda_c^-) \sim |\lambda - \lambda_c|^{\beta}$$
 (2)

in the ordered phase. Without loss of generality, the system is taken to be in the ordered phase for $\lambda < \lambda_c$ and the notation $\lambda \rightarrow \lambda_c^-$ means that λ is approaching λ_c from below, i.e., it is approaching in the ordered phase. In the disordered phase $\lambda > \lambda_c$, the order parameter by definition vanishes such that $m(\lambda \rightarrow \lambda_c^+) = 0$. The observables with their respective power-law singular behaviour, which is characterised by the critical exponents α , β , γ , δ , η , ν , and z, are summarised in Table 1 together with how they are commonly defined in terms of the free energy density f, the symmetry-breaking field H, which couples to the order parameter, and the reduced control parameter $r = (\lambda - \lambda_c)/\lambda_c$.

Table 1. Definitions of critical exponents by means of the singularities of thermodynamic quantities for a magnetic phase transition. The free energy density is denoted by *f*. Note that the control parameter susceptibility associated with the critical exponent α coincides with the heat capacity only for thermal phase transitions, where $r = (T - T_c)/T_c$, while for QPTs, the meaning depends on the control parameter triggering the phase transition [129].

Observable	Definition	Crit. Exp.	Singularity
Characteristic length ξ	via $G(\vec{r})$	ν	$\xi(r ightarrow 0, H=0) \sim r ^{- u}$
Energy gap Δ	via $G(\vec{r}, \omega)$	~1.	$\Delta(r o 0) \sim r ^{z u}$
Charact. time scale ξ_{τ}	$\xi_ au \sim \Delta^{-1}$	20	$\xi_{ au}(r ightarrow 0)\sim \xi^{z}\sim r ^{-z u}$
Order parameter <i>m</i>	$m = \frac{\partial f}{\partial H}\big _{H=0}$	β	$m(r ightarrow0^-$, $H=0)\sim r ^eta$
		δ	$m(r=0,H ightarrow 0)\sim H ^{1/\delta}$
Order-parameter	$\chi = \frac{\partial m}{\partial H}\big _{H=0}$	γ	$\chi(r ightarrow 0, H = 0) \sim r ^{-\gamma}$
susceptibility χ			
Control-parameter	$\chi_r = rac{\partial^2 f}{\partial r^2}$	α	γ $(r \rightarrow 0, H = 0) \sim r ^{-\alpha}$
susceptibility χ_r			$\chi_r(r \to 0, \Pi = 0) \stackrel{\sim}{\to} r $
Correlation	$\frac{\partial \langle m(\vec{r}) \rangle}{\partial H_{\vec{r}=0}}\Big _{H_0=0}$	η	$G(\vec{r} \to \infty, r = 0, H = 0)$
function $G(\vec{r})$			$\sim rac{1}{ert ec r ert^{d-2+\eta}}$

One usually defines reduced parameters like r that vanish at the critical point not only to shorten the notation, but also to express the power-law singularities independent of the microscopic details of the specific model one is looking at. While the value of λ_c depends on these details, the power-law singularities are empirically known to not depend on the microscopic details, but only on more general properties like the dimensionality, the symmetry that is being broken, and, with particular emphasis due to the focus of this review, on the range of the interaction. It is, therefore, common to classify continuous phase transitions in terms of universality classes. These universality classes share the same set of critical exponents. In terms of the RG, this behaviour is understood as distinct critical points of microscopically different models flowing to the same renormalisation group fixed point, which determines the criticality of the system [6,7,10]. Prominent examples for universality classes of magnets are the 2D and 3D Ising (\mathbb{Z}_2 symmetry), 3D XY (O(2) symmetry), and 3D Heisenberg (O(3) symmetry) universality classes [113]. It is important to mention that the dimension in the classifications is referring to classical and not quantum systems, and they should not be confused with each other. In fact, the universality class of a short-range interacting non-frustrated quantum Ising model of dimension d lies in the universality of the classical d + 1-dimensional Ising model.

There are only a few dimensions for which a separate universality class is defined for the different models. For lower dimensions, the fluctuations are too strong in order for a spontaneous symmetry breaking to occur. In the case of the classical Ising model, there is no phase transition for 1D, while for the classical XY and Heisenberg models with continuous symmetries, there is not even a phase transitions for 2D due to the Hohenberg–Mermin–Wagner (HWM) theorem [130,131]. This dimensional boundary is referred to as lower critical dimension d_{lc} . The lower critical dimension is the highest dimension for which no transition occurs, i.e., $d_{lc} = 1$ for the Ising model and $d_{lc} = 2$ for the XY and Heisenberg model. For higher dimensions $d \ge 4$, the critical exponents of the mentioned models do not depend on the dimensionality any longer, and they take on the mean-field critical exponents in all dimensions. The underlying reason is that, with increasing dimensions, the local fluctuations become smaller due to the higher connectivity of the system [132]. This has been also exploited in diagrammatic and series expansions in 1/d [133–135]. This dimensional boundary, at which the criticality becomes the mean-field one, is called upper critical dimension d_{uc} . Usually, the upper critical dimension is too large to realise a system above its upper critical dimension in the real world. However, long-rang interactions can increase the connectivity of a system in a similar sense as the dimensionality. A sufficiently long-range interaction can, therefore, lower the upper critical dimension to a value that is accessible in experiments.

Finally, it is worth mentioning that the critical exponents are not independent of each other, but obey certain relations [129], namely

$$2 - \alpha = (d + z)\nu, \tag{3}$$

$$2 - \alpha = 2\beta + \gamma \,, \tag{4}$$

$$\gamma = \beta(\delta - 1) \,, \tag{5}$$

$$\gamma = (2 - \eta)\nu. \tag{6}$$

The first relation in Equation (3) is the so-called hyperscaling relation, whose classical analogue (without *z*) was introduced by Widom [10,136]. The Essam–Fisher relation in Equation (4) [137,138] is reminiscent of a similar inequality, which was proven rigorously by Rushbrooke using thermodynamic stability arguments. Equation (5) is called the Widom relation. The last relation in Equation (6) is the Fisher scaling relation, which can be derived using the fluctuation–dissipation theorem [10,129,138]. Those relations were originally obtained from scaling assumptions of observables close to the critical point, which were only later derived rigorously when the RG formalism introduced to critical phenomena [10,129]. Due to these relations, it is sufficient to calculate three, albeit not arbitrary, exponents to obtain the full set of critical exponents.

The hyperscaling relation Equation (3) is the only relation containing the dimension of the system and is, therefore, often said to break down above the upper critical dimension, where one expects the same mean-field critical exponents independent of the dimension d [10]. It, therefore, deserves special focus in this review since the long-range models discussed will be above the upper critical dimension in certain parameter regimes. Personally, we would not agree that the hyperscaling relation breaks down above the upper critical dimension, but we would rather call Equation (3) a special case of a more general hyperscaling relation:

$$2 - \alpha = \left(\frac{d}{q} + z\right)\nu,\tag{7}$$

with the pseudo-critical exponent o ("koppa") [34]:

$$\varphi = \begin{cases} 1 & \text{for } d \le d_{\text{uc}} \\ \frac{d}{d_{\text{uc}}} & \text{for } d > d_{\text{uc}} . \end{cases}$$
(8)

Below the upper critical dimension, the general hyperscaling relation, therefore, relaxes to Equation (3). Above the upper critical dimension, the relation becomes

$$2 - \alpha = (d_{\rm uc} + z)\nu, \tag{9}$$

which is independent of the dimension of the system. For the derivation of this generalised version of the hyperscaling relation for QPTs, see Section 2.3 or Ref. [34]. The derivation of the classical counterpart can be found in Ref. [15] and is reviewed in Ref. [41].

2.2. Finite-Size Scaling below the Upper Critical Dimension

Even though the singular behaviour of observables at the critical point is only present in the thermodynamic limit, it is possible to study the criticality of an infinite system by investigating their finite counterparts. In finite systems, the power-law singularities of the infinite system are rounded and shifted with respect to the critical point, e.g., the susceptibility with its characteristic divergence at the critical point λ_c is deformed to a broadened peak of finite height. The peak's position $r_L = (\lambda_L - \lambda_c)/\lambda_c$ is shifted with respect to the critical point r = 0. A possible definition of a pseudo-critical point of a finite system is the peak position λ_L . As the characteristic length scale of fluctuations ξ diverges at the critical point, the finite system approaching the critical point will at some point begin to "feel" its finite extent and the observables start to deviate from the ones in the thermodynamic limit. As ξ diverges with the exponent ν like $\xi \sim |r|^{-\nu}$ at the critical point, the extent of rounding in a finite system is related to the value of ν . Similarly, the peak magnitude of finite-size observables at the pseudo-critical point will depend on how strong the singularity in the thermodynamic limit is, which means it depends on the respective critical exponents α , β , γ , and δ . The shifting, rounding, and varying peak magnitude are shown for the susceptibility of the long-range transverse-field Ising model in Figure 1. This dependence of observables in finite systems on the criticality of the infinite system is the basis of finite-size scaling.



Figure 1. Susceptibility χ_L of the long-range transverse-field Ising chain for different linear system sizes from L = 64 to L = 724. The smaller the system, the farther away from the critical point the susceptibility starts to deviate from the thermodynamic limit and the farther the peak position shifts away from the critical point marked by the black dotted line.

In a more mathematical sense, the relation between critical exponents and finite-size observables has its origin in the renormalisation group (RG) flow close to the corresponding RG fixed point that determines the criticality [139]. Close to this fixed point, one can linearise the RG flow so that the free energy density and the characteristic length ξ become generalised homogeneous functions (GHFs) in their parameters [124,128,129,140,141]. For a thorough discussion of the mathematical properties of GHFs, we refer to Ref. [124] and Appendix B. This means that the free energy density *f* and characteristic length scale ξ as functions of the couplings *r*, *H*, *T*, *u* and the inverse system length L^{-1} obey the relations:

$$f(r, H, T, L^{-1}, u) = b^{-(d+z)} f(b^{y_r} r, b^{y_H} H, b^z T, bL^{-1}, b^{y_u} u)$$
(10)

$$\xi(r, H, T, L^{-1}, u) = b\xi(b^{y_r}r, b^{y_H}H, b^zT, bL^{-1}, b^{y_u}u)$$
(11)

with the respective scaling dimensions y_r , y_H , z > 0, $y_L = 1$, and $y_u < 0$ governing the linearised RG flow with spatial rescaling factor b > 1 around the RG fixed point, at which all couplings vanish by definition. All of those couplings are relevant except for u, which denotes the leading irrelevant coupling [10,113]. Relevant couplings are related to real-world parameters that can be used to tune our system away from the critical point like the temperature T, a symmetry-breaking field H, or simply the control parameter r. The irrelevant couplings u do not per se vanish at the critical point like the relevant ones do. However, they flow to zero under the RG transformation and are commonly set to zero in the scaling laws:

$$f(r, H, T, L^{-1}) = b^{-(d+z)} f(b^{y_r} r, b^{y_H} H, b^z T, bL^{-1})$$
(12)

$$\xi(r, H, T, L^{-1}) = b\xi(b^{y_r}r, b^{y_H}H, b^zT, bL^{-1})$$
(13)

by assuming analyticity in these parameters. The generalised homogeneity of thermodynamic observables can be derived from the one of the free energy density f. For example, the generalised homogeneity of the magnetisation:

$$m(r, H, T, L^{-1}) = b^{-(d+z)+y_H} m(b^{y_r} r, b^{y_H} H, b^z T, bL^{-1})$$
(14)

can be derived by taking the derivative of f with respect to the symmetry-breaking field H.

By investigating the singularity of $\xi(r) = \xi(r, H = 0, T = 0, L^{-1} = 0)$ in r via Equation (13), one can show that the scaling power y_r of the control parameter r is related to the critical exponent v by $y_r = 1/v$ [113]. For this, one fixes the value $b^{y_r}r$ of the first argument to ± 1 in the right-hand side of Equation (13) by setting $b = |r|^{-1/y_r}$ such that

$$\xi(r) = |r|^{-1/y_r} \xi(\pm 1) \sim |r|^{-\nu}$$
 (15)

Analogously, further relations between the scaling powers and other critical exponents can be derived by looking at the singular behaviour of the respective observables in the corresponding parameters. Overall, one further obtains

$$\alpha = -\frac{d+z-2y_r}{y_r}, \quad \beta = \frac{d+z-y_H}{y_r}, \quad \delta = \frac{y_H}{d+z-y_H}, \quad \gamma = -\frac{d+z-2y_H}{y_r}.$$
(16)

From these equations, one can already tell that the critical exponents are not independent of each other. In fact, the scaling relations $2 - \alpha = (d + z)\nu$, $2 - \alpha = 2\beta + \gamma$ and $\gamma = \beta(\delta - 1)$ (see Equations (3)–(5)) can be derived from Equation (16) and $y_r^{-1} = \nu$. By expressing the RG scaling powers y_x in terms of critical exponents, the homogeneity law for an observable \mathcal{O} with a bulk divergence $\mathcal{O}(r, 0, 0, 0) \sim |r|^{\omega}$ is given by

$$\mathcal{O}(r, H, T, L^{-1}) = b^{-\omega y_r} \mathcal{O}(b^{y_r} r, b^{y_H} H, b^z T, bL^{-1})$$
(17)

$$= b^{-\omega/\nu} \mathcal{O}(b^{1/\nu}r, b^{(\beta+\gamma)/\nu}H, b^{z}T, bL^{-1}).$$
(18)

In order to investigate the dependence on the linear system size *L*, the last argument in the homogeneity law is fixed to $bL^{-1} = 1$ by inserting b = L. This readily gives the finite-size scaling form

$$\mathcal{O}_L(r,H,T) = L^{-\omega/\nu} \Psi(L^{1/\nu}r, L^{(\beta+\gamma)/\nu}H, L^zT)$$
⁽¹⁹⁾

with Ψ being the universal scaling function of the observables \mathcal{O} . The scaling function Ψ itself does not depend on L any longer, but in order to compare different linear system sizes, one has to rescale its arguments. To extract the critical exponents from finite systems, the observable $\mathcal{O}_L(r, H, T)$ is measured for different system sizes L and parameters (r, H, T) close to the critical point (r, H, T) = (0, 0, 0). The L-dependence according to Equation (19) is then fit with the critical exponents ω , ν , $\beta + \gamma$, and z, as well as the critical point λ_c , which is hidden in the definition of r, as free parameters. It is advisable to fix two of the three parameters r, H, T to their critical values in order to minimise the amount of free parameters in the fit. For example, with H = T = 0 and only $r \neq 0$, one can extract the two critical exponents ω and ν alongside λ_c . For further details on a fitting procedure, we refer to Ref. [32]. If one knows the critical point, one can also set (r, H, T) = (0, 0, 0) and look at the L-dependent scaling $\mathcal{O}_L \sim L^{-\omega/\nu}$ directly at the critical point to extract the exponent ratio ω/ν . There are many more possible approaches to extract critical exponents from the FSS law in Equation (19) [142–144]. For relatively small system sizes, it might be required to take corrections to scaling into account [143,144].

2.3. Finite-Size Scaling above the Upper Critical Dimension

In the derivation of finite-size scaling below the upper critical dimension, it was assumed that the free energy density is an analytic function in the leading irrelevant coupling u and, therefore, one can set it to zero. However, this is not the case above the upper critical dimension any longer, and the free energy density f is singular at u = 0. Due to this singular behaviour u is referred to as a dangerous irrelevant variable (DIV).

As a consequence, one has to take the scaling of u close to the RG fixed point into account. This is achieved by absorbing the scaling of f in u for small u into the scaling of the other variables [128]:

$$f(r, H, T, L^{-1}, u) = u^{p_{(d+z)}} \overline{f}(u^{p_r} r, u^{p_H} H, u^{p_T} T, u^{p_L} L^{-1}),$$
(20)

up to a global power $p_{(d+z)}$ of *u*. This leads to a modification of the scaling powers in the homogeneity law for the free energy density [128]:

$$f(r, H, T, L^{-1}) = b^{-(d+z)^*} f(b^{y_r^*} r, b^{y_H^*} H, b^{z^*} T, b^{y_L^*} L^{-1})$$
(21)

$$= L^{-(d+z)^{*}/y_{L}^{*}} \mathcal{F}(L^{y_{r}^{*}/y_{L}^{*}}r, L^{y_{H}^{*}/y_{L}^{*}}H, L^{z^{*}/y_{L}^{*}}T)$$
(22)

with the modified scaling powers [34,128]:

$$(d+z)^* = (d+z) - p_{(d+z)}y_u, \qquad \begin{aligned} y_r^* &= y_r + p_r y_u, \quad y_H^* &= y_H + p_H y_u, \\ z^* &= z + p_z y_u, \quad y_L^* &= 1 + p_L y_u. \end{aligned}$$
(23)

In the classical case [128], y_L^* was commonly set to 1 by choice. This is justified because the scaling powers of a GHF are only determined up to a common non-zero factor [124]. However, for the quantum case [34], this was kept general as it has no impact on the FSS.

As the predictions from the Gaussian field theory and mean field differed for the critical exponents α , β , and δ , but not for the "correlation" critical exponents ν , z, η , and γ [145], the correlation sector was thought not to be affected by DIVs at first [128,142,145]. Later, the Q-FSS, another approach to FSS above the upper critical dimension, pioneered by Ralph Kenna and his colleagues, was developed for classical [15,19,125], as well as for quantum systems [34], which explicitly allowed the correlation sector to also be affected by the DIV. In analogy to the free energy density, the homogeneity law of the characteristic length scale is then also modified to

$$\xi(r, H, T, L^{-1}) = b^{-y_{\xi}^{*}} \xi(b^{y_{r}^{*}}r, b^{y_{H}^{*}}H, b^{z^{*}}T, b^{y_{L}^{*}}L^{-1})$$
(24)

$$= L^{\mathsf{q}} \Xi (L^{y_r^*/y_L^*} r, L^{y_H^*/y_L^*} H, L^{z^*/y_L^*} T)$$
(25)

with $y_{\xi}^* = -1 - p_{\xi}y_u = -y_r^*/y_r$ in order to reproduce the correct bulk singularity $\xi \sim |r|^{-\nu}$. A new pseudo-critical exponent φ ("koppa"):

$$\varphi = -\frac{y_{\xi}^*}{y_L^*} = \frac{y_r^*}{y_r y_L^*} = \nu \frac{y_r^*}{y_L^*}$$
(26)

is introduced. This exponent describes the scaling of the characteristic length scale with the linear system size. This non-linear scaling of ξ with *L* is one of the key differences to the previous treatments above the upper critical dimension in Ref. [128].

Analogous to the case below the upper critical dimension, the modified scaling powers y_x^* can be related to the critical exponents:

$$x = -\frac{(d+z)^* - 2y_r^*}{y_r^*},$$
(27)

$$\beta = \frac{(d+z)^* - y_H^*}{y_r^*} \,, \tag{28}$$

$$\delta = \frac{y_H^*}{(d+z)^* - y_H^*},$$
(29)

$$\gamma = -\frac{(d+z)^* - 2y_H^*}{y_r^*} \,. \tag{30}$$

By using the mean-field critical exponents for the O(n) quantum rotor model, one obtains restrictions for the ratios of modified scaling powers:

$$y_r^* = \frac{(d+z)^*}{2}, \qquad \qquad y_H^* = \frac{3(d+z)^*}{4}$$
(31)

Furthermore, one can link the bulk scaling powers y_r^* , y_H^* , and $(d + z)^*$ to the scaling power y_L^* of the inverse linear system size [34]:

$$(d+z)^* = y_L^* d + \frac{y_r^*}{y_r} z, (32)$$

by looking at the scaling of the susceptibility in a finite system [34,128]. This relation is crucial for deriving an FSS form above the upper critical dimension as the modified scaling power y_L^* , or rather, its relation to the other scaling powers determines the scaling with the linear system size *L*. For details on the derivation, we refer to Ref. [34]. We want to stress again that the scaling powers of GHFs are only determined up to a common non-zero factor [124]. Therefore, it is evident that one can only determine the ratios of the modified scaling powers, but not their absolute value. The absolute values are subject to choice. Different choices were discussed in Ref. [34], but these choices rather correspond to taking on different perspectives and have no impact on the FSS nor the physics.

From Equation (32) together with Equations (26) and (27), a generalised hyperscaling relation:

$$2 - \alpha = \left(\frac{d}{q} + z\right)\nu,\tag{33}$$

can be derived. This also determines the pseudo-critical exponent:

$$\varphi = \frac{d}{d_{\rm uc}} \qquad \text{for } d > d_{\rm uc} \,. \tag{34}$$

Finally, we can express the modified scaling powers in the FSS law for an observable O with power-law singularity $O(r, 0, 0, 0) \sim |r|^{\omega}$:

$$\mathcal{O}(r, H, T, L^{-1}) = b^{-\omega y_r^*} \mathcal{O}(b^{y_r^*} r, b^{(\beta + \gamma) y_r^*} H, b^{z^*} T, b^{y_L^*} L^{-1})$$
(35)

$$= L^{-\omega \varrho/\nu} \Psi(L^{\varrho/\nu} r, L^{(\beta+\gamma)\varrho/\nu} H, L^{z^*/y_L^*} T).$$
(36)

For $\varphi = 1$ below the upper critical dimension, Equation (36) relaxes to the standard FSS law Equation (19). The scaling in temperature has not yet been studied for finite quantum systems above the upper critical dimension. However, in Ref. [34], it was conjectured that $z^* = y_r^*/y_r z$ based on Equation (32), which is also in agreement with *z* being the dynamical critical exponent that determines the space–time anisotropy $\xi_{\tau} \sim \xi^z$, as we will shortly see. This means that the finite-size gap scales as $\Delta_L \sim L^{-z^*/y_L^*} \sim L^{-\varphi z}$ with the system size [34]. Of particular interest is the scaling of the characteristic length scale above the

upper critical dimension, for which the modified scaling law Equation (36) also holds with $\omega = -\nu$. Hence, the characteristic length scale in dependence of the control parameter *r* scales like

$$\xi_L(r) = L^{\mathsf{Q}} \Xi(L^{\mathsf{Q}/\nu} r) \tag{37}$$

with the scaling function Ξ . Directly at the critical point r = 0, this leads to $\xi_L \sim L^{\varphi}$. Comparing this with the scaling of the inverse finite-size gap $\Delta_L^{-1} \sim \xi_{L,\tau} \sim \xi_L^z$ verifies that z still determines the space–time anisotropy. Prior to the Q-FSS [15,17], the characteristic length scale was thought to be bound by the linear system size L [128]. However, this was shown not to be the case by measurements of the characteristic length scale for the classical five-dimensional Ising model [14] and for long-range transverse-field Ising models [34].

For the latter, the data collapse of the correlation length according to Equation (37) is shown in Figure 2 as an example.



Figure 2. Rescaled correlation length according to Equation (37) for a model above the upper critical dimension with $\varphi = 20/9 \approx 2.\overline{2}$ (to be specific, the long-range transverse-field Ising chain with $\sigma = 0.3$). The control parameter $r \sim h - h_c$ is proportional to the transverse field. The collapse of the data around the critical point r = 0 verifies the scaling Equation (37) and, therefore, demonstrates that ξ_L is indeed—in contrast to the prior belief—not bound by the linear system size, but $\xi_L \sim L^{\varphi}$.

3. Monte Carlo Integration

In this section, we provide a brief introduction to Monte Carlo integration (MCI). We focus on the aspects of Markov chain MCI as the basis to formulate the white graph Monte Carlo embedding scheme of the pCUT+MC method in Section 4 and the stochastic series expansion (SSE) quantum Monte Carlo (QMC) algorithm in Section 5 in a self-contained fashion. MCI is the foundation for countless numerical applications, which require the integration over high-dimensional integration spaces. As this review has a focus on "Monte Carlo-based techniques for quantum magnets with long-range interactions", we forward readers with a deeper interest in the fundamental aspects of MCI and Markov chains to Refs. [146–148].

MCI summarises numerical techniques to find estimators for integrals of functions $f : C \to \mathbb{R}$ over an integration space C using random numbers. The underlying idea behind MCI is to estimate the integral, or the sum in the case of discrete variables, of the function f over the configuration space by an expectation value:

$$I = \int_{\mathcal{C}} d\omega f(\omega) = \int_{\mathcal{C}} d\omega \frac{P(\omega)}{P(\omega)} f(\omega) = \int_{\mathcal{C}} d\omega P(\omega) \tilde{f}(\omega) = \lim_{S \to \infty} \frac{1}{S} \sum_{i=1}^{i=S} \tilde{f}(\omega_i)$$
(38)

with $\omega_i \in C$ sampled according to a probability density function $P : C \to \mathbb{R}_{\geq 0}$ (PDF) and the function $\tilde{f}(\omega) = f(\omega)/P(\omega)$ reweighted by the PDF. A famous direct application of this idea is the calculation of the number "pi", which is discussed in great detail in Ref. [148].

In this review, MCI is used for the embedding of white graphs on a lattice to evaluate high orders of a perturbative series expansion or to calculate thermodynamic observables using the SSE framework. In both cases, non-normalised relative weights $\pi(\omega)$ within a configuration space C arise, which are used for the sampling of the PDF *P*:

$$P(\omega) = \frac{\pi(\omega)}{\int_{\mathcal{C}} d\omega \, \pi(\omega)} \,, \tag{39}$$

being oftentimes not directly accessible. In the context of statistical physics, $\pi(\omega)$ is often chosen to be the relative Boltzmann weight $e^{-\beta E(\omega)}$ of each configuration ω . While this relative Boltzmann weight is accessible as long as $E(\omega)$ is known, the full partition function to normalise the weights is in general not.

In order to efficiently sample the configuration space C according to the relative weights, the methods in this review use a Markovian random walk to generate $\{\omega_1, \ldots, \omega_m\}$. Let ω_n be the random state of a random walk at a discrete step n. The state ω_{n+1} at the next step is randomly determined according to the conditional probabilities $T(\omega \rightarrow \omega')$ (transition probabilities). These transition probabilities are normalised by

$$\int_{\mathcal{C}} d\omega' \, T(\omega \to \omega') = 1 \,. \tag{40}$$

Markovian random walks obey the Markov property, which means the random walk is memory-free and the transition probability for multiple steps factorises into a product over all time steps:

$$T(\omega^{(0)} \to \omega^{(1)} \to \dots \to \omega^{(m-1)} \to \omega^{(m)}) = \prod_{i=0}^{m-1} T(\omega^{(i)} \to \omega^{(i+1)})$$
(41)

with $\omega^{(0)}$ the start configuration. We require the Markovian random walk to fulfil the following conditions: First, the random walk should have a certain PDF $P(\omega)$ defined by the weights $\pi(\omega)$ in Equation (39) as a stationary distribution. By definition, $P(\omega)$ is a stationary distribution of the Markov chain if it satisfies the global balance condition:

$$\int_{\mathcal{C}} d\omega P(\omega) T(\omega \to \omega') = P(\omega') .$$
(42)

Second, we require the random walk to be irreducible, which means that the transition graph must be connected and every configuration $\omega \in C$ can be reached from any configuration $\omega' \in C$ in a finite number of steps. This property is necessary for the uniqueness of the stationary distribution [147]. Lastly, we require the random walk to be aperiodic (see Ref. [147] for a rigorous definition). Together with the irreducibility condition, this ensures convergence to the stationary distribution [147].

There are several possibilities to design a Markov chain with a desired stationary distribution [146–151]. Commonly, the Markov chain is constructed to be reversible. This means that it satisfies the detailed balance condition:

$$P(\omega)T(\omega \to \omega') = P(\omega')T(\omega' \to \omega), \qquad (43)$$

which is a stronger condition for the stationarity of *P* than the global balance condition in Equation (42). One popular choice for the transition probabilities $T(\omega \rightarrow \omega')$ that satisfies the detailed balance condition is given by the Metropolis–Hastings algorithm. Most applications of MCI reviewed in this work are based on the Metropolis–Hastings algo-

rithm [149,150]. In this approach, the transition probabilities $T(\omega \to \omega')$ are decomposed into propositions $\tilde{T}(\omega \to \omega')$ and acceptance probabilities $p_{acc}(\omega \to \omega')$ as follows:

$$T(\omega \to \omega') = \tilde{T}(\omega \to \omega') p_{\rm acc}(\omega \to \omega') .$$
(44)

The probabilities to propose a move $\tilde{T}(\omega \rightarrow \omega')$ can be any random walk satisfying the irreducibility and aperiodicity condition. By inserting the decomposition of the transition probabilities Equation (44) into the detailed balance condition Equation (43), one obtains for the acceptance probabilities:

$$\frac{p_{\rm acc}(\omega \to \omega')}{p_{\rm acc}(\omega' \to \omega)} = \frac{P(\omega')\tilde{T}(\omega' \to \omega)}{P(\omega)\tilde{T}(\omega \to \omega')} = \frac{\pi(\omega')\tilde{T}(\omega' \to \omega)}{\pi(\omega)\tilde{T}(\omega \to \omega')},$$
(45)

where, in the last step, the idea that the unknown normalisation factors (see Equation (39)) of the PDF cancel was used. The condition in Equation (45) is fulfilled by the Metropolis–Hastings acceptance probabilities [150]:

$$p_{\rm acc}(\omega \to \omega') = \min\left(1, \frac{\pi(\omega')\tilde{T}(\omega' \to \omega)}{\pi(\omega)\tilde{T}(\omega \to \omega')}\right).$$
(46)

For the special case, for which the proposition probabilities are symmetric $\tilde{T}(\omega \rightarrow \omega') = \tilde{T}(\omega' \rightarrow \omega)$, Equation (46) reduces to the Metropolis acceptance probabilities:

$$p_{\rm acc}(\omega \to \omega') = \min\left(1, \frac{\pi(\omega')}{\pi(\omega)}\right).$$
 (47)

As an example, we regard a classical thermodynamic system with Boltzmann weights $e^{-\beta E(\omega)}$ given by the energies of configurations and the inverse temperature to give an intuitive interpretation of the Metropolis acceptance probabilities in Equation (47). The proposition to move from a configuration ω to a configuration ω' with a smaller energy $E(\omega') < E(\omega)$ is always accepted independent of the temperature. On the other hand, the proposition to move to a configuration ω' with a larger energy than ω is only accepted with a probability depending on the ratio of the Boltzmann weights. If the temperature is higher, it is more likely to move to states with a larger energy. This reflects the physics of the system in the algorithm, focusing on the low-energy states at low temperatures and going to the maximum entropy state at large temperatures.

4. Series-Expansion Monte Carlo Embedding

In this section, we provide a self-contained and comprehensive overview of linked-cluster expansions for long-range interacting systems [25,29–31,34,35] using white graphs [152] in combination with Monte Carlo integration for the graph embedding. First, we introduce linked-cluster expansions (LCEs) and discuss perturbative continuous unitary transformations (pCUTs) [153,154] as a suitable high-order series expansion method. We then establish an adequate formalism for setting up LCEs and discuss the calculation of suitable physical quantities in practice. With the help of white graphs, we can employ LCEs for models with long-range interactions and use the resulting contributions in a Monte Carlo algorithm to deal with the embedding problem posed by long-range interactions. This approach we dub pCUT + MC.

4.1. Motivation and Basic Concepts

The goal of all our efforts is to calculate physical observables in the thermodynamic limit, i.e., for an infinite lattice \mathcal{L} , using high-order series expansions. The starting point of every perturbative problem is

$$\mathcal{H} = \mathcal{H}_0 + \lambda \mathcal{V},\tag{48}$$

where the Hamiltonian describing the full physical problem \mathcal{H} can be split up into an unperturbed part \mathcal{H}_0 , which is readily diagonalisable, and a perturbation \mathcal{V} associated with a perturbation parameter λ , which is small compared to the energy scales of \mathcal{H}_0 . We aim to obtain a power series up to a maximal reachable order \mathfrak{o}_{max} as an approximation of a desired physical quantity:

$$f(\lambda) \approx p_0 + p_1 \lambda + p_2 \lambda^2 + \dots p_{\mathfrak{o}_{\max}} \lambda^{\mathfrak{o}_{\max}} , \qquad (49)$$

where the coefficients p_i are to be determined by the series expansion. We want to use the information contained in the power series to infer the properties of the approximated function $f(\lambda)$ [155]. The cost of determining the coefficients is associated with an exponential growth in complexity with increasing order [155]. Hence, calculations are performed with the help of a computer programme. Obviously, the computer cannot deal with an infinitely large lattice. Instead, we must look at finite cut-outs consisting of a finite set of lattice sites that are connected by bonds (or links) symbolising the interactions of the Hamiltonian on the lattice. We term these cut-outs *clusters*. If two clusters *A* and *B* do not share a common site or conterminously do not have a link that connects any site of A and B with each other ($A \cap B = \emptyset$), then the cluster $C = A \cup B$ is called a *disconnected cluster*. Otherwise, if no such partition into disconnected clusters *A* and *B* exists ($A \cap B \neq \emptyset$), the cluster *C* is called *connected*. We can define quantum operators \mathcal{M} (e.g., a physical observable) on these clusters just as on the infinite lattice.

There are essentially two ways of performing high-order series expansions. The first one is the naive approach of taking a single finite cluster $C \subset \mathcal{L}$ [153,154,156–158] and designing it such that the contribution of $\mathcal{M}(C)$ coincides with the contributions on the infinite lattice $\mathcal{M}(\mathcal{L}) = \mathcal{M}(C)$ up to the considered order in the perturbation parameter. The cluster needs to be chosen large enough such that the perturbative calculations up to the considered order are not affected by the boundaries of the cluster. Another way of performing calculations is to construct the operator contribution on a cluster—coinciding with the infinite lattice contributions up to a given order—by decomposing it into all possible contributions on smaller clusters [155,159–170]. Now, the contributions on many, but smaller clusters must be determined and added up to obtain the contribution on the infinite lattice:

$$\mathcal{M}(\mathcal{L}) = \mathcal{M}(C) = \sum_{C' \subset C} \mathcal{M}(C') .$$
(50)

In contrast to the previous approach, we willingly accept boundary effects for the many subclusters. Such a cluster decomposition is known to be computationally more efficient because it suffices to calculate the contributions on the Hilbert space of the smaller clusters reducing the overhead of non-contributing processes, and it also suffices to perform the calculations only on a few clusters as many give identical contributions due to symmetries. This can significantly reduce the overhead.

However, there are subtleties about the validity of performing calculations on finite clusters, e.g., when setting up a linked-cluster expansion (linked-cluster means only connected clusters contribute), the operator \mathcal{M} must satisfy a certain property, namely cluster additivity. The quantity \mathcal{M} is called *cluster additive* if and only if the contribution on disconnected clusters $A \cup B$ solely comes from the contributions on its constituent connected clusters A and B. This means we can simply add the contributions of A and B from the smaller connected clusters to obtain the one for the disconnected cluster, i.e.,

$$\mathcal{M}(A \cup B) = \mathcal{M}(A) + \mathcal{M}(B), \tag{51}$$

as illustrated schematically in Figure 3.



Figure 3. Schematic illustration of cluster additivity. The contribution of a disconnected cluster $A \cup B$ (clusters within dashed circle) made up of individual connected clusters A and B (yellow areas) is the sum of its individual parts.

We can also understand cluster additivity in the language of perturbation theory where acting on bonds in every order forms a cluster of perturbatively active bonds. If such a cluster is connected, we call these processes *linked*. So, cluster additivity simultaneously means that only linked processes will contribute. Cluster additivity is at the heart of the *linked-cluster theorem*, which states that only linked processes will contribute to the overall contribution in the thermodynamic limit. To set up a linked-cluster expansion, we want to exploit cluster additivity and the linked-cluster theorem so that we can "simply" add up the contributions from individual connected clusters to obtain the desired power series in the thermodynamic limit.

An example of a cluster additive quantity is the ground-state energy E_0 . Imagine we want to calculate the ground-state energy of a non-degenerate ground-state subspace, then cluster additivity is naturally fulfilled:

$$E_0(A \cup B) = E_0(A) + E_0(B), \tag{52}$$

and we can calculate the ground-state energy on $A \cup B$ from its individual parts A and B. However, cluster additivity is not satisfied in general. We can construct a counterexample by considering the first excited state with energy E_1 . For example, consider the first excitation above the ferromagnetic ground state of the transverse-field Ising model in the low-field limit that is a single spin flip dressed with quantum fluctuations induced by the transverse field [113]. We usually refer to such excitations as *quasiparticles* (qp). Here, we cannot add the contributions on clusters A and B to obtain the excitation energy on cluster $A \cup B$:

$$E_1(A \cup B) \neq E_1(A) + E_1(B)$$
. (53)

How to set up a linked-cluster expansion for intensive properties is not obvious, and it seemed out of reach after the introduction of linked-cluster expansions in the 1980s [159–161]. Only several years later, it was noticed by Gelfand [162] that additivity can be restored for excited states when properly subtracting the ground-state energy. This approach was later generalised to multiparticle excitations [154,156,164,165] and observables [154,163].

In the following, we first introduce a perturbation theory method that maps the original problem in Equation (48) to an effective one. We will show that the derived effective Hamiltonian and observables satisfy cluster additivity. In the subsequent section, we make use of the property and show how we can set up a linked-cluster expansion for energies of excited states and observables by properly subtracting contributions from lower energy states.

4.2. Perturbation Method: Perturbative Continuous Unitary Transformations

The first step towards setting up a linked-cluster expansion is to find a perturbation method that satisfies cluster additivity, which is generically not given and a non-trivial task [171]. Here, we use perturbative continuous unitary transformations (pCUTs) [153,154] that transform the original Hamiltonian perturbatively order by order into a quasiparticle-conserving Hamiltonian, reducing the original many-body problem to an effective few-body problem. We start discussing how to solve the flow equation to obtain the pCUT method and show afterwards how the Hamiltonian decomposes into additive parts that can be used for a linked-cluster expansion.

We strive to solve the usual problem of perturbation theory of Equation (48). The unperturbed part \mathcal{H}_0 can be easily diagonalised exactly with a spectrum that has to be equidistant and bounded from below. Additionally, the perturbation \mathcal{V} must be a sum of operators T_n :

$$\mathcal{V} = \sum_{n=-N}^{N} T_n \,, \tag{54}$$

containing all processes changing the energy by *n* quanta and—if properly rescaled corresponding to the same number of quasiparticles *n*. The goal of the pCUT method is to find an optimal basis in which the many-body problem of the original Hamiltonian reduces to an effective few-body problem. For that, we introduce a unitary transformation depending on a continuous flow parameter ℓ and define

$$\mathcal{H}(\ell) = U^{\dagger}(\ell)\mathcal{H}U(\ell) .$$
(55)

In the limiting case $\ell = 0$, we require $\mathcal{H}(0) = \mathcal{H}$ to recover the original Hamiltonian, and for $\ell = \infty$, we require $\lim_{\ell \to \infty} \mathcal{H}(\ell) = \mathcal{H}_{eff}$ so that the unitary transformation maps the original to the desired effective Hamiltonian. We can rewrite the unitary transformation as

$$U(\ell) = \mathcal{T}_{\ell} \exp\left(-\int_{0}^{\ell} \eta(\ell') d\ell'\right), \qquad (56)$$

where η is the anti-hermitian generator generating the unitary transformation and T_{ℓ} the ordering operator for the flow parameter. Taking the derivatives of Equation (55) and Equation (56) in ℓ , we eventually arrive at the flow equation:

$$\frac{\mathrm{d}\mathcal{H}(\ell)}{\mathrm{d}\ell} = \left[\eta(\ell), \mathcal{H}(\ell)\right]. \tag{57}$$

Flow equations have been studied for quite some time in mathematics and physics with a variety of applications [172–180]. It was Knetter and Uhrig [153] who proposed a perturbative ansatz for the generator of continuous unitary transformations along the lines of Mielke [180], introducing the quasiparticle generator (also known as the "MKU generator") for the pCUT method:

$$\eta^{\mathrm{qp}}(\ell)_{i,j} := \mathrm{sgn}(\mathcal{H}_{0\,i,i} - \mathcal{H}_{0\,i,j})\mathcal{H}_{i,j}(\ell) , \qquad (58)$$

where the indices *i*, *j* refer to blocks of the Hamiltonian labelling the quasiparticle number. Diagonal blocks $\mathcal{H}_{i,i}$ contain all processes conserving the number of quasiparticles *i*, while off-diagonal blocks $\mathcal{H}_{i,j}$ contain all processes changing the quasiparticle number from *i* to *j*. The reasoning behind the ansatz can be explained by looking at $\text{sgn}(\mathcal{H}_{0\,i,i} - \mathcal{H}_{0\,j,j})$, where processes $i \to j$ in $\mathcal{H}_{i,j}$ are assigned the opposite sign of the inverse processes $j \to i$, and therefore, the idea is to "rotate away" off-diagonal blocks by the unitary transformation during the flow of ℓ , while processes that do not change the quasiparticle number are not transformed away due to sgn(0) = 0, but get renormalised during the flow. Consequently, in the limit $\ell \to \infty$, we obtain an effective Hamiltonian \mathcal{H}_{eff} that is block diagonal in *n*. This idea is depicted in Figure 4.

Next, we make a perturbative ansatz for the Hamiltonian during the flow:

$$\mathcal{H}(\ell) = \mathcal{H}_0 + \sum_{\substack{k=1\\\sum_j n_j = k}}^{\infty} \lambda_1^{n_1} \dots \lambda_{N_\lambda}^{n_{N_\lambda}} \sum_{\dim(\boldsymbol{m}) = k} F(\ell; \boldsymbol{m}) T(\boldsymbol{m}) , \qquad (59)$$

with the notation

$$m = (m_1, m_2, m_3, \dots, m_k),$$
 (60)

$$m_i \in \{0, \pm 1, \pm 2, \dots, \pm N\}$$
, (61)

$$\dim(m) = k , \tag{62}$$

$$T(m) = T_{m_1} T_{m_2} T_{m_3} \dots T_{m_k} , \qquad (63)$$

with $F(\ell; m)$ being undetermined real functions. We introduce N_{λ} distinct expansion parameters instead of just a single λ to keep the notation as general as possible because in Section 4.6.3, about white graphs, we will need multiple expansion parameters to encode additional information. Inserting Equations (58) and (59) into the flow equation (57), we can solve the equation perturbatively order-by-order as we obtain a recursive set of differential equations for $F(\ell; m)$.



Figure 4. Illustration of the method of perturbative continuous unitary transformations (pCUTs) transforming the original Hamiltonian on the left to a block-diagonal quasiparticle-conserving effective Hamiltonian on the right. The desired effective Hamiltonian is given in the limit $\ell \to \infty$ of the flow parameter ℓ of the continuous unitary transformation $\mathcal{H}(\ell) = U^{\dagger}(\ell)\mathcal{H}(0)U(\ell)$. While the different quasiparticle sectors interact with each other by the off-diagonal blocks in the original Hamiltonian, the off-diagonal blocks are zero in the effective Hamiltonian as they are "rotated away" during the flow.

To recover the original Hamiltonian \mathcal{H} , we have to demand the correct initial conditions $F(0; \mathbf{m}) = 1$ for $|\mathbf{m}| = 1$ and $F(0; \mathbf{m}) = 0$ for $|\mathbf{m}| > 1$. We can solve the differential equations (cf. Ref. [153]) exactly for $\ell \to \infty$, yielding

$$\mathcal{H}_{\text{eff}} = \mathcal{H}_0 + \sum_{\substack{k=1\\\sum_j n_j = k}}^{\infty} \lambda_1^{n_1} \dots \lambda_{N_\lambda}^{n_{N_\lambda}} \sum_{\substack{\dim(m) = k,\\M(m) = 0}} \mathcal{C}(m) T(m), \tag{64}$$

with $F(\infty; \mathbf{m}) = C(\mathbf{m}) \in \mathbb{Q}$ being exact rational coefficients and the restriction $M(\mathbf{m}) = \sum_{i=1}^{k} m_i = 0$ making the products $T(\mathbf{m})$ quasiparticle-conserving [153]. Hence, the commutator of the effective Hamiltonian with the unperturbed diagonal part of the original Hamiltonian vanishes ($[\mathcal{H}_{eff}, \mathcal{H}_0] = 0$). Note that, so far, the effective Hamiltonian (64) is model independent. It only depends on the overall structure of Equation (54). The generic form of the Hamiltonian comes at the cost of an additional normal ordering usually by applying the Hamiltonian to a cluster. Of course, it could also be performed explicitly by using the hard-core bosonic commutation relations, but the former approach can be handled much easier by a computer programme. Yet, we achieved our goal of obtaining a block-diagonal Hamiltonian:

$$\mathcal{H}_{\rm eff} = \bigoplus_{n=0}^{\infty} \mathcal{H}_{\rm eff\,n} \,, \tag{65}$$

where $\mathcal{H}_{\text{eff}n}$ is the effective irreducible Hamiltonian of *n* quasiparticle processes (see also Section 4.3). Let us emphasise again that this block-diagonal structure allows us to solve the *n* quasiparticle blocks individually, which significantly reduces the complexity of the

original many-body problem to an effective one that is block-diagonal in the quasiparticle number *n*.

If we want to calculate an effective observable, we can make an ansatz along the same lines [154]. We insert the perturbative ansatz:

$$\mathcal{O}(\ell) = \sum_{\substack{k=1\\\sum_{j}n_{j}=k}}^{\infty} \lambda_{1}^{n_{1}} \dots \lambda_{N_{\lambda}}^{n_{N_{\lambda}}} \sum_{i=1}^{k+1} \sum_{\dim(\boldsymbol{m})=k} G(\ell; \boldsymbol{m}; i) \mathcal{O}(\boldsymbol{m}; i)$$
(66)

with undetermined functions $G(\ell; m; i)$. The operator product is defined as

$$\mathcal{O}(\boldsymbol{m}; i) = T_{m_1} \dots T_{m_{i-1}} \mathcal{O} T_{m_{i+1}} \dots T_{m_k} .$$
(67)

Inserting exactly the same generator (58) and the ansatz for the observable in Equation (66) instead of the Hamiltonian into the flow Equation (57), we arrive at

$$\mathcal{O}_{\text{eff}} = \sum_{\substack{k=1\\\sum_{j}n_{j}=k}}^{\infty} \lambda_{1}^{n_{1}} \dots \lambda_{N_{\lambda}}^{n_{N_{\lambda}}} \sum_{i=1}^{k+1} \sum_{\dim(\boldsymbol{m})=k} \tilde{\mathcal{C}}(\boldsymbol{m};i) \mathcal{O}(\boldsymbol{m};i)$$
(68)

with $\tilde{C}(m; i) = G(\infty; m; i) \in \mathbb{Q}$ by solving the resulting set of differential equations for $\ell \to \infty$ [154]. Note that the last sum does not contain a restriction M(m) = 0, and therefore—in contrast to the effective Hamiltonian—effective observables are not (necessarily) quasiparticle-conserving.

We have just derived the effective form of the Hamiltonian and observables in the pCUT method that have a very generic form depending only on the structure of the perturbation of Equation (54). As already stated, the model dependence of our approach comes into play when performing a linked-cluster expansion by applying the effective Hamiltonian or observable to finite clusters. But how do we know if the effective quantities are cluster additive? We follow the argumentation of Refs. [181,182] by looking at the original Hamiltonian (48) that trivially satisfies cluster additivity as long as all bonds represent a non-vanishing term in \mathcal{V} between sites ("bond equals interaction"). Thus, the Hamiltonian on a disconnected cluster $A \cup B$:

$$\mathcal{H}|_{A\cup B} = \mathcal{H}|_A + \mathcal{H}|_B \tag{69}$$

is cluster additive because $\mathcal{H}|_A$ and $\mathcal{H}|_B$ are non-interacting. Here, we denote the restriction of the Hamiltonian \mathcal{H} to a cluster *C* as $\mathcal{H}|_C$. We can further insert this property into the flow equation:

$$\frac{d\mathcal{H}(\ell)|_{A\cup B}}{d\ell} = [\eta(\ell)|_{A\cup B}, \mathcal{H}(\ell)|_{A\cup B}]$$

$$\frac{d\mathcal{H}(\ell)|_{A}}{d\ell} + \frac{d\mathcal{H}(\ell)|_{B}}{d\ell} = [\eta(\ell)|_{A} + \eta(\ell)|_{B}, \mathcal{H}(\ell)|_{A} + \mathcal{H}(\ell)|_{B}]$$

$$= [\eta(\ell)|_{A}, \mathcal{H}(\ell)|_{A}] + [\eta(\ell)|_{B}, \mathcal{H}(\ell)|_{B}].$$
(70)

Here, we used the property of $\mathcal{H}(0)$ that it commutes on disconnected clusters and the fact that the Hamiltonian is continuously transformed during the flow starting from $\ell = 0$. Therefore, the derivative and the commutator can be split up, acting on each cluster individually and preserving cluster additivity during the flow. Consequently, the effective Hamiltonian:

$$\mathcal{H}_{\rm eff}|_{A\cup B} = \mathcal{H}_{\rm eff}|_A + \mathcal{H}_{\rm eff}|_B \tag{71}$$

in the limit $\ell \to \infty$ is cluster-additive as well. The same proof holds for effective observables. Another more physical argument is that the effective pCUT Hamiltonian can be written as a sum of nested commutators of *T*-operators [152,167]. For instance, considering the perturbation $\mathcal{V} = T_{-1} + T_0 + T_{+1}$, the effective Hamiltonian looks like

$$\mathcal{H}_{\rm eff} = \mathcal{H}_0 + \lambda T_0 + \lambda^2 [T_{+1}, T_{-1}] + \frac{\lambda^3}{2} ([[T_{+1}, T_0], T_{-1}] + [T_{+1}, [T_0, T_{-1}]]) + \dots$$
(72)

Splitting up $T_n = \sum_l \tau_{n,l}$ into local operators acting on bonds l, the nested commutators vanish for processes that are not linked. Hence, the linked-cluster theorem is fulfilled, and the effective Hamiltonian is cluster-additive. To emphasise the linked-cluster property, the generic effective Hamiltonian is often written as

$$\mathcal{H}_{\text{eff}} = \mathcal{H}_0 + \sum_{\substack{k=1\\\sum_j n_j = k}}^{\infty} \lambda_1^{n_1} \dots \lambda_{N_\lambda}^{n_{N_\lambda}} \sum_{\substack{\dim(m) = k\\M(m) = 0}} \sum_{\substack{C\\|\mathcal{E}_C| \le k}} \mathcal{C}(m) \sum_{\substack{l_1,\dots,l_k\\\bigcup_{i=1}^k l_i = C}} \tau_{m_1,l_1} \dots \tau_{m_k,l_k} , \qquad (73)$$

where the sum over *C* runs over all possible connected clusters with maximal *k* bonds $(k \ge |\mathcal{E}_C|)$ [152]. The notation \mathcal{E}_C will be clarified in Section 4.6.2, where graphs are formally introduced. In this context, it is simply the set of bonds of a connected cluster *C* and $|\mathcal{E}_C|$ the number of bonds in this set. The condition $\bigcup_{i=1}^{k} l_i = C$ arising from the linked-cluster theorem ensures that the cluster consisting of active links and sites during a process must match with the bonds and sites of the connected cluster *C*. For observables, the generalised condition $\bigcup_{i=1}^{k} l_i \cup x = C$ holds, where the index *x* can either refer to a site (local observable) or a link (non-local observable), and we have

$$\mathcal{O}_{\text{eff}} = \sum_{\substack{k=1\\\sum_{j}n_{j}=k}}^{\infty} \lambda_{1}^{n_{1}} \dots \lambda_{N_{\lambda}}^{n_{N_{\lambda}}} \sum_{i=1}^{k+1} \sum_{\dim(\boldsymbol{m})=k} \sum_{\substack{C\\|\mathcal{E}_{C}|\leq k}} \tilde{\mathcal{C}}(\boldsymbol{m}; i) \times \sum_{\substack{l_{1},\dots,l_{k}\\\bigcup_{i=1}^{k}l_{i}\cup x=C}} \tau_{m_{1},l_{1}} \dots \tau_{m_{i-1},l_{i-1}} \mathcal{O}_{x} \tau_{m_{i},l_{i}} \dots \tau_{m_{k},l_{k}}.$$

$$(74)$$

Although we showed that the effective Hamiltonian and observables are cluster-additive and, therefore, fulfil the linked-cluster theorem, to set up a linked-cluster expansion, there are important subtleties remaining when we restrict the effective Hamiltonian and observables to the quasiparticle basis, which we need to address before we can discuss how to perform the calculations in practice.

4.3. Unravelling Cluster Additivity

In this subsection, we need to clarify how we can use the cluster-additive property of the effective pCUT Hamiltonian and observables to set up a linked-cluster expansion not only for the ground-state energy, but also for the energies of excited states. Many aspects of this section are based on the original work of Ref. [154], in which a general formalism was developed for how to derive suitable quantities for the calculation of multiparticle excitations and observables. We further develop this formalism by inferring the concept of cluster additivity for the quasiparticle basis, introducing the notion of particle additivity. The term "additivity" in this context was recently introduced by Ref. [171].

We start by recalling that the effective Hamiltonian is block-diagonal, and we can write the Hamiltonian operator as a sum of irreducible operators of *n* quasiparticle processes:

$$\mathcal{H}_{\rm eff} = \sum_{n=0} \mathcal{H}_{\rm eff,n} \,. \tag{75}$$

We can express the *n* quasiparticle processes in second quantisation in terms of local (hardcore) bosonic operators b_i^{\dagger} creating and b_i annihilating a quasiparticle at site *i*. When considering quantum magnets like we do in this review, a hard-core repulsion comes into play allowing only a single quasiparticle at a given site [154]. For instance, in the ferromagnetic ground-state of the 2D Ising model, an elementary excitation is given by a single spin flip, which can be interpreted as a quasiparticle excitation [113]. Obviously, at most one excitation on the same site is allowed. Different particle flavours τ can also be accounted for by incorporating an additional index τ of the operator $b_{i,\tau}^{(\dagger)}$. To keep the notation simple, we will drop this additional index in the following. The irreducible operators in the second quantisation and normal-ordered form then read

$$\begin{aligned}
\mathcal{H}_{\text{eff }0} &= \epsilon_0 \mathbb{1}, \\
\mathcal{H}_{\text{eff }1} &= \sum_{i} \sum_{j} t_{i;j} b_{j}^{\dagger} b_{i} , \\
\mathcal{H}_{\text{eff }2} &= \sum_{i_{1}, i_{2}} \sum_{j_{1}, j_{2}} t_{i_{1}, i_{2}; j_{1}, j_{2}} b_{j_{2}}^{\dagger} b_{j_{1}}^{\dagger} b_{i_{2}} b_{i_{1}} , \\
&\vdots \\
\mathcal{H}_{\text{eff }n} &= \sum_{i_{1}, \dots, i_{n}} \sum_{j_{1}, \dots, j_{n}} t_{i_{1}, \dots, j_{n}} b_{j_{n}}^{\dagger} \dots b_{j_{1}}^{\dagger} b_{i_{n}} \dots b_{i_{1}} .
\end{aligned}$$
(76)

Written in normal order, the meaning of these processes is directly clear when acting on states in the quasiparticle basis. The prefactors $t_{i_1...,i_n,j_1,...,j_n}$ are to be determined by applying the effective pCUT Hamiltonian (64) to an appropriately designed cluster *C*. Let us consider the quasiparticle basis on a connected cluster *C*:

$$\{|0\rangle_{C}, |1\rangle_{C}, |2\rangle_{C}, \dots, |n\rangle_{C}\} = \{|0\rangle_{C}, |1; i\rangle_{C}, |2; i_{1}, i_{2}\rangle_{C}, \dots, |n; i_{1}, i_{2}, \dots, i_{n}\rangle_{C}\},$$
(77)

where the number *n* specifies the number of particle excitations and the indices i_j denote the positions of the *n* (local) excitations. The effective pCUT Hamiltonian is quasiparticle conserving, so let us restrict it to *N* particle states, like when evaluating its matrix elements in this basis. If we evaluate by acting on a state with fewer particles than particles involved in the process, then the irreducible operator annihilates more particles than there are and the contribution is zero, that is $\mathcal{H}_n|^N = 0$ for N < n. When we determine the action of $\mathcal{H}_{\text{eff} n}|^n$ for N = n, this allows us to determine all prefactors $t_{i_1...,i_n;j_1,...,j_n}$ defining the action of $\mathcal{H}_{\text{eff} n}$ on the entire unrestricted Hilbert space. The second quantisation presents a natural generalisation of the Hamiltonian restricted to a finite number of particles to an arbitrary number of particles [154]. We can construct $\mathcal{H}_n|^N$ for N > n from $\mathcal{H}_n|^n$ since the latter completely defines the action $\mathcal{H}_{\text{eff} n}$ on the entire Hilbert space.

Although everything seems fine so far to set up a linked-cluster expansions, let us tell you that it is not. We finished the motivation in Section 4.1 with the statement that we cannot simply add up contributions for energies of excited states (cf. Equation (53)). The reason is that, although we showed that \mathcal{H}_{eff} is cluster additive, the irreducible operators restricted to the *N* particle basis $\mathcal{H}_{\text{eff} n}|^N$ are in fact not. To grasp a better understanding of the abstract concept of cluster additivity and why setting up a linked cluster expansion for higher particle channels usually fails, let us consider the following basis on a disconnected cluster $A \cup B$:

$$\{|0\rangle_{A\cup B}, |1\rangle_{A\cup B}, |2\rangle_{A\cup B}, \dots\} = \{|0\rangle_A \otimes |0\rangle_B, |0\rangle_A \otimes |1\rangle_B, |1\rangle_A \otimes |0\rangle_B, |0\rangle_A \otimes |2\rangle_B, |1\rangle_A \otimes |1\rangle_B, |2\rangle_A \otimes |0\rangle_B, \dots\},$$
(78)

where $|n\rangle_C$ represents all possible *n*-particle states living on a cluster *C*. While there is only one way to decompose the zero-particle states $|0\rangle_{A\cup B}$ on the disconnected cluster $A \cup B$, one-particle states $|1\rangle_{A\cup B}$ decompose into two sets of states with a particle on cluster A $(|1\rangle_A \otimes |0\rangle_B)$ and a particle on B $(|0\rangle_A \otimes |1\rangle_B)$. For two-particle excitations $|2\rangle_{A\cup B}$, there are three possibilities to distribute the particles. In general, *N*-particle states have the form $|k\rangle_A \otimes |N-k\rangle_B$ with $k \in \{0, 1, ..., N\}$, and there are N + 1 possibilities to decompose the states. When restricting Equation (51) to these *N*-quasiparticle states, a cluster-additive Hamiltonian must decompose as

$$\mathcal{H}_{\text{eff}}^{\text{cl. add.}}|_{A\cup B}^{N} = \bigoplus_{k=0}^{N} \left(\mathcal{H}_{\text{eff}}|_{A}^{k} + \mathcal{H}_{\text{eff}}|_{B}^{N-k} \right),$$
(79)

where we introduce the notation $|_{C}^{n}$ restricting the Hamiltonian to all *n*-quasiparticle states on a cluster *C*. The direct sum in Equation (79) is introduced to emphasise that, for a cluster additive Hamiltonian, there must not be any particle processes between the two disconnected clusters *A* and *B*. The Hilbert space on the disconnected cluster $A \cup B$ can be seen as the natural extension of the Hilbert spaces on cluster *A* and *B*, and we can define the operators on the clusters *A* and *B* in terms of the Hilbert space on $A \cup B$ as a tensor product:

$$\begin{aligned} \mathcal{H}_{\text{eff}}|_{A}^{k} &:= \mathcal{H}_{\text{eff}}|^{k} \otimes \mathbb{1}|^{N-k} ,\\ \mathcal{H}_{\text{eff}}|_{B}^{k} &:= \mathbb{1}|^{N-k} \otimes \mathcal{H}_{\text{eff}}|^{k} , \end{aligned}$$

$$(80)$$

where operators on the left of \otimes are defined on the Hilbert space of A and the operators to the right on the Hilbert space of B. The issue with Equation (79) is that, when we restrict the particle basis to N on the disconnected cluster $A \cup B$, there are contributions from lower particles channels coming from the N + 1 possibilities to distribute the N particles on the two clusters. For example, if we look at the one-particle space:

$$\mathcal{H}_{\mathrm{eff}}^{\mathrm{cl.\,add.}}|_{A\cup B}^{1} = \left(\mathcal{H}_{\mathrm{eff}}|_{A}^{1} + \mathcal{H}_{\mathrm{eff}}|_{B}^{0}\right) \oplus \left(\mathcal{H}_{\mathrm{eff}}|_{A}^{0} + \mathcal{H}_{\mathrm{eff}}|_{B}^{1}\right),\tag{81}$$

we see that, in addition to the one-particle contributions, we obtain additional zero-particle contributions. The left part of the direct sum stems from acting on $|1\rangle_A \otimes |0\rangle_B$ and the right side from acting on $|0\rangle_A \otimes |1\rangle_B$. There are always two possibilities to distribute the one-particle excitation on the two clusters where the other cluster is always unoccupied, which gives an additional zero-particle contribution. This fact is schematically illustrated in Figure 5a.



Figure 5. Schematic illustration of cluster additivity (**a**) and particle additivity in the one-particle basis (**b**) on a disconnected cluster $A \cup B$ (grey area) consisting of individual connected clusters A and B (yellow areas). (**a**) Cluster additivity in the one-particle basis translates to one particle being on cluster A and zero on B and vice versa. To calculate the contribution on the disconnected cluster $A \cup B$, both contributions need to be considered including the cases of zero occupancy. (**b**) Particle additivity is fulfilled when the one-particle contribution on a disconnected cluster $A \cup B$ is simply the sum of one-particle contributions on the connected clusters A and B.

This is not the desired behaviour for a linked-cluster expansion. We would like the general notion of Equation (51) to directly translate to the particle-restricted basis, which is illustrated in Figure 5b. In other words, our goal is to find a notion of (cluster) additivity in the restricted particle basis such that we can simply add up the *N*-particle contributions

of individual clusters without caring about lower particle channels. We define *particle additivity* as

$$\mathcal{H}_{\text{eff}}^{\text{add.}}|_{A\cup B}^{N} := \mathcal{H}_{\text{eff}}|_{A}^{N} \oplus \mathcal{H}_{\text{eff}}|_{B}^{N}, \tag{82}$$

where we demand the other contributions from Equation (79) to vanish. The crucial thing to notice is that irreducible operators (76) in the particle basis:

$$\mathcal{H}_{\mathrm{eff}\,N}|_{A\cup B}^{N} \equiv \mathcal{H}_{\mathrm{eff}}^{\mathrm{add}}|_{A\cup B}^{N} \tag{83}$$

are, in fact, particle-additive [154,171]. In the following, we will show that this is indeed the case. First, we remember that, for the ground-state energy, we can trivially add up the contributions. Starting from the definition for cluster additivity (79), we have

$$\begin{aligned} \mathcal{H}_{\mathrm{eff}}^{\mathrm{cl.\,add}}|_{A\cup B}^{0} &= \mathcal{H}_{\mathrm{eff}}|_{A}^{0} + \mathcal{H}_{\mathrm{eff}}|_{B}^{0} \\ &\equiv \mathcal{H}_{\mathrm{eff}}|_{A}^{0} + \mathcal{H}_{\mathrm{eff}}|_{B}^{0} \equiv \mathcal{H}_{\mathrm{eff}}^{\mathrm{add}}|_{A\cup B}^{0} . \end{aligned}$$

$$(84)$$

Second, from restricting the decomposition of Equation (75) to the N = 1 particle channel, we can express the irreducible one-particle operator as

$$\mathcal{H}_{\text{eff 1}}|_{A\cup B}^{1} = \mathcal{H}_{\text{eff}}^{\text{cl. add.}}|_{A\cup B}^{1} - \mathcal{H}_{\text{eff 0}}|_{A\cup B}^{1}.$$
(85)

We recall that, by calculating $\mathcal{H}_{eff0}|^0$, we can automatically derive \mathcal{H}_{eff0} on the entire Hilbert space, which subsequently defines $\mathcal{H}_{eff0}|^1$. Therefore, by inserting the definition for cluster additivity (79), we obtain

$$\mathcal{H}_{\mathrm{eff}}^{\mathrm{cl.\,add.\,}}|_{A\cup B}^{1} - \mathcal{H}_{\mathrm{eff\,0}}|_{A\cup B}^{1} = \left[\left(\mathcal{H}_{\mathrm{eff}}|_{A}^{1} + \mathcal{H}_{\mathrm{eff}}|_{B}^{0} \right) \oplus \left(\mathcal{H}_{\mathrm{eff}}|_{A}^{0} + \mathcal{H}_{\mathrm{eff}}|_{B}^{1} \right) \right] \\ - \left[\left(\mathcal{H}_{\mathrm{eff\,0}}|_{A}^{1} + \mathcal{H}_{\mathrm{eff\,0}}|_{B}^{0} \right) \oplus \left(\mathcal{H}_{\mathrm{eff\,0}}|_{A}^{0} + \mathcal{H}_{\mathrm{eff\,0}}|_{B}^{1} \right) \right] \\ = \left(\mathcal{H}_{\mathrm{eff\,1}}|_{A}^{1} - \mathcal{H}_{\mathrm{eff\,0}}|_{A}^{1} \right) \oplus \left(\mathcal{H}_{\mathrm{eff\,0}}|_{B}^{1} - \mathcal{H}_{\mathrm{eff\,0}}|_{B}^{1} \right) \\ = \mathcal{H}_{\mathrm{eff\,1}}|_{A}^{1} \oplus \mathcal{H}_{\mathrm{eff\,1}}|_{B}^{1} \\ \equiv \mathcal{H}_{\mathrm{eff\,1}}^{\mathrm{add.\,1}} |_{A\cup B}^{1} , \qquad (86)$$

where we used the definition of Equation (82) in the last line. Hence, we have proven Equation (83) for N < 2. The above proof can be readily extended to $N \ge 2$.

We achieved our goal of finding a notion of cluster additivity in the particle basis, which we termed particle additivity. We can determine the desired particle-additive quantities by using the subtraction scheme:

$$\mathcal{H}_{\text{eff}}^{\text{add.}}|^{N} \equiv \mathcal{H}_{\text{eff}N}|^{N} = \mathcal{H}_{\text{eff}}|^{N} - \sum_{n=0}^{N-1} \mathcal{H}_{\text{eff}n}|^{N},$$
(87)

which comes from Equation (75) by restricting it to an *N*-particle basis. This is an inductive scheme starting from N = 0 calculating the irreducible additive quantity $\mathcal{H}_{eff0}|^0$. This result can be used to calculate the subsequent irreducible additive quantity $\mathcal{H}_{eff1}|^1$ for N = 1. Then, for N = 2, we use the results from N = 0 and N = 1 to calculate $\mathcal{H}_{eff2}|^2$, and so on. Again, it is important that $\mathcal{H}_{effn}|^n$ completely defines the operator \mathcal{H}_{effn} and, therefore, any $\mathcal{H}_{effn}|^m$.

When considering effective observables, the particle number is no longer conserved and more types of processes are allowed. We need to generalise Equation (75) for effective observables by introducing an additional sum over d that is the change in the quasiparticle number. An effective observable, thus, decomposes as

$$\mathcal{O}_{\text{eff}} = \sum_{n=0}^{\infty} \sum_{d \ge -n} \mathcal{O}_{\text{eff}\,n,d} , \qquad (88)$$

where $\mathcal{O}_{\text{eff }n,d}$ are irreducible contributions [154]. When writing them in second quantisation, we have

$$\mathcal{O}_{\text{eff}\,n,d} = \sum_{i_1,\dots,i_n} \sum_{j_1,\dots,j_{n+d}} \tilde{t}_{i_1\dots,i_n;j_1,\dots,j_{n+d}} b^{\dagger}_{j_{n+d}} \dots b^{\dagger}_{j_1} b_{i_n} \dots b_{i_1} .$$
(89)

We can directly see that the d quasiparticles are created because there are d additional creation operators. When d is negative, d quasiparticles are annihilated. We can infer a notion for cluster additivity and particle additivity along the same lines:

$$\mathcal{O}_{\text{eff}}^{\text{cl. add.}}|_{A\cup B}^{N\to N+d} = \bigoplus_{k=0}^{N} \left(\mathcal{O}_{\text{eff}}|_{A}^{k\to k+d} + \mathcal{O}_{\text{eff}}|_{B}^{N-k\to N-k+d} \right), \tag{90}$$

$$\mathcal{O}_{\text{eff}}^{\text{add.}}|_{A\cup B}^{N\to N+d} = \mathcal{O}_{\text{eff}}|_{A}^{N\to N+d} \oplus \mathcal{O}_{\text{eff}}|_{B}^{N\to N+d} .$$
(91)

To determine the particle additive parts, we can use an analogue subtraction scheme as described in Equation (87), which can be denoted as

$$\mathcal{O}_{\mathrm{eff}\,N,d}|^{N\to N+d} = \mathcal{O}_{\mathrm{eff}}|^{N\to N+d} - \sum_{n=0}^{N-1} \mathcal{O}_{\mathrm{eff}\,n,d}|^{N\to N+d} \,.$$
(92)

If we want to calculate $\mathcal{O}_{\text{eff }N,d}|^{N \to N+d}$, then we have to inductively apply Equation (92).

There are several things to be noted at this point. First, not all perturbation theory methods satisfy cluster additivity (79), and in this case, we cannot write operators as a direct sum any longer. There will be quasiparticle processes between one and the other cluster changing the number of particles on each cluster [155,162]. This is sketched in Figure 6 by the presence of an additional term.



Figure 6. For operators that are not cluster additive, the contribution on the disconnected cluster $A \cup B$ originates not only from the sum of the contributions where a single particle is on *A* or *B*, but also from contributions where the particle can hop between the two connected clusters.

When falsely performing a linked-cluster expansion, it can be noticed immediately that the approach breaks down. A symptom of non-cluster additivity is the presence of contributions of lower orders than expected from the number of edges of the graph [155]. When calculating reduced contributions in a linked-cluster expansion, we subtract only contributions of connected subgraphs, which leaves non-zero contributions of disconnected clusters when the perturbation theory method is not cluster-additive. However, there are notable exceptions when a linked-cluster expansion for energies of excited states is still correct even though the perturbation theory method is not cluster-additive [163–165,183–186]. This is only possible when the considered excitation does not couple with a lower particle channel, i.e., lower lying states are described by a distinct set of quantum numbers lying in another symmetry sector [155,165,187]. For instance, consider the elementary quasiparticle excitation in a high-field expansion for the transverse-field Ising model (TFIM), then the structure of the perturbation is $T_{-2} + T_0 + T_2$, and therefore, the first excited state does not couple directly with the ground state (there is no $T_{\pm 1}$, which is due to symmetry). If one wants to draw a comparison, we can think of this as being similar to the case when calculating the excitations in the density matrix renormalisation group. It is no problem to

target an excited state if it is in a different symmetry sector than the ground state, but if it is in the same symmetry sector described by the same set of quantum numbers, then the Hamiltonian needs to be modified to project out the ground state [188].

Recently, a minimal transformation to an effective Hamiltonian was discovered that preserves cluster additivity. This method, called "projective cluster-additive transformation" [171], can be used analogously and is even more efficient for the calculation of high-order perturbative series. In this review, however, we stick to the well-established pCUT method.

4.4. Calculating Cluster Contributions

At this point, we may ask how to evaluate physical quantities on finite clusters in practice. To evaluate these quantities, we must evaluate them in the quasiparticle basis. In general, when setting up a cluster expansion, may it be non-linked or linked, it is important to subtract the contributions from all possible subclusters to prevent overcounting. Mathematically, for a quantity $\mathcal{M}(\mathcal{H}_{\text{eff}} \text{ or } \mathcal{O}_{\text{eff}})$, this can be written as

$$\overline{\mathcal{M}}|_{C}^{\cdots} = \mathcal{M}|_{C}^{\cdots} - \sum_{C' \subset C} \overline{\mathcal{M}}|_{C'}^{\cdots}, \qquad (93)$$

where the sum runs over all real subclusters C' in cluster C, and we call the resulting quantity $\overline{\mathcal{M}}|_{C}^{\cdots}$ reduced. Starting from the smallest possible cluster (e.g., a single bond between two sites), this formula can be inductively applied to determine the reduced quantity on increasingly big clusters. An essential observation to make is that reduced operators vanish on disconnected clusters by construction if the operator \mathcal{M} is additive since we subtract all contributions from individual subclusters. As the linked-cluster theorem applies, we can set up a cluster expansion:

$$\overline{\mathcal{M}}(\mathcal{L})|_{C}^{\dots} = \sum_{C \subset \mathcal{L}} \overline{\mathcal{M}}(C)|_{C}^{\dots}$$
(94)

of connected clusters, but we need to consider reduced quantities $\overline{\mathcal{M}}$ to prevent overcounting. For a light notation, we will drop the bar in the sections below as we will only consider reduced contributions on graphs anyway.

Now, we are ready to look at the problem from a more practical point of view. From the previous subsection, we know how cluster additivity translates into the particle basis and how to construct particle-additive parts, namely the irreducible quasiparticle contributions. We decompose the effective Hamiltonian into its irreducible contributions by explicitly calculating:

$$\mathcal{H}_{\rm eff\,0}|^0 = \mathcal{H}_{\rm eff\,}|^0 \tag{95}$$

$$\mathcal{H}_{\text{eff 1}}|^1 = \mathcal{H}_{\text{eff }}|^1 - \mathcal{H}_{\text{eff }0}|^1 , \qquad (96)$$

$$\mathcal{H}_{\text{eff}\,2}|^2 = \mathcal{H}_{\text{eff}\,}|^2 - \mathcal{H}_{\text{eff}\,1}|^2 - \mathcal{H}_{\text{eff}\,0}|^2 \tag{97}$$

(98)

$$\mathcal{H}_{\mathrm{eff}\,N}|^{N} = \mathcal{H}_{\mathrm{eff}}|^{N} - \sum_{n=0}^{N-1} \mathcal{H}_{\mathrm{eff}\,n}|^{N} \,.$$
(99)

Again, consider the effective Hamiltonian in the second quantisation made up of hard-core bosonic operators $b_i^{(\dagger)}$ annihilating (creating) quasiparticles and the quasiparticle counting operator $n_i = b_i^{\dagger} b_i^{}$ occurring in the unperturbed Hamiltonian \mathcal{H}_0 . We also consider a connected cluster C, and we denote n quasiparticle states on this cluster as $|n; i_1, \ldots, i_n\rangle_C$ with the quasiparticles on the sites i_1 to i_n . Note that, for multiple quasiparticle flavours or multiple sites within a lattice unit cell, this notation can be generalised to $|n; i_1, \ldots, i_n, \tau_1, \ldots, \tau_n\rangle_C$ by introducing additional indices τ_i . To lighten the notation, in the following, we stick to the former case. Let us consider the three lowest particle channels:

n = 0 We can directly calculate the ground-state energy $E_0(C)$ on a cluster *C* as it is already additive:

$$E_0(C) = \langle 0 | \mathcal{H}_{\text{eff}} | 0 \rangle_C, \tag{100}$$

as can be seen from Equation (95).

n = 1 To calculate the irreducible amplitudes $t_{i;j}^{(1)}(C)$ associated with the hopping process $b_j^{\dagger}b_i$ in \mathcal{H}_{eff1} , we need to subtract the zero-particle channel, as can be seen from Equation (96). However, we only need to subtract the ground-state energy if the hopping process is local, $b_i^{\dagger}b_i$, since the ground-state energy only contributes to diagonal processes. Thus, we calculate

$$t_{i;j}^{(1)}(C) = \langle 1; j | \mathcal{H}_{\text{eff}} | 1; i \rangle_C \qquad \text{if } i \neq j ,$$

$$t_{i;i}^{(1)}(C) = \langle 1; i | \mathcal{H}_{\text{eff}} | 1; i \rangle_C - E_0(C) \quad \text{else} .$$
(101)

n = 2 In the two-particle case, we have to distinguish between three processes: pair hoppings $(t_{i,j;k,l}^{(2)}(C) b_l^{\dagger} b_k^{\dagger} b_j b_i^{\dagger})$ with four distinct indices), correlated hoppings $(t_{i,j;k,l}^{(2)}(C) b_k^{\dagger} b_j n_i)$, and density–density interactions $(t_{i,j;i,j}^{(2)}(C) n_j n_i)$. The free quasiparticle hopping is already irreducible, and nothing has to be done, but for the correlated hopping contribution, we have to subtract the free one-particle hopping. In the case of the two-particle density–density interactions, we need to subtract the local one-particle hoppings, as well as the ground-state energy, as this process is diagonal (cf. Equation (97)). Therefore, we calculate

$$\begin{aligned} t_{i,j;k,l}^{(2)}(C) &= \langle 2;k,l | \mathcal{H}_{\text{eff}} | 2;i,j \rangle_{C} & \text{if } i \neq j \neq k \neq l , \\ t_{i,j;k,k}^{(2)}(C) &= \langle 2;i,k | \mathcal{H}_{\text{eff}} | 2;i,j \rangle_{C} - t_{j,k}^{(1)}(C) & \text{if } i \neq j \neq k , \\ t_{i,j;i,j}^{(2)}(C) &= \langle 2;i,j | \mathcal{H}_{\text{eff}} | 2;i,j \rangle_{C} - t_{i;i}^{(1)}(C) - t_{j;j}^{(1)}(C) - E_{0}(C) & \text{if } i \neq j . \end{aligned}$$
(102)

An analogous procedure can be applied for effective observables. Here, we need to determine the irreducible contributions for a fixed *d*. The subtraction scheme is given by

$$\mathcal{O}_{\text{eff }0,d}|^{0\to d} = \mathcal{O}_{\text{eff}}|^{0\to d} \tag{103}$$

$$\mathcal{O}_{\text{eff }1,d}|^{1\to 1+d} = \mathcal{O}_{\text{eff}}|^{1\to 1+d} - \mathcal{O}_{\text{eff }0,d}|^{1\to 1+d}$$
(104)

$$\mathcal{O}_{\text{eff}\,2,d}|^{2\to2+d} = \mathcal{O}_{\text{eff}}|^{2\to2+d} - \mathcal{O}_{\text{eff}\,1,d}|^{2\to2+d} - \mathcal{O}_{\text{eff}\,0,d}|^{2\to2+d} \tag{105}$$

$$\mathcal{O}_{\text{eff }N,d}|^{N \to N+d} = \mathcal{O}_{\text{eff}}|^{N \to N+d} - \sum_{n=0}^{N-1} \mathcal{O}_{\text{eff }n,d}|^{N \to N+d} .$$
(107)

For d = 0, we recover exactly the same subtraction procedure as before. It is straightforward to generalise this procedure for $d \neq 0$. Let us specifically consider the example we will encounter in the next section, when calculating correlations for the spectral weight. The effective observable is applied to the unperturbed ground state $|0\rangle$. Hence, there are only contributions out of the ground state (N = 0), and the effective observables decomposes into

$$\mathcal{O}_{\text{eff}} = \mathcal{O}_{\text{eff}\,0,0} + \mathcal{O}_{\text{eff}\,0,1} + \mathcal{O}_{\text{eff}\,0,2} + \dots \,. \tag{108}$$

Since only N = 0 processes contribute, nothing needs to be subtracted and the effective observable $\mathcal{O}_{\text{eff}}|^{0 \to d} = \mathcal{O}_{\text{eff}0,d}|^{0 \to d}$ is irreducible and already particle-additive.

Although, for these types of calculations, the lowest orders can be analytically determined by hand, the calculations usually become cumbersome quickly and must be evaluated using a computer programme to push to higher perturbative orders (in most cases, a maximal order ranging from 8 to 20 is achievable). Such a programme reads the information of the cluster (site and bond information), the "bra" and "ket" states, the coefficient lists C(m) or $\tilde{C}(m; i)$ from Equations (64) and (68) up to the desired order, the structure of the Hamiltonian \mathcal{H} , the local τ -operators from the perturbation \mathcal{V} , and if necessary, the observable. The input states, as well as the τ -operators should be efficiently implemented in the eigenbasis of \mathcal{H}_0 bitwise encoding the information as known, for instance, from exact diagonalisation. If possible, the calculation should be performed with rational coefficients for the exact representation of the perturbative series up to a desired order. The routine of the programme is then to iterate through the operator sequences from the coefficient list C or \tilde{C} and to consecutively apply the τ -operators by systematically iterating over all bonds of the cluster and calculating the action of the operator, saving the intermediate states for the action of the next operator in the sequence. As intermediate states are superpositions of basis states, they are saved in associative containers:

$$|\psi\rangle = \sum_{j} c_{j} |j\rangle , \qquad (109)$$

where $|j\rangle$ is the bit representation of a basis state. The key of the associative container is the basis state $|j\rangle$ and the associated value, the prefactor c_j . The bitwise representation of the basis states $|j\rangle$, as well as the τ -operators allow for a fast access and modification during the calculation.

So far, we have introduced the pCUT method for calculating the perturbative contributions on clusters. We demonstrated that the resulting effective quasiparticle-conserving Hamiltonian is cluster-additive and showed how to extract particle-additive irreducible contributions. We introduced a subtraction scheme for the effective Hamiltonian and observables that can be easily applied to calculate additive quantities to set up a linked-cluster expansion. Finally, we briefly discussed an efficient implementation of the pCUT method.

4.5. Energy Spectrum and Observables

Having established the basic theoretical framework for the pCUT method, we want to give a short overview of the physical quantities that are most frequently calculated with this approach. To this end, we assume that we consider a single suitably designed cluster for the calculations instead of setting up an LCE as a full graph decomposition. We will see how we can calculate the desired quantities without thinking about the abstract concepts necessary for linked-cluster expansions, and with the insights from this section, it will be easier to recognise what we are aiming at. Here, we first consider the energy spectrum of the Hamiltonian. We derive both the control-parameter susceptibility as the second derivative of the ground-state energy, as well as the elementary excitation gap from the effective one-quasiparticle Hamiltonian. Second, we consider observables. In the pCUT approach, often, spectral weights are calculated, which are of great importance for inelastic neutron scattering experiments.

4.5.1. Ground-State Energy and Elementary Excitation Gap

Following the above-described recursive scheme, we start with the zero-quasiparticle channel assuming a non-degenerate unperturbed ground state, which is the situation in all applications discussed in this review. The ground-state energy can be directly calculated from the cluster as in Equation (100). We consider a suitably designed cluster that is large enough to accommodate all fluctuations of a given maximal order \mathfrak{o}_{max} and has periodic boundary conditions to correctly account for translational invariance. We calculate $E_0 = \langle 0 | \mathcal{H}_{eff} | 0 \rangle$ and obtain a high-order perturbative series:

$$E_0 = \sum_{\mathbf{o}=0}^{\mathbf{o}_{\max}} p_{\mathbf{o}} \lambda^{\mathbf{o}}$$
(110)

in the expansion parameter λ , which is valid in the thermodynamic limit up to the given maximal order \mathfrak{o}_{max} . We can extract the ground-state energy per site by dividing E_0 by the number of sites of the cluster $\epsilon_0 = E_0/N$. By taking the second derivative, we obtain the control-parameter susceptibility:

$$\chi = -\frac{\mathrm{d}^2 \epsilon_0}{\mathrm{d}\lambda^2} \,. \tag{111}$$

We are usually interested in the quantum-critical properties of the model and, therefore, analyse the behaviour about the quantum-critical point λ_c . The control-parameter susceptibility shows the diverging power-law behaviour:

$$\chi \propto |\lambda - \lambda_c|^{-\alpha} \tag{112}$$

with the associated critical exponent α , as we know from Table 1.

Turning to the one-quasiparticle channel, we calculate the hopping amplitudes following Equation (101). Here, we use open boundary conditions, again with a cluster large enough to accommodate all fluctuations contributing to the hopping process. Note that, in our notation, we denote $t_{i;j}$ for hopping amplitudes on a graph or cluster level and the character *a* for processes in the thermodynamic limit. As for the ground-state energy, we can directly infer the contribution in the thermodynamic limit if the contributing fluctuations do not feel finite-size effects, and thus, we use a(j - i) in the following. Further, we can generalise our notation to multiple particle types or larger unit cells containing multiple sites, as mentioned earlier, by introducing additional indices ξ , τ . We denote a hopping from unit cell *i* to *j* by $\delta = j - i$ and within the unit cells from ξ to τ . We calculate $a_{\xi,\tau}(\delta) = \langle 1; j, \tau | \mathcal{H}_{eff} | 1; j - \delta, \xi \rangle$ by fixing *j*, due to translational symmetry. The effective one-quasiparticle Hamiltonian in the second quantisation then is

$$\mathcal{H}_{\text{eff}}^{\text{1qp}} := \mathcal{H}_{\text{eff}}|^1 = \mathcal{H}_{\text{eff}0}|^1 + \mathcal{H}_{\text{eff}1}|^1 = \epsilon_0 N + \sum_{j,\delta,\xi,\tau} a_{\xi,\tau}(\delta) b_{j,\tau}^{\dagger} b_{j-\delta,\xi}.$$
 (113)

Applying the Fourier transform for a discrete lattice:

$$b_{j,\tau} = \frac{1}{\sqrt{N}} \sum_{k} b_{j,\tau} \exp(\mathbf{i}kj) \qquad b_{j,\tau}^{\dagger} = \frac{1}{\sqrt{N}} \sum_{k} b_{j,\tau}^{\dagger} \exp(-\mathbf{i}kj), \tag{114}$$

we can diagonalise the resulting Hamiltonian in momentum space:

$$\mathcal{F}\left(\mathcal{H}_{\text{eff}}^{1\text{qp}}\right) = \epsilon_0 N + \sum_{k} b_k^{\dagger} \Omega(k) b_k = \sum_{k} \Omega_{m,n}(k) b_{k,m}^{\dagger} b_{k,n}$$

$$= \sum_{k} \omega_{\nu}(k) \beta_{k,\nu}^{\dagger} \beta_{k,\nu'}$$
(115)

introducing the operators $\beta_{k,\nu}^{(\dagger)}(k)$ that diagonalise the matrix $\Omega_{m,n}$. The eigenenergies $\omega_{\nu}(k)$ are the associated bands of the one-quasiparticle dispersion. In case of a trivial unit cell or a single particle flavour, the dispersion matrix becomes a scalar, and we can directly express the single-banded dispersion as

$$\omega(\mathbf{k}) = a(\mathbf{0}) + 2\sum_{\delta} a(\delta) \cos(\mathbf{k}\delta) , \qquad (116)$$

where the sum over δ is restricted to symmetry representatives and we assumed real hopping amplitudes $a(\delta)$. We determine the hopping amplitudes $a(\delta)$ as a perturbative series by performing the calculations on the properly designed cluster. Note that, even for a single cluster, we need to perform a subtraction scheme because, in order to obtain the irreducible contribution $a(\mathbf{0})$ explicitly, we need to subtract the ground-state energy

from $\langle i | \mathcal{H}_{eff} | i \rangle$. When we determine the elementary excitation gap at the minimum of the lowest band:

$$\Delta = \min_{\boldsymbol{k},\nu} \omega_{\nu}(\boldsymbol{k}) = \sum_{\boldsymbol{\mathfrak{o}}=0}^{\boldsymbol{\mathfrak{o}}_{\max}} p_{\boldsymbol{\mathfrak{o}}} \lambda^{\boldsymbol{\mathfrak{o}}} , \qquad (117)$$

we can as well extract the gap directly as such a series. The gap closing shows a powerlaw behaviour:

$$\Delta \propto |\lambda - \lambda_c|^{z\nu} \tag{118}$$

about the critical point with the critical exponent zv.

4.5.2. Spectral Properties

Neutron scattering is a powerful method resolving spatial and dynamical structures in condensed matter physics since thermal neutrons have a de Broglie wavelength of similar length scale as interatomic distances and their energy is of the same order of magnitude as typical excitations [189]. By measuring the change in momentum and kinetic energy in inelastic neutron scattering experiments determining the dynamic response, not only static properties like magnetic order can be resolved, but also dynamic properties like spin–spin correlations [190]. The dynamic response $S_{\alpha\beta}(k, \omega)$ can be determined as it is proportional to the cross-section:

$$\frac{\mathrm{d}^2\sigma}{\mathrm{d}\Omega\mathrm{d}\omega} \propto \sum_{\alpha,\beta} \mathcal{S}_{\alpha,\beta}(k,\omega) \tag{119}$$

of inelastic neutron scattering [189,190]. We follow the derivations in Refs. [155,191,192] and start with the definition of the *dynamic response*:

$$S_{\alpha,\beta}(\mathbf{k},\omega) = \frac{1}{2\pi N} \sum_{i,j} \int_{-\infty}^{\infty} \mathrm{d}t \exp\{\mathrm{i}[\omega t - \mathbf{k}(j-i)]\} \langle S_j^{\alpha}(t) S_i^{\beta}(0) \rangle_T, \qquad (120)$$

which is the space and time Fourier transform of the spin correlation function $\langle S^{\alpha}(t)S^{\beta}(0)\rangle_T$ with $\alpha, \beta \in \{x, y, z, +, -\}$ and $\langle \cdot \rangle_T$ referring to the thermal expectation value. In the limit of vanishing temperature T = 0, the expectation value simplifies to $\langle \cdot \rangle = \langle \psi_0 | \cdot | \psi_0 \rangle$ with $| \psi_0 \rangle$ being the ground state. Then, we call $S_{\alpha,\beta}(k,\omega)$ the *dynamic structure factor*. We introduce a complete set of energy eigenstates $\{ | \psi_\Lambda \rangle \}$, where Λ denotes a set of quantum numbers, for instance *n*, *k*, where *n* is the number of quasiparticles and *k* the lattice momentum. Writing the dynamic structure factor in terms of these energy eigenstates, this yields the dynamic structure factor in the spectral form as a sum:

$$S_{\alpha,\beta}(\boldsymbol{k},\omega) = \sum_{\Lambda} S^{\Lambda}_{\alpha,\beta}(\boldsymbol{k},\omega) , \qquad (121)$$

where $S^{\Lambda}_{\alpha,\beta}(k,\omega)$ is called the *exclusive structure factor* or *spectral weight* associated with the quantum numbers Λ . We insert $\mathbb{1} = \sum_{\Lambda} |\psi_{\Lambda}\rangle \langle \psi_{\Lambda}|$ into the correlation function and switch to the Heisenberg picture, where $S^{\alpha}_{j}(t) = e^{i\mathcal{H}t}S^{\alpha}_{j}(0)e^{-i\mathcal{H}t}$. The spectral weight then reads

$$S^{\Lambda}_{\alpha,\beta}(\boldsymbol{k},\omega) = \frac{1}{2\pi N} \sum_{\boldsymbol{i},\boldsymbol{j}} \int_{-\infty}^{\infty} \mathrm{d}t \exp[\mathrm{i}(\omega - E_{\Lambda} + E_{0})t] \exp[\mathrm{i}\boldsymbol{k}(\boldsymbol{j} - \boldsymbol{i})] \times \langle \psi_{0} | S^{\alpha}_{\boldsymbol{j}}(0) | \psi_{\Lambda} \rangle \langle \psi_{\Lambda} | S^{\beta}_{\boldsymbol{i}}(0) | \psi_{0} \rangle , \qquad (122)$$

which, after integrating over time *t*, yields

$$S^{\Lambda}_{\alpha,\beta}(\mathbf{k},\omega) = \delta(\omega - E_{\Lambda} + E_0) \frac{1}{N} \sum_{i,j} \langle \psi_0 | S^{\alpha}_j(0) | \psi_{\Lambda} \rangle \langle \psi_{\Lambda} | S^{\beta}_i(0) | \psi_0 \rangle \exp[\mathrm{i}\mathbf{k}(j-i)] \,. \tag{123}$$

If we consider the case $(S_i^{\alpha})^{\dagger} = S_i^{\beta}$ or some observable with $(\mathcal{O}_i)^{\dagger} = \mathcal{O}_i$, which could be a linear combination of spin operators, then the above expression further simplifies to

$$S^{\Lambda}(\mathbf{k},\omega) = \delta(\omega - E_{\Lambda} + E_0) \frac{1}{N} \left| \sum_{\mathbf{i}} \langle \psi_{\Lambda} | \mathcal{O}_{\mathbf{i}} | \psi_0 \rangle e^{-\mathbf{i}\mathbf{k}\mathbf{i}} \right|^2.$$
(124)

By the above equations, we can identify the spectral weights $S^{\Lambda}(k)$ as

$$S^{\Lambda}(k) = \frac{1}{N} \left| \sum_{i} \langle \psi_{\Lambda} | \mathcal{O}_{i} | \psi_{0} \rangle e^{-iki} \right|^{2}.$$
(125)

These quantities are usually visualised as a heat map where the dispersion $\omega(k)$ is plotted against the momentum k and the value $S^{\Lambda}(k, \omega)$, that is the intensity of the scattering signal associated with $|\psi_{\Lambda}\rangle$, is colour-coded.

In the pCUT approach, we want to reformulate the observable in terms of an effective one \mathcal{O}_{eff} . Here, we want to restrict ourselves to the one-quasiparticle spectral weight. In Ref. [191], you can also find a formulation for the two-quasiparticle case. For the 1qp spectral weight, Λ are the quantum numbers defining one-quasiparticle states, and we denote $S^{\Lambda}(k) \equiv S^{1\text{qp}}_{\tau}(k)$ in the following. Since the pCUT method is a perturbative approach, we want to reformulate the problem in the language of \mathcal{H}_0 states:

$$|\psi_0
angle = U|0
angle$$
 , (126)

$$\left|\psi_{1\rm qp}\right\rangle = U\left|1;k,\tau\right\rangle,\tag{127}$$

where we introduce the momentum states $|1; k, \tau\rangle$ with additional index τ denoting a quantum number like a flavour of the excitation or denoting a site in a unit cell. The momentum states are defined via the Fourier transform:

$$|1;k,\tau\rangle = \frac{1}{\sqrt{N}} \sum_{j} \exp(ikj) |1;j,\tau\rangle .$$
(128)

Inserting these identities, we obtain

$$S_{\tau}^{1qp}(\mathbf{k}) = \frac{1}{N} \left| \sum_{i} \langle \psi_{1qp} | \mathcal{O}_{i} | \psi_{0} \rangle \exp(-i\mathbf{k}i) \right|^{2} = \frac{1}{N} \left| \sum_{i} \langle 1; \mathbf{k}, \tau | U^{\dagger} \mathcal{O}_{i} U | 0 \rangle \exp(-i\mathbf{k}i) \right|^{2}$$

$$= \frac{1}{N^{2}} \left| \sum_{i,j} \langle 1; j, \tau | \mathcal{O}_{\text{eff},i} | 0 \rangle \exp[i\mathbf{k}(j-i)] \right|^{2},$$
(129)

where we defined $\mathcal{O}_{\text{eff},i} = U^{\dagger}\mathcal{O}_{i}U$. For a problem with translational invariance, we can fix the site *i* of the observable and introduce $\delta = j - i$, which yields

$$\mathcal{S}_{\tau}^{1\mathrm{qp}}(\boldsymbol{k}) = \frac{1}{N^2} \left| \sum_{\boldsymbol{i},\boldsymbol{\delta}} \langle 1; \boldsymbol{i} + \boldsymbol{\delta}, \tau | \mathcal{O}_{\mathrm{eff},\boldsymbol{i}} | 0 \rangle \exp(\mathrm{i}\boldsymbol{k}\boldsymbol{\delta}) \right|^2 = \left| \sum_{\boldsymbol{\delta}} \tilde{a}_{\tau}(\boldsymbol{\delta}) \exp(\mathrm{i}\boldsymbol{k}\boldsymbol{\delta}) \right|^2, \quad (130)$$

where we used $\tilde{a}_{\tau}(\delta) = \langle 1; i + \delta, \tau | \mathcal{O}_{\text{eff},i} | 0 \rangle$. Note that the form of the irreducible contribution of the effective observable is $\mathcal{O}_{\text{eff}\,0,1} = \sum_{\delta} \tilde{a}_{\tau}(\delta) b^{\dagger}_{\delta,\tau'}$ exactly as denoted above in Equation (108). As long as the processes $\tilde{a}(\delta)$ and $\tilde{a}(-\delta)$ are equivalent by means of the model symmetries, we can use $\tilde{a}(\delta) = \tilde{a}(-\delta)$ and further simplify $\mathcal{S}_{\tau}^{1\text{qp}}(k)$ to

$$S_{\tau}^{1\mathrm{qp}}(\boldsymbol{k}) = \left| \tilde{a}_{\tau}(\boldsymbol{0}) + 2\sum_{\delta} \tilde{a}_{\tau}(\delta) \cos(\boldsymbol{k}\delta) \right|^2 \equiv |s_{\tau}(\boldsymbol{k})|^2 .$$
(131)

So, in the pCUT method, we determine $\tilde{a}_{\tau}(\delta)$ as a perturbative series with which we can determine $S_{\tau}^{1qp}(\mathbf{k})$. Note that this observable is already an irreducible contribution according to Equation (92), and no subtraction scheme needs to be performed. Because we consider the contribution $\mathcal{O}_{\text{eff}}|^{0\to 1}$ of the observable, the initial particle number is n = 0, and nothing needs to be subtracted. To extract the quantum-critical behaviour, we need to evaluate the expression at the critical momentum \mathbf{k}_c as

$$S_{\tau}^{1\mathrm{qp}}(k_{c}) = \sum_{\mathfrak{o}=0}^{\mathfrak{o}_{\max}} p_{\mathfrak{o}} \lambda^{\mathfrak{o}}, \qquad (132)$$

yielding a perturbative series for this quantity as well. It shows a diverging critical behaviour that goes as

$$S_{\tau}^{1\mathrm{qp}}(\mathbf{k}_c) \propto |\lambda - \lambda_c|^{-(2-z-\eta)\nu}$$
(133)

with the associated critical exponent $(2 - z - \eta)\nu$. After determining the three quantities χ , Δ , and $S_{\tau}^{1qp}(k_c)$ described above, we can use the extrapolation techniques described in Section 4.8 to extract estimates for the critical point λ_c and the associated critical exponents. However, we continue with the description of a linked-cluster expansion as a full graph decomposition for long-range interacting systems. The next step on the way is to formally introduce graphs, discuss their generation, and the associated concept of white graphs.

4.6. White Graph Decomposition

In this section, we first give a brief introduction to graph theory, which forms the basis for understanding how linked-cluster expansions as a graph decomposition work. Then, we discuss how to generate all topologically distinct graphs up to a certain number of edges corresponding to the maximal number of active edges at a given perturbative order. We conclude this section by explaining the concept of white graphs where edge colours are ignored as a topological attribute in the classification of graphs. White graphs are essential for tackling long-range interactions, and therefore, every graph decomposition in this review is, in fact, a white graph decomposition.

4.6.1. Graph Theory

So far, we already defined clusters as a cut-out of the infinite lattice with a finite set of sites and a set of bonds connecting those sites. More generally, only considering the topology of clusters without restricting them to the geometry of lattices, we can define a *graph* as a tuple $\mathcal{G} = (\mathcal{V}_{\mathcal{G}}, \mathcal{E}_{\mathcal{G}})$ consisting of a (finite) set of vertices $\mathcal{V}_{\mathcal{G}}$ and a (finite) set of edges $\mathcal{E}_{\mathcal{G}}$. An edge $e \in \mathcal{E}_{\mathcal{G}}$ consists of a pair of vertices $\{\mu, \nu\}$, and these vertices $\mu, \nu \in \mathcal{V}_{\mathcal{G}}$ are called *adjacent*. The *degree* of a vertex is the number of edges connecting it to other vertices of the graph. In the following, we only consider *undirected*, *simple*, and *connected* graphs, which means there are neither directed edges, multiple edges between two vertices, nor loops (no edge that is joining a vertex to itself), and there always exists a path of edges connecting any two vertices of a graph [155,193,194]. As an example, we depict a graph $\mathcal{G} = (\mathcal{V}_{\mathcal{G}}, \mathcal{E}_{\mathcal{G}})$ with $\mathcal{V}_{\mathcal{G}} = \{0, 1, 2, 3, 4, 5\}$ and $\mathcal{E}_{\mathcal{G}} = \{\{0, 1\}, \{0, 2\}, \{0, 4\}, \{0, 5\}, \{1, 3\}, \{2, 3\}\}$ in Figure 7.

A subgraph \mathcal{G}' of a graph \mathcal{G} (we write $\mathcal{G}' \subset \mathcal{G}$) is defined as a subset of $\mathcal{V}'_{\mathcal{G}} \subset \mathcal{V}_{\mathcal{G}}$ and $\mathcal{E}'_{\mathcal{G}} \subset \mathcal{E}_{\mathcal{G}}$ [193–195]. We call \mathcal{G}' a proper subgraph if $\mathcal{V}'_{\mathcal{G}}$ and $\mathcal{E}'_{\mathcal{G}}$ are proper subsets of $\mathcal{V}_{\mathcal{G}}$ and $\mathcal{E}_{\mathcal{G}}$.



Figure 7. Illustration of a graph isomorphism φ_{Iso} and automorphism φ_{Auto} map for an example graph. (a) The mapping $\varphi_{Iso}(3) = 4$, $\varphi_{Iso}(4) = 3$ and identity for the remaining vertices is a graph isomorphism preserving the adjacency of vertices. If such an isomorphism exists between two graphs, they are topologically equivalent. (b) Under a graph automorphism, the edge set $\mathcal{E}_{\mathcal{G}}$ remains invariant, i.e., the graph is mapped onto itself. Here, it is exemplified for the mapping $\varphi_{Auto}(1) = 2$, $\varphi_{Auto}(2) = 1$ and identity for the remaining vertices. A graph automorphism is, therefore, a special case of a graph isomorphism, which leaves the edge set invariant.

To set up a full graph decomposition, it is essential to define how to distinguish different graphs. If there exists an isomorphism between two graphs, we call them *topologically equivalent*; otherwise, they are *topologically distinct*. A *graph isomorphism* Iso(G_1, G_2) is a bijective map φ_{Iso} between the vertex sets of two graphs, such that

$$\begin{aligned} \varphi_{\text{Iso}} : \mathcal{V}_{\mathcal{G}_1} \to \mathcal{V}_{\mathcal{G}_2} \\ \{\mu, \nu\} \in \mathcal{E}_{\mathcal{G}_1} \Leftrightarrow \{\varphi_{\text{Iso}}(\mu), \varphi_{\text{Iso}}(\nu)\} \in \mathcal{E}_{\mathcal{G}_2} . \end{aligned}$$

$$(134)$$

So, an isomorphism preserves adjacency and non-adjacency, i.e., μ and ν are adjacent if and only if the vertices $\varphi(\mu)$ and $\varphi(\nu)$ are adjacent for any vertices $\mu, \nu \in \mathcal{V}_{\mathcal{G}}$ [193,194,196]. A special case are *graph automorphisms* Auto(\mathcal{G}), which are maps of a graph on itself, so we have

$$\begin{aligned} \varphi_{\text{Auto}} : \ \mathcal{V}_{\mathcal{G}} \to \mathcal{V}_{\mathcal{G}} \\ \{\mu, \nu\} \in \mathcal{E}_{\mathcal{G}} \Leftrightarrow \{\varphi_{\text{Auto}}(\mu), \varphi_{\text{Auto}}(\nu)\} \in \mathcal{E}_{\mathcal{G}} . \end{aligned}$$

$$(135)$$

In other words, a graph automorphism is a permutation of the vertex set preserving adjacency [194,196]. The number of graph automorphisms $|\operatorname{Auto}(\mathcal{G})|$ of a given graph gives the number of its symmetries, and we call it the *symmetry number* $s_{\mathcal{G}} = |\operatorname{Auto}(\mathcal{G})|$ of a graph [155,195]. Examples of a graph isomorphism and automorphism are depicted in Figure 7. Further, if $\mathcal{G}_1 \subset \mathcal{G}_2$, the mapping (134) is injective instead of bijective such that

$$p_{\text{Mono}} : \mathcal{V}_{\mathcal{G}_1} \to \mathcal{V}_{\mathcal{G}_2} \{\mu, \nu\} \in \mathcal{E}_{\mathcal{G}_1} \Rightarrow \{\varphi_{\text{Mono}}(\mu), \varphi_{\text{Mono}}(\nu)\} \in \mathcal{E}_{\mathcal{G}_2},$$
(136)

and we call it a *subgraph isomorphism* or *monomorphism* Mono($\mathcal{G}_1, \mathcal{G}_2$) [195,197,198].

An example of a monomorphism is depicted in Figure 8a. Monomorphisms will later become of utmost importance as they give the number of embeddings of the subgraph onto a graph, which is the infinite lattice.



Figure 8. Depiction of a graph monomorphism φ_{Mono} and a graph with colour attributes. (**a**) An example showing a graph monomorphism. The smaller graph on the left is mapped onto the bigger graph on the right, which is the same graph as in Figure 7. Explicitly, the mapping φ_{Mono} is given by $\varphi_{Mono}(2) = 4$, $\varphi_{Mono}(3) = 0$, $\varphi_{Mono}(4) = 2$, and $\varphi_{Mono}(5) = 5$. (**b**) A coloured graph with the "green" colour attribute assigned to vertices 0 and 2 ($A_V = \{\{0, \text{green}\}, \{2, \text{green}\}\}$). If mappings are applied to coloured graphs, the colour set must be left invariant, i.e., vertices with a colour must be mapped onto each other and vertices with no colour as well.

To account for hopping processes during the embedding of graphs, we need to assign additional attributes to graphs like colouring their vertices [195]. Then, a coloured graph \mathcal{G}_{c} is a tuple $(\mathcal{V}_{\mathcal{G}}, \mathcal{E}_{\mathcal{G}}, \mathcal{A}_{\mathcal{V}})$, where $\mathcal{A}_{\mathcal{V}}$ is a set of pairs $\{\mu, a\}$ with $\mu \in \mathcal{V}$ and a the colour attribute. In Figure 8b, an example is depicted with $A_V = \{\{0, \text{green}\}, \{2, \text{green}\}\}$. We can extend the above definitions for isomorphisms and automorphisms and say that they must preserve the vertex colour, i.e., only vertices of the same colour can be mapped onto each other. Of course, this reduces the cardinality $|Auto(\mathcal{G}_c)|$ and, therefore, the symmetry number $s_{\mathcal{G}_c}$ associated with the coloured graph. We can later exploit the colour information for matching hopping vertices of the graph with the actual hopping of the quasiparticle on the lattice. Note that, also, colouring edges of graphs with attributes $\mathcal{A}_{\mathcal{E}}$ is useful to distinguish different types of interactions of a Hamiltonian on the graph level and very similar properties as for coloured vertices hold. As stated above, two graphs are topologically distinct if there does not exist a graph isomorphism between the two graphs. Thus, for coloured graphs, the colour attribute serves as another topological attribute. The importance of this will become apparent later in Section 4.6.3. For a more elaborate overview of graph theory in the context of linked-cluster expansions, we recommend Ref. [195].

4.6.2. Graph Generation

For the graph generation of (undirected) simple, connected graphs, we need to define an ordering between all graph isomorphs such that it is possible to pick a unique representative of all graph isomorphs that is called *canonical representative* [199]. One challenge lays in efficiently generating graphs as the number of connected graphs grows exponentially with the number of edges and the idea behind every algorithm must be to restrict the number of possible graph isomorph candidates when generating new edge sets by permutation. A well known algorithm is McKay's algorithm [199,200], which exploits the degree information of the vertices, sets up a search tree for vertices with the same degree, and uses the ordering to check if the canonical representative of the current graph already exists. We recommend using open-source libraries for the graph generation and calculation of additional symmetries. For instance, there is "*nauty*" [201] and "*networkX*" [202] or the "*Boost Graph Library*" [203].

There are various conventions for how to write the graph information to a file. One could simply save the site and edge set as lists or save its *adjacency matrix*, where the rows and columns refer to the sites and a non-zero entry in the corresponding matrix element marks the existence of an edge between the sites [155,193,194]. Here, we suggest to simply use a *bond list*. Each entry in the list contains the edge information of the graph, denoted as n_e , μ , ν , where $e \in \mathcal{E}_{\mathcal{G}}$ with μ , $\nu \in \mathcal{V}_{\mathcal{G}}$ adjacent to e and n_e is just a number associated with the edge e. The number n_e can be interpreted as a specific expansion parameter corresponding to this bond. In the simplest case, there is just one expansion parameter

and therefore a single number n_e for all edges. Assigning multiple expansion parameters becomes especially important in the next Section 4.6.3. Usually, the symmetry number s_G of a graph is calculated on the fly when generating a set of graphs and should be saved as well. In our workflow, we save the generated graphs into bond lists and create a list containing all connected graphs (e.g., the file names) along with their symmetry number s_{G} . These types of lists suffice for the calculation of the ground-state energy. When calculating 1qp irreducible processes for the dispersion or for the spectral weight observable, we can think of these processes breaking the graph symmetry. Therefore, after the graph generation, we consider all possible processes on a graph and assign colour attributes to the start and end vertices. Due to the symmetry of the processes, we assign the same colour for hoppings and distinct colours for processes of the spectral weight (When calculating the hopping processes of the effective 1qp Hamiltonian from vertex μ to ν , it has the symmetry $t_{\mu;\nu} = t_{\nu;\mu}$ with the inverse process giving the same contributions as long as the prefactors are real due to the Hermiticity of the Hamiltonian. Thus, we can use the same colours for start and end vertex. The 1qp process for the spectral weight $\tilde{t}_{\mu;\nu}$ is distinct because a quasiparticle is created at vertex μ and then subsequently hops to vertex ν . Therefore, $\tilde{t}_{\mu;\nu}$ and $\tilde{t}_{\nu;\mu}$ are, in general, not equivalent processes $\bar{t}_{\mu;\nu} \neq \bar{t}_{\nu;\mu}$, and we have to use distinct colours for the start and end vertices. But, as mentioned in Section 4.5.2, the processes can become equivalent by means of general model symmetries, for instance due to underlying graph symmetries). We calculate the symmetry number $s_{\mathcal{G}_c}$ associated with the coloured graph. In the end, we create a list, where each entry contains the graph, its symmetry number $s_{\mathcal{G}}$, a representative process (start and end vertex), and the associated symmetry number of the coloured graph s_{G_c} , counting the number of processes that give the same contributions as the representative process due to symmetries.

After generating the graphs and the lists, we can employ perturbation theory calculations on the graph level, viewing graphs as abstract clusters with vertices as lattice sites connected by bonds. A programme as described above reads in the graph and process information and repeatedly performs the pCUT calculations on every graph. The resulting graph contributions must be added up in such a way that the information of the lattice geometry is restored by weighting the graph contributions with embedding factors. That is, how many ways a graph can be fit onto a lattice apart from translational invariance (cf. Ref. [155]). For the conventional embedding of contributions from models with just nearest-neighbour interactions, the number of graphs can be reduced as the lattice geometry puts a restriction on the graphs. For example, graphs containing cycles of odd length cannot be embedded on the square lattice. In contrast, for long-range interactions, no such restriction exists, and therefore, the lattice can be seen as a single fully connected graph where every site interacts with each other. Hence, we have to generate every simple connected graph up to a given number of edges as all graphs contribute. Before we turn to the embedding problem in more detail, we will first deal with the challenges long-range interactions pose and address the problem by using white graphs.

4.6.3. White Graphs for Long-Range Interactions

In many cases in perturbation theory, there may be more types of expansion parameters, e.g., due to the geometry of the lattice, as can be seen in the *n*-leg Heisenberg ladder [152]:

$$\mathcal{H} = \mathcal{H}_0 + \lambda_\perp \mathcal{V} + \lambda_{\shortparallel} \mathcal{V} \tag{137}$$

with an expansion parameter λ_{\perp} associated with the rungs and another one λ_{\parallel} associated with the legs. The interaction \mathcal{V} is given by XY interactions, and the series expansion is performed about the Ising limit \mathcal{H}_0 . Setting up a graph decomposition for this model, the canonical approach would be to associate each expansion parameter with a distinct edge colour, blue for λ_{\perp} and purple for λ_{\parallel} .

In Figure 9a (left), all graphs with two edges and two colours (one for λ_{\perp} and one for λ_{\parallel}) are depicted on the left. It is necessary to incorporate the edge colour information as the graphs can only be embedded correctly on the infinite lattice when the edge colour

matches the bond type of the lattice. Another common type of perturbation problem is where the perturbation splits into different types of interactions, although associated with the same perturbation parameter. For instance, the problem can look like

$$\begin{aligned} \mathcal{H} &= \mathcal{H}_0 + \lambda \mathcal{V} \\ &= \mathcal{H}_0 + \lambda (\mathcal{V}_r + \mathcal{V}_r + \mathcal{V}_z) , \end{aligned}$$
 (138)

which is essentially the form of the Hamiltonian when performing a high-field series expansion [204] for Kitaev's honeycomb model [205], where each site has one x-, y-, and z-type Ising interaction bond to one of its neighbours on the honeycomb lattice, respectively. Here, we associate the three different interaction types with three types of edge colours, blue, green, and purple for x-, y-, and z-bonds. See Figure 9b (left) for an illustration of all three graphs with two edges and three colours (there are only three possibilities because colours must alternate due to the constraints posed by Kitaev's honeycomb model). In both cases, the edge colour is an additional topological attribute of the graph, leading to exponentially more graphs with the number of colours, which becomes relevant when pushing to high orders in perturbation.



Figure 9. In contrast to the conventional approach using coloured graphs (left), where different expansion parameters or different interaction types are associated with an edge colour, for white graphs (centre) the edge colour is ignored in the topological classification of graphs. Instead additional information is tracked, e.g., by associating each link with abstract expansion parameters and only substituting these abstract contribution during the embedding procedure, reintroducing the correct colour information (right), hence the name white graphs. (a) For the problem of Equation (137) on linear graphs with two edges, there are three distinct graphs as the expansion parameters λ_{\perp} and λ_{\parallel} are associated with individual edge colours (left), but there is only one white graph, as we associated one abstract expansion parameter for each edge (centre). When substituting the abstract expansion parameters with the physical one, reintroducing the correct colour, we can recover the polynomial contributions of the conventional approach (right) (cf. Ref. [152]). (b) For the problem of Equation (138) also on linear graphs with two edges, there are also three topologically distinct graphs (left), but for the white graph contribution, we have to introduce multiple abstract expansion parameters for each edge, due to the three flavours $f \in \{x, y, z\}$ (centre). The substitution works analogously to recover the polynomial contribution form the conventional approach (right). Parameters that are not explicitly set are set to zero (cf. Ref. [204]).

In the case of long-range interactions, such an approach becomes unfeasible. A Hamiltonian with long-range interactions is of the form

$$\begin{aligned} \mathcal{H} &= \mathcal{H}_0 + \lambda \sum_{\delta} \frac{1}{|\delta|^{d+\sigma}} \mathcal{V} \\ &= \mathcal{H}_0 + \sum_{\delta} \lambda_{\delta} \mathcal{V}, \end{aligned}$$
(139)
where δ is the distance between interacting sites, d the dimension of the lattice, and σ the long-range decay exponent and $\lambda_{\delta} = \lambda |\delta|^{-(d+\sigma)}$. Applying the conventional approach from above, we would associate each of the infinitely many perturbations V_{δ} with its own edge colour. The only obvious way to resolve this problem would be to truncate the sum over δ only considering very small distances. Instead, the use of *white graphs* [152] can be a solution to problems of this kind [25,29]. The idea is to change our view of how to tackle these Hamiltonians. We ignore the edge colours of the graph—significantly reducing the number of graphs for a given order—and instead, encode the colour information in the expansion parameters on the graph level in a more abstract way. This is performed by associating each edge e of the graph \mathcal{G} with a different "abstract" perturbation parameter λ_e such that we can track how often τ -operators acted on each edge e, yielding a multivariable polynomial:

$$P_{\mathcal{G}}(\{\lambda_e\}) = \sum_{m} v_m(\mathcal{G}) M_{\mathcal{G},m} = \sum_{m} v_m(\mathcal{G}) \prod_{e \in \mathcal{E}_{\mathcal{G}}} \lambda_e^{n_{e,m}} , \qquad (140)$$

with the sum over all monomial contributions. The individual monomial contributions consist of a prefactor $v_m(\mathcal{G})$, and its monomial dependency $M_{\mathcal{G},m}$. $M_{\mathcal{G},m}$ comprises a product of expansion parameters λ_e associated with edge *e* and their respective integer powers $n_{e,m} \ge 1$ tracking how often each bond was active during the calculation. Let us emphasise that we simply wrote the white graph contribution for ϵ_0 , $t_{\mu;\nu}$, or $\tilde{t}_{\mu;\nu}$ explicitly as a polynomial $P_G(\{\lambda_e\})$. It is only later during the embedding of these abstract generalised contributions when the proper link colour is reintroduced by substituting the expansion parameters by the actual expansion parameters for each realisation on the actual lattice. This is the origin of the name "white graphs" because, during the calculation on the graph, the colour of the links is unspecified (but encoded in the multivariable polynomial), and only during the embedding, the colour of the edges is reintroduced. In Figure 9, we illustrate the white graph concept in the middle. On the right, we depict how to recover the polynomial of the conventional graph contribution from the abstract white graph contribution for the models in Equations (137) and (138) by correctly substituting the abstract expansion parameters. The main difference between the model (138) compared to (137) is that, for the different interaction types, multiple parameters are associated with each edge. To account for three types of interaction flavours, we have to consider three expansion parameters $\lambda_{e,f}$ with $f \in \{x, y, z\}$ per edge *e*.

For long-range interactions, we can straightforwardly apply the exact same white graph scheme. We substitute the abstract expansion parameter with the correct algebraically decaying interaction strength depending on the distance between the interacting sites on the lattice [25,29]. For this, we have to use the substitution:

$$\lambda_e \mapsto \frac{1}{|\boldsymbol{\delta}|^{(d+\sigma)}} \tag{141}$$

with $\delta = i_{\nu} - i_{\mu}$ and $e = \{\mu, \nu\} \in \mathcal{E}_{\mathcal{G}}$. It is also possible to incorporate multiple interaction types, but the substitution for the different interaction types must then be performed before the embedding, as was done in Refs. [31,35,206].

So far, we explained how to resolve the problem of infinitely many perturbation parameters by introducing white graphs. We managed to reduce the number of expansion parameters from infinity to the number of edges, i.e., the order of the perturbation. Yet, the polynomial in Equation (140) still grows exponentially with the number of expansion parameters. Following the description in Ref. [152], we can further mitigate this issue by an efficient representation of white graph contributions. The abstract concept is to track the relevant information in a quantity M(n) as a generalised monomial with the property $M(n_1 + n_2) = M(n_1)M(n_2)$ and use n_i as an abstract parameter that encodes the tracked information. In Equation (140), the monomial quantity M(n) is just $M_{\mathcal{G},m}$, tracking how often each edge was active during the calculation by associating each edge with its own

abstract expansion parameter. We can generalise the expression of Equation (109) for states comprising additional information:

$$|\psi\rangle = \sum_{i,j} c_{i,j} M(\boldsymbol{n}_i) |j\rangle = \sum_j |j\rangle \left(\sum_i c_{i,j} M(\boldsymbol{n}_i)\right).$$
(142)

Instead of a simple superposition of states $|j\rangle$ with associated prefactor c_j as in Equation (109), we have an additional superposition over all monomials $M(n_i)$ comprising the information of all the distinct processes encoded in $M(n_i)$ leading to the state $|j\rangle$. We can make use of this factorisation property by using nested containers. The key of the outer container is the basis state $|j\rangle$, and the value contains an inner container with the monomial $M(n_i)$ as the key and the prefactor $c_{i,j}$ as the value. Thus, the action of a τ -operator on a state $|j\rangle$ can be calculated independent of the action on the monomial $M(n_i)$. For a flat container, we would have to calculate the action on the same state $|j\rangle$ multiple times. To further improve the efficiency of the calculation, we can directly encode into the programme which edge of a graph was active in a bitwise representation. Using the information of the number of edges of a graph and tracking the applied perturbative order during the calculation, we can neglect subcluster contributions on the fly and reduce the computational overhead even further. Therefore, we can directly calculate the reduced contribution on the graph without the need for explicitly subtracting subcluster contributions.

Wrapping things up, we explained how the use of white graphs can be applied to models with long-range interactions resolving the issue of infinitely many graphs or expansion parameters at any finite perturbative order. Instead of treating each perturbation of the long-range interaction as another topological attribute in the classification of graphs, we associate an abstract expansion parameter with each edge of a graph, and only during the embedding on the lattice, we substitute these expansion parameters with the actual bond information of long-range interacting sites. An efficient representation of white graphs can further help to reduce the computational overhead.

4.7. Monte Carlo Embedding of White Graphs

We discussed ways to set up a linked-cluster expansion, either by designing a single appropriately large cluster hosting all relevant fluctuations up to a given perturbative order or by setting up a linked-cluster expansion as a full graph decomposition, where the latter is the more efficient way to perform high-order series expansions. Of course, a quantity \mathcal{M} must be cluster-additive in the first place. To decompose $\mathcal{M}(\mathcal{L})$ on the lattice \mathcal{L} into many smaller subcluster contributions, we can add up the contributions:

$$\mathcal{M}(\mathcal{L}) = \sum_{C \subset \mathcal{L}} \mathcal{M}(C) , \qquad (143)$$

where $\mathcal{M}(C)$ are reduced contributions on a cluster *C* to prevent overcounting. It is not necessary to calculate the contribution of every possible cluster since many clusters have the same contribution. It suffices to calculate the contribution of a representative cluster, only containing the relevant topological information, that is a graph defined by its vertex set $\mathcal{V}_{\mathcal{G}}$ and its edge set $\mathcal{E}_{\mathcal{G}}$. We can see a graph \mathcal{G} as representing an equivalence class $[\mathcal{G}]$, whose elements are all clusters *C* realising all possible embeddings of a graph on the lattice [195]. Figuratively, all elements of the equivalence class are related by translational and rotational symmetry or are different geometric realisations on the lattice. We can now split the sum over all clusters into one sum over all possible graphs on the lattice and another one over all elements in the equivalence class:

$$\mathcal{M}(\mathcal{L}) = \sum_{C \subset \mathcal{L}} \mathcal{M}(C) = \sum_{\mathcal{G} \subset \mathcal{L}} \sum_{C \in [\mathcal{G}]} \mathcal{M}(C) , \qquad (144)$$

where it suffices to calculate $\mathcal{M}(C)$ once for all $C \in [\mathcal{G}]$ [195]. We are left counting the number of elements *C* in the equivalence class such that we can write

$$\mathcal{M}(\mathcal{L}) = \sum_{\mathcal{G} \subset \mathcal{L}} \mathcal{W}(\mathcal{G}, \mathcal{L}) \mathcal{M}(\mathcal{G}) , \qquad (145)$$

where the embedding factor $\mathcal{W}(\mathcal{G}, \mathcal{L})$ is simply a number counting the number of embeddings [195]. The important point in our line of thought is that we can calculate the quantity $\mathcal{M}(\mathcal{G})$ only once on the graph level and multiply the contribution with a weight that is the embedding factor $\mathcal W$ and sum up the resulting contribution for all graphs to obtain the desired quantity $\mathcal{M}(\mathcal{L})$ in the thermodynamic limit. We are essentially left with determining the embedding factors after calculating the graph contributions $\mathcal{M}(\mathcal{G})$. Depending on the topology of the graph, the number of possible embeddings is different and, therefore, also the embedding factor. When calculating one quasiparticle processes for the 1qp dispersion or the 1qp spectral weight, it is important that we account for the graph vertices' colour attributes for the definition of the equivalence class. If conventional graph contributions are considered, we directly obtain the correct contribution $M(\mathcal{G})$. If there are multiple physical parameters or different interaction flavours, the graph edges have to be colour matched with the bonds of the lattice. If white graphs are used, we do not have to match any colours and this can be ignored; however, the white graph contributions have to be evaluated appropriately substituting the abstract expansion parameters with the correct physical parameter for each embedding. Regarding continuative reading on the embedding, we want to point out the standard literature for linked-cluster expansions [152,155,187], as well as Ref. [195]. Nonetheless, we show in the following how the embedding procedure works for models with nearest-neighbour interactions specifically on the example of the ground-state energy and one-quasiparticle processes yielding the 1qp dispersion of the Hamiltonian. Likewise, we derive the 1qp spectral weight. With these findings, we can turn to the embedding problem of long-range interacting models and eventually describe the Monte Carlo algorithm as a solution to this problem.

4.7.1. Conventional Nearest-Neighbour Embedding

Ground-state energy: We start with the simplest case of calculating the ground-state energy on the infinite lattice \mathcal{L} . We know that \mathcal{H}_{eff} is cluster additive and that the ground-state energy is an extensive quantity, so we can directly calculate

$$E_0 = \sum_{\mathcal{G} \subset \mathcal{L}} \mathcal{W}(\mathcal{G}, \mathcal{L}) E_0(\mathcal{G}) .$$
(146)

In other words, this means we have to multiply the ground-state energy contributions on the graphs with the correct embedding factor and add up the resulting weighted contributions for every graph. For the definition of the embedding factor, we follow the formalism introduced by Ref. [195] and write

$$\mathcal{W}(\mathcal{G}, \mathcal{L}) = \frac{|\text{Mono}(\mathcal{G}, \mathcal{L})|}{|\text{Auto}(\mathcal{G})|} = \frac{1}{s_{\mathcal{G}}} |\text{Mono}(\mathcal{G}, \mathcal{L})|.$$
(147)

The embedding factor W is the number of subgraph isomorphisms (monomorphisms) of the graph G on the lattice \mathcal{L} divided by the number of graph automorphisms, i.e., the symmetry number of the graph. It is necessary to divide by the symmetry number s_G because the number of monomorphisms is in general not equal to the number of subgraphs because there may be multiple monomorphisms that map to the identical subgraph [195]. To properly account for this, we have to divide by the number of automorphisms.

In Figure 10, you see an example for the embedding problem. We can recognise the fact that there are multiple monomorphisms by the presence of arrows illustrating the

ambiguity of mapping onto a specific subgraph embedding. Further, we always have to consider reduced graph contributions subtracting all subgraph contributions as

$$E_{0}(\mathcal{G}) = \langle 0 | \mathcal{H}_{\text{eff}} | 0 \rangle - \sum_{\mathcal{G}' \subset \mathcal{G}} \mathcal{W}(\mathcal{G}', \mathcal{G}) E_{0}(\mathcal{G}') .$$
(148)

Here, $W(\mathcal{G}', \mathcal{G})$ is the embedding factor for subgraphs \mathcal{G}' with respect to the considered graph \mathcal{G} . Note again that, with an efficient white-graph implementation, reduced contributions are calculated on the fly without the need for explicit subtraction. We state the subtraction scheme for completeness. Going back to the embedding problem for the infinite lattice \mathcal{L} , the embedding factor will be extensive as the ground-state energy is extensive as well. Usually, intensive quantities such as the energy per site $\epsilon_0 = E_0/N$ are calculated, where *N* is the number of sites, and an arbitrary edge of the graph is fixed on the lattice. This then gives

$$\epsilon_0 = \sum_{\mathcal{G} \subset \mathcal{L}} w(\mathcal{G}, \mathcal{L}) E_0(\mathcal{G}), \tag{149}$$

where the normalised embedding factor is

$$w(\mathcal{G}, \mathcal{L}) = \frac{\mathcal{W}(\mathcal{G}, \mathcal{L})}{N} = \frac{q}{s_{\mathcal{G}}} |\text{Mono}(\mathcal{G}_c, \mathcal{L}_c)|,$$
(150)

denoted with a lower case w, where q is the coordination number of the lattice (the number of neighbours of any site) [155,195]. One can think about fixing sites as colouring two adjacent vertices on the graph and two adjacent sites on the lattice and considering the monomorphism with respect to the additional colour attributes (for instance, $A_{\mathcal{V}} = \{\{\mu, \text{yellow}\}, \{\nu, \text{blue}\}\})$ of the coloured graph \mathcal{G}_c and lattice \mathcal{L}_c . In Figure 10, we depict the number of embeddings for a given graph on the square lattice and show how the embedding factor is calculated for this example. Note that it would be equally valid to just fix a single site. Then, one would not have to account for the coordination number q and the number of monomorphism would be larger by a factor of q, making it computationally more expensive. In the end, we obtain the ground-state energy (per site) as a high-order perturbative series in an expansion parameter λ up to a computationally feasible maximal order \mathfrak{o}_{max} as in Equation (110).

1qp dispersion: We now turn to the one-quasiparticle channel and calculate the hopping amplitudes. For a quasiparticle hopping δ on the lattice with additional hopping within the unit cell from ξ to τ or a quasiparticle changing its flavour from ξ to τ , we need to calculate

$$a_{\xi,\tau}(\delta) = \langle 1; i + \delta, \tau | \mathcal{H}_{\text{eff}} | 1; i, \xi \rangle = \sum_{\mathcal{G} \subset \mathcal{L}} \sum_{(\mu,\nu) \in \mathcal{P}} w(\mathcal{G}_c, \mathcal{L}_c) t_{\mu;\nu}(\mathcal{G}) , \qquad (151)$$

where we fix the initial vertex μ and end vertex ν of the quasiparticle process on the graph to the initial site i and end site $i + \delta$ in real space. We can choose an arbitrary site i without loss of generality due to the translational symmetry of the models considered. (If we consider larger unit cells, we also need to fix the sites ξ , τ with the unit cells i and $i + \delta$.) The fixing of sites can be formally achieved by assigning colours to the initial and end sites on the graph and on the lattice. The second sum goes over all representative hopping processes \mathcal{P} , as described above in Section 4.6.2. The embedding factor here is similar to Equation (150) and reads

$$w(\mathcal{G}_c, \mathcal{L}_c) = \frac{s_{\mathcal{G}_c}}{s_{\mathcal{G}}} |\text{Mono}(\mathcal{G}_c, \mathcal{L}_c)|, \qquad (152)$$

where the colour attribute comes from fixing the sites to the hopping. We account for the reduced symmetry of coloured graphs stemming from the representative hopping by multiplying by the symmetry number of the coloured graph $s_{\mathcal{G}_c}$.



The embedding factor is then calculated with respect to the coloured graph G_c and lattice \mathcal{L}_c . Again, we need to be careful as we have to consider reduced and additive contributions, i.e., we have to determine

$$t_{\mu;\nu}(\mathcal{G}) = \langle 1; \nu | \mathcal{H}_{\text{eff}} | 1; \mu \rangle - \delta_{\mu,\nu} E_0(\mathcal{G}) - \sum_{\mathcal{G}' \subset \mathcal{G}} \sum_{(\mu,\nu) \in \mathcal{P}} \mathcal{W}(\mathcal{G}'_c, \mathcal{G}_c) t_{\mu;\nu}(\mathcal{G}') , \qquad (153)$$

for each contribution on a graph \mathcal{G} . After having determined the ground-state energy per site ϵ_0 and the hopping amplitudes $a_{\xi,\tau}(\delta)$ on the lattice \mathcal{L} , we can derive the effective one-quasiparticle Hamiltonian (113). As we have seen in Section 4.5, we can readily derive the one-quasiparticle gap as a series in the perturbation parameter λ as in Equation (117).

1qp spectral weight: Lastly, we can do the same for one-quasiparticle spectral weights. We calculate the process amplitudes:

$$\tilde{a}_{\tau}(\delta) = \langle 1; \delta, \tau | \mathcal{O}_{\text{eff}, i} | 0 \rangle = \sum_{\mathcal{G} \subset \mathcal{L}} \sum_{(\mu, \nu) \in \mathcal{P}} w(\mathcal{G}_c, \mathcal{L}_c) \, \tilde{t}_{\mu; \nu}(\mathcal{G}) \,.$$
(154)

Note that a process creating a quasiparticle at μ that subsequently hops to ν with the contribution $\tilde{t}_{\mu;\nu}(\mathcal{G})$ is in general distinct to the inverse process, i.e., $\tilde{t}_{\mu;\nu}(\mathcal{G}) \neq \tilde{t}_{\nu;\mu}(\mathcal{G})$. On the other hand, the 1qp hopping processes of the Hamiltonian fulfil $t_{\mu;\nu}(\mathcal{G}) = t_{\nu;\mu}(\mathcal{G})$ as long as the hopping amplitudes are real. While for hopping processes of the Hamiltonian, we can use the same colour (for start and end vertex) as a topological attribute, here, we must use two different colours, leading to a smaller symmetry number of the coloured graphs. On the graph level, only subgraph contributions must be subtracted:

$$\tilde{t}_{\mu;\nu}(\mathcal{G}) = \langle 1; \nu | \mathcal{O}_{\text{eff},\mu} | 0 \rangle - \sum_{\mathcal{G}' \subset \mathcal{G}} \sum_{(\mu,\nu) \in \mathcal{P}} \mathcal{W}(\mathcal{G}'_c, \mathcal{G}_c) t_{\mu;\nu}(\mathcal{G}') .$$
(155)

With the contributions $\tilde{a}_{\tau}(\delta)$, the spectral weight can be determined with Equation (131), and evaluating this quantity for example at the critical momentum k_c , we again obtain a series (132) in the perturbation parameter λ .

For the conventional embedding problem, we can also consider several generalisations, which we want to briefly mention. First, we could consider quasiparticle processes between different particle types. In a 1qp process, a particle flavour could change from ξ to τ and the graph contribution would be denoted by including the additional indices $t_{\mu,\xi;\nu,\tau}$ and

 $\tilde{t}_{\mu;\nu,\tau}$. The rest of the formalism is identical. Second, we may want to consider different interaction types like in Equation (138) or more expansion parameter like in Equation (137). In such a case, when considering coloured graphs, we also have to consider their coloured edges, which must be matched with the appropriate bonds on the lattice. Then, the coloured graphs \mathcal{G}_c are given by the tuple ($\mathcal{V}_{\mathcal{G}}, \mathcal{E}_{\mathcal{G}}, \mathcal{A}_{\mathcal{E}}, \mathcal{A}_{\mathcal{V}}$), where $\mathcal{A}_{\mathcal{E}}$ are the edge colour attributes and $\mathcal{A}_{\mathcal{V}}$ the vertex colour attributes. Now, every embedding factor w needs to be determined using coloured graphs with respect to $\mathcal{A}_{\mathcal{E}}$ and $\mathcal{A}_{\mathcal{V}}$. When using white graphs, the additional edge colour can be ignored and the embedding is as before; however, we need to appropriately substitute the abstract expansion parameters with the actual physical expansion parameters.

By now, we have everything together to calculate the ground-state energy, the 1qp dispersion, and the 1qp spectral weight in the thermodynamic limit from a linked-cluster expansion set up as a full graph decomposition. Therefore, we need to calculate the embedding factor, multiply them with the associated graph contribution, and add up the resulting weighted contributions. The embedding factor can be determined using available graph libraries like the *Boost Graph Library* [203] as only automorphisms and monomorphisms (with colours) need to be determined. For long-range interactions, however, we cannot just simply calculate the embedding factor because, for every graph, there are infinitely many possible embeddings, even when accounting for translational invariance.

4.7.2. Embedding for Models with Long-Range Interactions

In this section, we restrict ourselves to a single physical perturbation parameter. If we had more than one perturbation parameter, this would introduce an additional functional dependency, which we want to avoid as we perform the Monte Carlo summation. A convenient way around this is to sample the parameters and perform the Monte Carlo summation for each parameter ratio like it was performed in [31,35,206] for anisotropic XY interactions, distinct ladder interactions, and for XXZ interactions. In the following, we restrict ourselves to a single quasiparticle flavour due to simplicity and a trivial unit cell, as we have not generalised the algorithm to larger unit cells yet. The starting point to describe the embedding problem for long-range interactions is the embedding formulas given in the previous section in Equations (149), (151), and (154) for the ground-state energy per site, 1qp hopping processes, and the 1qp spectral weight, respectively, which we have to rewrite and adapt for long-range interactions.

Ground-state energy: Starting with the embedding of the ground-state energy per site (149), we can write

$$\epsilon_{0} = \sum_{\mathcal{G}} w(\mathcal{G}, \mathcal{L}) E_{0}(\mathcal{G})$$

$$= \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{\text{max}}} \sum_{\mathcal{G}} w(\mathcal{G}, \mathcal{L}) E_{0}^{(\mathfrak{o})}(\mathcal{G})$$

$$= \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{\text{max}}} \sum_{n_{s}=2}^{\mathfrak{o}} \sum_{\substack{\mathcal{G} \\ |\mathcal{V}_{\mathcal{G}}|=n_{s}}} w(\mathcal{G}, \mathcal{L}) E_{0}^{(\mathfrak{o})}(\mathcal{G}) .$$
(156)

We have replaced $\mathcal{G} \subset \mathcal{L}$ with \mathcal{G} in the sum to emphasise that the graphs are not restricted to the nearest-neighbour lattice geometry any longer due to the long-range interactions, making the lattice a fully connected graph. From the first to second line, we decomposed the ground-state energy contribution from graphs into contributions of individual orders $E_0(\mathcal{G}) = \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{max}} E_0^{(\mathfrak{o})}$. Note that the minimum order of the ground-state energy is $\mathfrak{o} = 2$. From the second to third line, we introduced a second sum over vertices n_s and restricted the sum over all graphs to a sum over graphs with a fixed number of vertices n_s . Next, we can reformulate the embedding factor in Equation (147) as

$$w(\mathcal{G}, \mathcal{L}) = \sum_{c \in \mathcal{C}} \frac{1}{s_{\mathcal{G}}}, \qquad (157)$$

where we replaced the expression for the number of monomorphisms (divided by N) with a sum over all possible configurations C. A configuration is nothing else than the current embedding of graphs. As we will see later, we can calculate the contributions of multiple graphs simultaneously as the Monte Carlo sum only depends on the number of sites (we use "site" also as a synonym for "vertex position"). When using the word configuration, we think about it as the current set of vertex positions on the lattice. The sum over all configurations comprises individual sums for each vertex over all lattice sites excluding configurations where vertices overlap, as shown in the next subsection in Equation (180). One may falsely conclude that a sum over all configurations should result in the un-normalised embedding factor \mathcal{W} , but as we will see in the following, by substituting the abstract expansion parameters with the physical long-range interactions, only the relative distance between sites is relevant for the contribution in the summand, irrespective of the absolute position of the sites on the lattice. We can also see the reason behind splitting the graph set into sets with graphs of a fixed number of vertices because we can now group all graph contributions with a given number of sites into a single integrand because for long-range interactions there are no constraints on the embeddings (except for overlaps) and the integrand only depends on the (relative) position of the vertices.. Further, the white graph contributions $E_0^{(\mathfrak{o})}(\mathcal{G})$ still need to be replaced with the correct colour, i.e., the general expansion parameters need to be substituted with the algebraically decaying long-range interaction depending on the current configuration. In reality, the contribution $E_0^{(o)}(\mathcal{G}, c)$ depends on the current configuration *c*. Thus, replacing the expression with an explicit sum is necessary as the contribution for each configuration is different and $w(\mathcal{G}, \mathcal{L})$ cannot just be a number. The substitution must look like

$$E_0^{(\mathfrak{o})}(\mathcal{G},c) := E_0^{(\mathfrak{o})}\left(\mathcal{G}; \{\lambda_e \mapsto |\delta|^{-(d+\sigma)}\}\right) = \sum_m v_m(\mathcal{G}) \prod_{e \in \mathcal{E}_{\mathcal{G}}} \frac{1}{|\delta|^{-n_{e,m}(d+\sigma)}} , \qquad (158)$$

where the index *m* of the sum runs over all monomials of the contribution $E_0^{(o)}(\mathcal{G}, c)$ and the product is over all edges $e = \{\mu, \nu\} \in \mathcal{E}_{\mathcal{G}}$ of the graph \mathcal{G} with the adjacent vertices $\mu, \nu \in \mathcal{V}_{\mathcal{G}}$. The power law in the product arises from substituting the expansion parameters $\lambda_e \mapsto |\delta|^{-\alpha}$ on the edges *e* with the appropriate algebraically decaying long-range interaction of the current embedding (cf. Equation (140)). The adjacent vertices μ, ν are embedded on the lattice sites i_{μ} and i_{ν} with the distance $\delta = i_{\nu} - i_{\mu}$, and the multiplicity $n_{e,m} \in \mathbb{N}$ comes from the power n_e of the associated expansion parameter λ_e . This way, we can reduce the many expansion parameters from the white graph contribution to a single physical perturbation parameter λ , and by reordering the expression (156) of the ground-state energy, we have

$$\epsilon_0 = \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{\max}} \sum_{n_s=2}^{\mathfrak{o}} \sum_{c \in \mathcal{C}} \sum_{\substack{\mathcal{G} \\ |\mathcal{V}_{\mathcal{G}}|=n_s}} \frac{1}{s_{\mathcal{G}}} E_0^{(\mathfrak{o})}(\mathcal{G}, c) \lambda^{\mathfrak{o}} .$$
(159)

We can define

$$f_{n_s}^{(\mathfrak{o})}(c) := \sum_{\substack{\mathcal{G} \\ |\mathcal{V}_{\mathcal{G}}|=n_s}} \frac{1}{s_{\mathcal{G}}} E_0^{(\mathfrak{o})}(\mathcal{G}, c) , \qquad (160)$$

$$S[f_{n_s}^{(\mathfrak{o})}] := \sum_{c \in \mathcal{C}} f_{n_s}^{(\mathfrak{o})}(c) , \qquad (161)$$

where $f_{n_s}^{(0)}(c)$ is the integrand function and $S[\cdot]$ denotes the associated sum over all possible embeddings on the lattice that will be evaluated using a classical Monte Carlo algorithm. Since Monte Carlo runs are usually performed for a batch of different seeds, we introduce an additional sum over seeds averaging the Monte Carlo runs, which we denote as

$$\overline{S}[\cdot] = \frac{1}{N_{\text{seeds}}} \sum_{s=1}^{N_{\text{seeds}}} S[\cdot] .$$
(162)

Eventually, this yields the expression:

$$\epsilon_0 = \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{\text{max}}} \sum_{n_s=2}^{\mathfrak{o}} \overline{S}[f_{n_s}^{(\mathfrak{o})}] \lambda^{\mathfrak{o}} .$$
(163)

To express the ground-state energy ϵ_0 as a perturbative series, we write

$$\epsilon_0 = p_0 + \sum_{\mathfrak{o}=2}^{\mathfrak{o}_{\text{max}}} p_{\mathfrak{o}} \lambda^{\mathfrak{o}} \quad \text{with} \quad p_i = \sum_{n_s=2}^{\mathfrak{o}} \overline{S}[f_{n_s}^{(\mathfrak{o})}] , \qquad (164)$$

where we have to sum up the contributions of multiple Monte Carlo runs to obtain the series prefactors p_i for i > 0. The zeroth prefactor p_0 is simply given by the ground-state energy of \mathcal{H}_0 . As a result of the Monte Carlo summation, these prefactors p_i carry an uncertainty that can be estimated from the standard deviation obtained by averaging over the results for all seeds.

1qp dispersion: We now turn to extracting the one-quasiparticle dispersion. In Equation (116), we have seen that the dispersion can be analytically determined in terms of the hopping amplitudes of Equation (151) up to some perturbative order. For nearest-neighbour interactions, it is an analytic function in k; however, for long-range interactions, there are infinitely many hoppings possible at any order, so we can neither explicitly determine the hopping amplitudes $a(\delta)$, nor is it possible to have the dispersion as an analytical function in k. This would introduce a functional dependence in the integrand that we cannot sample. Instead, we will have to evaluate the dispersion for certain values k^* to obtain an explicit series in the perturbation parameter λ . Evaluating Equation (116) at $k = k^*$ and inserting Equation (151), we can write

$$\omega(\mathbf{k} = \mathbf{k}^{\star}) = a(\mathbf{0}) + 2\sum_{\delta} a(\delta) \cos(\mathbf{k}^{\star}\delta)$$

= $\sum_{\delta} \Xi(\mathbf{k}^{\star}, \delta) a(\delta) = \sum_{\delta} \Xi(\mathbf{k}^{\star}, \delta) \sum_{\mathcal{G}} \sum_{(\mu, \nu) \in \mathcal{P}} w(\mathcal{G}_{c}, \mathcal{L}_{c}) t_{\mu;\nu}(\mathcal{G}),$ (165)

where we rewrote the sum over δ by introducing the function:

$$\Xi(k^*,\delta) = \begin{cases} 1 & \delta = \mathbf{0}, \\ 2\cos(k^*\delta) & \delta \neq \mathbf{0}. \end{cases}$$
(166)

Again, we can split $t_{\mu;\nu}$ into the contributions of individual orders $t_{\mu;\nu}^{(\mathfrak{o})}$ and split the graph set into subsets of a fixed number of sites $n_s = |\mathcal{V}_{\mathcal{G}}|$, yielding

$$\omega(\mathbf{k}^{\star}) = \sum_{\mathbf{o}=1}^{\mathbf{o}_{\max}} \sum_{n_s=2}^{\mathbf{o}+1} \sum_{\substack{\mathcal{G} \\ |\mathcal{V}_{\mathcal{G}}|=n_s}} \sum_{(\mu,\nu)\in\mathcal{P}} \sum_{\boldsymbol{\delta}} \Xi(\mathbf{k}^{\star},\boldsymbol{\delta}) w(\mathcal{G}_c,\mathcal{L}_c) t_{\mu;\nu}^{(\mathbf{o})}(\mathcal{G}) .$$
(167)

Note here that the second sum goes until $\mathfrak{o} + 1$, while for the ground-state energy, it runs only until \mathfrak{o} . The maximal number of sites at a given order is $\mathfrak{o} + 1$ because graphs with \mathfrak{o} edges can maximally have $\mathfrak{o} + 1$ sites. For the ground-state energy fluctuations, every

quasiparticle that is created has to be annihilated again, so at order o, a process can only touch maximal o - 1 edges, which restricts the sum to o sites.

Now, we argue that we can drop the sum over δ by thinking differently about this embedding problem for the dispersion. The information of the start and end vertex of the hopping process is encoded into vertex colours, and when finding the subgraph monomorphisms for the embedding on the infinite lattice \mathcal{L} , the colours of the vertices must match the coloured sites on the lattice, i.e., the hopping vertices are fixed to the hopping sites on the lattice. Since the long-range interactions allow any hopping—i.e., of any distance—at any order, it is not useful to think in this picture. Instead, we should think about the embedding problem analogous to the one for the ground-state energy, where no such hopping constraint exists and the embedding factor \mathcal{W} is simply proportional to a sum over all configurations. This is valid as we let the sum over all lattice sites and account for constraints on δ by multiplying with the symmetry number of the coloured graph $s_{\mathcal{G}_c}$. The relevant hopping information of the vertices, which was previously fixed by coloured vertices, is anyway encoded into the cosine terms. Hence, we can make the substitution:

$$\sum_{\delta} \Xi(\mathbf{k}^{\star}, \delta) w(\mathcal{G}_{c}, \mathcal{L}_{c}) = \sum_{c \in \mathcal{C}} \frac{s_{\mathcal{G}_{c}}}{s_{\mathcal{G}}} \cos(\mathbf{k}^{\star} \delta) , \qquad (168)$$

where we account for the reduced symmetry of the graph due to the hopping by multiplying with the symmetry number of the coloured graph $s_{\mathcal{G}_c}$. As before, we need to substitute the general white graph contribution with the actual algebraically decaying long-range interactions of the current embedding:

$$t_{\mu;\nu}^{(\mathfrak{o})}(\mathcal{G},c) := t_{\mu;\nu}^{(\mathfrak{o})}\big(\mathcal{G}; \{\lambda_e \mapsto |\delta|^{-\alpha}\}\big) = \sum_m v_m(\mathcal{G}) \prod_{e \in \mathcal{E}_{\mathcal{G}}} \frac{1}{|\delta|^{-n_{e,m}\alpha}} \,. \tag{169}$$

Inserting Equation (168) into Equation (167), we end up with the expression:

$$\omega(k^{\star}) = \sum_{\mathfrak{o}=1}^{\mathfrak{o}_{\max}} \sum_{n_s=2}^{\mathfrak{o}+1} \sum_{c\in\mathcal{C}} \sum_{\substack{\mathcal{G}\\|\mathcal{V}_{\mathcal{G}}|=n_s}} \sum_{\substack{(\mu,\nu)\in\mathcal{P}\\ s_{\mathcal{G}}}} \frac{s_{\mathcal{G}_c}}{s_{\mathcal{G}}} t_{\mu;\nu}^{(\mathfrak{o})}(\mathcal{G},c) \cos(k^{\star}\delta) \lambda^{\mathfrak{o}} \,.$$
(170)

For a lighter notation, we again define the integrand function and the Monte Carlo sum:

$$f_{n_s,\boldsymbol{k}^\star}^{(\boldsymbol{\mathfrak{o}})}(c) := \sum_{\substack{\mathcal{G}\\|\mathcal{V}_{\mathcal{G}}|=n_s}} \sum_{\substack{(\mu,\nu)\in\mathcal{P}}} \frac{s_{\mathcal{G}_c}}{s_{\mathcal{G}}} t_{\mu;\nu}^{(\boldsymbol{\mathfrak{o}})}(\mathcal{G},c) \cos(\boldsymbol{k}^\star \boldsymbol{\delta}) ,$$
(171)

$$S[f_{n_s,k^\star}^{(\mathfrak{o})}] := \sum_{c \in \mathcal{C}} f_{n_s,k^\star}^{(\mathfrak{o})}(c) .$$
(172)

Introducing an average over a batch of seeds for the MC sum $\overline{S}[\cdot]$, we obtain

$$\omega(\mathbf{k}^{\star}) = \sum_{\mathfrak{o}=1}^{\mathfrak{o}_{\max}} \sum_{n_s=2}^{\mathfrak{o}+1} \overline{S}[f_{n_s,\mathbf{k}^{\star}}^{(\mathfrak{o})}] \lambda^{\mathfrak{o}} .$$
(173)

The perturbative series of the dispersion evaluated at $k = k^{\star}$ can then be expressed as

$$\omega(k^{\star}) = p_0 + \sum_{\mathfrak{o}=1}^{\mathfrak{o}_{\max}} p_{\mathfrak{o}} \lambda^{\mathfrak{o}} \quad \text{with} \quad p_i = \sum_{n_s=2}^{\mathfrak{o}+1} \overline{S}[f_{n_s,k^{\star}}^{(\mathfrak{o})}] \,. \tag{174}$$

The sum prefactors p_i for i > 0 can be determined by summing up the individual contributions from the Monte Carlo runs and the prefactor p_0 is given by the energy gap of \mathcal{H}_0 .

1qp spectral weight: Lastly, the evaluation for the spectral weight observable is analogous to the 1qp dispersion. The integrand and Monte Carlo sum are defined as

$$f_{n_s,\boldsymbol{k}^{\star}}^{(\boldsymbol{\mathfrak{o}})}(c) := \sum_{\substack{\mathcal{G}\\|\mathcal{V}_{\mathcal{G}}|=n_s}} \sum_{\substack{(\mu,\nu)\in\mathcal{P}}} \frac{s_{\mathcal{G}_c}}{s_{\mathcal{G}}} \tilde{t}_{\mu;\nu}^{(\boldsymbol{\mathfrak{o}})}(\mathcal{G},c) \cos(\boldsymbol{k}^{\star}\delta) , \qquad (175)$$

$$S[f_{n_s,k^*}^{(o)}] := \sum_{c \in \mathcal{C}} f_{n_s,k^*}^{(o)}(c) , \qquad (176)$$

which we use to calculate

$$s(\boldsymbol{k}^{\star}) = p_0 + \sum_{\mathfrak{o}=1}^{\mathfrak{o}_{\max}} p_{\mathfrak{o}} \lambda^{\mathfrak{o}} \quad \text{with} \quad p_i = \sum_{n_s=2}^{\mathfrak{o}+1} \overline{S}[f_{n_s,\boldsymbol{k}^{\star}}^{(\mathfrak{o})}] \,. \tag{177}$$

We determine $s(\mathbf{k}^*)$ by again calculating the series prefactors p_i for i > 0 and then determine the 1qp spectral weight with $S^{1qp}(\mathbf{k}^*) = |s(\mathbf{k}^*)|^2$.

It should be noted that we have to perform

$$n = \frac{\mathfrak{o}_{\max}(\mathfrak{o}_{\max} - 1)}{2} \times N_{\text{seeds}}$$
(178)

Monte Carlo runs for a series of order o_{max} with N_{seeds} . This means the number of runs grows quadratically with the maximal order o_{max} .

So far, we have derived the necessary formalism for how to express the embedding problem of models with long-range interactions, but we have not talked about how to evaluate the Monte Carlo sums $S[\cdot]$ for the integrand functions f. In the next section, we investigate how to evaluate such sums by introducing a suitable Monte Carlo algorithm.

4.7.3. Monte Carlo Algorithm for the Long-Range Embedding Problem

We are left with evaluating the Monte Carlo sum $S[\cdot]$, which runs over all configurations C of graphs. The embeddings on the lattice depend only on the number of vertices of a graph G, and there is no constraint by the edge set as in the nearest-neighbour case because every site of the lattice interacts with any other site, making it a fully connected graph. The only restriction for models with long-range interactions is that the vertices of the graph are not allowed to overlap as they do not self-interact and an overlap would imply infinite interaction strength resulting from the term $|\delta|^{-(d+\sigma)}$. In conclusion, we can write the sum over all possible configurations as number-of-sites-many sums over all possible lattice positions:

$$S[f_{n_s}^{(o)}] = \sum_{c \in \mathcal{C}} f_{n_s}^{(o)}(c) = \sum_{i_1} \cdots \sum_{i_{n_s}} f_{n_s}^{(o)}(i_1, \dots, i_{n_s}), \qquad (179)$$

$$S[f_{n_{s},k^{\star}}^{(\mathfrak{o})}] = \sum_{c \in \mathcal{C}} f_{n_{s},k^{\star}}^{(\mathfrak{o})}(c) = \sum_{i_{1}} ' \cdots \sum_{i_{n_{s}}} ' f_{n_{s},k^{\star}}^{(\mathfrak{o})}(i_{1},\ldots,i_{n_{s}}) , \qquad (180)$$

where the primed sum \sum_{i} ' over vertex position *i* is the short notation for excluding overlaps with any other vertex position. For example, for the ground-state contribution with three sites on a one-dimensional chain, this sum would look like

$$S[f_{n_s}^{(\mathfrak{o})}] = \sum_{k=-\infty}^{\infty} \sum_{\substack{j=-\infty\\j\neq k}}^{\infty} \sum_{\substack{i=-\infty\\i\neq j\\i\neq k}}^{\infty} f_{n_s}^{(\mathfrak{o})}(i,j,k) .$$
(181)

Due to the overlap constraint, these are nested high-dimensional sums over the integrand functions $f_{n_s}^{(o)}$, which are in general hard to solve. The dimensionality of the MC problem is given by $d_{sum} = n_s \cdot d$ because higher dimensions d of the system introduce additional sums for each component. If we wanted to evaluate the Monte Carlo sum in two dimensions for contributions with eight sites, which already occurs in eighth-order perturbation theory, we would have to determine the integral value of 16 nested sums over the integrand function $f_8^{(8)}$. This makes it clear that the evaluation of such sums becomes challenging very quickly. In the following, we use the short notation f_{n_s} , interchangeable for both the ground-state energy integrand $f_{n_s}^{(o)}$ and the 1qp process integrands $f_{n_s,k^*}^{(o)}$.

The first approach to tackle the problem of evaluating these sums using conventional numerical integration techniques was pioneered by S. Fey and K.P. Schmidt already in 2016 [25]. They managed to successfully determine the phase diagram and critical exponents from the closing of the gap of the long-range transverse-field Ising model on a one-dimensional chain with ferro- and antiferromagnetic interactions. While they were successful with their approach over a large range of decay exponents in one dimension, the extraction of the critical properties for small decay exponents was challenging. The twodimensional problem was out of reach with this approach as the number of nested sums doubles and the sums converge significantly more slowly. Here, Monte Carlo integration came into play as it is known to be a powerful integration technique for high-dimensional problems where conventional integration fails. The reason behind the slow convergence of such high-dimensional sums is often that the configuration space where the integrand mainly contributes to the integral is significantly smaller than the entire configuration space. In 2019, Fey et al. [29] introduced a Markov chain Monte Carlo algorithm to efficiently sample the relevant configuration space. They were able to determine the quantum-critical properties of the long-range TFIM on two-dimensional lattices to even higher precision than previously for the one-dimensional chain, extending the accessible range of decay exponents without having to forfeit perturbative orders in higher dimensions.

In the following, we describe the Markov chain Monte Carlo algorithm introduced by Ref. [29] to evaluate the high-dimensional nested sums. To sample the relevant configuration space efficiently, we use importance sampling with respect to some convenient probability weight $\pi(c)$ with respect to a configuration *c* and the associated partition function $Z = \sum_{c} \pi(c)$. We can insert an identity into Equation (179) or Equation (180) and rewrite it as

$$S[f_{n_s}] = \sum_{c \in \mathcal{C}} \underbrace{\frac{\pi(c)}{Z}}_{=1} \underbrace{\frac{Z}{\pi(c)}}_{=1} f_{n_s}(c) = Z \left\langle \frac{f_{n_s}(c)}{\pi(c)} \right\rangle_{\pi},$$
(182)

where $\pi(c)/Z$ can be interpreted as the probability of being in configuration *c*. The integrand now reads as the contribution f_{n_s} (we dropped the order \mathfrak{o} and momentum k^* as indices to lighten the notation) from configuration *c* multiplied by its probability, which allows us to write the sum as the expectation value $\langle \cdot \rangle_{\pi}$ of $f_{n_s}(c)/\pi(c)$ with respect to the weight π . We later call this sum "target sum". Since the partition function *Z* is not known a priori, we also introduce a "reference sum":

$$S\left[f_{n_s}^{\text{ref}}\right] = \sum_{c \in \mathcal{C}} f_{n_s}^{\text{ref}}(c) = Z \left\langle \frac{f_{n_s}^{\text{ref}}(c)}{\pi(c)} \right\rangle_{\pi},$$
(183)

over a reference function $f_{n_s}^{\text{ref}}$. We require this sum to be analytically solvable to avoid introducing an additional source of error. We denote its analytical expression as $S_{n_s}^{\text{ref}}$. The

reference function $f_{n_s}^{\text{ref}}$ should behave similarly to the integrand function of interest f_{n_s} . This means that the reference sum and target sum should have considerable contributions in the same area of the configuration space and their asymptotic behaviour should be similar as well. Although we could make, in principle, an arbitrary choice for the reference function, the latter properties guarantee to lead to good convergence. In one dimension, we choose the reference integrand as

$$f_{n_s}^{\text{ref}}(c) = \prod_{n=1}^{n_s-1} \frac{1}{|i_{n+1} - i_n|^{\rho}} = \prod_{n=1}^{n_s-1} \frac{1}{|\delta_n|^{\rho}}$$
(184)

with $\delta_n = i_{n+1} - i_n$ and the reference integrand exponent ρ , which is a free simulation parameter. We can solve the reference sum as follows:

$$S\left[f_{n_{s}}^{\text{ref}}\right] = \sum_{c \in \mathcal{C}} \prod_{n=1}^{n_{s}-1} \frac{1}{|i_{n+1} - i_{n}|^{\rho}}$$

$$= \sum_{i_{1}}' \cdots \sum_{i_{N}}' \frac{1}{|i_{2} - i_{1}|^{\rho}} \cdots \frac{1}{|i_{n_{s}} - i_{n_{s}-1}|^{\rho}}$$

$$= n_{s}! \sum_{i_{1} < i_{2}} \cdots \sum_{i_{n_{s}-1} < i_{n_{s}}} \frac{1}{|i_{2} - i_{1}|^{\rho}} \cdots \frac{1}{|i_{n_{s}} - i_{n_{s}-1}|^{\rho}}$$

$$= n_{s}! \sum_{i_{1} < i_{2}} \cdots \sum_{i_{n_{s}-1} < i_{n_{s}}} \frac{1}{|\delta_{1}|^{\rho}} \cdots \frac{1}{|\delta_{n_{s}-1}|^{\rho}}$$

$$= n_{s}! \prod_{n=1}^{n_{s}-1} \sum_{\delta=1}^{\infty} \frac{1}{\delta^{\rho}} = n_{s}! \zeta(\rho)^{n_{s}-1},$$

(185)

where $\zeta(\rho)$ is the Riemann ζ function and we accounted for $n_s!$ possibilities to randomly embed the vertices by ordering the indices of the sums. One major difference between the reference and the target sum is that, in the target sum, many different graph contributions contribute. In fact, the reference sum above is exactly the contribution of order $\mathfrak{o} = n_s - 1$ of a chain graph with n_s vertices and the contribution from the associated target sum is the same up to a linear factor. In higher dimensions, we cannot choose a contribution proportional to the one of a chain graph any longer since it cannot be solved analytically. Instead, we make simplifications to the reference sum and require that the reference sum is still good enough to capture the same properties as the target sum. We choose to decouple the dimensions in the reference integrand:

$$f_{n_s}^{\text{ref}}(c) = \prod_{n=1}^{d} \prod_{\nu=1}^{n_s-1} \frac{1}{(1+|i_{\nu+1,n}-i_{\nu,n}|)^{\rho}} = \prod_{n=1}^{d} \prod_{\nu=1}^{n_s-1} \frac{1}{(1+|\delta_{\nu,n}|)^{\rho}}$$
(186)

and explicitly allow overlaps in the reference sum, such that it can be solved analytically as follows:

$$S\left[f_{n_{s}}^{\text{ref}}\right] = \sum_{i_{1,1}=-\infty}^{\infty} \cdots \sum_{i_{n_{s},d}=-\infty}^{\infty} \frac{1}{(1+|i_{2,1}-i_{1,1}|)^{\rho}} \cdots \frac{1}{(1+|i_{n_{s},d}-i_{n_{s}-1,d}|)^{\rho}}$$
$$= \prod_{n=1}^{d(n_{s}-1)} \sum_{\delta=-\infty}^{\infty} \frac{1}{(1+|\delta|)^{\rho}} = \prod_{n=1}^{d(n_{s}-1)} \left(2\sum_{\delta=0}^{\infty} \frac{1}{(1+\delta)^{\rho}} - 1\right)$$
$$= (2\zeta(\rho) - 1)^{d(n_{s}-1)}.$$
 (187)

Although the exponent ρ can be chosen freely, we want to achieve similar asymptotic behaviour as the target integrand; therefore, we choose $\rho = 1 + \sigma$ in one dimension as the reference sum exactly behaves like the target integrand of a chain graph. In two dimensions, we made some simplifications to the reference sum, and we have to adopt the parameter as

 $\rho = (d + \sigma)/2$ for $\sigma < 5$, $\rho = 3$ for $5 \le \sigma < 7$, and $\rho = 3.5$ for $\sigma \ge 7$. This is by no means a rigorous choice, but it empirically proved to produce good convergence [29,207]. Solving Equation (183) for *Z* and inserting it into Equation (182), we obtain

$$S[f_{n_s}] = \frac{\left\langle \frac{f_{n_s}(c)}{\pi(c)} \right\rangle_{\pi}}{\left\langle \frac{f_{n_s}^{\text{ref}}(c)}{\pi(c)} \right\rangle_{\pi}} S_{n_s}^{\text{ref}} \,.$$
(188)

We use the analytic expression of Equation (185) in 1D or Equation (187) in 2D for the reference sum $S_{n_s}^{\text{ref}}$. We got rid of the partition function *Z* and now can use this expression in our Monte Carlo run to determine the sum $S[\cdot]$ using the analytic expression of the reference sum $S_{n_s}^{\text{ref}}$ while tracking the running averages in the numerator and denominator expressions.

We are left with just one missing ingredient, which is the choice of the probability weight $\pi(c)$. For our choice to be a probability weight, it must fulfil $\pi(c) \ge 0$, and we want the weight to be the largest if both the reference and the target integrand contribute the most. An obvious choice may be the quadratic mean:

$$\pi(c) = \sqrt{\left(f_{n_s}^{\text{ref}}(c)\right)^2 + \left(f_{n_s}(c)\right)^2},$$
(189)

which is always ≥ 0 and rotationally invariant in $f_{n_s}^{\text{ref}}$ and f_{n_s} . However, we also want both quantities to contribute equally to the probability weight on average over all configurations. As the contributions of the target and reference sum may differ significantly, we introduce a factor for rescaling:

$$R = \left| \frac{S[f_{n_s}^{\text{ref}}]}{S[f_{n_s}]} \right| \,, \tag{190}$$

which can be estimated in an in-advance calibration run. We then use an adjusted probability weight:

$$\pi(c) = \sqrt{\left(f_{n_s}^{\text{ref}}(c)\right)^2 + R^2 (f_{n_s}(c))^2}$$
(191)

for the actual Monte Carlo run. The weight needs to be evaluated at every Monte Carlo step to track the running averages of the numerator and denominator in Equation (188).

Now, we have everything together to describe the actual Monte Carlo algorithm. We employ a Markov chain Monte Carlo algorithm, where we need to sample the configuration space C according to the probability weight π in Equation (191). Each configuration with a non-zero weight must be in principle accessible by the Markov chain. On the one hand, we propose a high acceptance rate of the Monte Carlo steps to sample the large configuration space efficiently, not staying in the same configuration for too long. On the other hand, we want to sample closely confined configurations with rather small distances between vertices more often than configurations that are farther apart (configurations with large distances between vertices) such that we capture the asymptotic behaviour of the model. The interaction strength decays algebraically with the distance between vertices, leading to smaller contributions for configurations in which sites are far apart. What we call a confined configuration, therefore, depends on the decay exponent of the long-range interaction σ . In the algorithm, we have the free exponent parameter ρ (for the reference sum) and γ (for probability distributions), which can be changed to tweak this behaviour, but are usually chosen similar or equal to $d + \sigma$. In two dimensions, we had to adapt the values of ρ to obtain a similar asymptotic behaviour for the reference and target sum due the approximations we made. An optimal choice of these parameters ensures fast convergence of the Monte Carlo sum. The current embedding configuration should be represented as an array container where the entries are the positions of the graph vertices. In one dimension, entries are simple integers, while in higher dimensions, the position needs to be represented as a mathematical vector. For small decay exponents σ , very large distances can

occur between vertices from time to time, which need to be squared when calculating the absolute distance. This can lead to an integer overflow, and therefore, the use of a 128-bit integer may be considered. Further, we define functions in the programme for the target integrand f_{n_s} and for the reference integrand $f_{n_s}^{\text{ref}}$, where the current configuration is passed as a parameter and the function returns the contribution from the integrand evaluated for the current configuration.

Turning back to the sampling scheme, the idea of the Markov chain is to interpret the vertex positions on the lattices as random walkers. We randomly select a graph vertex and then draw a new position from a probability distribution such that the move fulfils the detailed balance condition. In each Monte Carlo step, we perform the following two moves:

Shift move: This Monte Carlo move is implemented to introduce confined random fluctuations to the current configuration independent of the strength of the algebraically decaying long-range interactions. It is especially important for larger decay exponents σ when the configurations are much more likely to be confined. First, we randomly select a vertex $n_{sel} \in \{1, \ldots, n_s\}$ drawn from a discrete uniform distribution with $p^{sel} = 1/n_s$. Second, for the fluctuation, we draw a shift value $d^{\text{prop}} \in \{-n_s, \ldots, n_s\}$ from a discrete uniform distribution $p^{\text{shift}} = 1/(2n_s + 1)$. In one dimension, we have to draw a single time, and in higher dimensions, we draw repeatedly for each component. Subsequently, we add the shift to the position of the selected vertex and propose the position:

$$i^{\text{prop}} = i_{n_{\text{col}}} + d^{\text{prop}} . \tag{192}$$

We might have proposed a position that is already occupied by another vertex, so we have to check for overlaps. In one dimension, we reset the proposed position to the original one if there is an overlap, while in higher dimensions, we explicitly allow overlaps. As we remember from above, this distinction is also present in the reference sums in one dimension in Equation (185) compared to higher dimensions in Equation (187). If an overlap occurs in dimensions higher than one, then the target summand is explicitly set to zero such that these configurations cannot contribute (otherwise, the sum would become infinity). Then, we calculate the Metropolis acceptance probability:

$$p_{\rm acc}^{\rm shift} = \min\left(1, \frac{\pi(c^{\rm prop})}{\pi(c^{\rm curr})}\right) = \min\left(1, \frac{\sqrt{(f_{n_s}^{\rm ref}(c^{\rm prop}))^2 + R^2(f_{n_s}(c^{\rm prop}))^2}}{\sqrt{(f_{n_s}^{\rm ref}(c^{\rm curr}))^2 + R^2(f_{n_s}(c^{\rm curr}))^2}}\right),$$
(193)

by determining the probability weights π of the current and the proposed configuration. The result of the target and reference function calls should be saved into variables to prevent redundant and expensive function calls at each Monte Carlo step. Note that the transition weights $\tilde{T}(c^{\text{curr}} \rightarrow c^{\text{prop}}) = p^{\text{sel}} \times p^{\text{shift}}$ cancel out as we draw only from uniform distributions. Lastly, the minimum function is implemented by drawing a random number $y \in [0, 1)$, and we accept the proposed move if $y < p^{\text{shift}}_{\text{acc}}$ and update the current configuration. An example of such a shift move is depicted in Figure 11a.

Rift move: In contrast to the previous move, which should introduce fluctuations to the configuration independent of the current one and independent of the long-range interaction strength, "rift moves" are introduced to better capture the correct asymptotic behaviour induced by the algebraically decaying interactions. The moves are able to propose very large distances between vertices, but are also able to do the opposite, closing the "rift" between vertices when the configuration is split into essentially two clusters. At first, we select a site

 $n_{sel} \in \{1, ..., n_s - 1\}$ from the vertex set with discrete uniform probability $p^{sel} = 1/(n_s - 1)$, explicitly excluding the last site. In one dimension, we can order the vertex set such that the first vertex is the one with the smallest positional value and the last the one with the largest value, so we order by $i_n < i_m$, where n, m are vertex indices and i_n, i_m the associated sites on the lattice. The same ordering was also performed when we solved the reference sum in Equation (185). In higher dimensions, a similar ordering comes at a much higher computational cost, so we stick to the vertex numbering given by the array indices, i.e., the order is n < m. Here, it is also important that the vertex labelling of the reference sum coincides with the labelling of the chain graph. To capture the physical asymptotics of the system, we draw random values from a ζ -function distribution. In one dimension, we draw from

$$p^{\text{rift}}(r^{\text{prop}}) = \frac{(r^{\text{prop}})^{-\gamma}}{\zeta(\gamma)},$$
(194)

yielding a power-law distribution with $r^{\text{prop}} > 0$ with the free exponent parameter γ . We choose $\gamma = d + \sigma$ for obvious reasons. The distance to the next vertex is given by $r^{\text{curr}} = i_{n_{\text{sel}}+1} - i_{n_{\text{sel}}}$, and r^{prop} is the proposed distance drawn from the ζ distribution. Since we ordered by the position and only selected sites in $\{1, \ldots, n_s - 1\}$, it is sufficient to draw positive values only. We shift all indices $i_n > i_{n_{\text{sel}}}$ according to

$$i_n^{\text{prop}} = i_n + (r^{\text{prop}} - r^{\text{curr}}) .$$
(195)

In higher dimensions, we have no such ordering and, therefore, extend such a distribution to negative values (we refer to it as a "double-sided" ζ -function distribution) and draw random values from

$$p^{\text{rift}}(r^{\text{prop}}) = \frac{(1+|r^{\text{prop}}|)^{-\gamma}}{2\zeta(\gamma) - 1}$$
(196)

for each component. Note that the additional one is introduced to prevent divergence when sites overlap. After drawing the new distance r^{prop} , we shift all vertices componentwise with $n > n_{\text{sel}}$ according to

$$i_{n>n_{\text{sel}}}^{\text{prop}} = i_{n>n_{\text{sel}}} + \left(r^{\text{prop}} - r^{\text{curr}}\right).$$
(197)

The underlying idea is that, if there is a large distance between two vertices $i_{n_{sel}}$ and $i_{n_{sel}+1}$, we can close the "rift" of the entire configuration instead of introducing a new one between $i_{n_{sel}+1}$ and $i_{n_{sel}+2}$. The transition weights for this move are given by

$$\tilde{T}(c^{\text{curr}} \to c^{\text{prop}}) = p^{\text{sel}} \times p(r^{\text{new}})$$
, (198)

$$\check{r}(c^{\text{prop}} \to c^{\text{curr}}) = p^{\text{sel}} \times p(r^{\text{curr}})$$
 (199)

With these, we can calculate the Metropolis–Hastings acceptance probability in one dimension:

$$p_{\text{acc}}^{\text{rift}} = \min\left(1, \frac{\pi(c^{\text{prop}})}{\pi(c^{\text{curr}})} \frac{p(c^{\text{curr}})}{p(c^{\text{prop}})}\right)$$
$$= \min\left(1, \frac{\sqrt{(f_{n_s}^{\text{ref}}(c^{\text{prop}}))^2 + R^2(f_{n_s}(c^{\text{prop}}))^2}}{\sqrt{(f_{n_s}^{\text{ref}}(c^{\text{curr}}))^2 + R^2(f_{n_s}(c^{\text{curr}}))^2}} \frac{(r^{\text{prop}})^{\gamma}}{(r^{\text{curr}})^{\gamma}}\right)$$
(200)

and, likewise, in higher dimensions:

$$p_{\rm acc}^{\rm rift} = \min\left(1, \frac{\sqrt{(f_{n_s}^{\rm ref}(c^{\rm prop}))^2 + R^2(f_{n_s}(c^{\rm prop}))^2}}{\sqrt{(f_{n_s}^{\rm ref}(c^{\rm curr}))^2 + R^2(f_{n_s}(c^{\rm curr}))^2}} \frac{\prod_{n=1}^d (1 + |r_n^{\rm prop}|)^{\gamma}}{\prod_{n=1}^d (1 + |r_n^{\rm curr}|)^{\gamma}}\right).$$
(201)

As above, we randomly draw $y \in [0, 1)$, accept if $y < p_{acc}^{rift}$, and update the current configuration if the proposed configuration is accepted. In Figure 11b, you can find a typical rift move illustrated.

To implement the Monte Carlo algorithm, we just have to introduce a loop, where for each loop iteration we perform a Monte Carlo step consisting of those two moves.



Figure 11. Exemplary Monte Carlo moves for a linear graph on a two-dimensional square lattice. (a) During a shift move, a vertex $n_{sel} \in \{1, ..., n_s\}$ is selected randomly from a uniform distribution. Then, a shift vector d^{prop} is drawn (uniformly for each component), which moves the selected site to $i^{prop} = i_{n_{sel}} + d^{prop}$ if accepted. (b) For rift moves, a vertex is selected from $n_{sel} \in \{1, ..., n_s - 1\}$. Instead of drawing from uniform distributions, rift moves account for the correct asymptotic behaviour of the system by drawing a new distance to the next vertex from a ζ -function distribution (from a normal ζ function in one dimension and from a double-sided ζ function in higher dimension for each component). If accepted, the vertices $n > n_{sel}$ are shifted to the new position $i_{n>n_{sel}}^{prop} = i_{n>n_{sel}} + (r^{prop} - r^{curr})$.

To ensure that fluctuations to the current embedding introduce enough new configurations (shift moves) while making sure that the Monte Carlo algorithm does not spend too much time in physically unlikely configurations, the shift move probability was set to $p^{\text{shift}} = 0.7$ and the rift move probability to $p^{\text{rift}} = 0.3$ [29,207] (The "single-site rift move" was later completely replaced with the "multi-site" rift move presented in this review). This choice proved to produce good convergence behaviour over a large range of decay exponents and different models. After performing a move—accepted or not—we update the estimate for the target and reference sum in the numerator and denominator in Equation (188), respectively, and keep track of the statistics like the variance of target and reference sum. To determine the value $S[f_{n_s}]$ of the Monte Carlo run, the ratio of both quantities has to be multiplied by the analytical value of the reference sum as in Equation (188).

We can test the convergence of the MC algorithm by considering the 1qp gap $\Delta = \omega(k = 0)$ of the one-dimensional ferromagnetic long-range transverse-field Ising model (LRTFIM). We benchmark the second-order series coefficient p_2 . According to Equation (174), the coefficient from the MC approach is $p_2^{\text{MC}} = \overline{S}[f_{2,0}^{(2)}] + \overline{S}[f_{3,0}^{(2)}]$ and the exact value is given by $p_2^{\text{exact}} = 2(\zeta(2(1 + \sigma)) - \zeta(1 + \sigma)^2)$ (see Ref. [25]).

In Figure 12, we show the MC error:

$$\varepsilon_{\rm MC} = \left| \frac{p_2^{\rm exact} - p_2^{\rm MC}}{p_2^{\rm exact}} \right| \tag{202}$$

as a function of the number of steps N_{steps} for a hundred MC runs with a distinct seed for each. As can be seen from the figure, the MC error goes to zero with $N_{\text{steps}}^{-1/2}$, as generally expected for MC simulations.



Figure 12. Monte Carlo error $\varepsilon_{MC} = |p_2^{exact} - p_2^{MC}| / |p_2^{exact}|$ (coloured data points) on a log-log scale as a function of the number of steps N_{steps} . The plot shows data from a hundred 12-hour-long MC runs with a distinct seed each. The MC error ε_{MC} goes to zero following a $N_{steps}^{-1/2}$ convergence behaviour (indicated by the black line), as generally expected from an MC algorithm.

Further improvements to the algorithm can be made by recentring the configuration to the origin since the graphs on the lattice may drift away due to the random walk. Also, for integer powers occurring in the target integrand from powers of the expansion parameters, it may be better to use plain multiplication than using generic integer power functions. Most importantly, lookup tables should be used for the non-integer powers stemming from the algebraically decaying interaction strengths for distances $|i_1 - i_2| < \delta_{\max}$ within a cutoff δ_{\max} that occur most commonly during the MC run. The integrand function also depends on k as cosine terms are present. It is useful to compile the code for a desired k value, so lookup tables can be defined at compile time due to the periodicity of the cosine. For instance, in the simplest case k = 0, the cosine is always 1 and for $k = \pi$ the cosine has only values of 1 and -1.

Let us emphasise that, for the pCUT+MC approach, we need to perform individual runs for every perturbative order, for every possible number of sites at a given order, for different values of momentum k if necessary (e.g., for a dispersion), and for different seeds so that an average value can be calculated. Typical runs are performed with $N_{\text{seeds}} = 5-20$ seeds for 6–24 hours. For instance, for the long-range transverse-field Ising model, we were able to extract perturbative series up to order 10 for the 1qp spectral weight, order 11 for the 1qp gap, and order 13 for the ground-state energy. Much higher orders in perturbation are likely not feasible with the current implementation as the number of Monte Carlo runs scales quadratically with the order. In the future, to further improve the efficiency of the approach, one may come up with additional Monte Carlo moves for the

Markov chain that can change the number of sites. As a result, the algorithm would only scale linearly with the perturbative order, but potentially at the cost of slower convergence. Maybe a compromise would be a favourable option, where the lower orders that converge faster are computed using an algorithm with the additional move and higher orders with the algorithm presented above.

Another important issue worth mentioning is the fact that, so far, the above algorithm can only be applied to lattices with a trivial single-site unit cell. To generalise the algorithm to arbitrary lattice geometries in the future, we would need to introduce another Monte Carlo move. We would keep the moves introduced above for moving the vertices along the underlying Bravais lattice. A new move then changes the position within the unit cell. However, new subtleties emerge from introducing a larger unit cell, as we would have to fix two vertices within the unit cells for hopping processes, while the remaining vertices can be moved freely. A larger unit cell also means that we have to calculate the entire matrix of the dispersion, as can be seen in Equation (115). Entries of the matrix are in general complex-valued, which needs to be accounted for as well. Lastly, the behaviour of the Monte Carlo algorithm for the system is altered due to the additional Monte Carlo move, which may impact the convergence. So, the choice of the reference sum should probably be adapted as well to achieve the desired convergence. See also the discussion in Ref. [207].

4.8. Series Extrapolation

We are interested in extracting the quantum-critical properties from the perturbative series obtained from the Monte Carlo embedding. DlogPadé extrapolations are an established and powerful method that allows us to extrapolate high-order series even beyond the radius of convergence and determine the quantum-critical point and associated critical exponents. A more elaborate description of DlogPadés and its application to critical phenomena can be found in Refs. [208,209].

We have given a high-order perturbative series of a physical quantity $\kappa(\lambda)$ in the perturbation parameter λ . See Section 4.5 for typical quantities of interest. A Padé extrapolant is defined as

$$P[L,M]_{\kappa} = \frac{P_L(\lambda)}{Q_M(\lambda)} = \frac{p_0 + p_1\lambda + \dots + p_L\lambda^L}{1 + q_1\lambda + \dots + q_M\lambda^M},$$
(203)

where $p_i, q_i \in \mathbb{R}$ and the degrees L, M of the numerator polynomial $P_L(\lambda)$ and denominator polynomial $Q_M(\lambda)$ are restricted to $\mathfrak{o}_{max} = L + M$, where \mathfrak{o}_{max} is the maximal perturbative order. The coefficients p_i and q_i are fixed by a set of linear equations by cross-multiplying Equation (203) with $Q_M(\lambda)$ and requiring $\mathfrak{o}_{max} = L + M$, i.e., that all higher order terms must vanish on the left side of the equation [208]. We introduce

$$P[L, M]_{\mathcal{D}} = \frac{\mathrm{d}}{\mathrm{d}\lambda} \ln(\kappa) \tag{204}$$

as the Padé extrapolant of the logarithmic derivative of κ that must satisfy $\mathfrak{o}_{max} - 1 = L + M$, as we lose one perturbative order due to differentiation. The DlogPadé extrapolant of κ can now be defined as

$$dP[L, M]_{\kappa} = \exp\left(\int_{0}^{\lambda} P[L, M]_{\mathcal{D}} \, d\lambda'\right).$$
(205)

Given that the quantity of interest κ shows a second-order phase transition with a dominant power law $\kappa \sim |\lambda - \lambda_c|^{-\theta}$ about the critical point λ_c , we can extract the critical point λ_c and the associated critical exponent θ with DlogPadé extrapolants (Although we can extract the critical properties of a second-order quantum phase transition, we are blind to first-order phase transitions as the series expansion of a single physical quantity cannot capture level crossings of the analysed quantity at the critical point λ_c). We can determine estimates for the critical point λ_c by analysing the poles of the extrapolant $P[L, M]_{\mathcal{D}}$. We have to identify the physical pole whose position determines λ_c and exclude spurious extrapolants that have non-physical poles in the complex plane close to the real line with $\lambda < \lambda_c$. If λ_c is known, we can define biased DlogPadés by

$$P[L, M]_{\theta^{\star}} = (\lambda_c - \lambda) \frac{\mathrm{d}}{\mathrm{d}\lambda} \ln(\kappa) .$$
(206)

Here, defective extrapolants have to be removed as well by excluding all extrapolants that have poles in the vicinity of $\lambda < \lambda_c$. We can extract estimates for the critical exponent θ by calculating the residua. For an unbiased estimate, we calculate

$$\theta = \operatorname{Res} P[L, M]_{\mathcal{D}}|_{\lambda = \lambda_{c}} = \left. \frac{P_{L}(\lambda)}{\frac{\mathrm{d}}{\mathrm{d}\lambda}Q_{M}(\lambda)} \right|_{\lambda = \lambda_{c}},$$
(207)

and for the biased estimate, we calculate

$$\theta^{\star} = \operatorname{Res} P[L, M]_{\theta^{\star}}|_{\lambda = \lambda_{c}} = \left. \frac{P_{L}(\lambda)}{Q_{M}(\lambda)} \right|_{\lambda = \lambda_{c}}.$$
(208)

Biased DlogPadé extrapolants were also used in the past trying to extract the exponent of multiplicative logarithmic corrections at the upper critical dimension [25,29,31,35]. At the upper critical dimensions, there are corrections to the dominant power-law behaviour of the form

$$\kappa \sim |\lambda - \lambda_c|^{-\theta} (\ln(\lambda - \lambda_c))^{p_{\theta}} .$$
(209)

We can bias the critical point λ_c and the critical exponent θ to the known mean-field value and define the extrapolant:

$$P[L, M]_{p_{\theta}^{\star}} = -\ln(1 - \lambda/\lambda_c)[(\lambda_c - \lambda)\mathcal{D}(\lambda) - \theta], \qquad (210)$$

so we can determine the estimate p_{θ}^{\star} by calculating the residuum of the Padé extrapolant.

We take the series of mean coefficients p_i for the quantities given by Equations (164), (174), and (177) and calculate a set of extrapolants to obtain reliable estimates for the critical properties of interest. We calculate estimates that satisfy $L + M = \mathfrak{o} \leq \mathfrak{o}_{max}$. We exclude all defective extrapolants as briefly described above and arrange the remaining DlogPadés into families with L - M = const. Usually, we only allow $|L - M| \leq 3$ or $|L - M| \leq 2$ since more diagonal families are expected to converge faster to the real physical value as a function of the perturbative order $\mathfrak{o} = L + M$ [208]. We further take the mean over the highest order extrapolants of each family that has at least two or three extrapolants to obtain an estimate for the critical point and exponents. The uncertainty obtained from averaging over the extrapolants is by no means a rigorous error, but is, rather, a "subjective" measure for the uncertainty obtained from systematically analysing extrapolants [209]. From experience, the estimated critical values are relatively stable to a small uncertainty in the series, and the uncertainty obtained from averaging over the highest order extrapolants is comparatively large. Therefore, the series of mean coefficients can simply be used for extrapolating the observables as performed in all previous publications [29-31,34,35,206]. A more rigorous approach can also be applied by repeatedly applying the extrapolation scheme to the series of individual seeds and averaging the critical values afterwards.

4.9. Workflow of Series Expansion Monte Carlo Embedding

In the previous subsections, we comprehensively described the pCUT+MC method for models with long-range interactions. The approach in its entirety is quite involved. Hence, to conclude this section, we want to give a short overview of the individual steps necessary and the workflow associated.

The workflow is sketched in Figure 13.



Figure 13. The workflow of the presented pCUT+MC method consists of several steps. There are three major steps. First, there is the graph generation, which only has to be performed once. Second is the calculation of the graph contributions with the pCUT method, and third is the Monte Carlo algorithm to embed the contributions on the lattice to determine the perturbative series of the quantity of interest in the thermodynamic limit.

The approach starts with the generation of graphs with a software programme like "*nauty*" [201] and saving the graphs to bond files. The symmetry number $s_{\mathcal{G}}$ of the graph can be obtained on the fly as it is usually a by-product of searching for further graph isomorphs. For the 1qp dispersion or spectral weight, we want to calculate quasiparticle processes on these graphs; however, we do not want to calculate every process on a graph since many are related by symmetries and give the exact same contribution. Therefore, we calculate additional symmetry numbers $s_{\mathcal{G}_c}$ of coloured graphs. In a programme, we iterate over all possible vertex pairs and colours of the associated vertices (two colours for the 1qp spectral weight and one colour for 1qp processes of the Hamiltonian), choose a representative process, and save the associated vertices and the symmetry number of the coloured graph names suffices, for particle processes, we want to have a list that contains the graph name, symmetry number, start and end vertex of the process, and a symmetry number associated with the process counting the number of equivalent processes. On the upside, it suffices to generate these lists and graphs once.

The second step is to iterate over the list entries, read the graph as a cluster, read the associated input states, and apply the pCUT method. Of course, such a programme must also read the pCUT coefficients C(m) or $\tilde{C}(m; i)$ for observables and the model information, i.e., the τ -operators. The programme iterates over the different operator sequences, through the operators from each sequence, systematically iterates over all edges of a graph, and applies the τ -operators. We additionally associate each edge of a graph with its own expansion parameter, such that we obtain the white graph contributions.

Having calculated the white graph contributions over every representative process for every graph, we need to convert the white graph contributions to an integrand function—also embeddable function—which is a callable function in our Monte Carlo algorithm. We have a programme that substitutes the expansion parameter with the algebraic decay expression $(\lambda_e \mapsto |\delta|^{-d+\sigma})$. This must be an expression in the programming language of choice, such that the Monte Carlo programme can call this function. Further, we multiply the white graph contribution with the symmetry factor $s_{\mathcal{G}_c}/s_{\mathcal{G}}$ for 1qp processes and with $1/s_{\mathcal{G}}$ for the ground-state energy to properly account for the symmetries during the embedding. Also, all white graph contributions of a given order and number of vertices are grouped into one such function. The functions are saved in a file and included in the Monte Carlo algorithm code. If the white graph contributions are associated with models with different interaction flavours, we must sample for certain parameter values during this step.

The physical parameters are also fixed like the momentum k and the decay exponent $d + \sigma$, and of course, the code must be adapted for different lattice geometries. There are also simulation parameters like the number of seeds, the ratio between shift and rift moves, and the simulation exponents ρ of the reference sum and γ for the rift move ζ -function distribution. The Monte Carlo algorithm is then executed for each embeddable function with a fixed order, number of sites, and seed, yielding the target values of Equation (188).

We average the target values for multiple runs with different seeds, add them up according to Equations (164), (174), and (177), and obtain a perturbative series in λ . Afterwards, we employ DlogPadé extrapolation to extract critical quantities of interest like the critical point λ_c and the associated critical exponent α for the ground-state energy, zv for the 1qp gap, or $(2 - z - \eta)v$ for the 1qp spectral weight. We can also use the extrapolations to construct the dispersion close to the critical point.

5. Stochastic Series Expansion Quantum Monte Carlo

In this section, we discuss the method of stochastic series expansion (SSE) QMC. This class of QMC algorithms is closely related to path integral (PI) QMC and samples configurations according to the Boltzmann distribution of a quantum mechanical Hamiltonian. This sampling is achieved by extending the configuration space in the imaginary-time direction by operator sequences. The objective is to evaluate thermal expectation values for operators at a finite temperature on a finite system.

The canonical partition function of a system with a quantum mechanical Hamiltonian \mathcal{H} can be expressed as

$$Z = \text{Tr}\{\exp(-\beta\mathcal{H})\} = \sum_{|\alpha\rangle} \langle \alpha | \exp(-\beta\mathcal{H}) | \alpha \rangle$$
(211)

with β being the inverse temperature and the sum over an arbitrary orthonormal basis $\{|\alpha\rangle\}$. The task is to bring Equation (211) into the form of

$$Z = \sum_{\omega \in \mathcal{C}} \pi(\omega)$$
(212)

with all weights $\pi(\omega)$ required to be non-negative.

Of course, for a Hamiltonian that is traced over in its eigenbasis (or a classical system), Equation (211) is already in the form of Equation (212), and the system can be directly sampled by a Metropolis–Hastings algorithm (see Equation (46)). For a general quantum mechanical problem, we do not have access to the eigenstates of a system and require a reformulation of Equation (211).

The SSE QMC idea resolves this issue in the following way: Given a Hamiltonian \mathcal{H} , a computational orthonormal basis $\{|\alpha\rangle\}$ is chosen in which the trace is evaluated. Further, there should exist a decomposition of the Hamiltonian:

$$\mathcal{H} = -\sum_{i} \mathcal{H}_{i} \tag{213}$$

into operators \mathcal{H}_i . \mathcal{H}_i and $\{|\alpha\rangle\}$ are chosen such that the following two conditions are met:

No-branching rule:

$$\mathcal{H}_{i}|\beta\rangle \propto |\gamma\rangle \in \{|\alpha\rangle\} \qquad \forall \mathcal{H}_{i} \quad \forall |\beta\rangle \in \{|\alpha\rangle\}, \tag{214}$$

ensuring that no superpositions of basis states are created by acting with H_i .

Non-negative real matrix elements in the computational basis:

$$\langle \beta | \mathcal{H}_i | \gamma \rangle \ge 0 \qquad \forall \mathcal{H}_i \quad \forall | \beta \rangle, | \gamma \rangle \in \{ | \alpha \rangle \}.$$
 (215)

The second condition is not strictly necessary, but makes sure that no sign problem arises, which would lead to exponentially hard computational complexity. In general, it is not necessarily possible to find a computational basis in which this condition can be fulfilled for all Hamiltonians. However, if the negative matrix elements contribute in such a way that they always occur in pairs and the minus signs cancel, the condition can be relaxed without inducing a sign problem. We will encounter this case for the antiferromagnetic Heisenberg models in Section 5.2.

For operators \mathcal{H}_i that are diagonal in the computational basis, the conditions (214) and (215) never pose a problem as diagonal operators intrinsically obey the no-branching rule and can always be made non-negative by adding a suitable constant to the Hamiltonian. The main difficulty is to find a computational basis in which the off-diagonal matrix elements are non-negative, which is not necessarily possible, as mentioned above. In particular, for fermionic or frustrated systems, negative signs typically occur.

In order to reformulate Equation (211) in the form of Equation (212), a high-temperature expansion for the partition function is performed:

$$Z = \operatorname{Tr}\{\exp(-\beta\mathcal{H})\} = \sum_{|\alpha\rangle} \langle \alpha | \exp(-\beta\mathcal{H}) | \alpha \rangle$$
(216)

$$=\sum_{|\alpha\rangle}\sum_{n=0}^{\infty}\frac{\beta^{n}}{n!}\langle\alpha|(-\mathcal{H})^{n}|\alpha\rangle$$
(217)

$$=\sum_{|\alpha\rangle}\sum_{n=0}^{\infty}\frac{\beta^n}{n!}\langle\alpha|(\sum_i\mathcal{H}_i)^n|\alpha\rangle.$$
(218)

In general, the evaluation of $\langle \alpha | (\sum_i \mathcal{H}_i)^n | \alpha \rangle$ is not feasible. The way the SSE tackles this expression is by expanding the product of sums as

$$\left(\sum_{i} \mathcal{H}_{i}\right)^{n} = \sum_{S_{n}} \prod_{k=1}^{n} \mathcal{H}_{i(k)}$$
(219)

as a sum over all occurring operator sequences S_n resulting from the exponentiation. The additional dimension created by the operator sequence is usually referred to as imaginary time in analogy to path-integral formulations. Inserting Equation (219) into Equation (218), one obtains

$$Z = \sum_{|\alpha\rangle} \sum_{n=0}^{\infty} \sum_{S_n} \frac{\beta^n}{n!} \langle \alpha | \prod_{k=1}^n \mathcal{H}_{i(k)} | \alpha \rangle .$$
(220)

Note that each of the summands in Equation (220) is non-negative by design due to the condition (215) and can be interpreted as the relative weight of a configuration. By comparing Equation (220) with Equation (212), we see that it is of a suitable form for a Markov chain Monte Carlo sampling. We identify the direct product of the set of all basis states $\{|\alpha\rangle\}$ with the set of all sequences as the configuration space:

$$\mathcal{C} = \{ |\alpha\rangle \} \times \bigcup_{n=0}^{\infty} \{ S_n \} .$$
(221)

The weight of a configuration is given by

$$\pi(|\alpha\rangle, S_n; \beta) = \frac{\beta^n}{n!} \langle \alpha | \prod_{k=1}^n \mathcal{H}_{i(k)} | \alpha \rangle .$$
(222)

In the next step, we discuss the structure of each configuration consisting of a computational basis state $|\alpha\rangle$ and an operator sequence S_n . Regarding Equation (220), we stress that the action of the product of operators in the sequence onto the basis state is crucial. Due to the no-branching rule (see Equation (214)), the action of $\prod_{k=1}^{n} \mathcal{H}_{i(k)}$ onto the basis state can be interpreted as a discrete propagation of the state α in imaginary time according to the operators in the sequence. We define the state at propagation index $p \in \{0, ..., n\}$ to be

$$\alpha(p)\rangle = \prod_{k=1}^{p} \mathcal{H}_{i(k)} |\alpha\rangle .$$
(223)

Due to the periodic boundary condition of the trace and the orthogonality of the basis states $\{|\alpha\rangle\}$, only sequences for which $|\alpha(n)\rangle = |\alpha(0)\rangle$ have a non-zero weight (see Equation (222)).

At this point, we can demonstrate why it does not matter if some matrix elements are negative in some instances, e.g., for some antiferromagnetic spin models on bipartite lattices. If a matrix element of an operator is negative, but it is ensured by the Hamiltonian and the periodicity of the trace that there is always an even number of negative matrix elements in a sequence with non-vanishing weight, then the definition of weights as in Equation (222) is nevertheless possible.

In the discussion so far, we considered sequences of all possible lengths. In order to formulate algorithms sampling the configuration space efficiently, a scheme with a fixed sequence length \mathcal{L} can be introduced, in which all sequences with $n < \mathcal{L}$ are padded with identity operators 1 to length \mathcal{L} and all sequences $n > \mathcal{L}$ are discarded. The physical justification to discard all sequences above a certain fixed length \mathcal{L} is that they are exponentially suppressed for sufficiently large \mathcal{L} . In short, this is the case because the mean operator number $\langle n \rangle$ is proportional to the mean energy $\langle \mathcal{H} \rangle$ and the variance is related to the specific heat in the following way:

$$\langle \mathcal{H} \rangle = -\frac{\langle n \rangle}{\beta} \qquad \qquad C = \langle n^2 \rangle - \langle n \rangle^2 - \langle n \rangle .$$
 (224)

A derivation and discussion of these statements can be found in Refs. [39,210–213]. From Equation (224), we can infer that the infinite sum over all sequence lengths can be truncated at a finite $\mathcal{L} \propto \beta N$. From rearranging Equation (224) and using the extensivity of the mean energy, the mean sequence length scales as $\langle n \rangle \propto \beta N$ proportional to the inverse temperature and system size N. As the mean sequence length has a finite value, the idea is to choose an $\mathcal{L} > \langle n \rangle$ great enough to be able to sample all but a negligible amount of operator sequences. Further, we will argue that the introduction of a large enough cutoff results in an exponentially small and negligible error. In the limit of $\beta \rightarrow \infty$, the specific heat has to vanish. Therefore, the variance of the mean sequence length is proportional to $\langle n \rangle$. From this, it is concluded that the weights of sequences vanish exponentially for large enough sequence lengths n [212].

We, therefore, introduce a large enough cutoff for the sequence length \mathcal{L} and consider operator sequences with a fixed length \mathcal{L} . The expression for the partition function can then be written as

$$Z \approx \sum_{S_{\mathcal{L}}} \sum_{|\alpha\rangle} \frac{\beta^{n} (\mathcal{L} - n)!}{\mathcal{L}!} \langle \alpha | \prod_{k=1}^{\mathcal{L}} \mathcal{H}_{i(k)} | \alpha \rangle .$$
(225)

The new configuration space includes all sequences of length \mathcal{L} where the shorter sequences are padded by inserting unity operators. The random insertion of unity operators into a sequence of $n < \mathcal{L}$ non-trivial operators results in $\binom{\mathcal{L}}{n} = \frac{\mathcal{L}!}{n!(\mathcal{L}-n)!}$ sequences of length \mathcal{L} . The modified prefactor in Equation (225) accounts for this overcounting. Although Equation (225) is, strictly speaking, an approximation to the partition function, we want to note that, in practice, this does not cause a systematic error as \mathcal{L} can be chosen large enough in a dynamical fashion such that, during the finite simulation time, no sequence of length $n > \mathcal{L}$ would occur. Therefore, we will not consider the fixed-length scheme as an approximation below.

To summarise: Following the argumentation discussed in this section, it is possible to bring any partition function into the form of Equation (225), whereby all the summands are non-negative if a suitable decomposition and computational basis can be found. The next step is to implement a Markov chain MC sampling on the configuration space. As the configuration space largely depends on the model, the sampling is performed in a model-dependent way. In Section 5.1, we will introduce an algorithm to sample the (LR)TFIM. In Section 5.2, we will further describe an algorithm to sample unfrustrated (long-range) Heisenberg models. In Section 5.3, the measurement of a variety of operators is discussed. Further, we will discuss how to sample systems at effectively zero temperature in Section 5.4 as this review considers the zero-temperature physics of quantum phase transitions. In Section 5.5, we give a short overview of other MC algorithms for long-range models based on path integrals.

5.1. Algorithm for Arbitrary Transverse-Field Ising Models

In this section, we describe the SSE algorithm to sample arbitrary transverse-field Ising models (TFIMs) of the form

$$H = \sum_{i \neq j} J_{i,j} \sigma_i^z \sigma_j^z - \sum_i h_i \sigma_i^x$$
(226)

as introduced by A. Sandvik in Ref. [39]. The Pauli matrices σ_i^{κ} with $\kappa \in \{x, z\}$ describe N spins 1/2 located on the lattice sites i, j. The transverse-field strength at a lattice site i is $h_i > 0$, and the Ising couplings between sites i and j have the strength $J_{i,j} \in \mathbb{R}$. For $J_{i,j} > 0$, the interaction is antiferromagnetic and it is energetically favourable for the spins to anti-align, while for $J_{i,j} < 0$, the interaction is ferromagnetic and it is energetically favourable for the spins to align.

Choosing the σ^z -eigenbasis $\{|\alpha\rangle\} = \{|\sigma_1^z, \dots, \sigma_N^z\rangle\}$ for the SSE formulation avoids the sign problem for arbitrary Ising interactions. The Hamiltonian is decomposed using the following operators:

$$\mathcal{H}_{0,0} = \mathbb{1} \tag{227}$$

$$\mathcal{H}_{i,0} = h_i \sigma_i^x \quad i > 0 \tag{228}$$

$$\mathcal{H}_{i,i} = h_i \mathbb{1} \tag{229}$$

$$\mathcal{H}_{i,j} = |J_{i,j}| - J_{i,j}\sigma_i^z \sigma_j^z \quad i,j > 0, \ i \neq j.$$
(230)

We call $\mathcal{H}_{0,0}$ a trivial operator, $\mathcal{H}_{i,0}$ a field operator, $\mathcal{H}_{i,i}$ a constant operator, and $\mathcal{H}_{i,j}$ an Ising operator. The operator $\mathcal{H}_{i,i}$ is associated with the site *i*, even though it is proportional to 1. With these operators, we can rewrite Equation (226) up to an irrelevant constant as

$$\mathcal{H} = -\sum_{i=1}^{N} \sum_{j=0}^{N} \mathcal{H}_{i,j} .$$
(231)

Note that Equation (231) is a sum over field, constant, and Ising operators. The constant operators are not part of the original Hamiltonian, but will be relevant for algorithmic purposes. The trivial operators (Equation (227)) are not relevant to express the Hamiltonian, but are necessary for the fixed-length sampling scheme. The proposed decomposition fulfils the no-branching property, and there are no negative matrix elements of the operators in the computational basis due to the positive constant $|J_{i,j}|$ that is added to the Ising operators. The possible matrix elements for Ising operators $\mathcal{H}_{i,j} = |J_{i,j}| - J_{i,j}\sigma_i^z\sigma_j^z$ acting on a pair of spins i, j are given by

$$\langle \uparrow \uparrow_{i,j} | \mathcal{H}_{i,j} | \uparrow \uparrow_{i,j} \rangle = \langle \downarrow \downarrow_{i,j} | \mathcal{H}_{i,j} | \downarrow \downarrow_{i,j} \rangle = \begin{cases} 2|J_{i,j}| & \text{for } J_{i,j} < 0\\ 0 & \text{for } J_{i,j} > 0 \end{cases}$$
(232)

$$\langle \uparrow \downarrow_{i,j} | \mathcal{H}_{i,j} | \uparrow \downarrow_{i,j} \rangle = \langle \downarrow \uparrow_{i,j} | \mathcal{H}_{i,j} | \downarrow \uparrow_{i,j} \rangle = \begin{cases} 0 & \text{for } J_{i,j} < 0\\ 2|J_{i,j}| & \text{for } J_{i,j} > 0 \end{cases}$$
(233)

This implies that only sequences where (anti)ferromagnetic Ising bonds are placed on (anti)aligned spins have a non-vanishing weight.

The partition function (Equation (225)) in the fixed-length scheme reads

$$Z = \sum_{S_{\mathcal{L}}} \sum_{|\alpha\rangle} \frac{\beta^n (\mathcal{L} - n)!}{\mathcal{L}!} \prod_{k=1}^{\mathcal{L}} \langle \alpha(k) | \mathcal{H}_{i(k), j(k)} | \alpha(k-1) \rangle .$$
(234)

The propagated states $|\alpha(p)\rangle$ only change in the imaginary-time propagation if a field operator, the only off-diagonal operator in the chosen basis, acts on it. A configuration only has a non-zero weight if all the Ising operators in the sequence are placed on spins that have

the correct alignment. Constant operators are included in the Hamiltonian for algorithmic purposes, which will become important in the off-diagonal update described below.

Before going into the description of the Markov chain to sample the configuration space, it is illustrative to visualise a configuration with non-vanishing weight defined by a state $|\alpha\rangle$ and an operator sequence $S_{\mathcal{L}}$. In Figure 14, an exemplary configuration with non-vanishing weight is illustrated for the one-dimensional TFIM and visualisations of the different operators Equations (228)–(230) in such a configuration are shown.



Figure 14. (a) An SSE configuration of a transverse-field Ising chain with N = 10 sites and a sequence length $\mathcal{L} = 27$. The spatial spin direction goes from left to right, and the imaginary-time dimension goes from bottom to top. The number of trivial operators in the operator sequence $S_{\mathcal{L}}$ is 8. Filled (empty) circles represent spins aligned in the $\sigma_i^z = +1$ (-1) direction. The propagated states $|\alpha(p)\rangle$ correspond to the *p*-th row from below with the lowest row being state $|\alpha\rangle$. (b) A depiction of the possible vertices for field operators. (c) A depiction of the possible vertices for constant operators. (d) A depiction of the allowed vertices for ferromagnetic Ising operators. Note that a ferromagnetic Ising vertex can only connect sites that are connected by a ferromagnetic bond in the Hamiltonian. (e) A depiction of the allowed vertices for antiferromagnetic Ising operators. Note that an antiferromagnetic Ising vertex can only connect sites that are connected by an antiferromagnetic bond in the Hamiltonian.

Before going into the details of the algorithm, we introduce the main concept of the update scheme and the crucial obstacles that are encountered in setting up an efficient algorithm [39]. Each step of the Markov chain sampling of configurations is performed by performing a so-called diagonal update followed by an off-diagonal quantum cluster update [39]. In the diagonal update, one iterates over the sequence and exchanges trivial operators with constant or Ising operators and vice versa while propagating the states along the sequence. In the off-diagonal update, constant operators are exchanged with

field operators while preserving the weight of the configuration. The main obstacle that is circumvented with this update procedure is the following: It is a non-trivial and non-local task to insert off-diagonal operators into the sequence without creating a configuration with vanishing weight. The first problem is that one cannot simply insert a single field operator into the sequence as it breaks the periodicity of the propagated states $|\alpha(\mathcal{L})\rangle \neq |\alpha(0)\rangle$, leading to a vanishing weight due to the orthonormality of the computational basis. Therefore, field operators can only occur in an even number for each site to preserve the periodicity in imaginary time. The second issue involves Ising operators placed on a pair of sites *i* and *j*. If one places a field operator at one of the sites before and behind the Ising operator in the sequence, this preserves the periodicity in imaginary time, but the matrix element of the Ising operator becomes zero as the spins will be misaligned with respect to the sign of the Ising coupling (see Equation (232)). These issues can be tackled by a non-local off-diagonal update [39], which we will thoroughly discuss after the diagonal update.

In the diagonal update, the number of non-trivial operators *n* is altered by exchanging trivial operators with constant and Ising operators in the operator sequence and vice versa. This does not change the states $|\alpha(p)\rangle$, including the state $|\alpha\rangle = |\alpha(0)\rangle$. Starting from propagation step p = 0 and state $|\alpha(0)\rangle$, one iterates over the sequence step-by-step and conducts the exchange of trivial operators with non-trivial diagonal operators. If a field operator is encountered at the current propagation step, the state is propagated by flipping the respective spin, and the iteration proceeds. If a diagonal operator is encountered, a local update following the Metropolis–Hastings algorithm as described in Section 3 is performed and the total transition probability is made of a proposition probability \tilde{T} and an acceptance probability p_{acc} .

If a trivial operator is encountered at propagation step *k*, a non-trivial diagonal operator gets proposed with the probability

$$\tilde{T}(\mathcal{H}_{0,0} \to \mathcal{H}_{i,j}) = \frac{M_{i,j}}{C}$$
(235)

taking into account the weight $M_{i,j}$ of the proposed operator with the normalising constant *C*:

$$M_{i,j} = \begin{cases} 2|J_{ij}| & \text{for } i \neq j, \\ h_i & \text{for } i = j, \end{cases}$$
(236)

$$C = \sum_{i}^{n} h_{i} + 2 \sum_{i \neq j} |J_{ij}|.$$
(237)

The weight is essentially given by the matrix elements of the respective operators $\mathcal{H}_{i,j}$ with the special case that the Ising operators are a priori handled as if they were allowed. At a later stage, it is checked if the spins are correctly aligned at the considered propagation step k and state $|\alpha(k)\rangle$, and the operator gets rejected if this is not the case. This has the benefit that one does not need to check every single bond for correct alignment for the insertion of a single operator, which scales like $\mathcal{O}(N^2)$ for long-range models. This sampling can be performed in a constant time complexity in the number of elements in the distribution by using the so-called walker method of aliases [214]. Instructions on how to set up a walker sampler from a distribution of discrete weights and how to draw from this distribution can be found in Appendix C or in Ref. [214]. Drawing from the discrete distribution of weights, an operator $\mathcal{H}_{i,j}$ gets proposed to be inserted. On the other side, if a non-trivial diagonal operator is encountered, it is always proposed to be replaced by a trivial operator:

$$\tilde{T}(\mathcal{H}_{i,i} \to \mathcal{H}_{0,0}) = 1.$$
(238)

The acceptance probabilities are then chosen according to the Metropolis–Hastings algorithm in Equation (46). For a constant field operator $\mathcal{H}_{i,i}$, this gives the acceptance probability:

$$p_{\rm acc}(\mathcal{H}_{0,0} \to \mathcal{H}_{i,i}) = \min\left(1, \frac{\beta M_{i,i}}{\mathcal{L} - n} \frac{C}{M_{i,i}}\right)$$
(239)

$$= \min\left(1, \frac{\beta(\sum_{i} h_{i} + 2\sum_{i \neq j} |J_{ij}|)}{\mathcal{L} - n}\right).$$
(240)

Similarly, for an Ising operator $\mathcal{H}_{i,j}$, one has

$$p_{\rm acc}(\mathcal{H}_{0,0} \to \mathcal{H}_{i,j}) = \min\left(1, \frac{\beta \langle \alpha(k) | \mathcal{H}_{i,j} | \alpha(k-1) \rangle}{\mathcal{L} - n} \frac{C}{M_{i,j}}\right)$$
(241)

$$= \frac{\langle \alpha(k) | \mathcal{H}_{i,j} | \alpha(k-1) \rangle}{M_{i,j}} \min\left(1, \frac{\beta(\sum_{i} h_i + 2\sum_{i \neq j} |J_{ij}|)}{\mathcal{L} - n}\right)$$
(242)

up to a factor $\langle \alpha(k) | \mathcal{H}_{i,j} | \alpha(k-1) \rangle / M_{i,j}$, which is either 1 or 0 depending on if the spins at the current propagation step *k* are correctly aligned or misaligned. An Ising bond with misaligned spins would lead to a vanishing weight of the configuration and is, therefore, not allowed.

Up to this factor, the acceptance probability is the same independent of the type of non-trivial diagonal operator that is proposed to be inserted as the operator weight $M_{i,j}$ arising in the weight for the newly proposed configuration cancels with the same factor in the proposition probability. This is because we already chose to propose an operator by considering its respective weight $M_{i,j}$. As the acceptance probabilities do not differ, one can, therefore, also first check if one accepts to insert any non-trivial diagonal operator and, only if this is accepted, draw the precise operator to be inserted. Of course, if the chosen operator is an Ising operator, it still has to be checked if the spins are correctly aligned to prevent a non-vanishing weight.

The cancellation of the operator weights $M_{i,j}$ also makes it easier to perform the reverse process and replace a non-trivial diagonal operator with a trivial one in the sense that the acceptance probability for inserting $\mathcal{H}_{0,0}$:

$$p_{\rm acc}(\mathcal{H}_{i,j} \to \mathcal{H}_{0,0}) = \min\left(1, \frac{\mathcal{L} - (n-1)}{\beta(\sum_i h_i + 2\sum_{i \neq j} |J_{ij}|)}\right)$$
(243)

does not depend on the current non-trivial diagonal operator $\mathcal{H}_{i,j}$.

If a proposition gets rejected, the iteration along the sequence continues and the procedure starts again for the next operator. After each diagonal update sweep (iteration over the whole sequence) during the equilibration, trivial operators are appended to the end of the sequence such that $\mathcal{L} > 4n/3$ [39]. This allows dynamically adjusting the sequence length \mathcal{L} for the fixed-length scheme to ensure a sufficiently long sequence.

For an efficient implementation of the off-diagonal quantum cluster update, it is crucial to introduce the concept of operator legs. An operator at position p in the sequence has legs with the numbers $4p, \ldots, 4p + 3$. For an Ising operator, these legs are associated with two legs per site, one upper and one lower leg (see Figure 15a).

Figure 15. Assignment of leg numbers to operator legs for an operator at propagation step *p*. (a) Illustration of a two-site operator with four real vertex legs. (b) Illustration of a single-site operator with only two real vertex legs and two ghost legs.

For a constant or field operator, only two of the four legs are real vertex legs since these operators act only on a single site (see Figure 15b). The remaining two legs are called ghost

legs and are considered solely due to algorithmic reasons as it is numerically beneficial to let every operator have the same number of legs. This allows calculating the propagation index *p* of an operator from the leg numbers using integer division with $p = \lfloor leg/4 \rfloor$. Trivial operators can be described with four ghost legs or can be ignored entirely in the sequence for the off-diagonal update.

For the chosen representation of the Ising model within the SSE, one can subdivide the configuration into disjoint clusters that extend in space, as well as in imaginary time with Ising operators acting as bridges in real space and with constant and field operators acting as delimiters in imaginary time. These clusters of spins can be flipped by replacing the delimiting constant operators with field operators and vice versa without changing the weight $\pi(|\alpha\rangle, S_{\mathcal{L}}; \beta)$. If the cluster winds around the boundary in imaginary time, the respective spins in state $|\alpha\rangle$ have to be updated as well. In order to flip half of the configuration on average and obtain a good mixing, all clusters are constructed and the probability to flip a specific cluster is chosen to be 1/2 for each cluster separately.

For the construction of the clusters, the propagated spin states along imaginary time are not needed. The entire problem can be dealt with in the language of vertices with legs. The relevant information for the cluster formation is which legs are connected. This information about the connection between legs of different operators is stored in a doubly linked list. The list is set up in the following way: At the index of a vertex leg *i*, we store the index *j* of the leg it is connected to, i.e., list[i] = j and vice versa list[list[i]] = list[j] = i. This segmentation is illustrated for an exemplary configuration in Figure 16 together with the doubly linked leg list for the depicted configuration. The doubly linked list can already be set up during the diagonal update when the sequence is traversed either way. An efficient algorithm to set up the data structure is described in Refs. [213,215].

In comparison to general off-diagonal loop updates [215–217], the formation of clusters in the presented off-diagonal cluster update for the TFIM is fully deterministic. The whole configuration is divided into disjoint clusters, all of which will be built and flipped with probability 1/2. It is beneficial to already decide whether a cluster is flipped or not before constructing the cluster so the constant and field operators and spin states can already be processed during the construction. A leg that is processed during the construction of a cluster is marked as visited in order to not process the same leg twice. This is also the reason why all the clusters have to be constructed even if they will not be flipped, as otherwise, the same cluster would get constructed starting from another leg later on. Further, we introduce a stack for the legs that were visited during the construction, but whose connections are yet to be processed.

The formation of each cluster starts by choosing a leg that has not yet been visited in the current off-diagonal update. At the beginning of the cluster formation, it is randomly determined if the cluster is flipped or not. If the leg corresponds to an Ising vertex, the cluster branches out to all four legs of the vertex. This means that all four legs of the operator are put on the stack. If the leg corresponds to a constant or field operator, the type of operator is exchanged if the cluster is flipped and only this primal leg is put on the stack. Next, the following logic is repeated until the stack is empty. We pop a leg *l* from the stack. If the leg has been visited already, we continue with a new leg from the stack. Else, we determine the new leg l' = list[l] using the doubly linked list. We mark both legs l and l' as visited. If one passes by the periodic boundary in imaginary time while going from leg l to l' and the cluster will be flipped, the corresponding site belonging to the legs in the state $|\alpha\rangle$ has to be flipped. If l' belongs to an Ising operator, we add all legs of the Ising operator that have not been visited yet to the stack. If l' belongs to a constant or field operator, we exchange the operator if the cluster is said to be flipped. This procedure is repeated until the stack is empty. After that, one proceeds with the next cluster starting from a leg that has not yet been visited in the current cluster update. If no such leg is left, the cluster update is finished.



Figure 16. Illustration for the segmentation and construction of the doubly linked list for an exemplary configuration of the ferromagnetic transverse-field Ising model. (a) Segmentation of a configuration into disjoint clusters including the numbers of the operator legs. Ghost legs are not depicted. The coloured lines with ellipses at each end depict the operator legs that are linked. Each colour represents one cluster in the off-diagonal update. (b) Depiction of a doubly linked list for the configuration shown in (a). The left column represents the entry numbers in the list and the right column the corresponding legs to which the entry is connected. The colour represents the clusters in (a) to which the connection belongs. Ghost legs are linked to the value -1 and are shaded in grey.

In addition to this cluster update, spins that have no operators acting on them in the sequence can be flipped with a probability of 1/2 (thermal spin flip).

In summary, the off-diagonal update exchanges field $\mathcal{H}_{i,0}$ and constant operators $\mathcal{H}_{i,i}$ with each other and changes the state $|\alpha\rangle$. Combining the diagonal update with the off-diagonal cluster update samples the entire configuration space.

5.2. Algorithm for Unfrustrated Heisenberg Models

In this section, we describe the algorithm to sample arbitrary unfrustrated spin-1/2 Heisenberg models with the SSE framework [213]. By unfrustrated Heisenberg models, we mean Hamiltonians:

$$\mathcal{H} = \sum_{b \in AF} J_b \vec{S}_{i(b)} \vec{S}_{j(b)} + \sum_{b \in F} J_b \vec{S}_{i(b)} \vec{S}_{j(b)}$$
(244)

written as a sum over three-component interactions between two sites i(b) and j(b) connected by bond b, where each bond is either ferromagnetic (F) with $J_b < 0$ or antiferromagnetic (AF) with $J_b > 0$, with the property that there is no loop of lattice sites connected by bonds that contains an odd number of antiferromagnetic bonds, as this would lead to frustration. The spin operators are a compact notation for a three-component vector of spin operators $\vec{S}_i = (S_i^x, S_i^y, S_i^z)^T$. The coupling strengths J_b can have a priori arbitrary amplitudes. It is crucial to look at unfrustrated Heisenberg models in order to define non-negative weights for configurations as the off-diagonal components of the antiferromagnetic operators have a negative matrix element. In an unfrustrated model, these matrix elements always occur an even number of times in the operator sequence of any valid configuration, which makes it possible to construct a non-negative SSE weight [213].

For the SSE algorithm, the σ^z -eigenbasis $\{|\alpha\rangle\} = \{|\sigma_1^z, ..., \sigma_N^z\rangle\}$ is chosen, but the σ^x or σ^y -eigenbasis would work the same way. The Hamiltonian is decomposed into

$$\mathcal{H} = -\sum_{b \in AF} \left(\mathcal{H}_{1,b}^{AF} + \mathcal{H}_{2,b}^{AF} - \frac{|J_b|}{4} \right) - \sum_{b \in F} \left(\mathcal{H}_{1,b}^F + \mathcal{H}_{2,b}^F - \frac{|J_b|}{4} \right)$$
(245)

using the following operators:

$$\mathcal{H}_{1,b}^{\mathrm{F}} = \frac{|J_b|}{4} - J_b S_{i(b)}^z S_{j(b)}^z$$
(246)

$$\mathcal{H}_{2,b}^{\mathrm{F}} = -\frac{J_b}{2} (S_{i(b)}^+ S_{j(b)}^- + S_{i(b)}^- S_{j(b)}^+)$$
(247)

$$\mathcal{H}_{1,b}^{\rm AF} = \frac{|J_b|}{4} - J_b S_{i(b)}^z S_{j(b)}^z$$
(248)

$$\mathcal{H}_{2,b}^{\rm AF} = -\frac{J_b}{2} \left(S_{i(b)}^+ S_{j(b)}^- + S_{i(b)}^- S_{j(b)}^+ \right) \,. \tag{249}$$

The diagonal operators $\mathcal{H}_{1,b}^{F}$ and $\mathcal{H}_{1,b}^{AF}$ in Equations (246) and (248) are defined in the same fashion as for the TFIM up to the factor of 1/4 due to the usage of spin operators $S^{\kappa} = \sigma^{\kappa}/2$ instead of Pauli matrices σ^{κ} with $\kappa \in \{x, y, z\}$. The contribution to the weight of a sequence of these operators is either $|J_b|/2$ if the bond fulfils the (anti)ferromagnetic condition and zero otherwise. Although the expressions for the ferromagnetic and antiferromagnetic bonds look the same, we distinguish between these two bonds to highlight that these objects behave differently within the off-diagonal update.

As $J_b > 0$ for antiferromagnetic bonds, the off-diagonal operators $\mathcal{H}_{2,b}^{AF}$ do not fulfil the non-negativity of matrix elements (see Equation (215)) in the computational basis. Therefore, they must always appear in an even number of times in the operator sequence to avoid the sign problem. For the ferromagnetic bonds, this restriction is not necessary.

Analogous to the TFIM, the operator sequence additionally contains trivial operators $\mathcal{H}_{0,0} = \mathbb{1}$, which are not part of the Hamiltonian, but are used to pad the sequence to a fixed length \mathcal{L} . In contrast to the TFIM, there is no need for further artificial operators like the constant field operators in the TFIM used to limit the cluster in the cluster update. In the case of the Heisenberg model, the non-local off-diagonal update is constructed in the form of loops instead of a cluster with several branches that need to be limited in imaginary time. An exemplary SSE configuration for a Heisenberg chain using the decomposition from above can be seen in Figure 17.

Similar to the sampling of the TFIM in Section 5.1, each step of the Markov chain sampling of configurations is achieved by performing a diagonal update followed by a non-local off-diagonal update. The diagonal update exchanges trivial operators with diagonal operators, while the off-diagonal update exchanges diagonal bond operators with the respective off-diagonal bond operators.



Figure 17. (a) An SSE configuration of a Heisenberg chain with ten sites and a sequence length of 27. The number of trivial operators is 5. Filled (empty) circles represent spins in the $\sigma_i^z = +1(-1)$ direction. The propagated states $|\alpha(p)\rangle$ correspond to the *p*-th row in the configuration. (b) A depiction of the allowed vertices for ferromagnetic diagonal operators. (c) A depiction of the allowed vertices for antiferromagnetic off-diagonal operators. (d) A depiction of the allowed vertices for antiferromagnetic off-diagonal operators. (e) A depiction of the allowed vertices for antiferromagnetic off-diagonal operators. Note that (anti)ferromagnetic vertices can only connect sites that are connected by (anti)ferromagnetic bonds in the Hamiltonian.

Due to the structure of the diagonal operators (see Equations (246) and (248)), similar to the TFIM, the diagonal update is performed similarly to the diagonal update of the TFIM described in Section 5.1. Nonetheless, to make the description of the algorithm for unfrustrated Heisenberg models self-contained, we recapitulate the diagonal update and adapt it to the Heisenberg case.

The diagonal update exchanges trivial operators with diagonal operators in the sequence and vice versa. It will, therefore, not change the states $|\alpha(p)\rangle$, including the state $|\alpha\rangle = |\alpha(0)\rangle$. Starting from $|\alpha\rangle$ at k = 0, the sequence is again iterated over by k. If the operator at the current position k in the sequence is an off-diagonal operator operator, the state $|\alpha(k-1)\rangle$ is propagated by applying the off-diagonal operator and the iteration proceeds. If one encounters diagonal or trivial operators in the sequence, the Metropolis–Hastings algorithm (see Section 3) is performed and the transition probability to an altered configuration is again split into a proposition probability \tilde{T} and an acceptance probability p_{acc} .

Analogous to the algorithm for the TFIM, if a trivial operator is encountered, a nontrivial diagonal operator gets proposed with the probability:

$$\tilde{T}(\mathcal{H}_{0,0} \to \mathcal{H}_{1,b}^{AF/F}) = \frac{M_b}{C},$$
(250)

taking into account the weight M_b of the proposed operator and the normalising constant C:

$$M_b = \frac{|J_b|}{2} \qquad \qquad C = \frac{1}{2} \sum_{b} |J_b|.$$
(251)

The diagonal update for the Heisenberg model only differs from the one of the TFIM by the detailed probabilities M_b and the normalising constant *C*. The factor of 1/4 in the bond weights in comparison to the Ising model comes from using spin operators in contrast to Pauli matrices in the Hamiltonian. This sampling according to the weight M_b can be performed in a constant-time complexity in the number of elements in the distribution by using the walker method of aliases [214]. An introduction on how to set up a walker sampler from a distribution of discrete weights and how to draw from this distribution can be found in Appendix C or in Ref. [214]. Drawing from the discrete distribution of weights, an operator $\mathcal{H}_{1,b}^{AF/F}$ gets proposed to be inserted. On the other side, if a non-trivial diagonal operator is encountered, it is always proposed to be replaced by a trivial operator:

$$\tilde{I}(\mathcal{H}_{1\,h}^{\mathrm{AF/F}} \to \mathcal{H}_{0,0}) = 1, \tag{252}$$

exactly as for the TFIM. The acceptance probabilities are then chosen according to the Metropolis–Hastings algorithm Equation (46). The respective Metropolis–Hastings probabilities with which the proposed replacements at position k in the operator sequence are accepted are given by the ratios of the configuration weights before and after the potential replacement and the ratio of the proposition probabilities:

=

$$p_{\rm acc}(\mathcal{H}_{0,0} \to \mathcal{H}_{1,b}^{\rm AF/F}) = \min\left(1, \frac{\beta \langle \alpha(k) | \mathcal{H}_{1,b}^{\rm AF/F} | \alpha(k-1) \rangle}{\mathcal{L} - n} \frac{C}{M_b}\right)$$
(253)

$$=\frac{\langle \alpha(k)|\mathcal{H}_{1,b}^{\text{AF/F}}|\alpha(k-1)\rangle}{M_b}\min\left(1,\frac{\beta(\sum_b|J_b|)}{2(\mathcal{L}-n)}\right),\qquad(254)$$

$$p_{\rm acc}(\mathcal{H}_{1,b}^{\rm AF/F} \to \mathcal{H}_{0,0}) = \min\left(1, \frac{2(\mathcal{L} - (n-1))}{\beta(\sum_{b} |J_b|)}\right).$$
(255)

As in the case of the TFIM, the specific matrix element $\langle \alpha(k) | \mathcal{H}_{1,b}^{AF/F} | \alpha(k-1) \rangle$ for the propagated states $|\alpha(k-1)\rangle$ and $|\alpha(k)\rangle = |\alpha(k-1)\rangle$ has to be considered, which is either M_b , and cancels with this factor in the acceptance probability, or is 0, and the acceptance probability vanishes. Intuitively, this factor simply checks if the spins are correctly aligned or misaligned with respect to the sign of the coupling J_b and only allows the operator to be inserted if the spins i(b) and j(b) of the propagated state $|\alpha(k-1)\rangle$ are correctly aligned. If the proposition is accepted, the replacement of the operator at index k is performed, and one continues with the next propagation index k + 1 in imaginary time. If the proposition index k + 1 in imaginary time.

As for the TFIM, a fixed-length scheme is used and the sequence length \mathcal{L} limiting the amount of non-trivial operators *n* has to be dynamically adjusted. During the thermalisation of the sampling procedure, trivial operators are appended to the end of the sequence after each diagonal update (iteration over the whole sequence) such that $\mathcal{L} > 4n/3$ [39].

In order to efficiently implement an off-diagonal loop update, it is crucial to introduce the concept of operator legs. An operator at position p in the sequence has four legs with the numbers $4p, \ldots, 4p + 3$, two legs (an upper and a lower leg) for each of the spin sites. Trivial operators can be described with four ghost legs or can be ignored entirely in the sequence for the off-diagonal update.

For the unfrustrated Heisenberg models, the off-diagonal update is performed using a loop update. During the update, loops are constructed in the space–time configuration (see Figure 17 for an example) and the spin states on the loop are flipped along the way, which changes the operator vertex types. Thereby, diagonal operators become off-diagonal operators and vice versa. It is noteworthy that the weight of a configuration is not modified by this since the weights of diagonal and off-diagonal operators are the same in the isotropic Heisenberg model. Therefore, flipping loops with a fixed probability satisfies detailed balance as long as the probability to construct the reverse loop is the same.

For the Heisenberg model, the creation of the loops is deterministic. Whenever a loop enters an operator during the loop construction, there is only one exit leg that creates a valid operator for the bond type (AF or F). Flipping the spins along the loop exchanges an diagonal operator $\mathcal{H}_{1,b}^{AF/F}$ with its off-diagonal counterpart $\mathcal{H}_{2,b}^{AF/F}$ and vice versa. The entrance leg and exit leg never belong to the same site. For ferromagnetic operators, the propagation direction of the loop in imaginary time remains the same (see Figure 18). For antiferromagnetic operators, the propagation direction of the loop crosses the periodic boundary in imaginary time at a site *i*, the computational basis state $|\alpha\rangle$ has to be updated by flipping the spin at site *i*. As each leg only belongs to one loop, the configuration can be split into disjoint loops. Therefore, it is possible to construct all loops during the off-diagonal update and flip each loop independently with a probability of 1/2.

Combining the diagonal and off-diagonal update allows sampling arbitrary unfrustrated Heisenberg models. This includes antiferromagnetic Heisenberg ladders and bilayer systems with an unfrustrated long-range interaction.



Figure 18. Illustration showing the construction of the doubly linked vertex list and the off-diagonal deterministic loop update for unfrustrated Heisenberg models. As an example, a Heisenberg chain with periodic boundary conditions and nearest-neighbour antiferromagnetic and next-nearest-neighbour ferromagnetic interactions is considered. (a) Illustration of a configuration including the

numbers of the operator legs belonging to the respective operators. The coloured lines with ellipses at each end depict the operator legs that are linked. Each colour represents one loop in the off-diagonal update. (b) Depiction of a doubly linked list for the configuration shown in (a). The left column represents the entry numbers in the list and the right column the corresponding legs to which the entry is connected. The colour represents the loops in (a) to which the connection belongs.

5.3. Observables

In this section, we introduce some observables that can be easily measured within the SSE formalism introduced. Throughout this section, we do not use the fixed-length scheme, but keep the general form with operator sequences of fluctuating length to keep the notation short and simple. The sampling of observables can be performed analogously in the fixed-length scheme. When implementing the formulas for the observables in the fixed-length scheme, one just has to keep in mind that the sequences used in the formulas are the same ones, but without the padded trivial operators. This means, for instance, that a sum over all propagation steps p becomes a sum over all non-trivial propagation steps in the fixed-length scheme.

Up to this point in Section 5, we have focused on the partition function of a quantummechanical Hamiltonian:

$$Z = \sum_{\omega \in \mathcal{C}} \pi(\omega),$$

expressed as a sum over non-negative weights $\pi(\omega)$ of a configuration $\omega \in C$. However, we are actually not interested in calculating the partition function itself, but rather, the expectation values of observables.

Analogous to the partition function, quantum mechanical expectation values can be expanded into a high-temperature series:

$$\langle A \rangle = \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0} \sum_{S_n} \frac{(-\beta)^n}{n!} \langle \alpha | A \prod_{k=1}^n \mathcal{H}_{l(k)} | \alpha \rangle.$$
(256)

It is important to realise that configurations with a non-vanishing weight $\pi(\alpha, S_n; \beta)$ constituting the partition function do not necessarily contribute to the expectation value $\langle A \rangle$ with the same weight or a non-vanishing weight at all [213]. In general, only for operators *A* that are diagonal in the computational basis { $|\alpha\rangle$ }, the expectation value can be written as

$$\langle A \rangle = \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0} \sum_{S_n} A(\alpha) \pi(\alpha, S_n; \beta)$$
(257)

with $A(\alpha) = \langle \alpha | A | \alpha \rangle$. For instance, the magnetisation in the *z*-direction or any *n*-th moment thereof is such a diagonal operator when using the σ^z -eigenbasis $\{ |\alpha\rangle \} = \{ |\sigma_1^z, \dots, \sigma_N^z \rangle \}$ as a computational basis. The statistics of the MC estimates for such observables can be improved by realising that [218]

$$\pi(\alpha, S_n; \beta) = \pi(\alpha(p), S_n(p); \beta)$$
(258)

with $S_n(p)$ being the sequence obtained from cyclically permuting S_n for p times [218]. This means that the expectation value Equation (257) can be expressed as

$$\langle A \rangle = \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0} \sum_{S_n} \frac{1}{n} \sum_{p=0}^{n-1} A(\alpha(p)) \pi(\alpha, S_n; \beta),$$
(259)

where $A(\alpha)$ is additionally averaged over imaginary time with $A(\alpha(p)) = \langle \alpha(p) | A | \alpha(p) \rangle$.

For off-diagonal operators, one needs to find customised formulas for the expectation value. However, some of these expectation values are accessible in a quite general way. For instance, one can show that the mean energy has a rather simple formula [210,211,218]:

$$\langle \mathcal{H} \rangle = -\frac{\langle n \rangle}{\beta} \,.$$
 (260)

Moreover, for the operators \mathcal{H}_i (see Equation (213)), one finds [210,211,218]

$$\langle \mathcal{H}_i \rangle = -\frac{\langle n_i \rangle}{\beta} ,$$
 (261)

where n_i is the amount of operators \mathcal{H}_i occurring in the operator sequence. Similarly, one can derive a formula for the heat capacity [210,211,218]:

$$C = \langle n(n-1) \rangle - \langle n \rangle^2 = \left\langle n^2 \right\rangle - \langle n \rangle^2 - \langle n \rangle \,. \tag{262}$$

However, for small temperatures, the heat capacity is calculated as the small difference of large numbers.

Linear Response and Correlation Functions

An important class of observables is linear response functions like susceptibilities. The linear response of an observable *A* to a perturbation $\mathcal{H} \rightarrow \mathcal{H} - \lambda B$ tuned by perturbation parameter λ is given by a Kubo integral [212,219]:

$$\chi_{A,B} = \left. \frac{\partial \langle A \rangle}{\partial \lambda} \right|_{\lambda=0} \tag{263}$$

$$= \int_{0}^{\beta} \langle A(\tau)B(0)\rangle d\tau - \beta \langle A\rangle \langle B\rangle \,. \tag{264}$$

Once again, if *A* or *B* are off-diagonal operators, one has to consider case by case and find customised formulas that can be used to sample the specific observables within the SSE formulation. We will focus on the case where *A* and *B* are diagonal in the chosen computational basis. Important examples are the magnetic susceptibility or its local version of spin–spin correlation functions averaged over imaginary time, i.e., the correlation function at zero frequency when regarding Equation (264) as a Laplace transformation from imaginary time to frequency space. To make this type of observable accessible to a sampling in the SSE formulation, the imaginary-time correlation function $\langle A(\tau)B(0)\rangle$ is expanded in temperature analogous to the partition function:

$$\langle A(\tau)B(0)\rangle = \frac{1}{Z}\operatorname{Tr}\left(e^{-\beta\mathcal{H}}e^{\tau\mathcal{H}}Ae^{-\tau\mathcal{H}}B\right)$$
(265)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \left\langle \alpha \left| \sum_{k=0}^{\infty} \left(\frac{(\beta - \tau)^k}{k!} (-\mathcal{H})^k \right) A \sum_{l=0}^{\infty} \left(\frac{\tau^l}{l!} (-\mathcal{H})^l \right) B \right| \alpha \right\rangle$$
(266)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{l,k=0}^{\infty} \frac{(\beta - \tau)^k}{k!} \frac{\tau^l}{l!} \left\langle \alpha \left| (-\mathcal{H})^k A (-\mathcal{H})^l B \right| \alpha \right\rangle$$
(267)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{l,k=0}^{\infty} \frac{(\beta-\tau)^k}{k!} \frac{\tau^l}{l!} A_l B_0 \left\langle \alpha \left| (-\mathcal{H})^{l+k} \right| \alpha \right\rangle,$$
(268)

where, in the last step, the operators *A* and *B* were replaced by their respective eigenvalues $A_l = \langle \alpha(l) | A | \alpha(l) \rangle$ and $B_0 = \langle \alpha(0) | B | \alpha(0) \rangle$ of the state $|\alpha(p)\rangle$ at the propagation steps p = l, 0.
By replacing the sum over k by a sum over n := l + k and inserting the decomposition of the Hamiltonian, this takes the form

$$\langle A(\tau)B(0)\rangle = \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0}^{\infty} \sum_{l=0}^{n} \sum_{S_n} \frac{(\beta-\tau)^{n-l}}{(n-l)!} \frac{\tau^l}{l!} A_l B_0 \left\langle \alpha \left| \prod_{i=0}^{n-1} \mathcal{H}_{a_i, b_i} \right| \alpha \right\rangle$$
(269)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0}^{\infty} \sum_{l=0}^{n} \sum_{S_n} \frac{(\beta - \tau)^{n-l}}{(n-l)!} \frac{\tau^l}{l!} A_l B_0 \frac{n!}{\beta^n} \pi(\alpha, S_n; \beta)$$
(270)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0}^{\infty} \sum_{l=0}^{n} \sum_{S_n} \frac{n!}{(n-l)!l!} \left(1 - \frac{\tau}{\beta}\right)^{n-l} \left(\frac{\tau}{\beta}\right)^l A_l B_0 \pi(\alpha, S_n; \beta)$$
(271)

$$= \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0}^{\infty} \sum_{l=0}^{n} \sum_{S_n} \binom{n}{l} \left(1 - \frac{\tau}{\beta}\right)^{n-l} \left(\frac{\tau}{\beta}\right)^l \pi(\alpha, S_n; \beta) \frac{1}{n} \sum_{p=0}^{n-1} A_{p+l} B_p, \quad (272)$$

where, in the last step, the average over imaginary time was taken in order to improve the statistics. From Equation (272), the connection between the discrete propagation steps of the SSE and the continuous imaginary time τ becomes apparent [211,212]. An imaginary-time separation τ corresponds to a binomial distribution of separations *l* of SSE propagation steps:

$$B(l|\tau,n) = {\binom{n}{l}} \left(1 - \frac{\tau}{\beta}\right)^{n-l} \left(\frac{\tau}{\beta}\right)^{l}, \qquad (273)$$

which is peaked around $l = n\tau/\beta$ [211]. If one is interested in the spectral properties of the system, one can use this formula for sampling imaginary-time correlation functions $\langle A(\tau)B(0)\rangle$. However, there are more efficient ways to calculate imaginary-time correlation functions by embedding the SSE configuration into continuous imaginary time. We refer to Refs. [212,220] for details, as this is out of the scope of this review.

To get to the linear response function $\chi_{A,B}$ in Equation (264), the imaginary time integral of in Equation (272) has to be calculated. This integral can be analytically calculated:

$$\int_0^\beta \mathrm{d}\tau \left(1 - \frac{\tau}{\beta}\right)^{n-l} \left(\frac{\tau}{\beta}\right)^l = \beta \int_0^1 \mathrm{d}u \,(1-u)^{n-l} u^l \tag{274}$$

$$=\beta \frac{(n-l)!}{n!} l! \int_0^1 \mathrm{d}u \, (1-u)^n \tag{275}$$

$$=\beta \frac{(n-l)!}{n!} l! \frac{1}{n+1}$$
(276)

by performing *l* partial integrations. Inserting Equation (272) into the formula (264) and performing the imaginary time integral Equation (276) yield

$$\chi_{A,B} = \frac{1}{Z} \sum_{\{|\alpha\rangle\}} \sum_{n=0}^{\infty} \sum_{S_n} \pi(\alpha, S_n; \beta) \frac{\beta}{n(n+1)} \sum_{l=0}^n \sum_{p=0}^{n-1} A_{p+l} B_p - \beta \langle A \rangle \langle B \rangle.$$
(277)

One can further separate the l = n term while using the periodicity $A_{p+n} = A_p$ in imaginary time and rewrite the sums over l and p as a product of two sums. This eventually yields [210]

$$\chi_{A,B} = \left\langle \frac{\beta}{n(n+1)} \left[\sum_{p=0}^{n-1} A_p B_p + \left\{ \sum_{p=0}^{n-1} A_p \right\} \left\{ \sum_{p=0}^{n-1} B_p \right\} \right] \right\rangle_{\pi(\alpha,S_n;\beta)} - \beta \langle A \rangle \langle B \rangle \,.$$
(278)

From an algorithmic perspective, the two sums in Equation (278) need to be calculated by traversing the operator sequence. The effort for measuring $\chi_{A,B}$, therefore, scales like the algorithm with complexity $\mathcal{O}(\beta N)$ when the observables A_p , B_p , and A_pB_p are not calculated from scratch at every propagation step *p*, but gradually updated and averaged while propagating the state through the sequence.

For the special case of the susceptibility, one needs to set $A = m = \sum_i \sigma_i^z$ and $B = M = N \cdot m$. The last term $N\beta \langle m \rangle^2$ in Equation (278) can be dropped for simulations of finite systems due to $\langle m \rangle = 0$. Explicitly, this gives

$$\chi = N \left\langle \frac{\beta}{n(n+1)} \left[\sum_{p=0}^{n-1} m_p m_p + \left\{ \sum_{p=0}^{n-1} m_p \right\} \left\{ \sum_{p=0}^{n-1} m_p \right\} \right] \right\rangle_{\pi(\alpha, S_n; \beta)}$$
(279)

for the susceptibility. Once again, we want to stress that this formula is not formulated in the fixed-length scheme and the average over the propagation index *p*, therefore, refers to the propagation steps of non-trivial operators. Averaging over all propagation steps in the fixed-length scheme is incorrect.

For the spin–spin correlation functions $G_{i,j}(\omega = 0)$, the operators are set to $A = \sigma_i^z$ and $B = \sigma_i^z$, leading to

$$G_{i,j}(\omega=0) = \int_0^\beta \left\langle \sigma_i^z(\tau) \sigma_j^z(0) \right\rangle d\tau$$
(280)

$$= \left\langle \frac{\beta}{n(n+1)} \left[\sum_{p=0}^{n-1} \sigma_{i,p}^z \sigma_{j,p}^z + \left\{ \sum_{p=0}^{n-1} \sigma_{i,p}^z \right\} \left\{ \sum_{p=0}^{n-1} \sigma_{j,p}^z \right\} \right] \right\rangle_{\pi(\alpha, S_n; \beta)},$$
(281)

where it was already used that $\langle \sigma_i^z \rangle = 0$ for any finite system. Similar to the zero-frequency spin–spin correlation function, one can also calculate an equal-time spin–spin correlation function:

$$G_{i,j}(\tau=0) = \left\langle \sigma_i^z \sigma_j^z \right\rangle = \left\langle \frac{1}{n} \sum_{p=0}^{n-1} \sigma_{i,p}^z \sigma_{j,p}^z \right\rangle_{\pi(\alpha, S_n; \beta)}.$$
(282)

Sampling the correlations among all sites *i* and *j* at all propagation steps *p* leads to a complexity of $\mathcal{O}(\beta N^3)$, which is worse than the computational complexity $\mathcal{O}(\beta N)$ of the SSE QMC updates. Even though one could eliminate the average over imaginary time in the case of Equation (282) to reduce the complexity to $\mathcal{O}(N^2)$ at the expense of statistical accuracy, this is not possible for Equation (281) as the sum over imaginary time is intrinsic in the definition of the zero-frequency correlation function. However, by realising that the correlations $\sigma_i^z \sigma_j^z$ between two sites *i* and *j* are not altered at the order $\mathcal{O}(\beta N)$ -times in imaginary time, but only $\mathcal{O}(\beta)$, one can measure the correlation functions Equations (281) and (282) in $\mathcal{O}(\beta N^2)$ time. This is achieved by traversing imaginary time while memorising the propagation step last[*i*] of the last preceding spin flip operator for every site separately. When a spin flip occurs at a site *j* at propagation step *p*, one needs to update the correlation function:

$$G[i,j] \to G[i,j] + \sigma_{i,p}^z \sigma_{i,p}^z (p - \max(\operatorname{last}[i], \operatorname{last}[j]))$$
(283)

for every site *i* before the local magnetisation is propagated with $\sigma_{j,p+1}^z = -\sigma_{j,p}^z$. One further needs to update last[*j*] = *p*. In order to avoid that the measurement dominates the simulation for large systems, we only measure the correlations between a fixed site *i* = 1 and *j* \in {1,...,*N*}. This finally reduces the complexity of the measurement to $\mathcal{O}(\beta N)$.

Correlation functions are observables that contain much information about a system's state and, in particular, its order. Moreover, its decay at the critical point with the distance between two spins is connected to the critical exponent η (anomalous dimension). Away from the critical point, the correlation function usually decays exponentially with long-range models being one exception to this. The length scale of this exponential decay is given by the correlation length ξ , which itself is an interesting quantity at a continuous

phase transition as its divergence is the reason for systems becoming scale-free at the critical point. As the correlation function does not decay exponentially for long-range system, which we focus on in this review, we will use the more general term "characteristic length scale" to refer to the length scale at which the correlations switch to their long-distance behaviour. In particular, with respect to the Q-FSS for systems above the upper critical dimension, the characteristic length scale is a crucial quantity as its scaling was predicted to be different from the standard FSS with $\xi_L \sim L^{\varphi}$ instead of $\xi_L \sim L$ (see Section 2.3 or Ref. [34] for details on scaling above the upper critical dimension in quantum systems). This quantity, therefore, played a crucial role in confirming the Q-FSS hypothesis. However, the characteristic length scale is a subtle quantity, which is hard to extract or even define on a finite lattice. For a finite system, the definition of a characteristic length scale is not unique, and there are several definitions for ξ_L , which will converge to ξ_{∞} for $L \to \infty$ [221]. For long-range systems, finding a suitable definition for the characteristic length scale is even more difficult, as the correlation function decays algebraically even away from the critical point [222].

Common definitions that are tailored for correlation lengths, which specify the exponential decay of a correlation function at long distances, such as the second moment:

$$\tilde{\xi}_{\infty} = \sqrt{\frac{1}{2D} \frac{\int |\vec{r}|^2 G(\vec{r}) \,\mathrm{d}r}{\int G(\vec{r}) \,\mathrm{d}r}} \tag{284}$$

therefore, might yield $\xi_{\infty} = \infty$ in an infinite system not only at the critical point, but also away from the critical point [223].

One possible definition for long-range quantum systems is [224]

$$\xi_L^{(\mathrm{LR}\omega)} = \frac{1}{q_{\min}} \left[\frac{\tilde{G}_L(0,\omega=0) - \tilde{G}_L(q_{\min},\omega=0)}{\tilde{G}_L(q_{\min},\omega=0)} \right]^{1/\sigma}$$
(285)

with $G(q, \omega)$ being the Fourier transform of $G_{i,j}(\omega)$ from real space to momentum space and $q_{\min} = 2\pi/L$ the smallest wavevector fitting on the finite lattice. By inserting the Gaussian propagator $\tilde{G}(q, \omega = 0) \sim (b_{\sigma}q^{\sigma} + m^2)^{-1}$ for long-range interactions [20,21] into Equation (285), one obtains

$$\xi_L^{(\mathrm{LR}\omega)} = \frac{1}{q_{\min}} \left[\frac{b_\sigma q_{\min}^\sigma + m_L^2}{m_L^2} - 1 \right]^{1/\sigma}$$

$$= b_\sigma^{1/\sigma} m_L^{-2/\sigma},$$
(286)

so that the momentum dependency cancels. Another definition for the same quantity, but using the equal-time Gaussian propagator $G_L(0, \tau = 0) \sim (2\sqrt{\tilde{g}}\sqrt{b_\sigma q^\sigma + m^2})^{-1}$ [212], is

$$\xi_L^{(\mathrm{LR}\tau)} = \frac{1}{q_{\min}} \left[\frac{\tilde{G}_L^2(0, \tau = 0) - \tilde{G}_L^2(q_{\min}, \tau = 0)}{\tilde{G}_L^2(q_{\min}, \tau = 0)} \right]^{1/\sigma}$$

$$= b_{\sigma}^{1/\sigma} m_L^{-2/\sigma} .$$
(287)

By inserting the scaling of the gap $m_{\infty} \sim |r|^{z\nu} = |r|^{\sigma\nu/2}$ for the *n*-component quantum rotor model in the long-range mean-field regime [20,21], which is relevant for most applications discussed below, one obtains the scaling of $\xi^{(LR)}$ in the thermodynamic limit to be

$$\xi_{\infty}^{(LR)} \sim |r|^{-\nu}, \qquad (288)$$

1/-

which is the singularity one expects from the characteristic length scale.

5.4. Sampling at Effectively Zero Temperature

The SSE QMC approach can be used to stochastically calculate the thermal average of operators for systems of a finite size at a finite temperature. We discussed in Section 2 that QPTs are a concept defined in the thermodynamic limit at zero temperature. Further, we presented in Sections 2.2 and 2.3 how the critical point, as well as critical exponents can be extracted from certain observables calculated on finite systems at zero temperature. The goal of many works that use SSE QMC sampling is to calculate the expectation value of these observables on finite systems at an effective zero temperature [32,34]. We define the notion of effective zero temperature as a temperature at which, in practice, only the ground state contributes to the thermal average, as all excited states are exponentially suppressed [32,34]. This definition can be applied only to gapped systems as gapless systems at infinitely small, but finite temperature lead to a thermal average over the infinitesimally low-lying excitations. As the SSE QMC approach is sampling finite systems, there cannot be a non-analytic point in the ground-state energy associated with a secondorder QPT [113]. Closely related with this statement, the ground state of a finite system cannot spontaneously break the symmetry of the Hamiltonian, as it is the case at a secondorder QPT [113]. An intuition to this statement can be built from a perturbative point of view. On a finite system, there is always a finite-order perturbative process coupling the hypothetical ground states of a symmetry-broken phase. Due to this finite-order coupling, there will always be a level repulsion between the states and no true degeneracy occurs. This means that, on a finite system, there is always a finite-size energy gap between the ground state and the excitations. As discussed in Section 2.2 on a finite system, there is a pseudo-critical point close to the parameter values of the QPT in the thermodynamic limit. For a system with gapped phases on both sides of the QPT, we expect the relevant finite-size gap to be the smallest in the vicinity of the pseudo-critical point. The finite-size gap at the pseudo-critical point is expected scales as $L^{-z\rho}$ with the linear system size L to the power of the dynamical correlation length exponent z [34,113,129,212] and q = 1 for a QPT below the upper critical dimension (see Section 2.3). This statement can be directly derived from the finite-size scaling form of the energy gap:

$$\Delta_L(r) = L^{-z\varphi} \Psi_\Delta(L^{\varphi/\nu} r) \,. \tag{289}$$

Note that the scaling dependence for systems above the upper critical dimension has not been confirmed yet by numerical studies. To perform simulations at effective zero temperature, there are two main approaches in the contemporary literature. Firstly, one just scales the simulation temperature with L for different linear system sizes. This is a valid approach for many systems as z = 1 is a common value for the critical exponent, indicating a space-time symmetry in the correlations. For long-range interacting systems such as the LRTFIM or unfrustrated long-range Heisenberg models, one has $z \leq 1$, which makes the scaling sufficient, but overly ambitious [21]. An improved version of this naive approach is to scale the simulation temperature with L^{zo} for different linear system sizes. This follows the expected scaling of the finite-size gap. The correct prefactor of the scaling is not known and accounted for in the scaling, and corrections to scaling for small system sizes can lead to errors. A more sophisticated approach to determine a suitable effective zero temperature for a finite system is discussed in Refs. [32,34,225]. This technique was introduced in Ref. [225] as the beta-doubling method. The idea is to study the convergence of observables $\langle \mathcal{O}_L(r,\beta) \rangle$ for a fixed system size and parameter value *r* at successively doubled inverse temperatures $\beta = 2^n$ of the simulation. The fraction of $\langle \mathcal{O}_L(r,\beta) \rangle / \langle \mathcal{O}_L(r,\beta_{\max}) \rangle$ is considered with β_{\max} being the largest considered β value. This quantity is used to probe the convergence of the observable $\langle \mathcal{O}_L(r,\beta) \rangle$ towards zero temperature. The temperature convergence can be gauged using a plot, as demonstrated in Figure 19. Note that the value of $\langle \mathcal{O}_L(r, \beta_{\max}) \rangle / \langle \mathcal{O}_L(r, \beta_{\max}) \rangle$ is always at one. Therefore, as a rule of thumb, one can validate simulation temperatures for which at least the two points prior to the last point also have a value close to one.

Although the beta-doubling scheme has more overhead than a naive scaling with L^{zq} , it provides multiple advantages. First of all, one does not need to know the dynamical critical exponents *z* beforehand. Secondly, in most cases, it is easy to implement: if one uses either way a simulated annealing scheme by successively cooling the simulation temperature during the thermalisation of the algorithm, the beta-doubling scheme can be easily incorporated in this procedure. Thirdly, the beta doubling provides a direct demonstration that a simulation is sampling zero-temperature properties. It captures intrinsically the relevant scaling L^{zq} , as well as potentially large prefactors and corrections to this scaling. Fourthly, it does not need to be applied to all parameter values *r*, as it suffices to perform the beta-doubling procedure only for the parameter values where the relevant finite-size gap is expected to be the smallest and use this temperature for the whole parameter range.



Figure 19. Illustration of the beta-doubling method for the one-dimensional LRTFIM in the shortrange regime with decay exponent $\sigma = 2.5$ at a transverse field of h = 1.25. The simulation starts at $\beta = 1$ (leftmost points) for every system size. The inverse temperature is then doubled in every beta-doubling step until the maximum $\beta_{max} = 2048$ is reached. All of the shown magnetisation curves seem to be converging to zero temperature. Larger systems with L > 1024 were discarded as they do not appear to be fully converged yet.

In general, the beta-doubling scheme can help to reduce computational cost by performing simulations at a tailored effective zero temperature. If computational time is not a critical factor, in many practical applications, one can use the temperature that is sufficient for the largest system and sample all smaller systems at the same temperature.

5.5. Overview: Path Integral Quantum Monte Carlo Algorithms for Long-Range Models

The QMC methods discussed so far are both based on the SSE formulation [39,213]. In parallel with the development of the SSE-based QMC methods for long-range systems, path integral (PI) QMC methods operating in continuous imaginary time have been introduced for extended Bose–Hubbard models with long-range density–density interactions [215,226–228] (see (290)). The main application of these methods in the context of long-range interactions is the determination and classification of ground states [64,229,230]. Since we focus on reviewing QPTs in long-range interacting systems, which are usually simulated using the SSE, we only skim over the basic concepts relevant for the PI QMC methods. One motivation for the development of these methods was the study of the Mott insulator to superfluid transition in the presence of disorder [231–233]. Further, long-range density–density interactions were added to the algorithms [226]. A great application case for these algorithms is the study of extended Bose–Hubbard models [64,229,230], which

are an effective description of ultracold atomic or molecular gases trapped in optical lattice potentials [56,58,61,62,64,68,229,234–238]. For a comprehensive explanation on how to derive effective Bose–Hubbard models for these systems, see Refs. [56,236–238].

A basic Hamiltonian studied with the world-line QMC methods in this context [226–228] reads

$$\mathcal{H} = -\mu \sum_{i} n_{i} + \sum_{ij} V_{i,j} n_{i} n_{j} - t \sum_{\langle i,j \rangle} \left(a_{i}^{\dagger} a_{j} + a_{j}^{\dagger} a_{i} \right)$$
(290)

with (hard-core) bosonic degrees of freedom on the lattice site *i* described by creation a_i^{\dagger} , annihilation a_i , and particle number n_i operators. The coupling μ describes a chemical potential and $V_{i,j} \leq 0$ the density–density interaction between particles at sites *i* and *j*, and the hopping amplitude t > 0.

PI QMC methods formulated in imaginary time are based on the PI formulation of the partition function:

$$Z = \operatorname{Tr}\left\{e^{-\beta\mathcal{H}}\right\} = \lim_{K \to \infty} \operatorname{Tr}\left\{\left(e^{-\Delta\tau\mathcal{H}}\right)^{K}\right\} \quad \text{with} \quad \Delta\tau = \beta/K$$
(291)

using the Suzuki–Trotter decomposition [115]. The PI formulation is used to extend the configuration space to all imaginary-time trajectories $|\alpha(\tau)\rangle$ of computational basis states [39,226], which are periodically closed in β ($|\alpha(0)\rangle = |\alpha(\beta)\rangle$) and which are connected by an imaginary-time evolution ($\langle \alpha(\tau_2) | e^{-\mathcal{H}(\tau_2-\tau_1)} | \alpha(\tau_1) \rangle$). These methods can be implemented in discrete [239] or continuous [226,240] imaginary time.

In the SSE approach, the configuration space is extended using the operator sequences arising from a high-temperature expansion of the exponential in the partition function. These configurations can be regarded as a series of discretely propagated states $|\alpha(p)\rangle$ with an artificial propagation index *p* labelling its position in the sequence. This introduction of an additional dimension to obtain a configuration space that can be sampled using MC in the PI and SSE QMC approach hints that both approaches are closely related. Studies of the close relationships between the SSE and PI representations can be found in Refs. [211,241]. Concepts developed in one of the two pictures can be equivalently formulated in the other one. For example, the concept of loops [215,216,242] and directed loops [215,217,242] can be applied in both formulations. The sampling of the Heisenberg model as described in Section 5.2 can be transferred to the PI QMC picture, as it is a direct application of the directed loop idea [217,225].

Since we will focus in this review on quantum-critical properties of magnetic systems, we only briefly summarise the major applications of PI QMC methods for extended (hardcore) Bose–Hubbard models with dipolar density–density interactions [64,229,230]. The PI QMC is used in this field to determine ground-state phase diagrams studying the emergence of solid, supersolid, and superfluid phases [64,229,230]. It has been demonstrated in Ref. [64] that long-range interactions stabilise more solid phases than the chequerboard solid with a filling of 1/2 present in systems with nearest-neighbour interactions. It has also been shown that long-range interactions lead to the emergence of supersolid ground states [64]. The great benefit of the QMC simulations in comparison to other methods (e.g., mean-field calculations [111]) is the quantitative nature of these methods. Therefore, the numerical study of these models is highly relevant, since experimental progress in cooling and trapping atoms and molecules with dipolar electric or magnetic moments [67,69,72,73,243] enables experiments to realise extended Hubbard models with long-range interactions [68]. We expect the implementation of SSE QMC techniques for this kind of model to be straightforward using the directed loop approach [217,225].

6. Long-Range Transverse-Field Ising Models

In this section, we review the ground-state quantum phase diagrams of the long-range transverse-field Ising model (LRTFIM) with algebraically decaying long-range interactions.

We emphasise how the Monte Carlo-based techniques introduced in this work are a reliable way to obtain critical exponents of quantum phase transitions (QPTs) in this model. The Hamiltonian of the LRTFIM is given by

$$\mathcal{H} = \frac{J}{2} \sum_{i \neq j} \frac{1}{|\vec{r}_i - \vec{r}_j|^{d+\sigma}} \sigma_i^z \sigma_j^z - h \sum_i \sigma_i^x$$
(292)

with Pauli matrices σ_i^{κ} and $\kappa \in \{x, z\}$ describing spins 1/2 located on the lattice sites \vec{r}_i . The Ising coupling is tuned by the parameter *J*. For J > 0, the Ising interaction is antiferromagnetic and, for J < 0, ferromagnetic. The amplitude of the transverse field is denoted by *h*. The positive parameter $(d + \sigma)$ governs the algebraic decay of the Ising interaction. Here, *d* denotes the spatial dimension of the system and σ is a tunable real-valued parameter. The limiting cases of the algebraically decaying long-range interaction are the nearest-neighbour interaction for $\sigma = \infty$ and an all-to-all coupling for $(d + \sigma) = 0$. As a side note, there is literature where $\alpha \equiv d + \sigma$ is used as a parameter to tune the decay of the long-range interaction. For this review, we stay with $d + \sigma$ in order to treat systems with different spatial dimensions *d* on the same footing.

The Hamiltonian (292) can be treated using the pCUT+MC method as described in Section 4. A generic, perturbative starting point for the LRTFIM is the high-field limit. The transverse field term is regarded as the unperturbed Hamiltonian and the Ising interaction as the perturbation. Using the Matsubara–Matsuda transformation [25,29,89]:

$$\sigma_i^x = 1 - 2b_i^\dagger b_i \qquad \qquad \sigma_i^z = b_i^\dagger + b_i , \qquad (293)$$

the Hamiltonian (292) can be brought into a hard-core bosonic quasi-particle picture:

$$\mathcal{H} = \epsilon_0 N + \sum_i b_i^{\dagger} b_i^{\dagger} + \sum_{i \neq j} \lambda_{i,j} \left(b_i^{\dagger} b_j^{\dagger} + b_i^{\dagger} b_j^{\dagger} + b_i^{\dagger} b_j^{\dagger} + b_i^{\dagger} b_j^{\dagger} \right).$$
(294)

Here, ϵ_0 is a constant, *N* is the number of sites, and $b_i^{(\dagger)}$ is a hard-core bosonic quasiparticle annihilation (creation) operator. The perturbation parameters are given by

$$\lambda_{i,j} = \frac{J}{4h} \frac{1}{|\vec{r}_i - \vec{r}_j|^{(d+\sigma)}} \,. \tag{295}$$

In the pCUT language, the perturbation associated with the perturbation parameter λ can be written as $\mathcal{V} = T_{-2} + T_0 + T_2$, where the hopping processes $b_i^{\dagger}b_j$ are contained in T_0 and the pair creation (annihilation) processes are in T_2 (T_{-2}).

Further, the Hamiltonian (292) is of the form to straightforwardly apply the SSE QMC algorithm reviewed in Section 5.1 (see Equation (226)). We recall that this QMC algorithm is sign-problem-free for arbitrary, potentially frustrated, Ising interactions.

The studied effects of algebraically decaying long-range Ising interactions on the ground-state phase diagram can be categorised into three scenarios:

First, for ferromagnetic interactions, there is a QPT with \mathbb{Z}_2 symmetry breaking between a ferromagnetic low-field phase and a symmetric *x*-polarised high-field phase. The intriguing effect of long-range interaction is the emergence of three regimes with distinct types of universality of the QPT. For large values of the parameter $\sigma > 2 - \eta_{SR}$, the universality class is the same as in the nearest-neighbour model. For intermediate σ values of $2d/3 < \sigma < 2 - \eta_{SR}$, the critical exponents change as a function of σ and the criticality can be described by a non-trivial long-range theory. For $\sigma < 2d/3$, the long-range interaction lowers the upper critical dimension below the physical dimension of the model and the transition becomes a long-range mean-field transition. We review the recent studies investigating this model in Section 6.1 with a particular focus on the ϕ^4 -theory in Section 6.1.1 and on numerical studies investigating the critical properties in Section 6.1.2. The aspects regarding FSS above the upper critical dimension are summarised in Section 6.1.3. The second relevant category is antiferromagnetic Ising interactions where the nearestneighbour couplings form a bipartite lattice. In this case, there is a QPT with \mathbb{Z}_2 symmetry breaking between an antiferromagnetic low-field phase and a symmetric *x*-polarised highfield phase. The current state of literature indicates no change in the universality class of the phase transition in dependence of the long-range interaction. We summarise and review the recent progress in studying this case in Section 6.2.

The last aspect is scenarios where the nearest-neighbour couplings do not form a bipartite lattice. In this case, the long-range interactions may substantially alter the quantum phase diagram of the nearest-neighbour model. We summarise and review the recent progress regarding this case in Section 6.3.

6.1. Ferromagnetic Long-Range Transverse-Field Ising Models

The ferromagnetic TFIM with nearest-neighbour interactions in $d \ge 1$ dimensions is a paradigmatic model to display a QPT in a (d + 1)D-Ising universality class. This transition describes the non-analytic change in the ground-state between a \mathbb{Z}_2 symmetry-broken low-field phase and a symmetric *x*-polarised high-field phase. By adding algebraically decaying long-range interactions to this model, one can study how such interactions can alter non-universal, as well as universal properties of a QPT and which new features can emerge.

In the case of ferromagnetic interactions, the algebraically decaying long-range interaction stabilises the symmetry-broken phase and the QPT shifts to larger *h* values [25,29,32,34]. An intuitive way to understand this behaviour is that, due to the long-range interaction, more connections are introduced that align the spins in the *z*-direction, and the energy cost for a single spin flip increases. In the limit $\sigma \rightarrow 0$, the critical value of the transverse field diverges as the cost for a single spin flip becomes extensive.

In this review, we focus exclusively on the regime of so-called weak long-range interactions with an algebraic decay exponent $\sigma > 0$ [21]. The research field of strong long-range interactions considers $\sigma \leq 0$ [5,42–45]. For weak long-range interactions, the Hamiltonian (292) is well defined in the thermodynamic limit, and common properties such as the ground-state additivity and thermodynamic quantities are well defined [5]. For strong long-range interactions, the energy of the ferromagnetically aligned state is superextensive, and in order to study the model in the thermodynamic limit, an appropriate rescaling of the Ising coupling with the system size is required [5].

The significance of the ferromagnetic LRTFIM comes from its paradigmatic nature of displaying changes in the universal behaviour due to the long-range interaction [20,21,25,29,32]. It has been known since the advent of the theory of classical phase transitions that algebraically decaying long-range interactions are a potential knob to alter the fixed-point structure of the renormalisation flow and, therefore, change the universal properties (e.g., critical exponents) [1,2,140]. Coming from the limit $\sigma = \infty$ of nearest-neighbour interactions, the QPT of a *d*-dimensional model remains in the (d + 1)D-Ising universality class until $\sigma = 2 - \eta_{SR}$ (η_{SR} is the anomalous dimension critical exponent of the nearestneighbour universality class). This means that the critical exponents are constant as a function of σ and the fixed point associated with the QPT remains the one of the nearestneighbour model. For $\sigma = 2 - \eta_{SR}$, the RG fixed point associated with the QPT changes to the one of a non-trivial long-range interacting theory, and the critical exponents become σ -dependent [21]. The upper critical dimension of this theory with long-range interaction is lowered with respect to the short-range model as $d_{\rm uc} = 3\sigma/2$. If the (fixed) spatial system dimension d is larger than or equal to the σ -dependent upper critical dimension, the universality class describing the QPT enters a long-range mean-field regime. The three different regimes, as well as their boundaries in dependence of the dimensionality of the system are visualised in Figure 20.

In the following, we recapitulate the basic field-theoretical arguments leading to the distinction between three universality regimes (see Section 6.1.1). We will further emphasise the most relevant aspects for the two long-range regimes. In the non-trivial,

intermediate regime, the precise values of the critical exponents depend on σ and need to be determined numerically. Therefore, we review the recent numerical studies computing the critical exponents in Section 6.1.2. To conclude the discussion, we outline how to utilise ferromagnetic long-range Ising models to study QPTs above the upper critical dimension (see Section 6.1.3).



Figure 20. Sketch of the three distinct universality regimes of the QPT in the ferromagnetic LRTFIM. For one- and two-dimensional systems, all three regimes exist, and the boundaries can be obtained using the expressions in the figure. For $d \ge 3$, there is only the long-range mean-field and the nearest-neighbour mean-field regime with a boundary at $\sigma = 2$.

6.1.1. ϕ^4 -Theory for Quantum Rotor Models with Long-Range Interactions

Starting with the short-range interacting action (see Equation (A3)) of the *n*-component quantum rotor model, Dutta et al. [20] introduced an action for long-range interacting rotor models, by adding the algebraic decay between interacting fields as an additional term to the action in real space. Note the the action for the Ising model is equivalent to the one-component (scalar) rotor model [113]. By performing a Fourier transformation of the resulting action, the authors obtained the action:

$$S_{\tilde{\phi}} = \int \frac{d^d q}{(2\pi)^d} \int \frac{d\omega}{2\pi} [\tilde{g}\omega^2 + r + aq^\sigma + bq^2] \tilde{\phi}(q, i\omega) \tilde{\phi}(-q, -i\omega)$$

$$+ u \int \frac{d\omega_1}{2\pi} \dots \frac{d\omega_4}{2\pi} \int \frac{d^d q_1}{(2\pi)^d} \dots \frac{d^d q_4}{(2\pi)^d} \delta^d(q_1 + \dots + q_4) \delta(\omega_1 + \dots + \omega_4) \times$$

$$\times [\tilde{\phi}(q_1, i\omega_1) \tilde{\phi}(q_2, i\omega_2)] [\tilde{\phi}(q_3, i\omega_3) \tilde{\phi}(q_4, i\omega_4)]$$
(296)

with a, b > 0, σ from the decay exponent $d + \sigma$ of the microscopic model, and r, u coupling constants as in the nearest-neighbour case. The q^{σ} term, which is new compared to the nearest-neighbour theory, results from the Fourier transformation of the added long-range interacting term:

$$\int d^d x \, \int d^d y \, \int d\tau \, \frac{\phi(x,\tau)\phi(y,\tau)}{|x-y|^{d+\sigma}} \tag{297}$$

of the order-parameter field ϕ [40]. As the relevant modes for the QPT are the ones with long wavelengths, it is clear that, for $\sigma \ge 2$, the q^2 term gives the leading contribution in q; the q^{σ} term can be neglected and, therefore, the system is, in this case, described by the short-range interacting theory. In contrast to this, a different critical behaviour distinct from the short-range case is possible for $\sigma < 2$. With the same argument as above, the q^2 term becomes negligible for $\sigma < 2$. In the following, we will focus on this regime, where the behaviour differs from the nearest-neighbour case. In order to gain insight into the QPT affected by long-range interactions, an investigation of the Gaussian theory is a good starting point to enter the framework of perturbative renormalisation group calculations [20,21,113] like the ϵ -expansion in Ref. [20]. From Equation (296), one can directly read off the propagator of the Gaussian theory as

$$G_0(q,\omega)^{-1} = q^{\sigma} + \tilde{g}\omega^2 + r.$$
(298)

From field-theoretical power-counting arguments, Dutta et al. [20] derived several properties of the Gaussian theory. Using the divergence of the mass renormalisation term, the lower critical dimension below which no phase transition is occurring can be derived as $d_{\rm lc} = \sigma/2$ [20]. By regarding the Gaussian propagator, one finds directly that $\eta = 2 - \sigma$ as η is defined by the deviation from 2 of the leading power by which the momentum enters the propagator (see Equation (A37)). Note that, in the long-range case, the Gaussian theory has an $\eta \neq 0$. From further power-counting analysis, it is possible to derive more critical exponents from the Gaussian theory [20]:

$$\gamma = 1$$
 $\nu = \frac{1}{\sigma}$ $z = \frac{\sigma}{2}$ $\eta = 2 - \sigma.$ (299)

In the limit $\sigma \rightarrow 2$, the short-range mean-field exponents are recovered [20]. With the argument that the hyperscaling relation still holds directly at the upper critical dimension, Dutta et al. [20] derived the upper critical dimension by inserting the long-range Gaussian exponents (see Equation (299) and $\alpha = 0$).

$$2 - \alpha = \nu(d+z) \longrightarrow 2 = \frac{1}{\sigma}(d_{\rm uc} + \frac{\sigma}{2}) \Leftrightarrow d_{\rm uc} = \frac{3\sigma}{2}.$$
 (300)

In addition to the investigation of the Gaussian part of the action in Equation (296), one can derive non-trivial exponents below the upper critical dimension by deriving perturbative corrections to the Gaussian exponents in an ϵ -expansion around the upper critical dimension [113]. Dutta et al. [20] established with a one-loop renormalisation group expansion of flow equations that the Gaussian fixed point is stable for $d \ge d_{uc}$. Therefore, the long-range Gaussian exponents (see Equation (299)) are valid for $\sigma \le 2D/3$. For $d < d_{uc}$, the correction indicates a flow to a non-trivial fixed point. The first-order corrections of the ϵ -expansion for ν and γ are provided [20], and the first-order corrections to η and z are argued to be zero [20]. It should be noted that the dynamic correlation exponent $z \ne 1$ for $\sigma < 2$ and, therefore, correlations are not isotropic in the space and imaginary-time direction [20,21]. "For any value of $\sigma < 2$, η sticks to its mean-field value $2 - \sigma$, because the renormalization does not generate new q^{σ} terms" [20]. This also fits well with the claim that $\sigma = 2 - \eta_{SR}$, because, then, there is no discontinuity at the boundary [1–3,21,244].

A recent study by Defenu et al. [21] with a functional renormalisation group approach investigated the flow of couplings in an effective action with anomalous dimension effects. This analysis showed that the boundary between the short-range and the non-trivial longrange regime does not occur at $\sigma = 2$, but at $\sigma = 2 - \eta_{SR}$. So, it is shifted towards lower σ by the anomalous dimension of the short-range criticality. This anomalous dimension consideration in [21] is mainly driven by the following argument: From the functional renormalisation group ansatz, the authors derived two flow equations for the coupling of the long-range coupling Z_k (connected to *a* in Equation (296)) and the short-range coupling $Z_{2,k}$ (connected to *b* in Equation (296)):

$$\partial_t Z_k = (2 - \sigma - \eta) Z_k \tag{301}$$

$$\partial_t \eta = \frac{\partial_z Z_{2,k}}{Z_{2,k}} \,. \tag{302}$$

The first flow Equation (301) has fixed points for $\eta = 2 - \sigma$ or $\lim_{k\to 0} Z_k = 0$. The first fixed point results in long-range exponents, whereas the second fixed point means that the long-range coupling becomes irrelevant.

In addition to this anomalous dimension effect, the authors in [21] also calculated critical exponents in the non-trivial intermediate phase with their functional renormalisation group technique. They also compared their results with Monte Carlo simulations of the dissipative non-Ohmic spin chains, which can be mapped to the same behaviour as the long-range interacting quantum Ising model [245], showing reasonable agreement.

6.1.2. Critical Exponents and Critical Points for One- and Two-Dimensional Systems

The criticality of the ferromagnetic LRTFIM has been studied by a wide variety of methods with different strengths and weaknesses. Overall, all methods qualitatively

agree with each other and confirm the three expected regimes from the field-theoretical analysis [20,21], namely a long-range mean field, a short-range regime, and an intermediate non-trivial long-range regime, which connects the two limiting regimes with monotonously changing exponents. In the limit $\sigma \rightarrow 0$, the value of the critical field h_c diverges as the energy cost for a single spin flip in the low-field phase becomes extensive and the quantum fluctuations introduced by the transverse field fail to compete with the Ising interaction. In the limit $\sigma \rightarrow \infty$, the value of h_c converges towards the exact value for the nearest-neighbour transverse-field Ising model [25,29,32,246].

In the case of the LRTFIM on the one-dimensional linear chain, the model has been studied by perturbative continuous unitary transformation with Monte Carlo embedding (pCUT + MC) [29,31,34,247], the functional renormalisation group (FRG) [21], the density matrix renormalisation group (DMRG) [28,248], as well as stochastic series expansion (SSE) [32,34,247,249], path integral (PI) [33] QMC, and QMC with stochastic parameter optimisation (QMC+SPO) [250].

The critical value of the transverse field was calculated with the pCUT method [29,31,34] by estimating the gap closing using DlogPadé extrapolation of the perturbative gap series coming from the high-field phase, by the DMRG using finite-size scaling of the fidelity susceptibility [28], and by SSE QMC calculations using finite-size scaling of the magnetisation [32,249]. The finite-temperature transition points were studied by PI QMC calculations [33] for $\sigma = 0.5$ in the long-range mean field regime and $\sigma = -0.95$ in the strong long-range regime.

Depending on the method, different critical exponents were extracted. Only in the case of pCUT + MC and SSE QMC calculations[34,247], all three regimes were investigated. By extracting three independent critical exponents, the authors were able to provide the full set of critical exponents for each method individually. In the case of the pCUT + MC approach [34,247], the exponents were computed by (biased) DlogPadé extrapolants of highorder series of the gap $(z\nu)$, the one-quasiparticle static spectral weight $((2 - z - \eta)\nu)$, and the control-parameter susceptibility (α). In SSE QMC studies [34,247], the exponents were calculated using data collapses of the magnetisation $(\beta/\nu \text{ and } \nu)$ and the order-parameter susceptibility (γ/ν). The DMRG study [28] applied the method of data collapse to the fidelity susceptibility to extract the single exponent ν in the non-trivial long-range regime. Another DMRG study [248] extracted critical exponents by using finite-size scaling for the energy gap (z) and the squared order parameter ($2\beta/\nu$ and ν). Using the FRG [21], the exponents ν and z were calculated, from which one can additionally calculate the exponent α . In the PI QMC study [33], the finite-temperature criticality of the LRTFIM was investigated for $\sigma = 0.5$ in the long-range mean field regime and $\sigma = -0.95$ in the strong long-range regime. The study includes the extraction of an exponent θ_t , which can be related to ν , from the Binder cumulant and classical order-parameter susceptibility via finite-size scaling, as well as the exponent $\gamma \theta_t$ from finite-size scaling of the classical order-parameter susceptibility. In Figure 21, the critical field h_c and the canonical critical exponents are plotted for the discussed studies.

The two-dimensional case of the square lattice is less studied. The critical values of the transverse field were only calculated by the pCUT + MC [29] and SSE QMC method [32,249]. There is so far no study that has extracted the full set of critical exponents altogether. With pCUT [29], the gap exponents zv were also extracted by DlogPadé extrapolants of the gap series. In the FRG study [21], the critical exponents z and v were calculated, and in the SSE QMC study [32], the critical exponents β and v were extracted from the data collapses of the magnetisation. These results are summarised in Figure 22. Note that we present the results from pCUT + MC calculations from Ref. [29] with one additional order to the maximal order in the perturbation parameter.



Figure 21. Critical field values and exponents from numerical studies of the ferromagnetic LRTFIM on the linear chain. The panels display $h_c/|J|$ (top), α (second row left), β (second row right), γ (third row left), δ (third row right), η (fourth row left), ν (fourth row right), and z (bottom). The labels refer to the references in the following way: "SSE QMC (2021) [32,249]", "SSE QMC (2022) [34,247]", "pCUT+MC (2022) [34,247]", "pCUT+MC (2022) [34,247]", "pCUT+MC α from SSE (2022) [34,247]", "DMRG (2018) [28]", "DMRG (2019) [248]", "FRG (2017) [21]", "PI QMC (2021) [33]", and "QMC+SPO (2023) [250]". The values for the critical exponents of the transition in the nearest-neighbour model $\alpha = 0$, $\beta = 1/8$, $\gamma = 7/4$, $\delta = 15$, $\eta = 1/4$, $\nu = 1$, and z = 1 [246,251,252] and in the long-range mean-field regime $\alpha = 0$, $\beta = 1/2$, $\gamma = 1$, $\delta = 3$, $\eta = 2 - \sigma$, $\nu = 1/\sigma$, and $z = \sigma/2$ [20,21] are given by the dashed lines.



Figure 22. Critical field values and exponents from numerical studies of the ferromagnetic LRTFIM on the two-dimensional square lattice. The panels display $h_c/|J|$ (top), ν (middle left), z (middle right), $z\nu$ (bottom left), and β (bottom right). The data points "SSE QMC (2021)" for h_c/J , ν , and β are from Ref. [32,249]. The data points "pCUT+MC (2019)" for h_c/J and $z\nu$ are from Ref. [29]. The data points "FRG (2017)" for ν and z originate from the functional RG study in Ref. [21]. The values for the critical exponents of the transition in the nearest-neighbour model $\nu = 0.629971(4)$, $\beta = 0.326419(3)$, and z = 1 [253] and in the long-range mean-field regime $\nu = 1/\sigma$, $z = \sigma/2$, and $\beta = 0.5$ [20,21] are given by the dashed lines.

Overall, the different methods all perform well even for relatively small σ . For the pCUT and SSE QMC studies, where all three regimes were simulated, the limiting cases of

long-range mean-field criticality and nearest-neighbour criticality were correctly reproduced, which underlines the reliability of the presented results in the intermediate regime. The main challenge that the methods are facing is to sharply resolve the regime boundaries due to rounding effects, which are probably due to the limited length scales in the simulation. Additionally, the boundary to the long-range mean field regime is spoiled by logarithmic corrections to scaling at the upper critical dimension. Even in the case of pCUT + MC, which is a method operating in the thermodynamic limit, there is an intrinsic limit of the length scale due to the finite order of the series. In particular, the rounding makes it hard to verify (or falsify) the claim that the boundary between the short-range and non-trivial long-range regime is shifted $2 \rightarrow 2 - \eta_{SR}$. For 1D, the boundary is shifted to $\sigma = 1.75$, which could be resolvable in extensive simulation. However, in 2D, the boundary is only marginally shifted to $\sigma \approx 1.964$, which is probably not resolvable by the reviewed methods. The best boundary resolution is probably given by the SSE QMC study [34]. If one is interested in studying the shift of the boundary, one could, therefore, push the SSE QMC simulations of the one-dimensional chain to larger systems for $\sigma \lesssim 2$. Another difficulty for the methods operating on finite systems is the breakdown of the common hyperscaling relation and standard FSS above the upper critical dimension, which affects the processing of data in the long-range mean-field regime for $\sigma < 2d/3$. The scaling above the upper critical dimension was already outlined in Section 2.3. However, we will recapitulate the most important points in the following Section 6.1.3, not only because this modified scaling was used in several of the above-mentioned methods to extract critical exponents [32–34], but also because the LRTFIM constitutes a testing ground for quantum Q-FSS scaling and the modified scaling of the correlation length was confirmed in a study of the LRTFIM [34].

6.1.3. Scaling above the Upper Critical Dimension

As discussed in Section 6.1.1, for $\sigma < 2 - \eta_{SR}$, the action describing the QPT and its universality changes from short range to long range. In this long-range regime, the upper critical dimension $d_{uc} = 3\sigma/2$ was found to depend on the value of σ [20,21]. This means that the connectivity of the system increases for decreasing σ and a smaller spatial dimension d < 3 is already sufficient to reach or even exceed the upper critical dimension. The sigma below which the model is above its upper critical dimension is $\sigma_{uc} = 2d/3$. In this regime, it is not possible to correctly extract all critical exponents from the standard FSS, and the common hyperscaling relation:

$$2 - \alpha = (d + z)\nu \tag{303}$$

becomes invalid as the critical exponents become independent of the dimension *d*. The origin of this discrepancy comes from the effect of dangerous irrelevant variables that one has to take into account in the derivation of scaling above the upper critical dimension. When doing so, one obtains a modified hyperscaling relation:

$$2 - \alpha = \left(\frac{d}{9} + z\right)\nu\tag{304}$$

with the exponent $\varphi = \max(1, d/d_{uc})$. This relation can be used for the conversion of different critical exponents [34,35].

Additionally, the standard FSS scaling form of an observable \mathcal{O} with power-law singularity $\mathcal{O}(r, L^{-1} = 0) \sim |r|^{\omega}$ is modified to [34]

$$\mathcal{O}(r, L^{-1}) = L^{-\omega \varrho/\nu} \Psi(L^{\varrho/\nu} r)$$
(305)

following the formulation of the Q-FSS for classical [15–19,41] and quantum [34] systems. Finite-size methods that rely on the method of data collapse or other FSS-based techniques have to use this adapted formula to extract all the critical exponents successfully [34]. For a more elaborate description of quantum Q-FSS, please refer to Section 2.3 or to Ref. [34].

On the other side, the LRTFIM can also be used as a testing ground for quantum Q-FSS. One key difference that distinguishes the Q-FSS from standard FSS is that the correlation sector is also affected by the DIV in the case of the Q-FSS. This leads to a modified scaling of the characteristic length scale $\xi_L(r) = L^{\varrho} \Xi(L^{\varrho/\nu}r)$ with the system size instead of $\xi_L(r) \sim L\Xi(L^{1/\nu}r)$, as it is for standard FSS. The characteristic length scale was measured in the SSE QMC study [34] in the long-range mean-field regime, and the exponent ϱ was extracted by a data collapse with fixed mean field value $\nu = \sigma^{-1}$. The extracted values coming from Ref. [34] are presented in Figure 23 and are clearly in line with the predictions made by the Q-FSS.



Figure 23. Exponent φ extracted by the data collapse of $\xi^{(LR\omega)}$ and $\xi^{(LR\tau)}$ for different decay exponents of the LRTFIM on the linear chain [34]. The black dashed line depicts the prediction by the Q-FSS $\varphi = \max(1, d/d_{uc})$, while the grey dashed line shows the prediction from standard FSS $\varphi = 1$. In the regime above the upper critical dimension for $\sigma < 2/3$, the predictions start to deviate and the extracted values for φ are clearly in line with the Q-FSS scenario. Figure adapted from Ref. [34].

6.2. Antiferromagnetic Long-Range Transverse-Field Ising Models on Bipartite Lattices

In this section, we review results for the antiferromagnetic LRTFIM on lattices where the coupling structure of the nearest-neighbour couplings forms a bipartite lattice. A latticeor, more generally, a graph—is called bipartite if the sites can be split into two disjoint sets A and B so that there are no edges between sites within each of the two sets A and B. In the case of a lattice, A and B are called bipartite sublattices. Note that, only in the nearestneighbour limit on bipartite lattices, there is an exact duality between the ferromagnetic and antiferromagnetic TFIM, and the quantum-critical properties coincide. This duality comes from a sublattice rotation of π along the *x*-axis for the spins in one of the two sublattices. In the nearest-neighbour limit, there is an *x*-polarised symmetric high-field phase and a \mathbb{Z}_2 symmetry-broken low-field phase. The low-field phase is adiabatically connected to the zero-field limit. In the zero-field limit, there are two ground states associated with the states where the spins on the sites in one set of the bipartite sublattices are pointing in one direction while the others point in the other direction. Note that these states can be directly mapped onto the two ferromagnetic ground states by the above-mentioned sublattice rotation. As for the ferromagnetic nearest-neighbour TFIM, the QPT between these two phases is of (d + 1)D-Ising universality.

The antiferromagnetic long-range interaction beyond the nearest-neighbours induces a hierarchy of competing interactions. Due to the long-range interactions, there is no longer a bipartite coupling structure as the spins cannot be aligned such that the energy of all interactions is minimised. Nevertheless, there is evidence that the two ground states in the low-field phase of the nearest-neighbour limit are adiabatically connected to the ground states in the low-field phase for all $(d + \sigma) > 0$ [23–26,29,31,32,111].

For the all-to-all connected case $(d + \sigma) = 0$, every state with zero *z*-magnetisation is a ground state of the system [30,40] at h = 0. The Hamiltonian (292) can then be written in terms of total spin $\sigma_{\text{tot}}^{\kappa} = \sum_{i} \sigma_{i}^{\kappa}$ with $\kappa \in \{x, z\}$ as

$$\mathcal{H} = -h\sigma_{\rm tot}^x + \frac{J}{2}(\sigma_{\rm tot}^z)^2 - \frac{J}{2}N$$
(306)

with the total number of spins $N \rightarrow \infty$ [30,40]. For all finite transverse fields h > 0, the ground state of Equation (306) is directly located in the *x*-polarised phase [40,254].

The natural questions arising for the QPT in the antiferromagnetic LRTFIM on the bipartite lattices are similar to the ferromagnetic LRTFIM: How does the algebraically decaying long-range interaction influence the critical field value of the QPT? Is there a change in the universality class of the QPT depending on the precise values of the decay exponent $(d + \sigma)$?

Regarding the critical field values in one-dimensional chains, there are MC-based studies using the SSE QMC (see Section 5) [32] and pCUT+MC (see Section 4) [25,31], as well as the DMRG [24,26,248] and matrix product states [23]. We summarise the results regarding critical field values and selected critical exponents of the studies mentioned above in Figure 24. In two-dimensional systems, there are MC-based studies for the square lattice using SSE QMC (see Section 5) [32] and pCUT+MC (see Section 4) [29] (for selected results, see Figure 25). Variational methods in two dimensions such as projected entangled pair states [255,256] or DMRG calculations [257] have not been applied for antiferromagnetic systems on bipartite lattices. Entanglement-based methods become more challenging from the increased entanglement due to the area law in two dimensions [258]. Further, there are reports about the violation of the area law for one-dimensional systems with long-range interactions [23,24]. On general grounds, one cannot efficiently represent algebraically decaying long-range interactions in these techniques [259,260]. However, there exist DMRG calculations for antiferromagnetic LRTFIMs on non-bipartite lattices [27,54,55,109].

In general, it has been numerically demonstrated that the critical field values decrease monotonically as a function of σ from the nearest-neighbour value at $\sigma = \infty$ to zero at $(d + \sigma) = 0$ [23–26,29,31,32,248]. Regarding the QPT on the antiferromagnetic chain, the established picture is that it is of the (1 + 1)D-Ising type for all $\sigma \ge 1.25$ [23–26,31,32,248]. In the regime of ultra-long-range couplings $\sigma \leq 1.25$, recent finite-size DMRG findings by G. Sun [26] and R. Puebla et al. [248] suggest that the (1 + 1)D-Ising universality even holds for any $\sigma > -1$. The only study indicating a change in criticality below the boundary of $\sigma \leq 1.25$ uses matrix product states generalising the time-dependent variational principle (TDVP) [23]. Note that there is no further theoretical motivation for this boundary besides this single numerical study, which does not provide convincing evidence for its existence. The more recent results from Refs. [26,248] challenge these findings and establish the (1 + 1)D-Ising universality for all $\sigma > -1$. The studies using MC-based techniques [25,31,32] cannot extract reliable critical exponents in this regime. In the SSE QMC study [32], the increasing competition of the long-range interactions is the major problem making the numerical simulations in this regime impractical. If the long-range interaction becomes more prominent, this leads to similar algorithmic challenges as in frustrated systems [32,40,261,262]. The autocorrelation times of the SSE QMC algorithm increase as more and more bond operators are present in the operator sequence, and the field operators diffuse only slowly [32]. Nevertheless, following the trend of the results in Ref. [32], the scenario of a single universality class across the entire σ range is plausible. Regarding the pCUT+MC studies [25,31], the main obstacle to extract information about the quantum criticality in the regime $\sigma < 1$ is the fact that the critical field value is decreasing. Refs. [25,31] use a perturbative expansion around the high-field limit; therefore, if the critical field value increases, then the extrapolation of the gap series becomes increasingly challenging.



Figure 24. Critical field values and exponents from numerical studies of the antiferromagnetic LRTFIM on the chain. The upper panel displays critical field values h_c/J ; the lower left panel displays values for the critical exponents ν and $z\nu$ and, the lower right panel, values for the critical exponents β . The data points "SSE QMC (2021)" and "SSE QMC $L \le 64$ (2021)" for h_c/J , ν , and β are from Ref. [32,249]. The data points "pCUT+MC (2019)" for h_c/J and $z\nu$ are from Ref. [29]. The data points "TDVP (2012)" for h_c/J , ν , and β originate from Ref. [23]. The exponents ν and β are calculated from the scaling dimensions in Ref. [23] under the assumption of z = 1, which is reasonable according to [248]. The data points "DMRG (2016)" for h_c/J are from Ref. [24]. The data points "DMRG (2017)" for h_c/J and ν originate from Ref. [26]. The data points "DMRG (2019)" for h_c/J are from Ref. [248]. The values for the critical exponents of the transition in the nearest-neighbour model $\nu = 1$, $\beta = 1/8$, and z = 1 [246,251,252,263] are given by the black dashed lines.

To summarise, there are studies indicating a change from the (1 + 1)D-Ising criticality on the chain for small σ values [23,24]. These studies also report a possible breakdown on the area law of the entanglement entropy for the ground state in this regime [23,24], which is a vital prerequisite for the methods they implemented. Two more recent studies [26,248] demonstrate no change in the universality class, which is backed by [25,30,31]. For two-dimensional systems, the universality class is demonstrated to be (2 + 1)D-Ising for $\sigma > 0.5$ [29,32], and no trend is reported that the universality class should change for smaller values of σ .



Figure 25. Critical field values and exponents from numerical studies of the antiferromagnetic LRTFIM on the square lattice. The upper panel displays critical field values h_c/J ; the lower left panel displays values for the critical exponents ν and $z\nu$ and, the lower right panel, values for the critical exponents β . The data points "SSE QMC (2021)" for h_c/J , ν , and β are from Ref. [32,249]. The data points "pCUT+MC (2019)" for h_c/J and $z\nu$ originate from Ref. [29]. The values for the critical exponents of the transition in the nearest-neighbour model $\nu = 0.629971(4)$, $\beta = 0.326419(3)$, and z = 1 [253] are given by the dashed lines.

In general, a great algorithmic challenge that remains for the MC-based methods with regard to the antiferromagnetic LRTFIM on bipartite lattices is a reliable study of the regime of small values $(d + \sigma)$. Here, as discussed above, all of the commonly applied methods have a handicap in some way.

6.3. Antiferromagnetic Long-Range Transverse-Field Ising Models on Non-Bipartite Lattices

We start with an antiferromagnetic nearest-neighbour interacting TFIM on a nonbipartite lattice (e.g., the sawtooth chain, triangular lattice, Kagome lattice, pyrochlore lattice) at zero field. In the case of non-bipartite lattices, one can always find an odd number of Ising bonds (edges) that form a closed loop. The presence of a loop of odd length means it is impossible to satisfy all antiferromagnetic couplings simultaneously. The phenomenon of not being able to minimise all interactions due to geometrical lattice constraints is called geometrical frustration [264]. A notion for the strength of geometric frustration can be defined using the resulting ground-state degeneracy [265]. A theoretical tool to access the ground-state degeneracy is through Maxwell counting arguments [265–267]. For systems with a large degree of frustration, e.g., the antiferromagnetic nearest-neighbour TFIM on the triangular or Kagome lattice, there is a finite residual entropy per site in the thermodynamic limit [168,268–271].

In general, extensively degenerate ground-state spaces due to frustration pose a great resource for emergent exotic quantum phenomena. Quantum fluctuations introduced, e.g., by a transverse field result in a breakdown of the extensive ground-state degeneracy and the potential emergence of non-trivial ground states [270,271]. In general, it is useful to think about the low-field physics as a degenerate perturbation theory problem on the zero-field ground-state space [30,32,270,271]. We will later review that this is a reasonable framework in order to treat the breakdown of the degenerate subspace due to fluctuations and algebraically decaying long-range interactions on the same footing [30,32,112].

Perturbing extensively degenerate ground-state spaces with fluctuations may result in several distinct scenarios. First, a distinct symmetry-broken order can emerge for infinitesimal perturbations (order-by-disorder) [168,270–273]. Further, a direct realisation of a symmetry-unbroken phase may occur (disorder-by-disorder). This phase can either be trivial [168,274] or exotic, e.g., quantum spin liquids [99–101,275,276].

For the first part of this section, we review the antiferromagnetic LRTFIM on the triangular lattice [27,29,30,32,40,110,277]. In the nearest-neighbour limit of that model, the zero-field case has an extensively degenerate ground state [268,270,271,273]. When adding an infinitesimal transverse field, this ground-state degeneracy breaks down and a $\mathbb{Z}_2 \times \mathbb{Z}_3$ -symmetry-broken clock order emerges from an order-by-disorder mechanism [270,271,273]. The effective Hamiltonian describing the breakdown of the degeneracy in leading order can be expressed as a quantum dimer model [270,271,273]. By investigating the degenerate ground-state space, it was observed that there are local spin configurations in which spins can be flipped without leaving the ground-state space. These configurations, called flippable plaquettes, consist of a spin surrounded by six spins with alternating orientation. It can be easily seen that flipping the spin in the middle of the flippable plaquette results in turning the plaquette by $\pi/3$ (see Figure 26).



Figure 26. Illustration of the four local spin configurations called flippable plaquettes. The arrows denote the local spin orientation in the *z*-direction. The grey lines in the back visualise the triangular lattice. Solid (dotted) black lines are depicted on (anti)ferromagnetically aligned bonds. Note that flipping the spin in the centre of each configuration maps the left to the right configuration in each row and vice versa.

The effective low-field quantum dimer model [270,271,273], consisting of the plaquette rotation term, is, therefore, the hexagonal-lattice version of the Rokhsar–Kivelson quantum dimer model with t = h and v = 0 [270,271,278]:

$$H_{\text{QDM}} = t\left(\left|\mathbf{I}\right\rangle \left\langle \left\langle \mathbf{I}\right| + \text{h. c.}\right) + v\left(\left|\mathbf{I}\right\rangle \left\langle \mathbf{I}\right\rangle \right| + \left|\left\langle \mathbf{I}\right\rangle \left\langle \left\langle \mathbf{I}\right|\right)\right.$$
(307)

From this mapping, the nature of the state at h > 0 was predicted, as well as the breakdown of the order with a 3D-XY QPT [270,271]. Further numerical studies extended the insights into this system [168,262,273]. The critical field value at which the phase transition between the symmetric *x*-polarised paramagnetic high-field phase and the symmetry-broken gapped clock-ordered low-field phase occurs is at $h/J = 1.65 \pm 0.05$ [168,273].

To infer the nature of the full phase diagram including the transverse field and the long-range interaction, the next step is to discuss the zero-field limit of the Ising model with long-range interactions. As the antiferromagnetic long-range interactions introduce a hierarchy of constraints due to the interaction between sites further apart, the long-range interaction breaks the ground-state degeneracy of the nearest-neighbour zero-field as well [30,110,277,279]. It is important to emphasise that the long-range interaction lifts the degeneracy in a different way and stabilises a gapped stripe phase breaking the translational symmetry in a different way than the clock order promoted by the transverse field [30,110,277,279,280]. This plain stripe state is sixfold degenerate by rotations around $\pi/3$, as well as spin flips [30,110,277,279]. The spins are aligned in straight lines with alternating *z*-orientation [30,110,277,279]. An important aspect is that these stripe states are gapped and, therefore, stable against finite transverse fields [30,110,277].

Regarding the limit of large transverse fields, pCUT+MC was used to investigate the QPT between the *x*-polarised high-field phase and the clock-ordered phase [29,30]. Similar to the antiferromagnetic models on the bipartite lattices, the critical field value is reduced by the long-range interaction towards smaller field strengths [29,30] (see Figure 27). On the triangular lattice, the universality class of the transition from the high-field to a clock-ordered phase is of the (2+1)D-XY type [270,271,273]. It is reported to remain in this category for all σ values investigated ($\sigma \ge 1$) [29]. For small σ values, the extrapolation of high-field series expansion is inconclusive [29,30]. Infinite DMRG (iDMRG) investigations on triangular lattice cylinders suggest that the clock order vanishes for small values of σ and a direct transition to a low-field stripe phase occurs [27]. A first-order phase transition between the *x*-polarised high-field phase and a low-field stripe phase would be consistent with the incapability of the high-field gap to track the transition [27,29,30,277]. Note that the stripe phase determined from the iDMRG study [27] does not agree with subsequent studies focusing on the low-field ground states of the model [30,32,277,279,280].

Complementary to the high-field analysis and numerical iDMRG studies, approaches in the low-field limit have also been conducted [30,110,277] (see Figure 27). As already mentioned, it was discussed in the literature for some time [30,279,280] and recently demonstrated using a unit cell-based optimisation technique [110] that gapped plain stripes are the zero-field ground state as soon as long-range interactions are present. Therefore, in a phase diagram with a σ and a transverse-field axis, the clock order phase is wedged between the high-field phase from above and the plain stripe phase from below [27,29,30,277] (see Figure 27). The extent of both the stripe phase and the high-field phase increases with stronger long-range interactions [27,29,30,277]. This behaviour was also demonstrated perturbatively from the nearest-neighbour zero-field limit by treating the long-range interaction, as well as the transverse field as perturbations on the degenerate subspace [30,277].



Figure 27. Phase diagrams for the long-range transverse-field Ising model on the triangular lattice (left panel) and triangular lattice cylinders with infinite extent in the *x*-direction and a circumference of 6 sites in the *y*-direction (YC6) (right panel). Right panel: The critical field values h_c/J for the triangular lattice for the QPT between the high-field polarised phase originate from pCUT+MC calculations [29]. The inset of the right panel represents the critical exponents zv determined from the series expansion [29]. The black dashed line represents the critical value zv = 0.67175(10) of the 3D-XY universality class [281,282]. The critical exponent zv confirms the (2 + 1)D-XY universality class within the limitations of the series expansion. Left panel: The transitions values between the high-field series obtained from pCUT+MC [30]. The transition between the plain stripe low-field phase and the $\sqrt{3} \times \sqrt{3}$ -clock-ordered phase is terminated via a level crossing of both ground-state energies, which were calculated perturbatively [30]. The phase diagrams for small σ values are not yet conclusively determined [27,29,30].

We sketch a qualitative picture of the quantum phase diagram in Figure 28. The precise ground state at small values of σ is still an open research question. We believe that there is some evidence that, after a certain value of σ , only a transition between the high-field phase and the low-field stripes remains, which is expected to be of first order. However, there is, by now, no numerical technique to make reliable statements in this regime. In this context, the high-field series expansions cannot detect a first-order phase transition from the elementary excitation gap. There is an SSE QMC study of the system by S. Humeniuk [40] applying the algorithm discussed in Section 5.1. The direct QMC simulation was used to determine the critical point between the high-field phase and the clock-ordered phase for $\alpha = 3$ [40]. For smaller transverse fields, the author identified a "region dominated by classical ground states [40]", which are adiabatically connected to the zero-field states. Therefore, there is a qualitatively sketch of a phase diagram in Ref. [40] similar to Figure 28. Nevertheless, the naive application of the SSE QMC approach as discussed in Section 5.1 is not as efficient as for models with interactions that are not competing [40,261,262]. However, there are ideas on how to implement efficient quantum cluster algorithms for frustrated Ising models in a transverse field [261,262], which need to be adjusted for long-range interacting Ising models.



Figure 28. Sketch of a generic phase diagram of a long-range transverse-field Ising model (LRTFIM) for which a degenerate subspace at $\sigma = \infty$ and h = 0 breaks down in an order-by-disorder scenario for h > 0 and into a different crystalline state due to the long-range interactions (LRIs) $\sigma < \infty$. An example is the LRTFIM on the triangular lattice, where the order-by-disorder phase is the clock-ordered phase and the crystalline phase is the plain stripe phase [27,29,30,110,277,279,280]. The transition between the crystalline phase and the order-by-disorder phase is believed to be a first-order lever-crossing transition [29,30,110,277].

The antiferromagnetic LRTFIM on the triangular lattice is a well-studied example of the interplay between long-range interactions and order-by-disorder. Recently, several more examples arose from the field of Rydberg atom quantum simulators [54,55,109,110,112]. In Rydberg atom simulators, atoms are positioned in a desired configuration using optical tweezers [52] and are laser-driven to a Rydberg state [52]. The Rydberg blockade mechanism leads to an algebraically decaying interaction between Rydberg excitations, which decays with $(d + \sigma) = 6$ [52]. Using the Matsubara–Matsuda transformation [89], the excitations/non-excitations of a Rydberg atoms are associated with spin degrees of freedom [52]. The density-density interaction transforms into an Ising interaction [110,112]. Rydberg atom quantum simulators are capable of simulating the antiferromagnetic LRT-FIM with $(d + \sigma) = 6$ and a longitudinal field [52,54,55,109]. Order-by-disorder scenarios are described for the Kagome lattice at a filling of f = 2/9 Rydberg excitations [54], the Ruby lattice at a filling of f = 1/4 Rydberg excitations [55,109], and at vanishing longitudinal field [112]. Regarding these examples, the order-by-disorder was studied for a long-range interaction truncated after the third-nearest neighbours [54,55,109,112]. Considering the remaining long-range interaction as a perturbation to the degenerate ground-state space for most of these mechanisms, a similar scenario as for the triangular lattice is expected [54,55,109,110]. Using the same unit cell-based optimisation method as for the triangular lattice, Koziol et al. [110] have calculated ground states of the zero-field model using the full untruncated long-range interaction. For the Kagome lattice at a filling of f = 2/9 Rydberg excitations [54] and the Ruby lattice at a filling of f = 1/4 Rydberg excitations [54,55,109], gapped ground states were determined [110], which do not coincide with the order arising from the order-by-disorder mechanism. Interestingly, in all these examples, the zero-field ground state determined by the long-range interaction possess none of the motifs relevant for the leading-order order-by-disorder mechanism [110]. For example, on the triangular lattice, the plain stripe states contain zero flippable plaquettes, which are the relevant motif to lower the energy at a finite transverse field in leading

order. Recently, an order-by-disorder mechanism for the antiferromagnetic J_1 - J_2 - J_3 TFIM on the Ruby lattice was introduced by A. Duft et al. [112]. The remarkable aspect of the mechanism is that adding long-range interactions does stabilise the same order, which is also selected by the quantum fluctuations [112]. A general theory, under which conditions the long-range interactions stabilise the same order as quantum fluctuations or stabilise a completely disjoint order, is yet to be found.

Conclusively, the breakdown of extensively degenerate ground-state spaces due to an interplay of long-range interactions and quantum fluctuations is a highly vibrant and relevant field. Long-range interactions are relevant for a wide range of quantum–optical quantum–simulation platforms, including cold atoms [52,70,72] and ions [74–77,79–82,84]. These platforms can be used to realise exotic phases of matter, e.g., quantum spin liquids [54,55,109], glassy behaviour [283], or clock-ordered states [29,30,112,277]. To understand the emergence of these exotic states of matter, simplified models are oftentimes considered [54,55,109,112,283], which truncate the long-range interaction. Therefore, it is imminent to understand the effects of the full long-range interactions on the mechanisms driving the emergence of these exotic phases. At the moment, the commonly used tools to investigate the phase diagrams of these frustrated systems consist of the following:

- 1. The derivation of effective models [30,54,55,109,112]. These can be used in order to predict exotic emergent phases and the nature of QPTs [30,54,55,109,112].
- 2. DMRG calculations [27,54,55,109], which are used to obtain a numerical insight into the ground-state phase diagrams.
- 3. QMC calculations, in particular the SSE, as discussed in Section 5.1, can be used for an unbiased sampling of ground-state properties [40]. In order to omit the slowdown of the algorithm due to the geometric frustration, algorithmic improvements are required [261,262]. It is still an open research question how to set up an algorithm that samples long-range interaction and frustration efficiently.
- 4. High-order high-field series expansions using a graph decomposition. This is also a very capable tool to track the first QPT coming from the high-field limit. With this method, the critical field value, as well as critical exponents can be determined [29,30,112]. It is also possible to infer information about the phase on the other side of the phase transition by studying the momentum at which the elementary excitation gap is located [29,30,112]. An MC embedding of white graphs can be performed to study the entire algebraically decaying long-range interaction [29,30], while an ordinary graph decomposition can be used for systems with truncated interactions [112].

7. Long-Range Transverse-Field XY Chain

In this section, we review results on the XY chain with long-range interactions in a transverse field from Ref. [31]. The Hamiltonian of the long-range transverse-field anisotropic XY model (LRTFAXYM) on a one-dimensional chain is given by

$$H = h \sum_{i} \sigma_{i}^{z} - \frac{J}{4} \sum_{i \neq j} \frac{1}{|i-j|^{1+\sigma}} \left[(1+\beta)\sigma_{i}^{x}\sigma_{j}^{x} + (1-\beta)\sigma_{i}^{y}\sigma_{j}^{y} \right],$$
(308)

with Pauli matrices σ_i^{κ} and $\kappa \in \{x, y, z\}$ describing spin-1/2 degrees of freedom on the *i*-th lattice site of the chain. The transverse-field strength is given by h > 0, and the strength of the coupling between sites *i* and *j* is given by $\sim J/|i-j|^{1+\sigma}$. The coupling is (anti)ferromagnetic for J > 0 (J < 0). We include a continuous interpolation parameter $\beta \in [0, 1]$ to tune the system from the XY chain with isotropic interactions ($\beta = 0$) to Ising interactions ($\beta = 1$). The exponent σ determines the decay of the long-range interaction. The limit of short-range interactions is recovered for $\sigma = \infty$ and the limit of all-to-all couplings for $\sigma = -1$.

Similar to the discussion of the transverse-field Ising model, it is useful to express Equation (308) in terms of hard-core bosonic operators b_i , b_i^{\dagger} using the Matsubara-Matsuda transformation [89]

$$\sigma_i^z = 1 - 2b_i^{\dagger}b_i \qquad \sigma_i^x = b_i^{\dagger} + b_i \qquad \sigma_i^y = i(b_i^{\dagger} - b_i).$$
(309)

We obtain

$$H = \epsilon_0 N + \sum_i b_i^{\dagger} b_i - \sum_{i \neq j} \frac{\lambda}{4} \frac{1}{|i-j|^{1+\sigma}} [(1+\beta)(b_i^{\dagger} b_j + b_i^{\dagger} b_j^{\dagger}) + (1-\beta)(b_i^{\dagger} b_j - b_i^{\dagger} b_j^{\dagger}) + \text{h. c.}]$$
(310)

in units of 2*h* with $\epsilon_0 = -1/2$ being the unperturbed ground-state energy per site, *N* the number of sites, and $\lambda = J/(2h)$ the perturbation parameter associated with the perturbation \mathcal{V} (second sum) taking the form $\mathcal{V} = T_{-2} + T_0 + T_2$ in the pCUT language, so that a high-order series expansion about the high-field limit can be performed.

In the limit J = 0, the ground state is a nondegenerate *z*-polarised state that serves as an unperturbed reference state. Elementary excitations are called local spin flips at an arbitrary site *i*, which can be annihilated (created) by the hard-core bosonic operator $b_i^{(\dagger)}$. Upon increasing λ , these quasiparticles (qps) become spin flips dressed by quantum fluctuations induced by the perturbation \mathcal{V} . In the limit of h = 0, the system exhibits ferromagnetic or antiferromagnetic magnetic order depending on the sign of J in the *x*direction (*y*-direction) for $\beta > 0$ ($\beta < 0$). When tuning the interpolation parameter β from pure Ising to isotropic XY interactions, ordering in the *x*-direction starts to compete with ordering in the *y*-direction.

In order to better appreciate the results for $\sigma < \infty$, we briefly discuss the quantum phase diagram in Figure 29 of the model (308) with nearest-neighbour interactions [284–293]. For any $\beta \neq 0$, the system with nearest-neighbour interactions undergoes a (1+1)D-Ising QPT at $J = \pm h$ from the paramagnetic symmetric high-field phase to the (anti)ferromagnetically ordered low-field phase, where the ground state spontaneously breaks the \mathbb{Z}_2 symmetry of the Hamiltonian [284]. The transition along the $\beta = 0$ line between the two distinct \mathbb{Z}_2 symmetry-broken phases is of the (Ising)² type, which is conformally and U(1) invariant [284]. These critical lines meet at $\beta = 0$ and $J = \pm h$ in multicritical points $M_{1/2}$ with critical exponents z = 2 and $\nu = 1/2$ [284].



Figure 29. Quantum phase diagram of the ferromagnetic nearest-neighbour XY chain in a transverse field (see Ref. [284]). M_1 and M_2 denote multicritical points. The phase transition between the symmetric high-field polarised phases and magnetically ordered low-field phases are of (1+1)D-Ising universality for $\beta \neq 0$. For $\beta = 0$, the transition at the multicritical points M_1 and M_2 has critical exponents z = 2 and $\nu = 1/2$. The transition between $\langle \sigma_i^x \rangle \neq 0$ and $\langle \sigma_i^y \rangle \neq 0$ is of the (Ising)² type.

For $\beta = 0$, the Hamiltonian (310) becomes particle-conserving as the terms annihilating (creating) two quasiparticles $b_i^{(\dagger)} b_j^{(\dagger)}$ cancel each other, making the isotropic XY Hamiltonian U(1) symmetric. Due to this particle-conserving nature of Equation (310), there are no quantum fluctuations dressing the symmetric $\lambda = 0$ ground state for $\lambda > 0$ and the perturbative treatment of the high-field dispersion becomes exact in first order of the perturbation theory [31]. This distinguished symmetry property of the $\beta = 0$ case motivates that we review the properties of the isotropic transverse-field XY chain in the ferromagnetic and antiferromagnetic case first in Section 7.1. Subsequently, we discuss the anisotropic XY chain for ferromagnetic interactions in Section 7.2 and for antiferromagnetic interactions in Section 7.3. Here, we show improved results of the ones shown in Ref. [31] since we noticed right after their publication that the Monte Carlo runs were performed with a suboptimal choice of the simulation parameters.

7.1. Isotropic Long-Range XY Chain in a Transverse Field

To consider the isotropic XY chain, $\beta = 0$ is set in the Hamiltonians in Equations (308) and (310). The quantum criticality of this model was studied in Ref. [31] in an analytical fashion, by evaluating the ground-state energy and the dispersion of the elementary excitations analytically. Setting $\beta = 0$, Equation (310) reads

$$H = \epsilon_0 N + \sum_i b_i^{\dagger} b_i - \sum_{i \neq j} \frac{\lambda}{2} \frac{1}{|i - j|^{1 + \sigma}} [b_i^{\dagger} b_j + \text{h. c.}]$$
(311)

since the pair creation and annihilation terms $b_i^{(\dagger)}b_j^{(\dagger)}$ cancel, as stated above. The perturbation \mathcal{V} acting on the unperturbed *z*-polarised ground state does not introduce any quantum fluctuations. Therefore, the *z*-polarised state becomes an exact eigenstate for arbitrary λ and stays in the ground state until a QPT occurs [31]. Analogously, the one-quasiparticle dispersion is exact in first-order perturbation theory. The one-particle dispersion in the symmetric high-field phase reads [31]

$$\omega(k) = 1 - 2\lambda \sum_{\delta=1}^{\infty} \frac{\cos(k\delta)}{\delta^{1+\sigma}} .$$
(312)

The critical value λ_c can be determined from the dispersion (312) via the closing of the quasiparticle gap $\Delta(\lambda)$. The critical exponent $z\nu$ associated with the gap closing can be extracted from the dominant power-law behaviour of the gap near λ_c :

$$\Delta(\lambda) \propto |\lambda - \lambda_c|^{z\nu} \,. \tag{313}$$

Further, it is possible to separate the dynamic *z* and static *v* exponent by evaluating the 1qp dispersion $\omega(k)$ at the quantum-critical point such that

$$\omega(k)|_{\lambda=\lambda_c} \propto |k-k_c|^z , \qquad (314)$$

which allows the extraction of the critical dynamic exponent *z*. For ferromagnetic XY interactions, the minimum of the dispersion (312) is located at the momentum k = 0. Therefore, the gap series is given by $\Delta(\lambda) = 1 - 2\lambda\zeta(1 + \sigma)$ and $\lambda_c(\sigma) = (2\zeta(1 + \sigma))^{-1}$. The Riemann-zeta function $\zeta(1 + \sigma) := \sum_{n=1}^{\infty} n^{-(1+\sigma)}$ is convergent for all $\sigma > 0$ and diverges for $\sigma \to 0$. This mathematical observation coincides with the qualitative behaviour found in ferromagnetic Ising systems discussed in Section 6.1 in the sense that $\lim_{\sigma\to 0} \lambda_c(\sigma) = 0$. Since the expression for the gap $\Delta(\lambda)$ is linear in λ , the critical exponent is $z\nu = 1$, independent of the σ value. Using the expression for the dispersion (312), the knowledge about λ_c , and the definition of the dynamic critical exponent (314), Adelhardt et al. [31] evaluated *z* and, therefore, also ν as a function of σ . The results of their calculations are presented in Figure 30. The exponents resulting from the consideration of the dispersion can be categorised into two distinct regimes: For $\sigma > 2$, the critical exponents of the nearest-neighbour

transition are found (z = 2 and v = 1/2). For $\sigma < 2$, the exponent z decreases linearly from 2 to 0 and v increases like 1/z from 1/2 to ∞ . Note that this linear decrease of z and the divergence of v appear similar to the behaviour of the ferromagnetic LRTFIM at small σ values (see Section 6.1) [20,21]. To explain this behaviour, a quantum field theory was proposed in Ref. [31]. The idea was to add a k^{σ} term (analogous to [20,21]) to the well-studied bosonic action used for the study of the isotropic short-range transition [113,231,284,294]. The suggested action reads

$$S = \frac{1}{2} \int_{k,\omega} (ak^2 + bk^{\sigma} + ig\omega + r) |\psi_{k,\omega}|^2 + u \int_{k,\omega} |\psi_{k,\omega}|^4$$
(315)

with ψ being a complex *c*-number order-parameter field of the transition, *a*, *b* > 0, and the real constants *u*, *g* and *r* [31]. The notation of Equation (315) is taken from Ref. [31,231]. From power counting, one obtains the critical exponents:

$$z = \begin{cases} 2 & \text{for} \quad \sigma \ge 2\\ \sigma & \text{for} \quad \sigma < 2 \end{cases}$$
(316)

$$\nu = \begin{cases} 1/2 & \text{for} \quad \sigma \ge 2\\ 1/\sigma & \text{for} \quad \sigma < 2 \end{cases}$$
(317)

from the Gaussian part of the action in Equation (315) [31]. Following the arguments of Ref. [231], these exponents hold "presumably" [231] for $1 \le d \le d_{uc}$ below the upper critical dimension $d_{uc} = 2$. Refs. [231,294] provide arguments that the self-energy vanishes for every order in u and the renormalisation of u can be performed in all orders via ladder diagrams [31]. The vanishing self-energy is explained by the fact that, in each diagram, every pole in ω lies in the complex upper-half plane. The frequency integral can be deformed into the lower half-plane to give zero [294]. The fact that the free propagator of the field theory is not changed by a self-energy is a manifestation of the absence of fluctuations that do not preserve the particle number [31]. The predictions of the field theory (315) for z in Equation (316) and ν in Equation (317) are in perfect agreement with the results of the high-field excitation gap (see Figure 30) [31].

Note that, in the nearest-neighbour case, the low-field ground state does not break the U(1) symmetry of the Hamiltonian due to the Hohenberg–Mermin–Wagner (HMW) theorem, ruling out continuous symmetry breaking [113,130,131,284,295,296]. Long-range interactions are a known mechanism to circumvent the HMW theorem [22,35,297–306]. With the high-field approach discussed above, it is not possible to determine the nature of the low-field ground state and if a continuous symmetry breaking occurs for sufficiently small decay exponents. We are not aware of studies addressing this topic.

For the remainder of this section, we discuss the antiferromagnetic isotropic XY chain in a transverse field along the same lines as for the ferromagnetic case. The minimum of the dispersion is at momentum $k = \pi$, and the gap can be expressed as $\Delta(\lambda) = 1 - 2\lambda\eta(1 + \sigma)$ with $\eta(1 + \sigma)$ being the Dirichlet eta function (also known as the alternating ζ function) [31]. The Dirichlet eta function $\eta(1 + \sigma)$ is convergent for all $\sigma > -1$; therefore, the definition of an excitation gap is also possible in the regime of strong long-range interactions ($\sigma \le 0$). Analogous to the discussion of the Ising interaction in Section 6.2, the extent of the lowfield phase is diminishing for decreasing σ values. For $\sigma \rightarrow -1$, the energy gap closes at diverging $\lambda_c \rightarrow \infty$ [31]. With a similar analysis to the ferromagnetic case, the same critical exponents z = 2 and v = 1/2 are found along the entire σ line [31]. This is also analogous to the antiferromagnetic LRTFIM discussed in Section 6.2, where the nearest-neighbour criticality is believed to be the correct universality class for all values of σ .



Figure 30. Critical exponents *z* and *v* for the ferromagnetic long-range transverse-field XY model as a function of σ . The data points z_{ω} and v_{ω} were determined in Ref. [31] by studying the dispersion of the elementary excitations at the critical point that can be determined exactly in first-order perturbation theory from the high-field limit. The lines z_{ft} and v_{ft} are theoretical predictions from the QFT investigated in Ref. [31]. The region with the blue-shaded background denotes the regime in which the nearest-neighbour XY universality occurs [284], while for $\sigma < 2$, the QPT is in a long-range regime with continuously varying exponents.

7.2. Ferromagnetic Anisotropic Long-Range XY Chain in a Transverse Field

The fundamental difference between the isotropic $\beta = 0$ and anisotropic $\beta > 0$ XY chain in a transverse field is that, for $\beta \neq 0$, the Hamiltonian is no longer U(1)invariant (particle-conserving); see Equation (310). Therefore, a perturbative treatment of the anisotropic model with long-range interaction from the high-field limit requires the entire procedure described in Section 4 to perform pCUT+MC calculations. In Ref. [31], this procedure was applied to obtain the critical values λ_c and critical exponents $z\nu$ from the gap. We review these results in this section, followed by the results for the antiferromagnetic case in Section 7.3.

Critical values λ_c and critical exponents $z\nu$ for several β values as a function of σ are depicted in Figure 31. The overall σ dependence of λ_c for $\beta > 0$ is similar to the $\beta = 0$ case (see the inset of Figure 31). However, the low-field ferromagnetic phase becomes more stable for increasing β values [31]. Note that the perturbative parameter is $\lambda = J/2h$; therefore, a QPT at smaller λ values means a larger extent of the phase in h/J. In Ref. [307], the QPT of the anisotropic LRTFAXYM was studied with exact diagonalisation, and the critical point was determined for $\beta = 1/2$, which coincides perfectly with the value from pCUT+MC (see Figure 31). For all $\beta > 0$, Adelhardt et al. [31] identified the same three regimes for the universality of the QPT as for the ferromagnetic LRTFIM. The gap exponent behaves analogously to the LRTFIM ($\beta = 1$), as depicted in Figure 31. Solely the $\beta = 0$ exponent behaves in a special manner, being $z\nu = 1$ for all σ values, in full agreement with the existence of a multicritical point in the nearest-neighbour model. For $\beta > 0$, the first domain identified in Ref. [31] is the long-range mean-field regime for $\sigma < 2/3$ with $z\nu = 1/2$. For large values of σ , the gap exponent is $z\nu = 1$ and the transition belongs to the 2D-Ising universality class [31]. Third, in the intermediate regime, the critical exponents vary continuously between the two values in a non-trivial fashion [31].

The criticality regimes for $\beta > 0$ can be understood using the same field-theoretical argument (see Section 6.1.1) as for the ferromagnetic LRTFIM, since the underlying symmetry that is spontaneously broken is, in both cases, the same \mathbb{Z}_2 symmetry of the Hamiltonian. Therefore, at $\sigma = 2/3$, the upper critical dimension as a function of σ becomes smaller than

100 of 135

between nearest-neighbour criticality and the intermediate long-range non-trivial regime is expected to be at $\sigma = 2 - \eta_{SR}$ [21]. The data from Ref. [31], presented in Figure 31, are in good agreement with these three regimes. Nevertheless, the changes between the regimes cannot be determined accurately by the pCUT+MC approach. In Figure 31, we can observe that the interfaces between the nearest-neighbour and intermediate long-range regime become more pronounced for increasing β values, while at the interface between the long-range mean-field and intermediate regime, the deviation is the same for all values of $\beta > 0$. The reason for the distinct deviations at the nearest-neighbour interface may be due to the finite nature of the perturbative series or may also be indicative of different corrections to scaling. At $\sigma = 2/3$, where the system dimension equals the upper critical dimension of the transition, there are multiplicative logarithmic corrections to the critical exponents [125,135,308–310]. From the biased DlogPadé extrapolations at $\sigma = 2/3$, Adelhardt et al. [31] determined these corrections to scaling at the upper critical dimension with a qualitatively good agreement for different β values.



Figure 31. Critical gap exponent $z\nu$ and critical values of λ (see inset) as a function of σ for the anisotropic ferromagnetic transverse-field XY model. The data points "pCUT+MC" are improved results from Ref. [31], and the data point "ED" is from Ref. [307]. The anisotropy parameter β is tuned from $\beta = 1$ (Ising) to $\beta = 0$ (isotropic XY). As discussed in Section 7.1, the isotropic case is analytically solvable with $z\nu = 1$ and $\lambda_c = (2\zeta(\sigma + 1))^{-1}$. The black lines denote the values of $z\nu$ if the ferromagnetic LRTFIM is in the nearest-neighbour and the long-range mean-field regime.

To summarise the findings, we provide the following: Similar to the ferromagnetic LRTFIM, the ferromagnetic LRTFAXYM is a paradigmatic toy model extending the LRTFIM. With the pCUT+MC method described in Section 4, the critical exponents as a function of the decay exponent σ can be determined as demonstrated in Ref. [31]. The LRTFAXYM displays three distinct regimes of quantum criticality including a mean-field regime above the upper critical dimension for $\sigma < 2/3$ for all $\beta > 0$. The sensitivity of the series expansion method suffices to even study corrections to scaling at the upper critical dimension. The limiting case $\beta = 0$ is especially interesting, as in the high-field limit, the ground state contains no quantum fluctuations, the dispersion is exactly solvable in first-order perturbation theory, the underlying QFT is different from the case $\beta > 0$, and two instead of three critical regimes were found. We would expect that the setup of an SSE QMC algorithm following the directed loop idea [215,217,311] is conceptually possible for the model.

7.3. Antiferromagnetic Anisotropic Long-Range XY Chain in a Transverse Field

For frustrated antiferromagnetic interactions, we begin our discussion with the Ising case at $\beta = 1$. Here, we have already discussed in Section 6.2 that, for the antiferromagnetic LRTFIM, there is strong numerical evidence that it remains in the 2D-Ising universality class for all $\sigma > -1$. It was also demonstrated that it is possible to determine critical gap exponents up to $\sigma \approx 0.5$ with the pCUT+MC method [29,31,35].

Following the same procedure as for the ferromagnetic model, the critical values λ_c and the critical gap exponents $z\nu$ were studied in Ref. [31], and we summarise these results in Figure 32. The critical values λ_c shift towards larger values when decreasing σ , eventually diverging for $\sigma \rightarrow -1$. It is possible to reliably determine the critical exponents for the anisotropic XY chain until $\sigma \approx 0.5$, but it becomes increasingly challenging to extract the exponents for smaller σ due to the shifting of λ_c [31]. For $\beta \neq 0$, the critical values λ_c behave qualitatively in a similar way to $\beta = 0$. However, the larger β , the more the high-field phase is stabilised (see the inset of Figure 32). As a consequence, we observe more reliable exponent estimates in the regime $\sigma < 0.5$ for β values closer to zero. In general, the observations in Ref. [31] are in line with a constant gap exponent $z\nu = 1$ within the limitations of the method.

This promotes the scenario that the nearest-neighbour 2D-Ising universality class remains the correct universality class for all considered σ values and $\beta > 0$. For $\beta = 0$, the critical exponents $z\nu = 1$ and critical values $\lambda_c = (2\eta(1+\sigma))^{-1}$ were determined analytically in Section 7.1, and it was shown that the universality class of the nearest-neighbour isotropic chain remains for all $\sigma > -1$ [31]. A recent study [312] investigated the multicritical point using exact diagonalisation for comparably small system sizes. The critical points for $\beta = 1/2$ and $\beta = 1/5$ agree well with the exact limit and the pCUT+MC values. The study further confirms 2D-Ising universality up to $\sigma > -0.4$; however, in contrast to Ref. [31], signatures of a different crossover criticality at $\sigma = -0.4$ are claimed.



Figure 32. Critical gap exponent $z\nu$ and critical values of λ (see inset) as a function of σ for the anisotropic antiferromagnetic transverse-field XY model [31]. The anisotropy parameter β is tuned from $\beta = 1$ (Ising) to $\beta = 0$ (isotropic XY). As discussed in Section 7.1, the isotropic case is analytically solvable with $z\nu = 1$ and $\lambda_c = (2\eta(\sigma + 1))^{-1}$.

8. Long-Range Heisenberg Models

We review the results obtained from SSE and pCUT+MC calculations for Heisenberg models with long-range interactions. The starting point is the nearest-neighbour Heisenberg model, which can be written as

$$\mathcal{H} = \sum_{\langle i,j \rangle} \vec{S}_i \vec{S}_j = \sum_{\langle i,j \rangle} \left(S_i^x S_j^x + S_i^y S_j^y + S_i^z S_j^z \right)$$
(318)

with the spin-1/2 operators $S^{\kappa} = \sigma^{\kappa}/2$ and $\kappa \in \{x, y, z\}$. Instead of a transverse field that induces quantum fluctuations in the (anti)ferromagnetic phase of the (LR)TFIM or (LR)TFAXYM, a magnetic field would immediately break the *SU*(2)-symmetry of the Heisenberg Hamiltonian. Here, we introduce a dimerisation limit. We consider two layers of lattices with Heisenberg spins on each site stacked on top of each other, where each spin couples to its nearest-neighbour. The nearest-neighbour coupling between the layers gives rise to interlayer dimers, also referred to as rungs. In one dimension, the layers are just two chains, resulting in a ladder system, while in two dimensions, we consider two square lattice layers, resulting in a square lattice bilayer model. Generalising this model for long-range interactions gives rise to intralayer, as well interlayer long-range couplings. The generic antiferromagnetic Hamiltonian is then given by

$$\mathcal{H} = J_{\perp} \sum_{i} \vec{S}_{i,1} \vec{S}_{i,2} - \frac{1}{2} \sum_{i \neq j} \left[J_{\parallel}(i-j) \underbrace{\left(\vec{S}_{i,1} \vec{S}_{j,1} + \vec{S}_{i,2} \vec{S}_{j,2}\right)}_{\text{intralayer}} + J_{\times}(i-j) \underbrace{\left(\vec{S}_{i,1} \vec{S}_{j,2} + \vec{S}_{i,2} \vec{S}_{j,1}\right)}_{\text{interlayer}} \right], \quad (319)$$

where the first term couples the nearest-neighbour spins between the layers with coupling strength $J_{\perp} > 0$, the second term describes the long-range intralayer coupling with J_{\parallel} , and the third term the long-range interlayer coupling of Heisenberg spins with J_{\times} . The long-range interaction is given by

$$J_{\mu,\nu}(i-j) = J \frac{(-1)^{\|\vec{r}_{i,\mu} - \vec{r}_{j,\nu}\|_1}}{|\vec{r}_{i,\mu} - \vec{r}_{j,\nu}|^{d+\sigma}},$$
(320)

where *J* is the coupling constant, *i* and *j* are the rung dimer indices, and μ and ν are the indices of the respective legs (layers) of the ladder (bilayer) model. In Equation (319), we use the notation $J_{||}(i - j) = J_{1,1}(i - j) = J_{2,2}(i - j)$ and $J_{\times}(i - j) = J_{1,2}(i - j) = J_{2,1}(i - j)$. The long-range interaction introduced is similar to the one introduced in the sections before, however with the difference that the alternating sign in the numerator gives rise to staggered, non-frustrating antiferromagnetic long-range interactions, as we chose J > 0. The exponent of the numerator is given by the one-norm $\|\cdot\|_1$, while the usual geometric distance in the denominator given by $|\cdot|$ can be identified as the two-norm $|\cdot| = \|\cdot\|_2$. When increasing the overall interaction strength *J* of intralayer and interlayer coupling and, likewise, making the long-range decay exponent σ smaller, the Heisenberg spins will anti-align at odd distances and align at even distances, inducing a Néel-ordered antiferromagnetic LRTFIM in Section 6.1), and for $\sigma \to \infty$, we recover the antiferromagnetic short-range ladder or bilayer model.

In order to employ the pCUT+MC method, we need an exactly soluble limit about which we can perform the series expansion. We introduce the perturbation parameter $\lambda = J/J_{\perp}$ rescaling the Hamiltonian by J_{\perp} and identify the first term of Equation (319) as the unperturbed part \mathcal{H}_0 , which corresponds to a sum over Heisenberg dimers that can be easily diagonalised. For a single dimer, the lowest lying state with total spin S = 0 is the antisymmetric combination of S = 1/2 spins:

$$|s\rangle = \frac{1}{\sqrt{2}} (|\uparrow\downarrow\rangle - |\downarrow\uparrow\rangle), \qquad (321)$$

also called a rung singlet in our case with the associated energy $\epsilon_0 = -3/4$. The total spin S = 1 states:

$$|t_{-}\rangle = |\downarrow\downarrow\rangle, \quad |t_{0}\rangle = \frac{1}{\sqrt{2}}(|\uparrow\downarrow\rangle + |\downarrow\uparrow\rangle), \quad |t_{+}\rangle = |\uparrow\uparrow\rangle$$
(322)

are three-fold degenerate with the energy 1/4 and are called triplets. Alternatively, we can use an SU(2)-symmetric basis, where the singlet and triplet states read

$$|s\rangle = \frac{1}{\sqrt{2}}(|\uparrow\downarrow\rangle - |\downarrow\uparrow\rangle), \qquad (323)$$
$$|t_x\rangle = -\frac{1}{\sqrt{2}}(|\uparrow\uparrow\rangle - |\downarrow\downarrow\rangle), \quad |t_y\rangle = \frac{i}{\sqrt{2}}(|\uparrow\uparrow\rangle + |\downarrow\downarrow\rangle), \quad |t_z\rangle = \frac{1}{\sqrt{2}}(|\uparrow\downarrow\rangle + |\downarrow\uparrow\rangle).$$

There is also a convenient mapping from the spin operators to bosonic, SU(2)-symmetric operators creating and annihilating these states introduced in Ref. [313], which can be readily adapted to hard-core bosonic operators for triplet excitations [314]. We use the mapping

$$\vec{S}_{i,1}^{\alpha} = \frac{1}{2} \left(t_{i,\alpha}^{\dagger} + t_{i,\alpha} - i\epsilon_{\alpha,\beta,\gamma} t_{i,\beta}^{\dagger} t_{i,\gamma} \right),$$

$$\vec{S}_{i,2}^{\alpha} = \frac{1}{2} \left(t_{i,\alpha}^{\dagger} + t_{i,\alpha} + i\epsilon_{\alpha,\beta,\gamma} t_{i,\beta}^{\dagger} t_{i,\gamma} \right),$$
(324)

where $\epsilon_{\alpha,\beta,\gamma}$ is the Levi-Civita symbol, and insert these expressions into Equation (319). After some straightforward manipulations, we obtain

$$\mathcal{H} = \epsilon_0 N_R + \sum_{i,\alpha} t^{\dagger}_{i,\alpha} t_{i,\alpha} + \frac{1}{4} \sum_{i \neq j} \lambda_{\shortparallel} (i-j) \left(t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\alpha} + t^{\dagger}_{i,\alpha} t_{j,\alpha} - t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\alpha} t_{i,\beta} t_{j,\beta} + t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\beta} t_{i,\beta} t_{j,\alpha} + \text{h.c.} \right)$$

$$+ \frac{1}{4} \sum_{i \neq j} \lambda_{\times} (i-j) \left(-t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\alpha} - t^{\dagger}_{i,\alpha} t_{j,\alpha} - t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\alpha} t_{i,\beta} t_{j,\beta} + t^{\dagger}_{i,\alpha} t^{\dagger}_{j,\beta} t_{i,\beta} t_{j,\alpha} + \text{h.c.} \right),$$
(325)

where N_R is the number of rungs, $\lambda_{\shortparallel}(i-j) = J_{\shortparallel}(i-j)/J_{\perp}$, and $\lambda_{\times}(i-j) = J_{\times}(i-j)/J_{\perp}$. The second and third line constitute the perturbation $\mathcal{V} = T_{-2} + T_0 + T_2$ associated with the physical perturbation parameter λ . The T_1 and T_{-1} operators are absent, as terms contributing to these operators cancel out due to the reflection symmetry about the centre of the rung dimers.

For $\lambda = 0$, the ground state is given by a trivial product state of rung singlets, and for small, but finite λ , the ground state is still adiabatically connected to this product state. We refer to this adiabatically connected ground state as the rung-singlet ground state. Elementary excitations in this phase are called triplons [315], corresponding to dressed triplet excitations. For large $\lambda \gg 0$, we expect the non-frustrating antiferromagnetic long-range interaction to induce an antiferromagnetic phase giving rise to a quantum phase transition (QPT) between these two phases. In contrast to the transverse-field Ising model and the anisotropic XY model, where the ground state in the ordered phase spontaneously breaks the discrete \mathbb{Z}_2 symmetry of the Hamiltonian, the Néel-ordered antiferromagnetic ground state for strong long-range couplings must break the continuous SU(2) symmetry of the Heisenberg Hamiltonian. Related to this observation, we present in the following two prominent theorems of great importance that apply to Hamiltonians with continuous symmetries:

First, there is the Hohenberg–Mermin–Wagner (HMW) theorem [130,131,295], which rules out the spontaneous breaking of continuous symmetries in one- and two dimensional systems for T > 0. From the quantum–classical correspondence (see Ref. [113]), we can infer that, for one-dimensional systems, a QPT at T = 0 breaking such symmetry should be ruled out as well. Indeed, it was shown rigorously by Pitaevskii and Stringari in 1991 [296]

that it is prohibited in one dimension for T = 0 as well. However, there is the restriction that the interaction must be sufficiently short-ranged, i.e., the condition $\sigma > 2$ for T > 0must hold for the long-range decay exponent. A stronger condition was given by Ref. [316] with $\sigma \ge d$, where *d* is the spatial dimension of the system. For the quantum case, it was shown in Ref. [317] for a staggered antiferromagnetic long-range Heisenberg chain (d = 1) that long-range order is absent for all $\sigma > 2$.

Second, Goldstone's theorem [318-320] applies. The theorem states that, when spontaneous breaking of a continuous symmetry occurs, it gives rise to massless excitations, also known as Nambu–Goldstone modes [318–320]. For instance, magnons that are quantised spin-wave excitations are a manifestation of such gapless Nambu-Goldstone modes inside the ordered phase of Heisenberg systems. The recent findings in Ref. [321] actually go beyond the conventional scenario of Goldstone modes due to long-range interactions. The authors identified three regimes depending on the decay exponent σ : for ferromagnetic (antiferromagnetic) interactions, they found standard Goldstone modes for $\sigma \ge 2$ ($\sigma \ge 0$) and an anomalous Goldstone regime with $\omega \sim |k|^s$ and s < 2 (s < 1) for $\sigma < 2$. Interestingly, a third regime for strong long-range interactions $\sigma \leq 0$ ($\sigma \leq -2$) was found where the Goldstone modes become gapped via a generalised Higgs mechanism [321]. A recent large-scale QMC study [38] investigating the dynamic properties of the non-frustrating staggered antiferromagnetic square lattice Heisenberg model confirmed these three scenarios and found evidence that the Higgs regime already occurs in the regime $\sigma \leq 0.2$ when the Hamiltonian is still extensive [38]. Let us mention at this point that previous results from linear spin-wave theory [22,322] already indicated the existence of a sublinear dispersion in the staggered antiferromagnetic long-range Heisenberg chain a decade earlier. Remarkably, the existence of gapped Goldstone modes in the Heisenberg model with long-range interactions was already pointed out in the 1960s by Refs. [323,324].

The HMW theorem, as well as Nambu–Goldstone modes are a distinguishing feature of quantum systems with continuous symmetries like Heisenberg antiferromagnets. The quantum field theory describing dimerised Heisenberg antiferromagnets [113] is given by the action

$$S_{\phi} = \int d^{d}x \int_{0}^{\beta} d\tau \left[\{ g(\partial_{\tau} \phi(\tau, x))^{2} + (\nabla_{x} \phi(\tau, x))^{2} + r \phi(\tau, x)^{2} \} + u \phi(\tau, x)^{4} \right], \quad (326)$$

which is the same *n*-component ϕ^4 -theory as the one describing the transverse-field Ising model. Here, the order-parameter field $\phi(\tau, x)$ is now a 3-component field instead of a 1-component one. We can readily include long-range interactions by adding the term

$$\int d^d x \, \int d^d y \, \int d\tau \, a \frac{\phi(\tau, x)\phi(\tau, y)}{|x - y|^{d + \sigma}} \tag{327}$$

to the action (326). For a more detailed discussion of the short-range ϕ^4 action, we refer to Appendix A and to Section 6.1.1, where the implications of long-range interactions were already discussed for n = 1. The classical equivalent of the n = 3 action can describe the finite temperature phase transition of the ferromagnetic Heisenberg model [1,36,113,325]. Interestingly, the action of zero-temperature Heisenberg ferromagnets is not given by the above action [113]. In fact, the quantum field theory describes a class of dimerised Heisenberg antiferromagnets, which can be appropriately described by Equation (326) since their low-energy physics can be mapped onto the n = 3 quantum rotor model because pairwise antiferromagnetically coupled Heisenberg spins are an effective low-energy representation of quantum rotors [113]. We can see this correspondence by looking at the Hamiltonian of the O(3) quantum rotor model:

$$\mathcal{H} = \mathcal{H}_{\rm kin} + \mathcal{V} = \frac{K}{2} \sum_{i} L_i^2 - J \sum_{\langle i,j \rangle} n_i n_j , \qquad (328)$$

where the first part $\mathcal{H}_{kin} = K/2\sum_i L_i^2$ with K > 0 and *L* the angular momentum operator gives the kinetic energy of the rotors. The second part $\mathcal{V} = -J \sum_{(i,j)} n_i n_i$ with J > 0 is the interaction between the three-component quantum rotors inducing parallel ordering of the rotors [113]. The physical picture is that the kinetic energy is minimised when the orientation of the rotors is maximally uncertain and the energy of the interaction term is minimised when the rotors are aligned [113]. The eigenvalues of the kinetic term of a single quantum rotor are given by $E_{kin}(l) = K/2l(l+1)$ with $l \in \mathbb{N}$ corresponding to a (2l + 1)-fold degenerate state. Now, given two spins S interacting antiferromagnetically with coupling strength J_{\perp} forming a dimer, the total spin is $0 \le S_{tot} \le 2S$ and the eigenenergies are given by $E_{\text{dimer}}(S_{\text{tot}}) = J_{\perp}/2 \left(S_{\text{tot}}(S_{\text{tot}}+1) - 2S(S+1) \right)$ with a $(2S_{\text{tot}}+1)$ -fold degeneracy. This gives a one-to-one correspondence to the kinetic energy of a rotor, but with an upper bound for the energy. Thus, this mapping is only valid when considering the low-energy properties of the models [113] and holds certainly for large K/J. We just introduced a powerful mapping from the low-energy properties of the antiferromagnetic Heisenberg systems like ladder and bilayer models forming Heisenberg dimers on the rungs (see Equation (319)) to an O(3) quantum rotor model, which is described by the action (326). It turns out that this mapping is not only valid for large K/I, but also in the ordered phase and at the quantum critical point [113]. To include algebraically decaying long-range interactions, the term (327) must be added to the action. The implications of long-range interactions for an *n*-component order-parameter field in the ϕ^4 field theory was already extensively discussed in the 1970s [1-3] for the classical action, but it took until 2001 for the quantum analogue to be studied by Ref. [20]. More studies followed by N. Defenu et al. employing functional RG approaches [21,244,326].

The predictions for long-range Heisenberg models are in large part the same as for the LRTFIM. We expect long-range mean-field behaviour with a Gaussian fixed point for $\sigma \leq \sigma_{uc}$, a regime of nearest-neighbour criticality for $\sigma \geq \sigma_{*}$, and a non-trivial regime with continuously varying critical exponents in between [20,21,244,326]. For d = 1, however, there is the major difference that a QPT associated with the spontaneous breaking of continuous symmetries is ruled out by the HMW theorem [130,131,295,296], at least for $\sigma > 2$ [21,244,317,326]. The boundary is then called lower critical exponent σ_{lc} . For the Heisenberg model, the previously discussed action (326) with long-range interactions (327) holds, which together with the HMW theorem rules out the breaking of the order-parameter field for sufficiently short-range interactions and, therefore, a QPT in the regime $\sigma > \sigma_{lc}$. For $\sigma \leq \sigma_{lc}$, a QPT breaking the continuous symmetry is predicted [20,21,244,326]. Indeed, it has been long confirmed by several numerical studies [22,297–306,327] that the HMW theorem can be circumvented if the long-range interaction is sufficiently strong. Notably, in recent experiments with trapped ions [328] and Rydberg atoms [329] continuous symmetry breaking was realized in one and two dimensions which would be prohibited in the abscence of long-range interactions.

So far, most of the studies considered only Heisenberg Hamiltonians with long-range interactions in one-dimensional systems [22,35,297–306,327], however often considering various modifications to the Hamiltonian. Some studies considered a one-dimensional Heisenberg chain with non-frustrating (staggered) antiferromagnetic long-range interactions [22,300,305], while others included an anisotropy along the *z*-components resulting in the *XXZ*-model [302,303,327]. An interesting example with frustrating long-range interactions is the exactly solvable Haldane–Shastry model [330,331] with ~ r^{-2} , which is known to show quasi-long-range order (QLRO) just like the conventional Heisenberg chain, but with vanishing logarithmic corrections [332]. For the dimerisation transition between the QLRO and a valence bond solid (VBS), the frustrated long-range interaction seems to play only a minor role compared to the J_1 - J_2 model [213]. Based on this, in Refs. [213,297–299,333], a model combining the frustrated J_1 - J_2 model with additional non-frustrating staggered long-range interaction was investigated. The initial intention was to realise a one-dimensional analogue of deconfined criticality [334–337] between a Néel-order phase and a VBS. As the transition was found to be of first order, the authors

in Ref. [338] later turned to the one-dimensional long-range *J*-*Q* model, indeed finding evidence for a continuous deconfined QPT. Interestingly, a very recent paper [327] also predicts a deconfined QPT in the antiferromagnetic long-range XXZ chain. There are also studies considering larger spin S = 1. In Ref. [304], a Heisenberg chain with single-ion anisotropy was studied, while in Ref. [301], the XXZ chain was under scrutiny. In both cases, a rich phase diagram was found with intriguing quantum critical properties. For instance, continuous symmetry breaking due to long-range interactions gives rise to a new, possibly exotic, tricritical point in the XXZ chain with no analogue in short-range one-dimensional spin systems [301].

In Sections 6 and 7, we reviewed the results of various numerical studies investigating the quantum-critical properties of the LRTFM [23–26,28–30,32–34] and the LRTFAXYM [31] with the focus on extracting the critical exponent as a function of the long-range decay σ . In comparison to the LRTFIM, for long-range Heisenberg models, just recently, efforts have been made to extract the critical exponents associated with the QPT as a function of the long-range decay exponent [22,35,37,206]. Reference [22], studying a one-dimensional long-range Heisenberg chain, is one notable exception in the sense that it was ahead of its time, performing large-scale SSE QMC simulations for long-range Heisenberg systems already in 2005, way before the rapid progress in implementing quantum simulators in quantum optics reignited the interest in long-range interacting systems.

In the following, we will first review the results from Refs. [37,206,339] for the longrange square-lattice Heisenberg bilayer model in Section 8.1. Then, we proceed with the discussion of the results from Ref. [35] for long-range Heisenberg ladders in Section 8.2. Finally, we conclude this section with the discussion of the long-range Heisenberg chain from Ref. [22] in Section 8.3.

8.1. Staggered Antiferromagnetic Long-Range Heisenberg Square Lattice Bilayer Model

In this section, we consider two square lattices stacked directly on top of each other, where the Heisenberg spins on each lattice site interact with their nearest neighbours. When the two spins on top of each other form a rung dimer interacting with coupling strength J_{\perp} , the Hamiltonian describing the bilayer system is given by Equation (319). Here, we neglect the long-range interlayer interactions, which results in the Hamiltonian

$$\mathcal{H} = J_{\perp} \sum_{i} \vec{S}_{i,1} \vec{S}_{i,2} - \frac{1}{2} \sum_{i \neq j} \left[J_{\parallel}(i-j) \left(\vec{S}_{i,1} \vec{S}_{j,1} + \vec{S}_{i,2} \vec{S}_{j,2} \right) \right].$$
(329)

We consider staggered non-frustrating long-range interactions along the layers $J_{\parallel}(i-j)$ of the form (320). The short-range model ($\sigma \rightarrow \infty$) was the subject of several studies from the 1990s onward [162,183,184,340–346] investigating its quantum-critical properties. For a critical coupling ratio $\lambda = J_{\parallel}/J_{\perp}$, the system undergoes a QPT breaking the continuous SU(2) symmetry of the Hamiltonian from a ground state adiabatically connected to the product singlet state towards an antiferromagnetic Néel ground state with gapless magnon excitations (Nambu–Goldstone modes). For long-range interactions, we expect that these Goldstone modes are altered [38,321] and that the criticality of the system changes as a function of the decay exponent σ . We expect short-range criticality until $\sigma \ge \sigma_* = 2 - \eta_{SR}$, a non-trivial regime of continuously varying criticality in $\sigma_{uc} \le \sigma < \sigma_*$ with $\sigma_{uc} = 4/3$, and a long-range mean-field regime for $\sigma < \sigma_{uc}$ [20,21,244,326].

Until recently, to the best of our knowledge, there was a complete absence of numerical confirmation of this scenario. The first data for the critical point and exponents as a function of σ were published by Song et al. [37] using the SSE QMC algorithm and is now supplemented by pCUT+MC data from Adelhardt and Schmidt [206,339].

In Figure 33, the critical values from Refs. [37,206,339] are plotted together with the results from functional renormalisation group (FRG) calculations from Ref. [21] for the O(3) quantum rotor model. The critical points λ_c for both methods are in excellent agreement over the entire σ -range. However, the critical exponents cannot be directly

compared since the FSS of the magnetisation curves from SSE QMC simulations give β and ν , while pCUT+MC calculations for the gap and spectral weight give $z\nu$ and $(2 - z - \eta)z$. While the QMC data show relatively large error bars in the long-range mean-field regime, the pCUT approach overestimates the critical exponents in the short-range regime. For the two-dimensional long-range Heisenberg bilayer, the deviation is larger than for the LRTFIM in two dimensions. This is no surprise, as a previous series expansion for the nearest-neighbour square-lattice Heisenberg bilayer model showed similar deviations [347]. Also, the critical exponents from QMC are better at capturing the boundaries of the long-range mean-field and the short-range regime. In general, both approaches show good agreement with the FRG results from Ref. [21] for the non-trivial intermediate regime (which apparently has its own shortcomings at the boundary to the short-range regime).



Figure 33. Critical values and exponents from numerical studies of the Néel-ordering transition in the unfrustrated antiferromagnetic long-range Heisenberg square-lattice bilayer model. The upper panel shows the critical values. The middle left panel displays critical exponent values zv, the middle right panel the exponent $(2 - z - \eta)v$, the lower right panel the exponent v, and the lower right panel the one-particle spectral weight exponent β . The data points "QMC (2024)" for v and β originate from Ref. [37]. The data points "pCUT+MC (2024)" for zv and $(2 - z - \eta)v$ originate from Ref. [206,339]. The data points "FRG (2017)" for v and zv originate from Ref. [21]. The black dashed lines denote the critical exponents in the regime of short-range O(3) criticality (v = 0.7116(10), $\beta = 0.36932(16)$, zv = 0.7116(10), and $(2 - z - \eta)v = 0.6847(10)$ [348,349]) and long-range mean-field criticality ($v = 1/\sigma$, $\beta = 1/2$, zv = 1/2, and $(2 - z - \eta)v = 1/2$ [20,21]).

In conclusion, we summarised the results from SSE QMC and pCUT+MC studies for the square-lattice Heisenberg bilayer model with staggered antiferromagnetic interactions. Both approaches are in good agreement and confirm the three critical regimes [1-3,20,21,244,326] from the two-dimensional O(3) quantum rotor model within their limitations.

8.2. Staggered Antiferromagnetic Long-Range Heisenberg Ladder Models

After the discussion of the square-lattice bilayer model, we can imagine the Heisenberg ladder models as effectively reducing the dimension of the square-lattice bilayer model by one now considering two linear chains coupled by rung interactions. The long-range ladder Hamiltonian then reads

$$\mathcal{H} = J_{\perp} \sum_{i} \vec{S}_{i,1} \vec{S}_{i,2} - \frac{1}{2} \sum_{i \neq j} \left[J_{\parallel}(i-j) \left(\vec{S}_{i,1} \vec{S}_{j,1} + \vec{S}_{i,2} \vec{S}_{j,2} \right) + J_{\times}(i-j) \left(\vec{S}_{i,1} \vec{S}_{j,2} + \vec{S}_{i,2} \vec{S}_{j,1} \right) \right],$$
(330)

with long-range coupling $J_{II}(i-j)$ along the legs of the ladder and $J_{\times}(i-j)$ between spins of different legs. Recalling the long-range interactions of Equation (320) and the definitions of $J_{II}(i-j)$ and $J_{\times}(i-j)$, thereafter, we define two distinct Hamiltonians $\mathcal{H}_{II} = \mathcal{H}|_{J_{\times=0}}$ and $\mathcal{H}_{\bowtie} = \mathcal{H}$, where the first one includes long-range interactions only along the legs, while the second one is the original Hamiltonian including long-range interactions both along and in between the legs.

A sketch of these ladders is provided in Figure 34. As for the bilayer model, for small J/J_{\perp} (cf. Equation (320)), the ground state is adiabatically connected to the product state of rung singlets (rung-singlet ground state), while for strong coupling ratios, the long-range interactions want to induce an antiferromagnetic ground state. However, in contrast to the bilayer model in Section 8.1, there is no QPT for the nearest-neighbour Heisenberg ladder [350–352], due to the HMW theorem ruling out continuous symmetry breaking for one-dimensional quantum models [296]. Note that the HMW theorem only rules out a QPT with continuous symmetry breaking and not a QPT in general. For instance, the nearest-neighbour isotropic XY model in a transverse field exhibits a QPT without breaking the U(1) symmetry [284–293], while in the nearest-neighbour Heisenberg ladder, there is no QPT at all [350–352]. In fact, a QPT was ruled out until $\sigma \ge 2$ in another one-dimensional model, namely the staggered antiferromagnetic long-range Heisenberg chain in Ref. [317]. As there is a one-to-one correspondence between antiferromagnetic Heisenberg ladders and the low-energy properties of the one-dimensional O(3) quantum rotor model [113], we can expect a QPT predicted by the quantum field theory given by the action (326) with Equation (327) for $\sigma \leq \sigma_{lc} = 2 - \eta_{sr}$ with $\eta_{SR} = 0$ [20,21,244]. The existence of a lower critical decay exponent σ_{lc} has the consequence that there are only two quantum-critical regimes. One is the long-range mean-field regime for $\sigma \leq 2/3$, and the other one is a regime of a continuously varying critical exponent with a non-trivial fixed point. The third regime in these models is non-critical.

The only study investigating the full parameter space of this model is Ref. [35] using pCUT+MC and complementary linear spin–wave calculations. A previous study [306] using QMC and DMRG investigated the $\lambda = 1$ parameter line for \mathcal{H}_{\bowtie} . There are also known limiting cases of decoupled staggered long-range Heisenberg chains at $\lambda = \infty$, where a QPT from a QLRO phase towards the same Néel-ordered antiferromagnetic phase occurs. We can compare the critical points determined by pCUT+MC for \mathcal{H}_{\parallel} and the linear spin–wave results to the SSE QMC data from Ref. [22] and also to the linear spin–wave calculations in Refs. [22,322] of the long-range Heisenberg chain. In fact, the linear spin–wave calculations for \mathcal{H}_{\parallel} can be seen as a generalisation of the ones for the Heisenberg chain and, therefore, exactly including them as a limiting case.

We can find a plot in Figure 35 showing a ground-state phase diagram for both \mathcal{H}_{\parallel} and \mathcal{H}_{\bowtie} from Ref. [35,353]. The figure also includes other known values from Refs. [22,306], which fit into the overall picture of the pCUT+MC results. We also show the critical exponents zv, $(2 - z - \eta)v$, and α determined by the pCUT+MC approach. It is easy to identify the two critical regimes of long-range mean-field behaviour and continuously
varying critical exponents as predicted by quantum field theory [20,21,244]. In the nontrivial regime, the exponents shown here diverge when approaching the lower critical dimension $\sigma \to \sigma_{\rm lc}$, which is in reasonable agreement with FRG results in Ref. [21,244]. The largest deviation can be seen for α , which is the hardest exponent to extract from DlogPadé extrapolations. Using the (hyper)scaling relations (3)-(9), the remaining critical exponents can be determined. See Figure A2 in Appendix D showing all critical exponents for the Heisenberg ladders. It should be noted that there are three main difficulties discussed in Ref. [35]. First, as just stated, the α exponent is difficult to determine. Second, it becomes increasingly hard to extrapolate the perturbative series in the regions $\sigma \gtrsim 1.1$ $(\sigma \gtrsim 1.2)$ for $\mathcal{H}_{\parallel}(\mathcal{H}_{\bowtie})$. Third, the presence of logarithmic corrections to the dominant power-law behaviour close to the critical point spoils the exponents around the upper critical dimension at $\sigma = 2/3$. In the end, all factors play a role when determining all critical exponents due to error propagation. We can observe in Figure A2 that several exponents from the pCUT+MC approach deviate from the FRG exponents significantly in the non-trivial regime. For instance, the ν exponent seems to approach a constant value $\nu \approx 1$ for $\sigma \rightarrow \sigma_{lc}$ for pCUT+MC, while the FRG predicts a diverging exponent. Another important finding is that the lower critical decay exponent σ_{lc} is apparently not universal in these two models and considerably smaller than the predicted value $\sigma_{lc} = 2$ from quantum field theory [21,244]. This claim is made in Ref. [35] due to the known limiting case of the long-range Heisenberg chain [22] from SSE QMC simulations and due to linear spin-wave calculations [35].



Figure 34. Illustration of the quantum spin ladders with long-range interactions. For nearestneighbour interactions ($\sigma = \infty$), both long-range ladders \mathcal{H}_{\parallel} (left) and \mathcal{H}_{\bowtie} (right) reduce to the same Heisenberg ladder. The coupling on the rungs $\sim J_{\perp}$ is illustrated with black lines, and the long-range coupling along the legs $\sim J_{\parallel}(i - j)$ and in between the legs $\sim J_{\times}(i - j)$ is depicted in blue for \mathcal{H}_{\parallel} and in purple for \mathcal{H}_{\bowtie} . The figure is adapted from Ref. [35,353].

Beyond these interesting discrepancies, there was speculation about a possible deconfined quantum critical point along the $\lambda = 1$ parameter line for \mathcal{H}_{\bowtie} in Ref. [306]. The reason for this was the fact that the staggered long-range Heisenberg ladder undergoes a QPT from a disordered phase with a non-local string order parameter towards a Néel-ordered phase with conventional order. Also, they found a sharp peak and a gap in the dynamic structure factor at the ordering momentum k_c in the ordered phase, which could be indicative of deconfined excitations in terms of spinons [306,354]. Usually, when there is a QPT between two competing ordered phases, the system undergoes a first-order phase transition or there must be a coexistence phase. There is also a much more exotic scenario beyond the Landau–Ginzburg–Wilson theory of phase transitions [334]. A deconfined quantum critical point [334–337] is a second-order QPT that is not described by a "confining" order parameter, but by an emergent U(1)-symmetric gauge field with "deconfined" degrees of freedom accompanied by a fractionalisation of the order parameters [334]. A paradigmatic example is the deconfined QPT between a Néel-ordered antiferromagnetic ground state and a VBS state on a two-dimensional lattice [334,335,337,355–360]. While deconfined criticality was originally proposed in two dimensions, similarities in terms of a conventional Luttinger Liquid theory description have been drawn in one dimension [361–368]. Interestingly, a very close analogy to a two-dimensional deconfined critical point was found in Ref. [338] using a toy model with six-spin Heisenberg interactions and long-range two-spin interactions inducing a continuous phase transition between a VBS phase and an antiferromagnetic phase. A scenario between a conventional order and non-local string order as proposed by Ref. [306] would go even beyond the one-dimensional scenarios found so far. However, it was argued in Ref. [35] that there should be no such deconfined critical point in the above long-range Heisenberg ladders due to the critical exponents found and the fact that the rung-singlet phase is adiabatically connected to the trivial product state of rung singlets not falling into the category of symmetry-protected topological phases despite the presence of a non-local string order parameter [35,369]. Another possible interpretation of the finding in Ref. [306] is probably along the lines of Refs. [38,321]. The observed gap in the dynamic structure factor is in agreement with strong finite-size artefacts arising from the altered dispersion $\omega \sim |k|^s$ of sublinear behaviour s < 1 in the anomalous Goldstone regime $\sigma \leq 2$.



Figure 35. Critical values and exponents from numerical studies of the Néel-ordering transition in the unfrustrated antiferromagnetic long-range Heisenberg ladders. The panels show the critical values λ_c (upper left), $z\nu$ (upper right), $(2 - z - \eta)\nu$ (lower left), and α (lower right). The data points "pCUT+MC H_{II} ", "SW H_{II} ", "pCUT+MC H_{IX} ", and "SW H_{IX} " originate from Ref. [35,353] and refer to parallel (II) and parallel + diagonal (IX) interactions. The "SSE QMC (2005)" data point from Ref. [22] shows a $\lambda_c = \infty$ value on the long-range Heisenberg chain, which corresponds to the limiting case of decoupled legs. The "QMC (2022)" data [306] shows a $\lambda_c = 1$ value for \mathcal{H}_{XI} . The data points "FRG 2020" are from Ref. [244] and show the critical exponents for the one-dimensional O(3) quantum rotor model. The blue-shaded region denotes the σ regime in which long-range mean-field criticality is expected. The black-dashed lines denote long-range mean field critical exponents.

In this subsection, we have seen that the antiferromagnetic Heisenberg ladders with staggered long-range interactions show two quantum-critical regimes: one regime with long-range mean-field behaviour and a second non-trivial regime with continuously varying critical exponents. There is also a third regime that does not show any QPT because continuous symmetry breaking is ruled out by the HMW theorem. Despite the numerical confirmation of the two critical regimes predicted from quantum field theory [20,21,244,326], a discrepancy between the upper bounds of the Néel-ordered phase and the predicted lower critical exponent σ_{lc} was identified by Ref. [35]. On the other hand, the data in Ref. [35,353] are in agreement with other literature for the limiting case of long-range Heisenberg spin chains [22,322]. We also discussed briefly the possibility of a deconfined QPT and the presence of anomalous Goldstone modes in one dimension for the above ladder models. To put it briefly, the Hamiltonian (330) hosts some intriguing physics with several aspects that need further clarification.

8.3. Staggered Antiferromagnetic Long-Range Heisenberg Chain

Lastly, we consider another one-dimensional system, the Heisenberg chain with staggered non-frustrating antiferromagnetic long-range interactions:

$$\mathcal{H} = \sum_{i} \left[\vec{S}_{i} \vec{S}_{i+1} - \sum_{j=2}^{\infty} \lambda(i-j) \vec{S}_{i} \vec{S}_{i+j} \right].$$
(331)

In the previous two models (bilayer and ladders), the unperturbed part at $\lambda = 0$ consisted of uncoupled dimers with a trivial product singlet state as its ground state and local triplet excitations above. Here, the unperturbed part is the nearest-neighbour Heisenberg chain with a ground state exhibiting quasi-long-range order (QLRO) and fractionalised elementary excitations. These excitations are referred to as spinons and can be seen as propagating domain walls carrying S = 1/2 degrees of freedom. The perturbation consists of long-range interactions that couple sites beyond the nearest-neighbours. This interaction is of the same algebraic form as Equation (320). Because of the non-frustrating nature of the antiferromagnetic long-range interactions, it induces a Néel-ordered antiferromagnetic phase upon increasing its coupling strength. Again, a QPT breaking the SU(2)-symmetry of the Hamiltonian is only allowed when the long-range decay exponent satisfies $\sigma < 2$ due to the HMW theorem [22,317], and thus, such a transition can be ruled out for larger decay exponents. In this model, a QPT from a QLRO towards an ordered phase is expected to occur, and therefore, the ϕ^4 theory of Equation (326) does not apply. The k = 1 Wess– Zumino–Witten non-linear σ model [370,371] is known to describe the low-energy physics of Heisenberg chains and includes topological coupling to account for the presence of QLRO in the lattice model [22,213]. As for the ϕ^4 theory, a long-range coupling analogous to Equation (327) can be added to describe the Hamiltonian (331) [22].

Reference [22] is a comprehensive study of the Heisenberg chain (331) using largescale SSE QMC simulations to extract the critical properties of the QPT. The results for the critical point, as well as the critical exponents η and z can be found in Figure 36. The overall behaviour of the critical point as a function of the decay exponent σ is very similar to the one found in the previous subsection for the Heisenberg ladder. Here, the critical point diverges at about $\sigma \approx 1.8$ when approaching the lower critical exponent $\sigma \rightarrow \sigma_{lc}$. The hard boundary for a QPT is again given by the HMW theorem. Yet, there is a significant difference. While, for the Heisenberg ladders, the disordered rung-singlet phase exists for any $\sigma > 0$, for the Heisenberg chain, the QLRO phase only exists for $\sigma > 1$. Thus, for the Heisenberg chain, the critical line terminates in a marginal point at $\sigma = 1$ and $\lambda = 0$. For any $\sigma < 1$, the perturbation parameter λ becomes irrelevant, and the system is always in the antiferromagnetic phase [22]. The long-range Heisenberg chain (331) was also studied in Ref. [213,297,333] in the context of a Heisenberg chain with frustrated next-nearest-neighbour and non-frustrating long-range interactions, i.e., the $J_1 - J_2$ chain with staggered long-range interactions. In both models, the QLRO-Néel QPT can be identified by a level crossing between a triplet S = 0 and quintuplet S = 2excitations (The transition was initially misidentified as a level crossing between two S = 0states in Refs. [213,297] until it was later clarified to be a level crossing between the S = 0and S = 2 states [333]). The critical point from finite-size scaling of the level crossing from the ED [297] and DMRG [333] is in very good agreement with the QMC results at $\lambda = 1$ (see Figure 36). Proceeding with the critical exponents, we can see that the critical exponent η matches well with the field-theoretical expectations $\eta = 2 - \sigma$ (linear in at least leading order, but also a simple scaling argument predicts the linear behaviour [22]) in the range $1 \le \sigma \le 1.3$. For $\sigma > 1.3$, the η from SSE QMC simulations starts to deviate from the linear behaviour. One interesting observation was pointed out in Ref. [35]. In both the Heisenberg ladders and the Heisenberg chain, the linear behaviour $\eta = 2 - \sigma$ is expected from the underlying quantum field theory, yet the data from pCUT+MC and SSE QMC studies indicate a deviation from this with $\eta \leq 2 - \sigma$ for the Heisenberg ladders and $\eta \geq 2 - \sigma$ for the Heisenberg chain. It should be noted, however, that the exponent from the pCUT+MC approach is determined using scaling relations and, therefore, can suffer from unfavourable error propagation, especially when the α exponent is involved. The dynamical exponent z of the Heisenberg chain was extracted from SSE QMC simulations as well. The exponent z is one at the marginal point $\sigma = 1$ and then quickly drops to $z \approx 0.75$, where it seems to be constant within the error bars up to $\sigma = 1.7$. This finding is also in contrast to the RG prediction, where z = 1 in leading order, and the exponent is expected to be constant even in higher orders [22]. In Ref. [213], for $\lambda = 1$, the dynamic exponent z was determined in excellent agreement with the QMC results (see Figure 36). Also, the results for the $J_1 - J_2$ model with non-frustrating long-range interactions, where the same QLRO-Néel transition is realised, the exponent is in agreement with $z \approx 0.75$ [213,297]. Further, the RG analysis in Ref. [22] gave also a prediction for ν . However, this could not be compared with QMC as it was not possible to obtain accurate estimates of ν [22]. Note, also, that neither the long-range transverse-field XY chain nor the Heisenberg chain show long-range mean-field behaviour.

The Heisenberg chain with staggered long-range interactions is another prime example of long-range models hosting intriguing critical behaviour. The remaining discrepancy between the numerical SSE QMC results and the underlying field-theoretical description shows that the critical properties are not yet fully settled and further exploration of the model is necessary.



Figure 36. Critical values and exponents from numerical studies of the QLRO-Néel transition in the unfrustrated antiferromagnetic long-range Heisenberg chain. The upper panel shows critical values λ_c ; the lower left shows the critical exponent *z*; the lower right shows η . The data points "SSE QMC 2005" are from Ref. [22]; "ED 2010" are from Refs. [213,297]; the single data point "DMRG 2018" is from Ref. [333]. The dashed line is for η and is the prediction from the first-order RG and scaling arguments provided in Ref. [22].

9. Summary and Outlook

In this review, we gave an overview of recent advances in the investigation of the quantum-critical properties of quantum magnets with long-range interactions focusing on two techniques, both based on Monte Carlo integration, but complementary in spirit. On the one hand, we described pCUT+MC, where classical Monte Carlo integration is decisive in the embedding scheme of white graphs. This allows extracting series expansions of relevant physical quantities directly in the thermodynamic limit. On the other hand, SSE QMC enables calculations on large finite systems where finite-size scaling can be used to determine the physical properties of the infinite system. Both quantitative and unbiased approaches take the full long-range interaction into account and can be used a priori in any spatial dimension for any geometry.

In recent years, both techniques, alongside other methods, have been applied successfully to one- and two-dimensional quantum magnets involving long-range Ising, XY, and Heisenberg interactions on various bipartite and non-bipartite lattices. In this work, we have summarised the obtained quantum-critical properties including quantum phase diagrams and the (full sets of) critical exponents for all these systems coherently. Further, we reviewed how long-range interactions are used to study quantum phase transitions above the upper critical dimension and how the scaling techniques are extended to extract these quantum critical properties from the numerical calculations. This is indeed generically the case for all unfrustrated systems in this review, with the exception of the one-dimensional isotropic XY and Heisenberg chain. For frustrated systems, one can apply both MC techniques successfully for Ising interactions, while in general, only the pCUT+MC method is applicable (if an appropriate perturbative limit exists) due to the sign problem of SSE QMC. Nevertheless, in all frustrated cases, the small- σ regime of long-range interactions is challenging for both approaches, and further technical developments are desirable.

In the future, several extensions and research directions are interesting. As mentioned in Section 6.3 of this review, the interplay between long-range interactions and geometric frustration is a vibrant research field at the moment. It has been demonstrated numerically and experimentally that this interplay provides a great resource to engineer exotic phases of matter [54,55,109,283,372] with the most spectacular example being the \mathbb{Z}_2 quantum spin liquid on the Ruby lattice [55,109,372]. We expect further rapid development in the field since many promising theoretical proposals can be realised in analogue quantum simulation platforms (e.g., programmable Rydberg atom quantum simulators [52,55]).

In terms of methods, pCUT+MC is yet to be extended to arbitrary unit cells, larger spin values, and multi-spin interactions. The access to larger unit cells will enable the investigation of the interplay between long-range interactions and frustration on even more relevant lattice structures, e.g., the Kagome or Ruby lattice. There are systems with multi-spin interactions hosting deconfined quantum criticality [338,355–359], and an introduction of long-range interactions to this type of systems seems to be an interesting research topic [338]. We hope to spark further interest in the development and application of pCUT+MC by other users.

The SSE QMC approach is a widely used numerical tool for the calculation of unbiased thermal averages of observables. A large variety of distinct QPTs is potentially accessible by these QMC simulations. We envision that, for all systems that do not suffer from a sign problem, SSE QMC, in combination with appropriate zero-temperature protocols and finite-size scaling, can be used to study how long-range interactions affect QPTs beyond the standard O(n)-symmetry. The SSE QMC method has also been extended to tackle frustrated systems in a more efficient way [261,262,373]. However, an efficient treatment of both the long-range interaction and frustration has not been introduced yet [32,40]. Reference [374] developed an SSE QMC approach to access the toric code quantum spin liquid regime [55,109,372]. An application of the SSE QMC approach to extended longrange interacting Bose–Hubbard models (see Section 5.5) along the lines of directed loop updates [215–217] would also be a natural development. This would enable numerically calculating observables with SSE QMC for ultracold gas experiments with optical lattices. A possible application could be the study of complex crystalline phases and their breakdown in frustrated Bose–Hubbard systems with long-range interactions [64,73,111,230].

Finally, in the context of the long-range mean-field regime above the upper critical dimension, much research has been conducted regarding finite-size scaling in classical systems [12–19,41,125,128,142,145], including the study of multiplicative logarithmic corrections for the characteristic length scale at the upper critical dimension [17] and the investigation of the role of Fourier modes and boundary conditions [19]. On the contrary, its quantum counterpart has only been treated successfully in recent years [32,34]. In addition to the transfer of established concepts from classical to quantum Q-FSS, one interesting open question is a detailed understanding of the crossover regime between classical and quantum Q-FSS for small temperatures. Even though we focused on the quantum version of the Q-FSS of Ref. [34] due to the quantum nature of the models analysed in this review, the ground work has been conducted by the inventors of classical Q-FSS (Refs. [15–19,41,125]) and, in general, many other researchers who provided valuable insight into the scaling above the upper critical dimension, e.g., Refs. [10,12–14,127,128,142]. Overall, it is exciting that the abstract concept of dangerous irrelevant variables and the

physics above the upper critical dimension are accessible in quantum–optical platforms realising long-range interactions.

Author Contributions: Conceptualisation: all authors; investigation: all authors; writing—original draft preparation: introduction (J.A.K. and K.P.S.), quantum phase transitions (J.A.K. and A.L.), Monte Carlo integration (J.A.K. and A.L.), series expansion Monte Carlo embedding (P.A.), stochastic series expansion quantum Monte Carlo (J.A.K. and A.L.), long-range transverse-field Ising models (J.A.K. and A.L.), long-range transverse-field XY chain (P.A. and J.A.K.), long-range Heisenberg models (P.A.), and conclusion and outlook (J.A.K. and K.P.S.); writing—review and editing: all authors; visualisation: P.A., J.A.K. and A.L.; supervision: K.P.S. All authors have read and agreed to the published version of the manuscript.

Funding: This work was funded by the Deutsche Forschungsgemeinschaft (DFG) (German Research Foundation)—Project-ID 429529648—TRR 306 QuCoLiMa (Quantum Cooperativity of Light and Matter). K.P.S. acknowledges the support by the Munich Quantum Valley, which is supported by the Bavarian state government with funds from the Hightech Agenda Bayern Plus. The authors gratefully acknowledge the scientific support and HPC resources provided by the Erlangen National High Performance Computing Center (NHR@FAU) of the Friedrich-Alexander-Universität Erlangen-Nürnberg (FAU) under the NHR project b177dc ("SELRIQS"). NHR funding is provided by federal and Bavarian state authorities. NHR@FAU hardware is partially funded by the German Research Foundation (DFG)–440719683.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Acknowledgments: We thankfully acknowledge the valuable input on parts of the manuscript by Antonia Duft, Matthias Mühlhauser, Calvin Krämer, and Andreas Schellenberger. We also thank Jiarui Zhao and Nicolas Laflorencie for providing the data from the publications [22,37]. We are grateful for the fruitful collaboration with Sebastian Fey and Sebastian C. Kapfer regarding many of the projects reviewed in this work. It is with great dismay that we express our deepest condolences on the death of Ralph Kenna, who invited and encouraged us to write this review as part of his Special Issue on the "violation of hyperscaling above the upper critical dimension" in the journal "*Entropy*". We will always remember his encouraging and cheerful spirit in our correspondence with him.

Conflicts of Interest: The authors declare no conflicts of interest.

Abbreviations

The following abbreviations are used in this manuscript:

AF	antiferromagnetic
DIV	dangerous irrelevant variable
DMRG	density matrix renormalisation group
DOF	degrees of freedom
F	ferromagnetic
FRG	functional renormalisation group
FSS	finite-size scaling
GHF	generalised homogeneous function
HMW theorem	Hohenberg-Mermin-Wagner theorem
iDMRG	infinite density matrix renormalisation group
LCE	linked-cluster expansion
LRI	long-range interactions
LRTFAXYM	long-range transverse-field anisotropic XY model
LRTFIM	long-range transverse-field Ising model
MC	Monte Carlo
MCI	Monte Carlo integration
pCUT	perturbative continuous unitary transformation
PDF	probability density function
PI	path integral
QLRO	quasi long-range order
QMC	quantum Monte Carlo

QPT	quantum phase transition
qp	quasiparticle
RG	renormalisation group
SSE	stochastic series expansion
SPO	stochastic parameter optimisation
TDVP	time-dependent variational principle
VBS	valence bond solid

Appendix A. Short-Range $\mathcal{O}(n)$ Transition ϕ^4 -Theory

For ferromagnetic short-range interacting transverse-field Ising models (see Equation (292)) the underlying \mathbb{Z}_2 symmetry of the Hamiltonian is spontaneously broken at the quantum phase transition [113]. The associated order parameter is the magnetisation in coupling direction which is microscopically defined as $\sum_i \sigma_i^z$ for Ising couplings in *z*-direction. Rewriting the partition function of the TFIM in the picture of Feynman path integrals, coarse-graining and going to the continuum limit is the general approach to derive a field theoretical description at criticality [113]. The fields of the quantum field theory $\phi(x, \tau)$ are obtained by coarse-graining the order parameters at imaginary time τ for a coarse-graining neighbourhood $\mathcal{N}(x)$ [113]

$$\phi(x,\tau) \propto \sum_{i \in \mathcal{N}(x)} \sigma_i^z$$
(A1)

By properly [113] taking care of the microscopic degrees of freedom leading to terms of order $(\phi^2)^k$ with $k \in \mathbb{N}$, the resulting sufficiently truncated action leading to the right partition function to describe the phase transition is given by

$$\mathcal{Z} = \int \mathcal{D}\phi(x,\tau) \, e^{-\mathcal{S}_{\phi}} \tag{A2}$$

$$S_{\phi} = \int d^{d}x \int_{0}^{\beta} d\tau \left[\{ g(\partial_{\tau}\phi)^{2} + (\nabla_{x}\phi)^{2} + r\phi^{2} \} + u\phi^{4} \right]$$
(A3)

with periodic boundary conditions in imaginary time τ [113]. Note that the dependencies of $\phi(x, \tau)$ remain, but are dropped to lighten the notation. Gradient and time derivative terms are accounting for the spatial and temporal direction fluctuations [113,132]. The initial couplings in Equation (A3) are depending on the values of the couplings in the microscopical Hamiltonian [113]. Regarding the action in Equation (A3) the symmetry-breaking quantum phase transition between $\langle \phi \rangle = 0$ and $\langle \phi \rangle \neq 0$ becomes visualised as tuning the coupling *r* to a negative sign changes the minimum structure of the action into the symmetry broken Mexican hat. The field theory defined in Equation (A3) is a distilled representation of the key features of the quantum short range Ising criticality with ϕ having the interpretation of an "order parameter field".

In general, field theories as in Equation (A3) are used to study QPT with a spontaneous breaking of a $\mathcal{O}(n)$ symmetry. Here, $\mathcal{O}(n)$ refers to the orthogonal group (the group of all orthogonal $(n \times n)$ -matrices). The study of a $\mathcal{O}(1)$ -symmetry breaking is equivalent to the study of a \mathcal{Z}_{\in} -symmetry breaking since both groups are isomorphic. Dependent on *n* of the orthogonal group $\mathcal{O}(n)$ the quantum fields $\phi(x, \tau)$ of the action Equation (A3) become multicomponent fields [113,132]. In order to study a $\mathcal{O}(n)$ symmetry breaking QPT real-valued fields $\phi(x, \tau)$ with *n* components are considered.

After introducing actions of the form Equation (A3) to describe quantum criticality, we review basic computations that can be performed within this framework. We start with the derivation of field correlators

$$G^{(n)}(x_1,\tau_1;\ldots;x_n,\tau_n) = \frac{1}{\mathcal{Z}} \int \mathcal{D}\phi(x,\tau) \,\phi(x_1,\tau_1)\ldots\phi(x_n,\tau_n) \,e^{-S_{\phi}} \tag{A4}$$

from a field-path-integral formulated partition function is technically accomplished by additionally coupling the order parameter field $\phi(x, \tau)$ linearly to an auxiliary field $j(x, \tau)$ [132]

$$S_{\phi,j} = \int d^d x \int_0^\beta d\tau \left[\{ g(\partial_\tau \phi)^2 + (\nabla_x \phi)^2 + r\phi^2 \} + u\phi^4 - \phi(x,\tau)j(x,\tau) \right].$$
(A5)

The *n*-field correlator can now be understood as the functional derivative of the partition function with respect to the respective auxiliary fields [132]

$$G^{(n)}(x_1,\tau_1;\ldots;x_n,\tau_n) = \frac{1}{\mathcal{Z}} \left[\frac{\delta}{\delta j(x_1,\tau_1)} \dots \frac{\delta}{\delta j(x_n,\tau_n)} \int \mathcal{D}\phi(x,\tau) \, e^{-\mathcal{S}_{\phi,j}} \right]_{j=0}.$$
 (A6)

From the expression for the spatial correlator in Table 1 we see that at criticality the system is scale-free therefore for distances $x \gg a$ much larger than the lattice spacing the theory should be invariant under a scaling transformation of coordinates in space and time

$$x \to x' = x/b \tag{A7}$$

$$\tau \to \tau' = x/b^z \tag{A8}$$

with *b* being a rescaling factor [113]. We define the power by which a quantity O has to be rescaled in order to preserve the structure as the scaling dimension [O] of that quantity [113]. The Gaussian part of the action in Equation (A3) keeps its structure under scaling transformations if the field ϕ and the mass coupling *r* are also scaled appropriately. One can calculate the scaling dimensions of ϕ and *r* as follows,

$$\int d^{d}x' \int_{0}^{\beta} d\tau' (\partial_{\tau'} \phi'(x',\tau'))^{2} = b^{-d-z+2z+2[\phi]} \int d^{d}x \int_{0}^{\beta} d\tau (\partial_{\tau} \phi(x,\tau))^{2}$$
(A9)

$$\int d^{d}x' \int_{0}^{\beta} d\tau' (\nabla_{x'}\phi'(x',\tau'))^{2} = b^{-d-z+2+2[\phi]} \int d^{d}x \int_{0}^{\beta} d\tau (\nabla_{x}\phi(x,\tau))^{2}$$
(A10)

$$d - z + 2z + 2[\phi] = 0 \tag{A12}$$

$$-d - z + 2 + 2[\phi] = 0 \tag{A13}$$

$$-d - z + [r] + 2[\phi] = 0 \tag{A14}$$

$$z = 1 \tag{A15}$$

$$[\phi] = (d-z)/2 = (d-1)/2$$
(A16)

$$[r] = 2z = 2 \tag{A17}$$

$$\phi \rightarrow \phi'(x',\tau') = b^{(d-1)/2}\phi(x,\tau) \tag{A18}$$

$$r \to r' = b^2 r \,. \tag{A19}$$

So far the following naively obtained power counting scaling dimensions were encountered,

∜

$$[\mathrm{d}^d x] = -d \tag{A20}$$

$$[\mathrm{d}\tau] = -z = -1 \tag{A21}$$

$$\left[\nabla_{x}\right] = 1 \tag{A22}$$

$$[\partial \tau] = z = 1 \tag{A23}$$

$$[\phi] = (d-z)/2 = (d-1)/2 \tag{A24}$$

$$[r] = 2. (A25)$$

$$[ddx] + [d\tau] + [u] + 4[\phi] = 0.$$
(A26)

Therefore, the ϕ^4 term remains invariant for a scaling dimension of [u] = 3z - d = 3 - d. We can now regard the scaling in terms of differential equations representing the renormalisation group flow of the coupling constants using b = 1 + dl for $dl \ll 1$ and building up a finite rescaling $b = e^l$ by repeated action of infinitesimal rescalings [113]

∜

$$r + dr = e^{dl[r]}r = (1 + dl[r])r = r + dl[r]r$$
(A27)

$$u + du = e^{dl[u]}u = (1 + dl[u])u = u + dl[u]u$$
(A28)

$$\frac{\mathrm{d}r}{\mathrm{d}l} = 2r \tag{A29}$$

$$\frac{\mathrm{d}u}{\mathrm{d}l} = (3-d)u \,. \tag{A30}$$

Focusing only on the flow of r (see Equation (A29)) one sees that for r > 0 it flows towards the stable fixed point at $r = \infty$, and towards the stable fixed point at $-\infty$ for r < 0, while there is an unstable fixed point at $r = r^* = 0$. Regarding also the coupling u one sees directly one fixed point of the coupled flow at $(r^* = 0, u^* = 0)$. For d > 3 the flow in u is stable towards the fixed point while for d < 3 it is unstable. That is a hint of the irrelevance of the coupling u for systems with d > 3 as independently of the starting condition the system is always in the attraction basin of $u^* = 0$.

A coupling *c* with scaling dimension [c] < 0 always has one stable fixed point at $c^* = 0$. Independent of the starting point it always flows to zero and is called irrelevant [10,127]. As long as the coupling is not dangerously irrelevant (see Section 2.3), meaning that there is no additional singular behaviour in observables approaching $c \to 0$ [127], it can be neglected. That is the reason why it is sufficient to regard an action with ϕ^4 as its highest power as all higher powers have irrelevant couplings [10].

The spatial dimension *d* at which the coupling of *u* has scaling dimension zero [u] = 0 is called the upper critical dimension. Below the upper critical dimension the coupling *u* needs to be included perturbatively and a " true" decimation of degrees of freedom is necessary [113]. In this thesis we will not deal with the technicalities, but in order to understand some arguments from other resources a basic understanding of the principle and the results of renormalisation group techniques are required. The decimation of degrees of freedom (DOF) is often done in this context by a momentum shell renormalisation [113]. Here as a first step " fast" DOF, i.e., DOF with large momentum/frequency, are partially integrated out [113,132,375]. One chooses a momentum cutoff Λ and frequency cutoff \mathcal{J} and defines $\tilde{\Lambda} = \Lambda/b$ and $\tilde{\mathcal{J}} = \mathcal{J}/b^z$. After that one integrates the " fast" DOF between Λ and $\tilde{\Lambda}$ and respectively \mathcal{J} and $\tilde{\mathcal{J}}$ [113,132,375]. After that one rescales x' = x/b which is equivalent to $\Lambda' = b\tilde{\Lambda} = \Lambda$ and $\tau' = \tau/b^z$ which is equivalent to $\mathcal{J}' = b^z \tilde{\mathcal{J}} = \mathcal{J}$ [113,132,375]. The most important result of this scheme are new modified couplings

$$\tilde{r} = r + f_r(u, r) \tag{A31}$$

$$\tilde{u} = u + f_u(u, r) \tag{A32}$$

which are related to the original couplings plus some contributions $f_{r/u}$ which can be calculated order by order in *u* from occurring Feynman diagrams [113]. The modified couplings scale as [113]

$$r' = b^2 \tilde{r} \tag{A33}$$

$$u' = b^{3-d}\tilde{u} . \tag{A34}$$

One can now also calculate the flow equations from those quantities,

$$\frac{\mathrm{d}r}{\mathrm{d}l} = 2r + g_r(r, u) \tag{A35}$$

$$\frac{\mathrm{d}u}{\mathrm{d}l} = (3-d)u + g_u(r,u) \tag{A36}$$

for which we see that the equations are not decoupled like in the case of the Gaussian theory with an irrelevant u (see Equations (A29) and (A30)) and there is a second fixed point $(r^{\star\star}, u^{\star\star})$ differing from the trivial one $r^{\star} = u^{\star} = 0$ [113]. The new non-trivial fixed point is called Wilson-Fisher fixed point [113]. One can express the fixed point and flow equations in terms of $\epsilon = d_{uc} - d$ and perturbatively extract the values of $(r^{\star\star}, u^{\star\star})$ as well as the eigenstates and eigenvalues of the linearised flow equation around the Wilson-Fisher fixed point [113].

For calculations higher than first-order it is also necessary to rescale the field at least for the apparent ϕ^4 theory of the TFIM due to self-energy contributions changing the propagator [113]. This could be taken care of by defining a new scaling dimension of the field $[\phi]_{\text{new}} = [\phi]_{\text{naive}} + h(\eta)$ from the naive scaling dimension (see Equation (A24)) and a further functional dependence $h(\eta)$ on the anomalous dimension exponent to be consistent with the scaling of the two-point correlators η [113]. The underlying mechanism relating the anomalous dimension to the self-energy $\Sigma(q)$ is as follows: We define the self-energy as the difference between the inverse propagator of the Gaussian field theory and the propagator including the u terms $G_0^{-1}(q, \omega) - G^{-1}(q, \omega) = \Sigma(q, \omega)$ and we find,

$$G(q,\omega) = \frac{1}{G_0^{-1}(q,\omega) - \Sigma(q,\omega)} = \frac{1}{q^2 + \Sigma(q,\omega) + \tilde{g}\omega^2 + r} \propto \frac{1}{q^{2-\eta}}$$
(A37)

that the self-energy potentially changes the leading q dependence for small q values [113]. This difference is incorporated into the anomalous dimension [113]. The Fourier transform of the propagator gives the two-point correlators which again are proportional to two times the field scaling dimension. Concluding the scaling dimension,

$$[\phi] = [\phi]_{\text{naive}} + h(\eta) \tag{A38}$$

is always necessary if the anomalous dimension is non-zero [113].

As already seen for the anomalous dimension exponent η in Equation (A37) and the dynamical correlation length exponent *z* it is possible to connect the canonical exponents (see Table 1) to the results of scaling/renormalisation group theory in the field-theoretical framework. We will now discuss the relations between canonical exponents which characterise a phase transition and scaling results. We will also briefly skim the origin of the fluctuation-dissipation relation and the analogy between the auxiliary field in Equation (A5) and the conjugate field to coupling to the order parameter in the microscopic model.

• The correlation length exponent ν is given by the inverse of the relevant eigenvalue λ_1 of the linearised flow equation spectrum at the relevant fixed point of the phase transition [113]. That means $\nu = 1/\lambda_1$ given by the non-trivial eigenvalue λ_1 of the linearised flow equations at the Wilson-Fisher fixed point below the upper critical dimension and $\nu = 1/[r]$ from the Gaussian fixed point above the upper critical dimension [113].

- The dynamical correlation length exponent stands for the space–time anisotropy. So far the short-range ϕ^4 -theory defined in Equation (A3) did not have such an anisotropy, but in the case of long-range interaction models [20,21,31] or the Bose–Hubbard superfluidity onset transition [31,113,231] such space–time anisotropy occurs.
- The anomalous dimension exponent η is obtained by the q proportionality of the propagator of the theory $G(q, \omega) \propto q^{\eta-2}$ and is defined to stand for the difference to two in the proportionality [113]. The anomalous dimension η is therefore affected by self-energy contributions. We will also see in the long-range theory that even in the free theory without any self-energy contributions an η can occur if the spatial fluctuations in the action go with a q^{σ} term with $\sigma < 2$ [20,21,31].
- The conjugate field used in the consideration of phase transitions in the microscopical model translates to the auxiliary field in Equation (A5) [113,132]. Therefore the scaling dimension of the auxiliary field is in the Gaussian theory given by the scale invariance of

$$\int d^d x \int d\tau \phi(x,\tau) j(x,\tau) \tag{A39}$$

$$[d^d x] + [d\tau] + [\phi] + [j] = 0.$$
(A40)

- According to Equation (A6) one can derive the expectation value of the field φ and the two-point correlator via functional derivatives with respect to the auxiliary field *j* and setting *j* to zero afterwards [132]. This is in full analogy to the derivation of the magnetisation and the susceptibility from the free energy via derivatives with respect to the longitudinal field.
- In the linear coupling formalism, Equation (A6) leads to the later used fluctuationdissipation relation between the two-point correlator and the susceptibility [113],

$$\Im\mathfrak{m}[\chi(q,\omega)] = f(\omega/T)S(q,\omega) \tag{A41}$$

with $f(\omega/T)$ being a function implementing the temperature occupation, $S(q, \omega)$ the dynamic structure factor and $\Im m[\chi(q, \omega)]$ the imaginary part of the susceptibility in momentum and frequency space.

Appendix B. Generalised Homogenous Functions

The following definitions, theorems and comments on generalised homogeneous functions are replicated from [124]. Proofs to all presented theorems can be found in [124].

Definition A1. A function $f(x_1, x_2, ..., x_n)$ is a generalised homogeneous function (GHF) with *n* variables if there exist *n* numbers $a_1, a_2, ..., a_n$ such that for all positive $\lambda \in \mathbb{R}^+$,

$$f(\lambda^{a_1}x_1, \lambda^{a_2}x_2, \dots, \lambda^{a_n}x_n) = \lambda^{a_f}f(x_1, x_2, \dots, x_n).$$
(A42)

 a_1, a_2, \ldots, a_n are being referred to as scaling powers of the respective variables x_1, x_2, \ldots, x_n and a_f is the scaling power of $f(x_1, x_2, \ldots, x_n)$. The scaling powers a_1, \ldots, a_n are non-zero.

Remark A1. Regarding a GHF with *n* variables there are only *n* independent scaling powers. This can be seen by setting $\lambda = \tilde{\lambda}^p$, leading to

$$f(\tilde{\lambda}^{pa_1}x_1, \tilde{\lambda}^{pa_2}x_2, \dots, \tilde{\lambda}^{pa_n}x_n) = \tilde{\lambda}^{pa_f}f(x_1, x_2, \dots, x_n).$$
(A43)

If $a_f = 0$ the statement that there are only *n* independent scaling powers is trivial. If $a_f \neq 0$ one can set $p = 1/a_f$ resulting in

$$f(\tilde{\lambda}^{a_1/a_f}x_1, \tilde{\lambda}^{a_2/a_f}x_2, \dots, \tilde{\lambda}^{a_n/a_f}x_n) = \tilde{\lambda}f(x_1, x_2, \dots, x_n).$$
(A44)

Theorem A1. Let $f(x_1, x_2, ..., x_n)$ and $g(x_1, x_2, ..., x_n)$ be GHF with scaling powers a_1 , a_2 , ..., a_n and respective a_f and a_g then,

- $f \cdot g$ is a homogeneous function with scaling power $a_f + a_g$.
- f + g is only a homogeneous function if $a_f = a_g$. The scaling power of f + g is then a_f .

Theorem A2. If $f(x_1, x_2, ..., x_n)$ is a GHF with *n* variables with scaling power a_f , then

$$f^{(j_1,j_2,\ldots,j_n)}(x_1,x_2,\ldots,x_n) = \frac{\partial^{j_1}}{\partial x_1^{j_1}} \frac{\partial^{j_2}}{\partial x_2^{j_2}} \cdots \frac{\partial^{j_n}}{\partial x_n^{j_n}} f(x_1,x_2,\ldots,x_n)$$
(A45)

is also a GHF

$$f^{(j_1, j_2, \dots, j_n)}(\lambda^{a_1} x_1, \lambda^{a_2} x_2, \dots, \lambda^{a_n} x_n) = \lambda^{a_f - \sum_{i=1}^n j_i a_i} f^{(j_1, j_2, \dots, j_n)}(x_1, x_2, \dots, x_n)$$
(A46)

with scaling power $a_f - \sum_{i=1}^n j_i a_i$.

Theorem A3. Let $f(x_1, x_2, ..., x_n)$ be a GHF with *n* variables and scaling power a_f . The Legendre transform of $f(x_1, x_2, ..., x_n)$ in which x_i is replaced as a free parameter by its conjugate variable $\overline{x}_i = f^{(0,0,...,0,j_i=1,0,...,0)}(x_1, x_2, ..., x_n)$

$$\overline{f}(x_1, x_2, \dots, \overline{x}_i, \dots, x_n) = f(x_1, x_2, \dots, x_n) - x_i \overline{x}_i$$
(A47)

is also a GHF with the same scaling power $a_{\overline{f}} = a_f$. The scaling power of \overline{x}_i is given by Theorem A2 as $\overline{a}_i = a_f - a_i$.

Theorem A4. A function $f(x_1, x_2)$ is a two-variable GHF with scaling power a_f and independent scaling powers a_1, a_2 if there exists some function $g_{\pm}(u)$ such that

$$f(x_1, x_2) = |x_1|^{a_f/a_1} g_{\text{sgn}(x_1)} \left(\frac{x_2}{|x_1|^{a_2/a_1}} \right)$$
(A48)

or equally, if there exists some function $h_{\pm}(u)$ such that

$$f(x_1, x_2) = |x_2|^{a_f/a_2} h_{\operatorname{sgn}(x_2)} \left(\frac{x_1}{|x_2|^{a_1/a_2}}\right).$$
(A49)

Conversely if *f* is a GHF then,

$$g_{\pm}(u) = f(\pm 1, u)$$
 (A50)

$$h_{\pm}(u) = f(u, \pm 1)$$
 (A51)

The statement of this theorem is straightforward generalisable to *n*-variable GHF. Due to limitations in compact notation for the formulation two-variable GHF are used.

Theorem A5. All GHFs have power-law singularities at the origin when it is approached along one of the principal axes. The exponent for the path of the approach $|x_i| \rightarrow 0$ is given by a_f/a_i . For a function of *n* variables that results in,

$$f(0,\ldots,0,x_i,\ldots,0) \propto |x_i|^{a_f/a_i} f(0,\ldots,0,1,\ldots,0).$$
 (A52)

Theorem A6. Let $f(x_1, x_2, ..., x_n)$ be a *n*-variable GHF with scaling power a_f . Let

$$\tilde{f}(x_1, x_2, \dots, \tilde{x}_j, \dots, x_n) = \int \mathrm{d}^d x_j f(x_1, x_2, \dots, x_j, \dots, x_n) \exp(i x_j \tilde{x}_j)$$
(A53)

denote the Fourier transform in which x_j is replaced by \tilde{x}_j as the independent variable and d being the dimensionality of the variable. $\tilde{f}(x_1, x_2, ..., \tilde{x}_j, ..., x_n)$ is also a GHF with scaling power $\tilde{a}_f = a_f - d\tilde{a}_j$ and the scaling power of the transformed variable \tilde{x}_j being $\tilde{a}_j = -a_j$

$$\tilde{f}(\lambda^{a_1}x_1,\lambda^{a_2}x_2,\ldots,\lambda^{\tilde{a}_j}\tilde{x}_j,\ldots,\lambda^{a_n}x_n) = \lambda^{a_f-d\tilde{a}_j}\tilde{f}(x_1,x_2,\ldots,\tilde{x}_j,\ldots,x_n).$$
(A54)

Appendix C. Walker's Method of Alias

Walker's method of alias is a numerical technique to draw integer random numbers from a discrete PDF in constant time [212,214]. Let $P(\omega_j) = q_j$ be a discrete probability distribution with $j \in \{1, ..., N\}$ elements ω_j and their probabilities q_j . The key idea of Walker's method is to redistribute the unbalanced part of the probability distribution. A naive way to sample a discrete distribution would be the rejection method. This means to draw an integer $j \in \{1, ..., N\}$ uniformly and accept it with probability $q_j / \max(q_j)$ [214]. However, this leads to $\mathcal{O}(N)$ steps until a proposed move eventually gets accepted [214]. In the Walker's method one draws the integer j uniformly, possibly accepts it with a probability P_j and otherwise chooses its alias A_j . One therefore succeeds after one try. The probabilities P_j and the aliases A_j need to be determined in advance such that the correct distribution q_j is sampled. In Ref. [214] an algorithm to set up the tables for P_j and A_j with complexity $\mathcal{O}(N)$ is given. Note that the construction of the tables needs to be conducted only once before the actual simulation is performed. During the simulation we draw from q_j with the help of the precalculated P_j and A_j in $\mathcal{O}(1)$ time. The Walker's method therefore does not alter the complexity of the whole algorithm.

In Figure A1 we illustrate the algorithm by Fukui and Todo [214] to set up the tables P_j and A_j . The algorithm proceeds as follows: One starts by setting P_j to a preliminary distribution $P_j = N \cdot q_j$ and creates an empty alias table A_j . The table P_j is split into $P_j \ge 1$ and $P_j < 1$ (see Figure A1b). The elements of the partition $P_j < 1$ get filled up by the elements $P_j \ge 1$. Starting from the outermost right elements of both parts in Figure A1b one fills up the shortfall $1 - P_j$ (green blocks in Figure A1) of weight P_j and sets the alias A_j to the number corresponding to the weight left from the separating line of the partitions $(j = 5 \text{ and } A_j = 6 \text{ in Figure A1b})$. This alias is later chosen with probability $1 - P_j$ if j gets drawn. In order to compensate for this, one cuts the weight P_{A_j} of the alias by $1 - P_j$ (red blocks in Figure A1). If the new weight P_{A_j} falls below 1, it is transferred to the partition with $P_j < 1$ (see e.g., j = 6 in Figure A1d,e). This filling up of the shortfalls is performed as long as the partition $P_j < 1$ is not empty. Figure A1f therefore shows the last step.



Figure A1. Illustration of Walker's method. The horizontal dashed line corresponds to the mean of the probability distribution q_j with $j \in \{1, ..., N\}$. (a) Definition of a tentative probability $P_j = n \cdot q_j$ which compares the probabilities q_j with the mean q = 1/N. (b) The probabilities q_j are discarded, a table A_j for the aliases is created and the distribution is splitted into $P_j \ge 1$ and $P_j < 1$. (c–f) The weights of P_j are gradually redistributed by creating aliases until the distribution is flattened. The weights that are taken away are depicted with red bars and the added weights with green bars. The values of P_j and A_j that get modified during a step are written in red. At the right part of the distributions, the weights which are already filled up are gathered. In the middle part of the distribution there are the weights which still need to be filled and the left part contains all weights that are at least full from the beginning and yet need to be redistributed. (c) Weight of 6 is added to 2 and $A_2 = 6$ is set. (d) Weight of 6 is added to 4 and $A_4 = 6$ is set. (e) Weight of 6 is added to 2 and $A_2 = 6$ is set. (f) Weight of 3 is added to 6 and $A_6 = 3$ is set. After this, the middle part is empty and we are done with the redistribution.

For drawing ω_j corresponding to its original weight q_j , one draws a candidate k for j uniformly in 1, . . . , N and a uniform number $u \in [0, 1]$. If $u < P_k$ then j = k and otherwise $j = A_k$.



Appendix D. Critical Exponents for Long-Range Heisenberg Ladders

Figure A2. Critical exponents of the Néel-ordering transition in the unfrustrated antiferromagnetic long-range Heisenberg ladders. The panels display the exponents α (first row left), β (first row left), γ (second row left), δ (second row right), η (third row left), ν (third row right), and z (fourth row left). The data points 'pCUT+MC $H_{||}$ ' and 'pCUT+MC $H_{||}$ ' are from Ref. [35,353] and refer to parallel (1) and parallel + diagonal (\bowtie) interactions. The other data 'FRG 2020' is from Ref. [244] and show the critical exponents for the one-dimensional O(3) quantum rotor model. The blue shaded region denotes the σ regime in which long-range mean-field criticality is expected. The black dashed lines denote long-range mean field critical exponents.

References

- Sak, J. Recursion Relations and Fixed Points for Ferromagnets with Long-Range Interactions. *Phys. Rev. B* 1973, *8*, 281–285. [CrossRef]
- Sak, J. Low-temperature renormalization group for ferromagnets with long-range interactions. *Phys. Rev. B* 1977, 15, 4344–4347. [CrossRef]
- 3. Fisher, M.E.; Ma, S.k.; Nickel, B.G. Critical Exponents for Long-Range Interactions. Phys. Rev. Lett. 1972, 29, 917–920. [CrossRef]
- 4. Maity, S.; Bhattacharya, U.; Dutta, A. One-dimensional quantum many body systems with long-range interactions. *J. Phys. A Math. Theor.* **2019**, *53*, 013001. [CrossRef]
- Defenu, N.; Donner, T.; Macrì, T.; Pagano, G.; Ruffo, S.; Trombettoni, A. Long-range interacting quantum systems. *Rev. Mod. Phys.* 2023, 95, 035002. [CrossRef]
- 6. Wilson, K.G. Renormalization Group and Critical Phenomena. I. Renormalization Group and the Kadanoff Scaling Picture. *Phys. Rev. B* **1971**, *4*, 3174–3183. [CrossRef]
- Wilson, K.G. Renormalization Group and Critical Phenomena. II. Phase-Space Cell Analysis of Critical Behavior. *Phys. Rev. B* 1971, 4, 3184–3205. [CrossRef]
- 8. Wilson, K.G.; Kogut, J. The renormalization group and the epsilon expansion. *Phys. Rep.* 1974, 12, 75–199. [CrossRef]
- 9. Wilson, K.G. The renormalization group and critical phenomena. Rev. Mod. Phys. 1983, 55, 583-600. [CrossRef]
- 10. Fisher, M.E. Renormalization group theory: Its basis and formulation in statistical physics. *Rev. Mod. Phys.* **1998**, *70*, 653–681. [CrossRef]
- 11. Stanley, H.E. Scaling, universality, and renormalization: Three pillars of modern critical phenomena. *Rev. Mod. Phys.* **1999**, 71, S358–S366. [CrossRef]
- 12. Luijten, E.; Blöte, H.W.J. Monte Carlo method for spin models with long-range interactions. *Int. J. Mod. Phys. C* 1995, *6*, 359–370. [CrossRef]
- 13. Luijten, E.; Blöte, H.W.J. Classical critical behavior of spin models with long-range interactions. *Phys. Rev. B* 1997, *56*, 8945–8958. [CrossRef]
- 14. Jones, J.L.; Young, A.P. Finite-size scaling of the correlation length above the upper critical dimension in the five-dimensional Ising model. *Phys. Rev. B* 2005, *71*, 174438. [CrossRef]
- 15. Berche, B.; Kenna, R.; Walter, J.C. Hyperscaling above the upper critical dimension. Nucl. Phys. B 2012, 865, 115–132. [CrossRef]
- 16. Kenna, R.; Berche, B. Fisher's scaling relation above the upper critical dimension. *EPL (Europhys. Lett.)* **2014**, *105*, 26005. [CrossRef]
- 17. Kenna, R.; Berche, B. A new critical exponent koppa and its logarithmic counterpart koppa-hat. *arXiv* **2014**, arXiv:1411.2754. [CrossRef]
- 18. Flores-Sola, E.J.; Berche, B.; Kenna, R.; Weigel, M. Finite-size scaling above the upper critical dimension in Ising models with long-range interactions. *Eur. Phys. J. B* 2015, *88*, 28. [CrossRef]
- 19. Flores-Sola, E.; Berche, B.; Kenna, R.; Weigel, M. Role of Fourier Modes in Finite-Size Scaling above the Upper Critical Dimension. *Phys. Rev. Lett.* **2016**, *116*, 115701. [CrossRef] [PubMed]
- 20. Dutta, A.; Bhattacharjee, J.K. Phase transitions in the quantum Ising and rotor models with a long-range interaction. *Phys. Rev. B* **2001**, *64*, 184106. [CrossRef]
- 21. Defenu, N.; Trombettoni, A.; Ruffo, S. Criticality and phase diagram of quantum long-range O(*N*) models. *Phys. Rev. B* 2017, 96, 104432. [CrossRef]
- 22. Laflorencie, N.; Affleck, I.; Berciu, M. Critical phenomena and quantum phase transition in long range Heisenberg antiferromagnetic chains. *J. Stat. Mech. Theory Exp.* 2005, 2005, P12001. [CrossRef]
- 23. Koffel, T.; Lewenstein, M.; Tagliacozzo, L. Entanglement Entropy for the Long-Range Ising Chain in a Transverse Field. *Phys. Rev. Lett.* **2012**, *109*, 267203. [CrossRef] [PubMed]
- 24. Vodola, D.; Lepori, L.; Ercolessi, E.; Pupillo, G. Long-range Ising and Kitaev models: Phases, correlations and edge modes. *New J. Phys.* **2015**, *18*, 015001. [CrossRef]
- 25. Fey, S.; Schmidt, K.P. Critical behavior of quantum magnets with long-range interactions in the thermodynamic limit. *Phys. Rev.* B **2016**, *94*, 075156. [CrossRef]
- 26. Sun, G. Fidelity susceptibility study of quantum long-range antiferromagnetic Ising chain. *Phys. Rev. A* 2017, *96*, 043621. [CrossRef]
- 27. Saadatmand, S.N.; Bartlett, S.D.; McCulloch, I.P. Phase diagram of the quantum Ising model with long-range interactions on an infinite-cylinder triangular lattice. *Phys. Rev. B* 2018, *97*, 155116. [CrossRef]
- 28. Zhu, Z.; Sun, G.; You, W.L.; Shi, D.N. Fidelity and criticality of a quantum Ising chain with long-range interactions. *Phys. Rev. A* **2018**, *98*, 023607. [CrossRef]
- 29. Fey, S.; Kapfer, S.C.; Schmidt, K.P. Quantum Criticality of Two-Dimensional Quantum Magnets with Long-Range Interactions. *Phys. Rev. Lett.* **2019**, 122, 017203. [CrossRef] [PubMed]
- 30. Koziol, J.; Fey, S.; Kapfer, S.C.; Schmidt, K.P. Quantum criticality of the transverse-field Ising model with long-range interactions on triangular-lattice cylinders. *Phys. Rev. B* 2019, 100, 144411. [CrossRef]
- 31. Adelhardt, P.; Koziol, J.A.; Schellenberger, A.; Schmidt, K.P. Quantum criticality and excitations of a long-range anisotropic XY chain in a transverse field. *Phys. Rev. B* 2020, 102, 174424. [CrossRef]

- 32. Koziol, J.A.; Langheld, A.; Kapfer, S.C.; Schmidt, K.P. Quantum-critical properties of the long-range transverse-field Ising model from quantum Monte Carlo simulations. *Phys. Rev. B* **2021**, *103*, 245135. [CrossRef]
- 33. Gonzalez-Lazo, E.; Heyl, M.; Dalmonte, M.; Angelone, A. Finite-temperature critical behavior of long-range quantum Ising models. *SciPost Phys.* 2021, 11, 76. [CrossRef]
- 34. Langheld, A.; Koziol, J.A.; Adelhardt, P.; Kapfer, S.C.; Schmidt, K.P. Scaling at quantum phase transitions above the upper critical dimension. *SciPost Phys.* **2022**, *13*, 88. [CrossRef]
- Adelhardt, P.; Schmidt, K.P. Continuously varying critical exponents in long-range quantum spin ladders. *SciPost Phys.* 2023, 15, 87. [CrossRef]
- 36. Zhao, J.; Song, M.; Qi, Y.; Rong, J.; Meng, Z.Y. Finite-temperature critical behaviors in 2D long-range quantum Heisenberg model. *npj Quantum Mater.* **2023**, *8*, 59. [CrossRef]
- 37. Song, M.; Zhao, J.; Qi, Y.; Rong, J.; Meng, Z.Y. Quantum criticality and entanglement for the two-dimensional long-range Heisenberg bilayer. *Phys. Rev. B* 2024, *109*, L081114. [CrossRef]
- Song, M.; Zhao, J.; Zhou, C.; Meng, Z.Y. Dynamical properties of quantum many-body systems with long-range interactions. *Phys. Rev. Res.* 2023, *5*, 033046. [CrossRef]
- 39. Sandvik, A.W. Stochastic series expansion method for quantum Ising models with arbitrary interactions. *Phys. Rev. E* 2003, 68, 056701. [CrossRef] [PubMed]
- 40. Humeniuk, S. Quantum Monte Carlo study of long-range transverse-field Ising models on the triangular lattice. *Phys. Rev. B* **2016**, 93, 104412. [CrossRef]
- Berche, B.; Ellis, T.; Holovatch, Y.; Kenna, R. Phase transitions above the upper critical dimension. *SciPost Phys. Lect. Notes* 2022, 60. [CrossRef]
- Dauxois, T.; Ruffo, S.; Arimondo, E.; Wilkens, M. Dynamics and Thermodynamics of Systems with Long-Range Interactions: An Introduction. In *Dynamics and Thermodynamics of Systems with Long-Range Interactions*; Dauxois, T., Ruffo, S., Arimondo, E., Wilkens, M., Eds.; Springer: Berlin/Heidelberg, Germany, 2002; pp. 1–19. [CrossRef]
- Campa, A.; Dauxois, T.; Ruffo, S. Statistical mechanics and dynamics of solvable models with long-range interactions. *Phys. Rep.* 2009, 480, 57–159. [CrossRef]
- 44. French, R.H.; Parsegian, V.A.; Podgornik, R.; Rajter, R.F.; Jagota, A.; Luo, J.; Asthagiri, D.; Chaudhury, M.K.; Chiang, Y.m.; Granick, S.; et al. Long range interactions in nanoscale science. *Rev. Mod. Phys.* **2010**, *82*, 1887–1944. [CrossRef]
- 45. Bouchet, F.; Gupta, S.; Mukamel, D. Thermodynamics and dynamics of systems with long-range interactions. *Phys. A Stat. Mech. Its Appl.* **2010**, *389*, 4389–4405. [CrossRef]
- 46. Barredo, D.; Labuhn, H.; Ravets, S.; Lahaye, T.; Browaeys, A.; Adams, C.S. Coherent Excitation Transfer in a Spin Chain of Three Rydberg Atoms. *Phys. Rev. Lett.* **2015**, *114*, 113002. [CrossRef] [PubMed]
- 47. Barredo, D.; de Léséleuc, S.; Lienhard, V.; Lahaye, T.; Browaeys, A. An atom-by-atom assembler of defect-free arbitrary two-dimensional atomic arrays. *Science* **2016**, *354*, 1021–1023. [CrossRef]
- Labuhn, H.; Barredo, D.; Ravets, S.; de Léséleuc, S.; Macrì, T.; Lahaye, T.; Browaeys, A. Tunable two-dimensional arrays of single Rydberg atoms for realizing quantum Ising models. *Nature* 2016, 534, 667–670. [CrossRef] [PubMed]
- Lienhard, V.; de Léséleuc, S.; Barredo, D.; Lahaye, T.; Browaeys, A.; Schuler, M.; Henry, L.P.; Läuchli, A.M. Observing the Spaceand Time-Dependent Growth of Correlations in Dynamically Tuned Synthetic Ising Models with Antiferromagnetic Interactions. *Phys. Rev. X* 2018, *8*, 021070. [CrossRef]
- 50. Schauss, P. Quantum simulation of transverse Ising models with Rydberg atoms. Quantum Sci. Technol. 2018, 3, 023001. [CrossRef]
- de Léséleuc, S.; Lienhard, V.; Scholl, P.; Barredo, D.; Weber, S.; Lang, N.; Büchler, H.P.; Lahaye, T.; Browaeys, A. Observation of a symmetry-protected topological phase of interacting bosons with Rydberg atoms. *Science* 2019, 365, 775–780. [CrossRef] [PubMed]
- 52. Browaeys, A.; Lahaye, T. Many-body physics with individually controlled Rydberg atoms. *Nat. Phys.* **2020**, *16*, 132–142. [CrossRef]
- 53. Scholl, P.; Schuler, M.; Williams, H.J.; Eberharter, A.A.; Barredo, D.; Schymik, K.N.; Lienhard, V.; Henry, L.P.; Lang, T.C.; Lahaye, T.; et al. Quantum simulation of 2D antiferromagnets with hundreds of Rydberg atoms. *Nature* 2021, 595, 233–238. [CrossRef] [PubMed]
- 54. Samajdar, R.; Ho, W.W.; Pichler, H.; Lukin, M.D.; Sachdev, S. Quantum phases of Rydberg atoms on a kagome lattice. *Proc. Natl. Acad. Sci. USA* **2021**, *118*, e2015785118. [CrossRef] [PubMed]
- 55. Semeghini, G.; Levine, H.; Keesling, A.; Ebadi, S.; Wang, T.T.; Bluvstein, D.; Verresen, R.; Pichler, H.; Kalinowski, M.; Samajdar, R.; et al. Probing topological spin liquids on a programmable quantum simulator. *Science* **2021**, 374, 1242–1247. [CrossRef] [PubMed]
- 56. Jaksch, D.; Bruder, C.; Cirac, J.I.; Gardiner, C.W.; Zoller, P. Cold Bosonic Atoms in Optical Lattices. *Phys. Rev. Lett.* **1998**, *81*, 3108–3111. [CrossRef]
- 57. Greiner, M.; Mandel, O.; Esslinger, T.; Hänsch, T.W.; Bloch, I. Quantum phase transition from a superfluid to a Mott insulator in a gas of ultracold atoms. *Nature* **2002**, *415*, 39–44. [CrossRef] [PubMed]
- Góral, K.; Santos, L.; Lewenstein, M. Quantum Phases of Dipolar Bosons in Optical Lattices. *Phys. Rev. Lett.* 2002, *88*, 170406. [CrossRef] [PubMed]
- 59. Bloch, I. Ultracold quantum gases in optical lattices. Nat. Phys. 2005, 1, 23–30. [CrossRef]

- 60. Bloch, I.; Greiner, M. Exploring Quantum Matter with Ultracold Atoms in Optical Lattices. In *Advances in Atomic, Molecular, and Optical Physics*; Elsevier: Amsterdam, The Netherlands, 2005; pp. 1–47. [CrossRef]
- 61. Kovrizhin, D.L.; Pai, G.V.; Sinha, S. Density wave and supersolid phases of correlated bosons in an optical lattice. *Europhys. Lett.* (*EPL*) **2005**, *72*, 162–168. [CrossRef]
- 62. Menotti, C.; Trefzger, C.; Lewenstein, M. Metastable States of a Gas of Dipolar Bosons in a 2D Optical Lattice. *Phys. Rev. Lett.* **2007**, *98*, 235301. [CrossRef]
- 63. Bloch, I.; Dalibard, J.; Zwerger, W. Many-body physics with ultracold gases. Rev. Mod. Phys. 2008, 80, 885–964. [CrossRef]
- 64. Capogrosso-Sansone, B.; Trefzger, C.; Lewenstein, M.; Zoller, P.; Pupillo, G. Quantum Phases of Cold Polar Molecules in 2D Optical Lattices. *Phys. Rev. Lett.* **2010**, *104*, 125301. [CrossRef]
- 65. Bloch, I.; Dalibard, J.; Nascimbène, S. Quantum simulations with ultracold quantum gases. Nat. Phys. 2012, 8, 267–276. [CrossRef]
- 66. Yamamoto, D.; Masaki, A.; Danshita, I. Quantum phases of hard-core bosons with long-range interactions on a square lattice. *Phys. Rev. B* 2012, *86*, 054516. [CrossRef]
- Moses, S.A.; Covey, J.P.; Miecnikowski, M.T.; Yan, B.; Gadway, B.; Ye, J.; Jin, D.S. Creation of a low-entropy quantum gas of polar molecules in an optical lattice. *Science* 2015, 350, 659–662. [CrossRef] [PubMed]
- 68. Baier, S.; Mark, M.J.; Petter, D.; Aikawa, K.; Chomaz, L.; Cai, Z.; Baranov, M.; Zoller, P.; Ferlaino, F. Extended Bose–Hubbard models with ultracold magnetic atoms. *Science* 2016, 352, 201–205. [CrossRef] [PubMed]
- 69. Moses, S.A.; Covey, J.P.; Miecnikowski, M.T.; Jin, D.S.; Ye, J. New frontiers for quantum gases of polar molecules. *Nat. Phys.* 2017, 13, 13–20. [CrossRef]
- 70. Gross, C.; Bloch, I. Quantum simulations with ultracold atoms in optical lattices. Science 2017, 357, 995–1001. [CrossRef] [PubMed]
- 71. Schäfer, F.; Fukuhara, T.; Sugawa, S.; Takasu, Y.; Takahashi, Y. Tools for quantum simulation with ultracold atoms in optical lattices. *Nat. Rev. Phys.* 2020, 2, 411–425. [CrossRef]
- Chomaz, L.; Ferrier-Barbut, I.; Ferlaino, F.; Laburthe-Tolra, B.; Lev, B.L.; Pfau, T. Dipolar physics: A review of experiments with magnetic quantum gases. *Rep. Prog. Phys.* 2022, *86*, 026401. [CrossRef] [PubMed]
- 73. Su, L.; Douglas, A.; Szurek, M.; Groth, R.; Ozturk, S.F.; Krahn, A.; Hébert, A.H.; Phelps, G.A.; Ebadi, S.; Dickerson, S.; et al. Dipolar quantum solids emerging in a Hubbard quantum simulator. *Nature* **2023**, *622*, 724–729. [CrossRef] [PubMed]
- 74. Friedenauer, A.; Schmitz, H.; Glueckert, J.T.; Porras, D.; Schaetz, T. Simulating a quantum magnet with trapped ions. *Nat. Phys.* **2008**, *4*, 757–761. [CrossRef]
- 75. Kim, K.; Chang, M.S.; Islam, R.; Korenblit, S.; Duan, L.M.; Monroe, C. Entanglement and Tunable Spin-Spin Couplings between Trapped Ions Using Multiple Transverse Modes. *Phys. Rev. Lett.* **2009**, *103*, 120502. [CrossRef] [PubMed]
- 76. Kim, K.; Chang, M.S.; Korenblit, S.; Islam, R.; Edwards, E.E.; Freericks, J.K.; Lin, G.D.; Duan, L.M.; Monroe, C. Quantum simulation of frustrated Ising spins with trapped ions. *Nature* **2010**, *465*, 590–593. [CrossRef]
- 77. Islam, R.; Edwards, E.E.; Kim, K.; Korenblit, S.; Noh, C.; Carmichael, H.; Lin, G.D.; Duan, L.M.; Joseph Wang, C.C.; Freericks, J.K.; et al. Onset of a quantum phase transition with a trapped ion quantum simulator. *Nat. Commun.* 2011, 2, 377. [CrossRef] [PubMed]
- 78. Blatt, R.; Roos, C.F. Quantum simulations with trapped ions. Nat. Phys. 2012, 8, 277–284. [CrossRef]
- 79. Schneider, C.; Porras, D.; Schaetz, T. Experimental quantum simulations of many-body physics with trapped ions. *Rep. Prog. Phys.* **2012**, *75*, 024401. [CrossRef] [PubMed]
- Britton, J.W.; Sawyer, B.C.; Keith, A.C.; Wang, C.C.J.; Freericks, J.K.; Uys, H.; Biercuk, M.J.; Bollinger, J.J. Engineered twodimensional Ising interactions in a trapped-ion quantum simulator with hundreds of spins. *Nature* 2012, 484, 489–492. [CrossRef]
- Islam, R.; Senko, C.; Campbell, W.C.; Korenblit, S.; Smith, J.; Lee, A.; Edwards, E.E.; Wang, C.C.J.; Freericks, J.K.; Monroe, C. Emergence and Frustration of Magnetism with Variable-Range Interactions in a Quantum Simulator. *Science* 2013, 340, 583–587. [CrossRef] [PubMed]
- 82. Jurcevic, P.; Lanyon, B.P.; Hauke, P.; Hempel, C.; Zoller, P.; Blatt, R.; Roos, C.F. Quasiparticle engineering and entanglement propagation in a quantum many-body system. *Nature* **2014**, *511*, 202–205. [CrossRef] [PubMed]
- 83. Richerme, P.; Gong, Z.X.; Lee, A.; Senko, C.; Smith, J.; Foss-Feig, M.; Michalakis, S.; Gorshkov, A.V.; Monroe, C. Non-local propagation of correlations in quantum systems with long-range interactions. *Nature* **2014**, *511*, 198–201. [CrossRef]
- 84. Bohnet, J.G.; Sawyer, B.C.; Britton, J.W.; Wall, M.L.; Rey, A.M.; Foss-Feig, M.; Bollinger, J.J. Quantum spin dynamics and entanglement generation with hundreds of trapped ions. *Science* **2016**, *352*, 1297–1301. [CrossRef] [PubMed]
- 85. Monroe, C.; Campbell, W.; Duan, L.M.; Gong, Z.X.; Gorshkov, A.; Hess, P.; Islam, R.; Kim, K.; Linke, N.; Pagano, G.; et al. Programmable quantum simulations of spin systems with trapped ions. *Rev. Mod. Phys.* **2021**, *93*. [CrossRef]
- Douglas, J.S.; Habibian, H.; Hung, C.L.; Gorshkov, A.V.; Kimble, H.J.; Chang, D.E. Quantum many-body models with cold atoms coupled to photonic crystals. *Nat. Photonics* 2015, *9*, 326–331. [CrossRef]
- Vaidya, V.D.; Guo, Y.; Kroeze, R.M.; Ballantine, K.E.; Kollár, A.J.; Keeling, J.; Lev, B.L. Tunable-Range, Photon-Mediated Atomic Interactions in Multimode Cavity QED. *Phys. Rev. X* 2018, *8*, 011002. [CrossRef]
- Dubin, D.H.E.; O'Neil, T.M. Trapped nonneutral plasmas, liquids, and crystals (the thermal equilibrium states). *Rev. Mod. Phys.* 1999, 71, 87–172. [CrossRef]
- 89. Matsubara, T.; Matsuda, H. A Lattice Model of Liquid Helium, I. Prog. Theor. Phys. 1956, 16, 569-582. [CrossRef]
- Harris, M.J.; Bramwell, S.T.; McMorrow, D.F.; Zeiske, T.; Godfrey, K.W. Geometrical Frustration in the Ferromagnetic Pyrochlore Ho₂Ti₂O₇. *Phys. Rev. Lett.* **1997**, *79*, 2554–2557. [CrossRef]

- 91. Bramwell, S.T.; Harris, M.J. Frustration in Ising-type spin models on the pyrochlore lattice. *J. Phys. Condens. Matter* **1998**, 10, L215–L220. [CrossRef]
- 92. Ramirez, A.P.; Hayashi, A.; Cava, R.J.; Siddharthan, R.; Shastry, B.S. Zero-point entropy in 'spin ice'. *Nature* **1999**, 399, 333–335. [CrossRef]
- 93. Siddharthan, R.; Shastry, B.S.; Ramirez, A.P.; Hayashi, A.; Cava, R.J.; Rosenkranz, S. Ising Pyrochlore Magnets: Low-Temperature Properties, "Ice Rules," and Beyond. *Phys. Rev. Lett.* **1999**, *83*, 1854–1857. [CrossRef]
- 94. den Hertog, B.C.; Gingras, M.J.P. Dipolar Interactions and Origin of Spin Ice in Ising Pyrochlore Magnets. *Phys. Rev. Lett.* 2000, 84, 3430–3433. [CrossRef] [PubMed]
- 95. Melko, R.G.; den Hertog, B.C.; Gingras, M.J.P. Long-Range Order at Low Temperatures in Dipolar Spin Ice. *Phys. Rev. Lett.* 2001, 87, 067203. [CrossRef] [PubMed]
- 96. Melko, R.G.; Gingras, M.J.P. Monte Carlo studies of the dipolar spin ice model. J. Phys. Condens. Matter 2004, 16, R1277–R1319. [CrossRef]
- Bramwell, S.T.; Gingras, M.J.P. Spin Ice State in Frustrated Magnetic Pyrochlore Materials. Science 2001, 294, 1495–1501. [CrossRef] [PubMed]
- Ruff, J.P.C.; Melko, R.G.; Gingras, M.J.P. Finite-Temperature Transitions in Dipolar Spin Ice in a Large Magnetic Field. *Phys. Rev.* Lett. 2005, 95, 097202. [CrossRef] [PubMed]
- 99. Fennell, T.; Deen, P.P.; Wildes, A.R.; Schmalzl, K.; Prabhakaran, D.; Boothroyd, A.T.; Aldus, R.J.; McMorrow, D.F.; Bramwell, S.T. Magnetic Coulomb Phase in the Spin Ice Ho2Ti2O7. *Science* 2009, *326*, 415–417. [CrossRef] [PubMed]
- 100. Castelnovo, C.; Moessner, R.; Sondhi, S.L. Magnetic monopoles in spin ice. Nature 2008, 451, 42–45. [CrossRef] [PubMed]
- 101. Jaubert, L.D.C.; Holdsworth, P.C.W. Signature of magnetic monopole and Dirac string dynamics in spin ice. *Nat. Phys.* **2009**, 5, 258–261. [CrossRef]
- 102. Bitko, D.; Rosenbaum, T.F.; Aeppli, G. Quantum Critical Behavior for a Model Magnet. *Phys. Rev. Lett.* **1996**, 77, 940–943. [CrossRef] [PubMed]
- 103. Chakraborty, P.B.; Henelius, P.; Kjønsberg, H.; Sandvik, A.W.; Girvin, S.M. Theory of the magnetic phase diagram of LiHoF₄. *Phys. Rev. B* **2004**, *70*, 144411. [CrossRef]
- Ronnow, H.M.; Parthasarathy, R.; Jensen, J.; Aeppli, G.; Rosenbaum, T.F.; McMorrow, D.F. Quantum Phase Transition of a Magnet in a Spin Bath. *Science* 2005, 308, 389–392. [CrossRef] [PubMed]
- Gingras, M.J.P.; Henelius, P. Collective Phenomena in the LiHoxY1-xF4Quantum Ising Magnet: Recent Progress and Open Questions. J. Phys. Conf. Ser. 2011, 320, 012001. [CrossRef]
- 106. Tiwari, A.; Ahn, H.; Kumar, B.; Saini, J.; Srivastava, P.K.; Singh, B.; Lee, C.; Ghosh, S. Phase transition in the two-dimensional Heisenberg ferromagnet Fe₃GeTe₂ with long-range interaction. *Phys. Rev. B* **2024**, *109*, L020407. [CrossRef]
- Burch, K.S.; Mandrus, D.; Park, J.G. Magnetism in two-dimensional van der Waals materials. *Nature* 2018, 563, 47–52. [CrossRef]
 [PubMed]
- 108. Wang, Q.H.; Bedoya-Pinto, A.; Blei, M.; Dismukes, A.H.; Hamo, A.; Jenkins, S.; Koperski, M.; Liu, Y.; Sun, Q.C.; Telford, E.J.; et al. The Magnetic Genome of Two-Dimensional van der Waals Materials. ACS Nano 2022, 16, 6960–7079. [CrossRef] [PubMed]
- 109. Verresen, R.; Lukin, M.D.; Vishwanath, A. Prediction of Toric Code Topological Order from Rydberg Blockade. *Phys. Rev. X* 2021, 11, 031005. [CrossRef]
- 110. Koziol, J.A.; Duft, A.; Morigi, G.; Schmidt, K.P. Systematic analysis of crystalline phases in bosonic lattice models with algebraically decaying density–density interactions. *SciPost Phys.* 2023, 14, 136. [CrossRef]
- 111. Koziol, J.A.; Morigi, G.; Schmidt, K.P. Quantum phases of hard-core bosons with repulsive dipolar density–density interactions on two-dimensional lattices. *arXiv* 2023, arXiv:2311.10632. [CrossRef]
- 112. Duft, A.; Koziol, J.A.; Adelhardt, P.; Mühlhauser, M.; Schmidt, K.P. Order-by-disorder in the antiferromagnetic long-range transverse-field Ising model on the ruby lattice. *arXiv* **2023**, arXiv:2312.12941. [CrossRef]
- 113. Sachdev, S. Quantum Phase Transitions, 2nd ed.; Cambridge University Press: Cambridge, UK, 2011. [CrossRef]
- 114. Jaeger, G. The Ehrenfest Classification of Phase Transitions: Introduction and Evolution. *Arch. Hist. Exact Sci.* **1998**, 53, 51–81. [CrossRef]
- Suzuki, M. Relationship between d-Dimensional Quantal Spin Systems and (d+1)-Dimensional Ising Systems: Equivalence, Critical Exponents and Systematic Approximants of the Partition Function and Spin Correlations. *Prog. Theor. Phys.* 1976, 56, 1454–1469. [CrossRef]
- 116. Sondhi, S.L.; Girvin, S.M.; Carini, J.P.; Shahar, D. Continuous quantum phase transitions. *Rev. Mod. Phys.* **1997**, *69*, 315–333. [CrossRef]
- 117. Vojta, M. Quantum phase transitions. Rep. Prog. Phys. 2003, 66, 2069–2110. [CrossRef]
- 118. Widom, B. Equation of State in the Neighborhood of the Critical Point. J. Chem. Phys. 1965, 43, 3898–3905. [CrossRef]
- 119. Domb, C.; Hunter, D.L. On the critical behaviour of ferromagnets. Proc. Phys. Soc. 1965, 86, 1147. [CrossRef]
- 120. Kadanoff, L.P. Scaling laws for ising models near *T_c*. *Phys. Phys. Fiz.* **1966**, *2*, 263–272. [CrossRef]
- 121. Patashinskii, A.; Pokrovskii, V. Behavior of ordered systems near the transition point. Sov. Phys. JETP 1966, 23, 292–297.
- 122. Fisher, M.E. The theory of equilibrium critical phenomena. Rep. Prog. Phys. 1967, 30, 615–730. [CrossRef]
- 123. Kadanoff, L.P.; Götze, W.; Hamblen, D.; Hecht, R.; Lewis, E.A.S.; Palciauskas, V.V.; Rayl, M.; Swift, J.; Aspnes, D.; Kane, J. Static Phenomena Near Critical Points: Theory and Experiment. *Rev. Mod. Phys.* **1967**, *39*, 395–431. [CrossRef]

- 124. Hankey, A.; Stanley, H.E. Systematic Application of Generalized Homogeneous Functions to Static Scaling, Dynamic Scaling, and Universality. *Phys. Rev. B* 1972, *6*, 3515–3542. [CrossRef]
- 125. Kenna, R. Finite size scaling for O(N) Phi4-theory at the upper critical dimension. Nucl. Phys. B 2004, 691, 292–304. [CrossRef]
- 126. Brézin, E. An investigation of finite size scaling. J. Phys. Fr. 1982, 43, 15–22. [CrossRef]
- 127. Fisher, M.E. Scaling, universality and renormalization group theory. In *Critical Phenomena*; Hahne, F.J.W., Ed.; Springer: Berlin/Heidelberg, Germany, 1983; pp. 1–139.
- 128. Binder, K.; Nauenberg, M.; Privman, V.; Young, A.P. Finite-size tests of hyperscaling. *Phys. Rev. B* **1985**, *31*, 1498–1502. [CrossRef] [PubMed]
- 129. Kirkpatrick, T.R.; Belitz, D. Exponent relations at quantum phase transitions with applications to metallic quantum ferromagnets. *Phys. Rev. B* 2015, *91*, 214407. [CrossRef]
- Mermin, N.D.; Wagner, H. Absence of Ferromagnetism or Antiferromagnetism in One- or Two-Dimensional Isotropic Heisenberg Models. *Phys. Rev. Lett.* 1966, 17, 1133. Erratum in *Phys. Rev. Lett.* 1966, 17, 1307. [CrossRef]
- 131. Hohenberg, P.C. Existence of Long-Range Order in One and Two Dimensions. Phys. Rev. 1967, 158, 383–386. [CrossRef]
- 132. Kleinert, H.; Schulte-Frohlinde, V. Critical Properties of Phi4-Theories; World Scientific: Singapore, 2001. [CrossRef]
- Joshi, D.G.; Coester, K.; Schmidt, K.P.; Vojta, M. Nonlinear bond-operator theory and 1/d expansion for coupled-dimer magnets. I. Paramagnetic phase. *Phys. Rev. B* 2015, *91*, 094404. [CrossRef]
- 134. Joshi, D.G.; Vojta, M. Nonlinear bond-operator theory and 1/*d* expansion for coupled-dimer magnets. II. Antiferromagnetic phase and quantum phase transition. *Phys. Rev. B* 2015, *91*, 094405. [CrossRef]
- 135. Coester, K.; Joshi, D.G.; Vojta, M.; Schmidt, K.P. Linked-cluster expansions for quantum magnets on the hypercubic lattice. *Phys. Rev. B* 2016, 94, 125109. [CrossRef]
- 136. Widom, B. Surface Tension and Molecular Correlations near the Critical Point. J. Chem. Phys. 1965, 43, 3892–3897. [CrossRef]
- 137. Essam, J.W.; Fisher, M.E. Padé Approximant Studies of the Lattice Gas and Ising Ferromagnet below the Critical Point. *J. Chem. Phys.* **1963**, *38*, 802–812. [CrossRef]
- 138. Fisher, M.E. Correlation Functions and the Critical Region of Simple Fluids. J. Math. Phys. 1964, 5, 944–962. [CrossRef]
- 139. Cardy, J. Finite-Size Scaling; Elsevier: Amsterdam, The Netherlands, 1988.
- 140. Fisher, M.E. The renormalization group in the theory of critical behavior. Rev. Mod. Phys. 1974, 46, 597–616. [CrossRef]
- 141. Pelissetto, A.; Vicari, E. Critical phenomena and renormalization-group theory. Phys. Rep. 2002, 368, 549–727. [CrossRef]
- 142. Binder, K. Critical properties and finite-size effects of the five-dimensional Ising model. Z. Phys. B-Condens. Matter 1985, 61, 13–23. [CrossRef]
- 143. Binder, K. Finite size effects on phase transitions. *Ferroelectrics* 1987, 73, 43–67. [CrossRef]
- 144. Binder, K.; Heermann, D. *Monte Carlo Simulation in Statistical Physics*; Graduate Texts in Physics; Springer: Cham, Switzerland, 2019; Chapter 2.3.4, pp. 47–56. [CrossRef]
- 145. Cardy, J. *Scaling and Renormalization in Statistical Physics*; Cambridge Lecture Notes in Physics; Cambridge University Press: Cambridge, UK, 1996; Chapter 5.4, pp. 92–94. [CrossRef]
- 146. Chib, S. Chapter 57—Markov Chain Monte Carlo Methods: Computation and Inference. In *Handbook of Econometrics*; Elsevier: Amsterdam, The Netherlands, 2001; Volume 5, pp. 3569–3649. [CrossRef]
- 147. Levin, D.A.; Peres, Y.; Wilmer, E.L. Markov Chains and Mixing Times; American Mathematical Society: Providence, RI, USA, 2006.
- 148. Krauth, W. Statistical Mechanics: Algorithms and Computations; Oxford University Press: Oxford, UK, 2006; Volume 13.
- 149. Metropolis, N.; Rosenbluth, A.W.; Rosenbluth, M.N.; Teller, A.H.; Teller, E. Equation of State Calculations by Fast Computing Machines. J. Chem. Phys. 1953, 21, 1087–1092. [CrossRef]
- 150. Hastings, W.K. Monte Carlo sampling methods using Markov chains and their applications. *Biometrika* **1970**, *57*, 97–109. [CrossRef]
- 151. Krauth, W. Event-Chain Monte Carlo: Foundations, Applications, and Prospects. Front. Phys. 2021, 9, 663457. [CrossRef]
- 152. Coester, K.; Schmidt, K.P. Optimizing linked-cluster expansions by white graphs. *Phys. Rev. E* 2015, *92*, 022118. [CrossRef] [PubMed]
- 153. Knetter, C.; Uhrig, G. Perturbation theory by flow equations: Dimerized and frustrated S = 1/2 chain. *Eur. Phys. J. B* 2000, 13, 209–225. [CrossRef]
- 154. Knetter, C.; Schmidt, K.P.; Uhrig, G.S. The structure of operators in effective particle-conserving models. *J. Phys. A Math. Gen.* **2003**, *36*, 7889. [CrossRef]
- 155. Oitmaa, J.; Hamer, C.; Zheng, W. Series Expansion Methods for Strongly Interacting Lattice Models; Cambridge University Press: Cambridge, UK, 2006.
- 156. Knetter, C.; Bühler, A.; Müller-Hartmann, E.; Uhrig, G.S. Dispersion and Symmetry of Bound States in the Shastry-Sutherland Model. *Phys. Rev. Lett.* **2000**, *85*, 3958–3961. [CrossRef] [PubMed]
- 157. Dorier, J.; Schmidt, K.P.; Mila, F. Theory of Magnetization Plateaux in the Shastry-Sutherland Model. *Phys. Rev. Lett.* 2008, 101, 250402. [CrossRef] [PubMed]
- 158. Vidal, J.; Dusuel, S.; Schmidt, K.P. Low-energy effective theory of the toric code model in a parallel magnetic field. *Phys. Rev. B* **2009**, *79*, 033109. [CrossRef]
- 159. Kadanoff, L.P.; Kohmoto, M. Quantum mechanical ground states, nonlinear Schrodinger equations and linked cluster expansions. *J. Phys. Math. Gen.* **1981**, *14*, 1291. [CrossRef]

- 160. Marland, L.G. Series expansions for the zero-temperature transverse Ising model. J. Phys. Math. Gen. 1981, 14, 2047. [CrossRef]
- 161. Singh, R.R.P.; Gelfand, M.P.; Huse, D.A. Ground States of Low-Dimensional Quantum Antiferromagnets. *Phys. Rev. Lett.* **1988**, 61, 2484–2487. [CrossRef] [PubMed]
- 162. Gelfand, M.P. Series expansions for excited states of quantum lattice models. Solid State Commun. 1996, 98, 11–14. [CrossRef]
- Singh, R.R.P.; Gelfand, M.P. Spin-wave excitation spectra and spectral weights in square lattice antiferromagnets. *Phys. Rev. B* 1995, 52, R15695–R15698. [CrossRef] [PubMed]
- 164. Trebst, S.; Monien, H.; Hamer, C.J.; Weihong, Z.; Singh, R.R.P. Strong-Coupling Expansions for Multiparticle Excitations: Continuum and Bound States. *Phys. Rev. Lett.* **2000**, *85*, 4373–4376. [CrossRef]
- 165. Zheng, W.; Hamer, C.J.; Singh, R.R.P.; Trebst, S.; Monien, H. Linked cluster series expansions for two-particle bound states. *Phys. Rev. B* **2001**, *63*, 144410. [CrossRef]
- Yang, H.Y.; Läuchli, A.M.; Mila, F.; Schmidt, K.P. Effective Spin Model for the Spin-Liquid Phase of the Hubbard Model on the Triangular Lattice. *Phys. Rev. Lett.* 2010, 105, 267204. [CrossRef] [PubMed]
- Dusuel, S.; Kamfor, M.; Schmidt, K.P.; Thomale, R.; Vidal, J. Bound states in two-dimensional spin systems near the Ising limit: A quantum finite-lattice study. *Phys. Rev. B* 2010, *81*, 064412. [CrossRef]
- Powalski, M.; Coester, K.; Moessner, R.; Schmidt, K.P. Disorder by disorder and flat bands in the kagome transverse field Ising model. *Phys. Rev. B* 2013, *87*, 054404. [CrossRef]
- Schulz, M.D.; Dusuel, S.; Schmidt, K.P.; Vidal, J. Topological Phase Transitions in the Golden String-Net Model. *Phys. Rev. Lett.* 2013, 110, 147203. [CrossRef] [PubMed]
- 170. Schulz, M.D.; Dusuel, S.; Misguich, G.; Schmidt, K.P.; Vidal, J. Ising anyons with a string tension. *Phys. Rev. B* 2014, *89*, 201103. [CrossRef]
- 171. Hörmann, M.; Schmidt, K.P. Projective cluster-additive transformation for quantum lattice models. *SciPost Phys.* **2023**, *15*, 097. [CrossRef]
- 172. Rutishauser, H. Ein infinitesimales Analogon zum Quotienten-Differenzen-Algorithmus. Arch. Math. 1954, 5, 132–137. [CrossRef]
- 173. Toda, M. Vibration of a Chain with Nonlinear Interaction. J. Phys. Soc. Jpn. **1967**, 22, 431–436. [CrossRef]
- 174. Flaschka, H. The Toda lattice. II. Existence of integrals. Phys. Rev. B 1974, 9, 1924–1925. [CrossRef]
- 175. Moser, J. Finitely many mass points on the line under the influence of an exponential potential—An integrable system. In Dynamical Systems, Theory and Applications: Battelle Seattle 1974 Rencontres; Moser, J., Ed.; Springer: Berlin/Heidelberg, Germany, 1975; pp. 467–497. [CrossRef]
- 176. Brockett, R. Dynamical systems that sort lists, diagonalize matrices and solve linear programming problems. In Proceedings of the 27th IEEE Conference on Decision and Control, Austin, TX, USA, 7–9 December 1988; Volume 1, pp. 799–803. [CrossRef]
- 177. Głazek, S.D.; Wilson, K.G. Renormalization of Hamiltonians. Phys. Rev. D 1993, 48, 5863–5872. [CrossRef] [PubMed]
- 178. Wegner, F. Flow-equations for Hamiltonians. Ann. Phys. 1994, 506, 77-91. [CrossRef]
- 179. Stein, J. Flow equations and the strong-coupling expansion for the Hubbard model. J. Stat. Phys. 1997, 88, 487–511. [CrossRef]
- 180. Mielke, A. Flow equations for band-matrices. Eur. Phys. J. B 1998, 5, 605–611. [CrossRef]
- 181. Schulz, M.D. Topological Phase Transitions Driven by Non-Abelian Anyons. Ph.D. Thesis, TU Dortmund, Dortmund, Germany, Université Pierre-et-Marie-Curie, Paris, France, 2013.
- 182. Coester, K. Quasiparticle Pictures and Graphs—From Perturbative to Non-Perturbative Linked-Cluster Expansions. Ph.D. Thesis, TU Dortmund, Dortmund, Germany, 2015.
- 183. Weihong, Z. Various series expansions for the bilayer $S=\frac{1}{2}$ Heisenberg antiferromagnet. *Phys. Rev. B* **1997**, *55*, 12267–12275. [CrossRef]
- 184. Collins, A.; Hamer, C.J. Two-particle bound states and one-particle structure factor in a Heisenberg bilayer system. *Phys. Rev. B* **2008**, *78*, 054419. [CrossRef]
- Klagges, D.; Schmidt, K.P. Constraints on Measurement-Based Quantum Computation in Effective Cluster States. *Phys. Rev. Lett.* 2012, 108, 230508. [CrossRef] [PubMed]
- 186. Boos, C.; Ganahl, C.J.; Lajkó, M.; Nataf, P.; Läuchli, A.M.; Penc, K.; Schmidt, K.P.; Mila, F. Time-reversal symmetry breaking Abelian chiral spin liquid in Mott phases of three-component fermions on the triangular lattice. *Phys. Rev. Res.* 2020, 2, 023098. [CrossRef]
- 187. Gelfand, M.P.; Singh, R.R.P. High-order convergent expansions for quantum many particle systems. *Adv. Phys.* 2000, 49, 93–140.
 [.]. [CrossRef]
- 188. Feiguin, A.E. The Density Matrix Renormalization Group and its time-dependent variants. *AIP Conf. Proc.* 2011, 1419, 5–92. [CrossRef]
- 189. Squires, G.L. Introduction to the Theory of Thermal Neutron Scattering, 3rd ed.; Cambridge University Press: Cambridge, UK, 2012. [CrossRef]
- Zaliznyak, I.A.; Tranquada, J.M. Neutron Scattering and Its Application to Strongly Correlated Systems: Experimental Techniques; Avella, A., Mancini, F., Eds.; Springer: Berlin/Heidelberg, Germany, 2015; pp. 205–235. [CrossRef]
- 191. Hamer, C.J.; Zheng, W.; Singh, R.R.P. Dynamical structure factor for the alternating Heisenberg chain: A linked cluster calculation. *Phys. Rev. B* 2003, *68*, 214408. [CrossRef]

- 192. Hamer, C.J. Critical behaviour of structure factors at a quantum phase transition. *J. Phys. Condens. Matter* 2009, 21, 124205. [CrossRef] [PubMed]
- 193. Rahman, M.S. Basic Graph Theory; Springer: Berlin/Heidelberg, Germany, 2017; Volume 9.
- 194. Harary, F. Graph Theory; Addison-Wesley Series in Mathematics; Addison-Wesley Publishing Company: Boston, MA, USA, 1969.
- 195. Mühlhauser, M. Graph Decomposition Techniques and Their Application to Topological Systems. Ph.D. Thesis, Friedrich-Alexander-Universität Erlangen-Nürnberg, Erlangen, Germany, 2024; To be published.
- 196. Grohe, M.; Schweitzer, P. The graph isomorphism problem. Commun. ACM 2020, 63, 128–134. [CrossRef]
- 197. Bonnici, V.; Giugno, R.; Pulvirenti, A.; Shasha, D.; Ferro, A. A subgraph isomorphism algorithm and its application to biochemical data. *BMC Bioinform.* **2013**, *14*, S13. [CrossRef] [PubMed]
- Bonnici, V.; Giugno, R. On the Variable Ordering in Subgraph Isomorphism Algorithms. *IEEE/ACM Trans. Comput. Biol. Bioinform.* 2017, 14, 193–203. [CrossRef] [PubMed]
- 199. Hartke, S.G.; Radcliffe, A. Mckay's canonical graph labeling algorithm. Commun. Math. 2009, 479, 99–111.
- 200. McKay, B.D.; Piperno, A. Practical graph isomorphism. Congr. Numer. 1981, 30, 45-87. [CrossRef]
- 201. McKay, B.D.; Piperno, A. Practical graph isomorphism, II. J. Symb. Comput. 2014, 60, 94–112. [CrossRef]
- Hagberg, A.A.; Schult, D.A.; Swart, P.J. Exploring Network Structure, Dynamics, and Function using NetworkX. In Proceedings of the 7th Python in Science Conference, Pasadena, CA, USA, 19–24 August 2008; Varoquaux, G., Vaught, T., Millman, J., Eds.; pp. 11–15.
- 203. Siek, J.; Lee, L.; Lumsdaine, A. *The Boost Graph Library: User Guide and Reference Manual*; The C++ In-Depth Series; Pearson Education: London, UK, 2001.
- 204. Jahromi, S.S.; Hörmann, M.; Adelhardt, P.; Fey, S.; Orus, R.; Schmidt, K.P. Kitaev honeycomb antiferromagnet in a field: Quantum phase diagram for general spin. *arXiv* 2021, arXiv:2111.06132. [CrossRef]
- 205. Kitaev, A. Anyons in an exactly solved model and beyond. Ann. Phys. 2006, 321, 2–111. [CrossRef]
- 206. Adelhardt, P.; Schmidt, K.P. In Preparation 2024. Friedrich-Alexander Universität Erlangen-Nürnberg, Erlangen, Bavaria, Germany.
- 207. Fey, S. Investigation of Zero-Temperature Transverse-Field Ising Models with Long-Range Interactions. Ph.D. Thesis, Friedrich-Alexander-Universität Erlangen-Nürnberg, Erlangen, Germany, 2020.
- 208. Baker, G. Essentials of Padé Approximants; Elsevier Science: Amsterdam, The Netherlands, 1975.
- Guttmann, A.J. Asymptotic Analysis of Power-Series Expansions. In *Phase Transitions and Critical Phenomena*; Domb, C., Green, M.S., Lebowitz, J.L., Eds.; Academic Press: Cambridge, MA, USA, 1989; Volume 13.
- Sandvik, A.W.; Kurkijärvi, J. Quantum Monte Carlo simulation method for spin systems. *Phys. Rev. B* 1991, 43, 5950–5961.
 [CrossRef]
- Sandvik, A.W.; Singh, R.R.P.; Campbell, D.K. Quantum Monte Carlo in the interaction representation: Application to a spin-Peierls model. *Phys. Rev. B* 1997, *56*, 14510–14528. [CrossRef]
- Humeniuk, S. Quantum Monte Carlo Studies of Strongly Correlated Systems for Quantum Simulators. Ph.D. Thesis, Fakultät Mathematik und Physik, Universität Stuttgart, Stuttgart, Germany, 2018.
- Sandvik, A.W.; Avella, A.; Mancini, F. Computational Studies of Quantum Spin Systems. AIP Conf. Proc. 2010, 1297, 135. [CrossRef]
- Fukui, K.; Todo, S. Order-N cluster Monte Carlo method for spin systems with long-range interactions. J. Comput. Phys. 2009, 228, 2629–2642. [CrossRef]
- 215. Syljuåsen, O.F.; Sandvik, A.W. Quantum Monte Carlo with directed loops. Phys. Rev. E 2002, 66, 046701. [CrossRef] [PubMed]
- 216. Sandvik, A.W. Stochastic series expansion method with operator-loop update. Phys. Rev. B 1999, 59, R14157–R14160. [CrossRef]
- 217. Alet, F.; Wessel, S.; Troyer, M. Generalized directed loop method for quantum Monte Carlo simulations. *Phys. Rev. E* 2005, 71, 036706. [CrossRef] [PubMed]
- 218. Sandvik, A.W. A generalization of Handscomb's quantum Monte Carlo scheme-application to the 1D Hubbard model. *J. Phys. A Math. Gen.* **1992**, *25*, 3667–3682. [CrossRef]
- 219. Kubo, R. Statistical-Mechanical Theory of Irreversible Processes. I. General Theory and Simple Applications to Magnetic and Conduction Problems. *J. Phys. Soc. Jpn.* **1957**, *12*, 570–586. [CrossRef]
- 220. Michel, F.; Evertz, H.G. Lattice dynamics of the Heisenberg chain coupled to finite frequency bond phonons. *arXiv* 2007, arXiv:0705.0799. [CrossRef]
- 221. Caracciolo, S.; Gambassi, A.; Gubinelli, M.; Pelissetto, A. Finite-size correlation length and violations of finite-size scaling. *Eur. Phys. J. B* **2001**, *20*, 594–598. [CrossRef]
- 222. Sadhukhan, D.; Dziarmaga, J. Is there a correlation length in a model with long-range interactions? *arXiv* 2021, arXiv:2107.02508. [CrossRef]
- 223. Brankov, J.; Tonchev, N. Finite-size scaling for systems with long-range interactions. Phys. A 1992, 189, 583–610. [CrossRef]
- 224. Flores-Sola, E.J. Finite-Size Scaling above the Upper Critical Dimension. Ph.D. Thesis, Université de Lorraine, Nancy, France; Coventry University, Coventry, UK, 2016.
- 225. Sandvik, A.W. Classical percolation transition in the diluted two-dimensional $S = \frac{1}{2}$ Heisenberg antiferromagnet. *Phys. Rev. B* **2002**, *66*, 024418. [CrossRef]

- 226. Prokof'ev, N.; Svistunov, B.; Tupitsyn, I. "Worm" algorithm in quantum Monte Carlo simulations. *Phys. Lett. A* **1998**, 238, 253–257. [CrossRef]
- 227. Pollet, L.; Houcke, K.; Rombouts, S. Engineering local optimality in quantum Monte Carlo algorithms. *J. Comput. Phys.* 2007, 225, 2249–2266. [CrossRef]
- 228. Söyler, S.G.; Capogrosso-Sansone, B.; Prokof'ev, N.V.; Svistunov, B.V. Sign-alternating interaction mediated by strongly correlated lattice bosons. *New J. Phys.* 2009, 11, 073036. [CrossRef]
- 229. Pollet, L.; Picon, J.; Büchler, H.; Troyer, M. Supersolid Phase with Cold Polar Molecules on a Triangular Lattice. *Phys. Rev. Lett.* **2010**, *104*, 125302. [CrossRef]
- 230. Maik, M.; Hauke, P.; Dutta, O.; Zakrzewski, J.; Lewenstein, M. Quantum spin models with long-range interactions and tunnelings: A quantum Monte Carlo study. *New J. Phys.* **2012**, *14*, 113006. [CrossRef]
- Fisher, M.P.A.; Weichman, P.B.; Grinstein, G.; Fisher, D.S. Boson localization and the superfluid-insulator transition. *Phys. Rev. B* 1989, 40, 546–570. [CrossRef] [PubMed]
- 232. Scalettar, R.T.; Batrouni, G.G.; Zimanyi, G.T. Localization in interacting, disordered, Bose systems. *Phys. Rev. Lett.* **1991**, 66, 3144–3147. [CrossRef]
- Krauth, W.; Trivedi, N.; Ceperley, D. Superfluid-insulator transition in disordered boson systems. *Phys. Rev. Lett.* 1991, 67, 2307–2310. [CrossRef] [PubMed]
- 234. Batrouni, G.G.; Scalettar, R.T. Phase Separation in Supersolids. Phys. Rev. Lett. 2000, 84, 1599–1602. [CrossRef] [PubMed]
- 235. Hébert, F.; Batrouni, G.G.; Scalettar, R.T.; Schmid, G.; Troyer, M.; Dorneich, A. Quantum phase transitions in the two-dimensional hard-core boson model. *Phys. Rev. B* 2001, *65*, 014513. [CrossRef]
- Biedroń, K.; Łącki, M.; Zakrzewski, J. Extended Bose–Hubbard model with dipolar and contact interactions. *Phys. Rev. B* 2018, 97, 245102. [CrossRef]
- Kraus, R.; Biedroń, K.; Zakrzewski, J.; Morigi, G. Superfluid phases induced by dipolar interactions. *Phys. Rev. B* 2020, 101, 174505.
 [CrossRef]
- Kraus, R.; Chanda, T.; Zakrzewski, J.; Morigi, G. Quantum phases of dipolar bosons in one-dimensional optical lattices. *Phys. Rev. B* 2022, 106, 035144. [CrossRef]
- Hirsch, J.E.; Sugar, R.L.; Scalapino, D.J.; Blankenbecler, R. Monte Carlo simulations of one-dimensional fermion systems. *Phys. Rev. B* 1982, 26, 5033–5055. [CrossRef]
- Beard, B.B.; Wiese, U.J. Simulations of Discrete Quantum Systems in Continuous Euclidean Time. *Phys. Rev. Lett.* 1996, 77, 5130–5133. [CrossRef] [PubMed]
- 241. Sandvik, A.W. The Stochastic Series Expansion Method for Quantum Lattice Models. In Proceedings of the Computer Simulation Studies in Condensed-Matter Physics XIV, Athens, GA, USA, 19–24 February 2001; Landau, D.P., Lewis, S.P., Schüttler, H.B., Eds.; Springer: Berlin/Heidelberg, Germany, 2002; pp. 182–187.
- Assaad, F.; Evertz, H. World-line and Determinantal Quantum Monte Carlo Methods for Spins, Phonons and Electrons. In *Computational Many-Particle Physics*; Fehske, H., Schneider, R., Weiße, A., Eds.; Springer: Berlin/Heidelberg, Germany, 2008; pp. 277–356. [CrossRef]
- 243. Reichsöllner, L.; Schindewolf, A.; Takekoshi, T.; Grimm, R.; Nägerl, H.C. Quantum Engineering of a Low-Entropy Gas of Heteronuclear Bosonic Molecules in an Optical Lattice. *Phys. Rev. Lett.* **2017**, *118*, 073201. [CrossRef] [PubMed]
- 244. Defenu, N.; Codello, A.; Ruffo, S.; Trombettoni, A. Criticality of spin systems with weak long-range interactions. *J. Phys. A Math. Theor.* **2020**, *53*, 143001. [CrossRef]
- 245. Sperstad, I.B.; Stiansen, E.B.; Sudbø, A. Quantum criticality in spin chains with non-Ohmic dissipation. *Phys. Rev. B* 2012, 85, 214302. [CrossRef]
- 246. Pfeuty, P.; Elliott, R.J. The Ising model with a transverse field. II. Ground state properties. *J. Phys. C Solid State Phys.* **1971**, 4, 2370–2385. [CrossRef]
- 247. Langheld, A.; Koziol, J.A.; Adelhardt, P.; Kapfer, S.C.; Schmidt, K.P. Raw data to "Scaling at quantum phase transitions above the upper critical dimension". Zenodo 2021. [CrossRef]
- 248. Puebla, R.; Marty, O.; Plenio, M.B. Quantum Kibble-Zurek physics in long-range transverse-field Ising models. *Phys. Rev. A* 2019, 100, 032115. [CrossRef]
- 249. Koziol, J.A.; Langheld, A.; Kapfer, S.C.; Schmidt, K.P. Raw data to "Quantum-critical properties of the long-range transverse-field Ising model from quantum Monte Carlo simulations". *Zenodo* **2021**. [CrossRef]
- 250. Shiratani, S.; Todo, S. Stochastic approximation analysis of dynamical quantum critical phenomena in long-range transverse-field Ising chain. *arXiv* 2023, arXiv:2305.14121. [CrossRef]
- 251. Elliott, R.J.; Pfeuty, P.; Wood, C. Ising Model with a Transverse Field. Phys. Rev. Lett. 1970, 25, 443–446. [CrossRef]
- 252. Pfeuty, P. The one-dimensional Ising model with a transverse field. Ann. Phys. 1970, 57, 79–90. [CrossRef]
- 253. Kos, F.; Poland, D.; Simmons-Duffin, D.; Vichi, A. Precision islands in the Ising and O(N) models. J. High Energy Phys. 2016, 2016. [CrossRef]
- 254. Ribeiro, P.; Vidal, J.; Mosseri, R. Exact spectrum of the Lipkin-Meshkov-Glick model in the thermodynamic limit and finite-size corrections. *Phys. Rev. E* 2008, *78*, 021106. [CrossRef] [PubMed]
- 255. Verstraete, F.; Wolf, M.M.; Perez-Garcia, D.; Cirac, J.I. Criticality, the Area Law, and the Computational Power of Projected Entangled Pair States. *Phys. Rev. Lett.* **2006**, *96*, 220601. [CrossRef] [PubMed]

- 256. Cirac, J.I.; Pérez-García, D.; Schuch, N.; Verstraete, F. Matrix product states and projected entangled pair states: Concepts, symmetries, theorems. *Rev. Mod. Phys.* **2021**, *93*, 045003. [CrossRef]
- 257. Stoudenmire, E.; White, S.R. Studying Two-Dimensional Systems with the Density Matrix Renormalization Group. *Annu. Rev. Condens. Matter Phys.* **2012**, *3*, 111–128. [CrossRef]
- 258. Eisert, J.; Cramer, M.; Plenio, M.B. Colloquium: Area laws for the entanglement entropy. *Rev. Mod. Phys.* 2010, 82, 277–306. [CrossRef]
- 259. Crosswhite, G.M.; Doherty, A.C.; Vidal, G. Applying matrix product operators to model systems with long-range interactions. *Phys. Rev. B* 2008, *78*, 035116. [CrossRef]
- 260. Pirvu, B.; Murg, V.; Cirac, J.I.; Verstraete, F. Matrix product operator representations. New J. Phys. 2010, 12, 025012. [CrossRef]
- 261. Biswas, S.; Rakala, G.; Damle, K. Quantum cluster algorithm for frustrated Ising models in a transverse field. *Phys. Rev. B* 2016, 93, 235103. [CrossRef]
- 262. Biswas, S.; Damle, K. Singular ferromagnetic susceptibility of the transverse-field Ising antiferromagnet on the triangular lattice. *Phys. Rev. B* 2018, 97, 085114. [CrossRef]
- Onsager, L. Crystal Statistics. I. A Two-Dimensional Model with an Order-Disorder Transition. *Phys. Rev.* 1944, 65, 117–149.
 [CrossRef]
- 264. Moessner, R.; Ramirez, A.P. Geometrical frustration. Phys. Today 2006, 59, 24–29. [CrossRef]
- 265. Moessner, R.; Chalker, J.T. Low-temperature properties of classical geometrically frustrated antiferromagnets. *Phys. Rev. B* 1998, 58, 12049–12062. [CrossRef]
- 266. Maxwell, J.C. L. on the calculation of the equilibrium and stiffness of frames. *Lond. Edinb. Dublin Philos. Mag. J. Sci.* **1864**, 27, 294–299. [CrossRef]
- Reimers, J.N.; Berlinsky, A.J.; Shi, A.C. Mean-field approach to magnetic ordering in highly frustrated pyrochlores. *Phys. Rev. B* 1991, 43, 865–878. [CrossRef] [PubMed]
- 268. Wannier, G.H. Antiferromagnetism. The Triangular Ising Net. *Phys. Rev.* **1950**, *79*, 357–364. Erratum in *Phys. Rev. B* **1973**, *7*, 5017. [CrossRef]
- 269. Kano, K.; Naya, S. Antiferromagnetism. The Kagome Ising Net. Prog. Theor. Phys. 1953, 10, 158–172. [CrossRef]
- Moessner, R.; Sondhi, S.L.; Chandra, P. Two-Dimensional Periodic Frustrated Ising Models in a Transverse Field. *Phys. Rev. Lett.* 2000, *84*, 4457–4460. [CrossRef] [PubMed]
- 271. Moessner, R.; Sondhi, S.L. Ising models of quantum frustration. *Phys. Rev. B* 2001, 63, 224401. [CrossRef]
- 272. Villain, J.; Bidaux, R.; Carton, J.P.; Conte, R. Order as an effect of disorder. J. Phys. 1980, 41, 1263–1272. [CrossRef]
- 273. Isakov, S.V.; Moessner, R. Interplay of quantum and thermal fluctuations in a frustrated magnet. *Phys. Rev. B* 2003, *68*, 104409. [CrossRef]
- Priour, D.J.; Gelfand, M.P.; Sondhi, S.L. Disorder from disorder in a strongly frustrated transverse-field Ising chain. *Phys. Rev. B* 2001, 64, 134424. [CrossRef]
- 275. Balents, L. Spin liquids in frustrated magnets. Nature 2010, 464, 199–208. [CrossRef] [PubMed]
- 276. Röchner, J.; Balents, L.; Schmidt, K.P. Spin liquid and quantum phase transition without symmetry breaking in a frustrated three-dimensional Ising model. *Phys. Rev. B* 2016, *94*, 201111. [CrossRef]
- 277. Koziol, J.A.; Mühlhauser, M.; Schmidt, K.P. Order-by-disorder and long-range interactions in the antiferromagnetic transverse-field Ising model on the triangular lattice—A perturbative point of view. arXiv 2024, arXiv:2402.10584. https://doi.org/10.48550/arXiv.2402.10584.
- 278. Rokhsar, D.S.; Kivelson, S.A. Superconductivity and the Quantum Hard-Core Dimer Gas. *Phys. Rev. Lett.* **1988**, *61*, 2376–2379. [CrossRef] [PubMed]
- Smerald, A.; Korshunov, S.; Mila, F. Topological Aspects of Symmetry Breaking in Triangular-Lattice Ising Antiferromagnets. *Phys. Rev. Lett.* 2016, 116, 197201. [CrossRef] [PubMed]
- 280. Smerald, A.; Mila, F. Spin-liquid behaviour and the interplay between Pokrovsky-Talapov and Ising criticality in the distorted, triangular-lattice, dipolar Ising antiferromagnet. *SciPost Phys.* **2018**, *5*, 030. [CrossRef]
- 281. Hasenbusch, M. Monte Carlo study of an improved clock model in three dimensions. Phys. Rev. B 2019, 100, 224517. [CrossRef]
- Chester, S.M.; Landry, W.; Liu, J.; Poland, D.; Simmons-Duffin, D.; Su, N.; Vichi, A. Carving out OPE space and precise O(2) model critical exponents. J. High Energy Phys. 2020, 2020, 142. [CrossRef]
- Yan, Z.; Wang, Y.C.; Samajdar, R.; Sachdev, S.; Meng, Z.Y. Emergent Glassy Behavior in a Kagome Rydberg Atom Array. *Phys. Rev. Lett.* 2023, 130, 206501. [CrossRef]
- Damle, K.; Sachdev, S. Multicritical Crossovers near the Dilute Bose Gas Quantum Critical Point. *Phys. Rev. Lett.* 1996, 76, 4412–4415. [CrossRef]
- 285. Lieb, E.; Schultz, T.; Mattis, D. Two soluble models of an antiferromagnetic chain. Ann. Phys. 1961, 16, 407–466. [CrossRef]
- 286. McCoy, B.M. Spin Correlation Functions of the X Y Model. Phys. Rev. 1968, 173, 531–541. [CrossRef]
- 287. Barouch, E.; McCoy, B.M.; Dresden, M. Statistical Mechanics of the XY Model. I. Phys. Rev. A 1970, 2, 1075–1092. [CrossRef]
- 288. Barouch, E.; McCoy, B.M. Statistical Mechanics of the *XY* Model. II. Spin-Correlation Functions. *Phys. Rev. A* **1971**, *3*, 786–804. [CrossRef]
- 289. Barouch, E.; McCoy, B.M. Statistical Mechanics of the XY Model. III. Phys. Rev. A 1971, 3, 2137–2140. [CrossRef]

- McCoy, B.M.; Barouch, E.; Abraham, D.B. Statistical Mechanics of the XY Model. IV. Time-Dependent Spin-Correlation Functions. *Phys. Rev. A* 1971, 4, 2331–2341. [CrossRef]
- Jimbo, M.; Miwa, T.; Môri, Y.; Sato, M. Density matrix of an impenetrable Bose gas and the fifth Painlevé transcendent. *Phys. D Nonlinear Phenom.* 1980, 1, 80–158. [CrossRef]
- 292. Santos, R.R.d.; Stinchcombe, R.B. Finite size scaling and crossover phenomena: The XY chain in a transverse field at zero temperature. J. Phys. A Math. Gen. 1981, 14, 2741–2757. [CrossRef]
- 293. Ray, P.; Chakrabarti, B. Exact ground-state excitations of the XY model in a transverse field in one dimension. *Phys. Lett. A* **1983**, 98, 431–432. [CrossRef]
- 294. Uzunov, D. On the zero temperature critical behaviour of the nonideal Bose gas. Phys. Lett. A 1981, 87, 11–14. [CrossRef]
- 295. Coleman, S. There are no Goldstone bosons in two dimensions. Commun. Math. Phys. 1973, 31, 259–264. [CrossRef]
- 296. Pitaevskii, L.; Stringari, S. Uncertainty principle, quantum fluctuations, and broken symmetries. *J. Low Temp. Phys.* **1991**, 85, 377–388. [CrossRef]
- 297. Sandvik, A.W. Ground States of a Frustrated Quantum Spin Chain with Long-Range Interactions. *Phys. Rev. Lett.* 2010, 104, 137204. [CrossRef]
- Kumar, M.; Soos, Z.G. Decoupled phase of frustrated spin-¹/₂ antiferromagnetic chains with and without long-range order in the ground state. *Phys. Rev. B* 2013, *88*, 134412. [CrossRef]
- 299. Li, Z.H.; Wang, A.M. Matrix product state approach to a frustrated spin chain with long-range interactions. *Phys. Rev. B* 2015, *91*, 235110. [CrossRef]
- 300. Tang, Y.; Sandvik, A.W. Quantum Monte Carlo studies of spinons in one-dimensional spin systems. *Phys. Rev. B* 2015, 92, 184425. [CrossRef]
- Gong, Z.X.; Maghrebi, M.F.; Hu, A.; Foss-Feig, M.; Richerme, P.; Monroe, C.; Gorshkov, A.V. Kaleidoscope of quantum phases in a long-range interacting spin-1 chain. *Phys. Rev. B* 2016, 93, 205115. [CrossRef]
- Maghrebi, M.F.; Gong, Z.X.; Gorshkov, A.V. Continuous Symmetry Breaking in 1D Long-Range Interacting Quantum Systems. Phys. Rev. Lett. 2017, 119, 023001. [CrossRef]
- Frérot, I.; Naldesi, P.; Roscilde, T. Entanglement and fluctuations in the XXZ model with power-law interactions. *Phys. Rev. B* 2017, 95, 245111. [CrossRef]
- Ren, J.; You, W.L.; Oleś, A.M. Quantum phase transitions in a spin-1 antiferromagnetic chain with long-range interactions and modulated single-ion anisotropy. *Phys. Rev. B* 2020, 102, 024425. [CrossRef]

 Yang, L.; Feiguin, A.E. From deconfined spinons to coherent magnons in an antiferromagnetic Heisenberg chain with long range interactions. *SciPost Phys.* 2021, 10, 110. [CrossRef]

- Yang, L.; Weinberg, P.; Feiguin, A. Topological to magnetically ordered quantum phase transition in antiferromagnetic spin ladders with long-range interactions. *SciPost Phys.* 2022, 13, 60. [CrossRef]
- Nishiyama, Y. Fidelity-mediated analysis of the transverse-field XY chain with the long-range interactions: Anisotropy-driven multi-criticality. *Eur. Phys. J. B* 2021, 94, 226. [CrossRef]
- 308. Brezin, E.; Le Guillou, J.C.; Zinn-Justin, J. Approach to Scaling in Renormalized Perturbation Theory. *Phys. Rev. D* 1973, *8*, 2418–2430. [CrossRef]
- Wegner, F.J.; Riedel, E.K. Logarithmic Corrections to the Molecular-Field Behavior of Critical and Tricritical Systems. *Phys. Rev. B* 1973, 7, 248–256. [CrossRef]
- 310. Weihong, Z.; Oitmaa, J.; Hamer, C.J. Series expansions for the 3D transverse Ising model at T=0. *J. Phys. A Math. Gen.* **1994**, 27, 5425–5444. [CrossRef]
- 311. Wessel, S.; Troyer, M. Supersolid Hard-Core Bosons on the Triangular Lattice. Phys. Rev. Lett. 2005, 95, 127205. [CrossRef]
- 312. Nishiyama, Y. Transverse-field XY spin chain with the competing long-range interactions: Multi-criticality around the XX-symmetric point. *J. Stat. Mech. Theory Exp.* **2023**, 2023, 083102. [CrossRef]
- 313. Sachdev, S.; Bhatt, R.N. Bond-operator representation of quantum spins: Mean-field theory of frustrated quantum Heisenberg antiferromagnets. *Phys. Rev. B* 1990, 41, 9323–9329. [CrossRef]
- Hörmann, M.; Wunderlich, P.; Schmidt, K.P. Dynamic Structure Factor of Disordered Quantum Spin Ladders. *Phys. Rev. Lett.* 2018, 121, 167201. [CrossRef]
- 315. Schmidt, K.P.; Uhrig, G.S. Excitations in One-Dimensional $S = \frac{1}{2}$ Quantum Antiferromagnets. *Phys. Rev. Lett.* **2003**, *90*, 227204. [CrossRef]
- Bruno, P. Absence of Spontaneous Magnetic Order at Nonzero Temperature in One- and Two-Dimensional Heisenberg and XY Systems with Long-Range Interactions. *Phys. Rev. Lett.* 2001, 87, 137203. [CrossRef]
- 317. Parreira, J.R.; Bolina, O.; Perez, J.F. Néel order in the ground state of Heisenberg antiferromagnetic chains with long-range interactions. *J.-Phys.-Lond.-Math. Gen.* **1997**, *30*, 1095–1100. [CrossRef]
- 318. Nambu, Y. Quasi-Particles and Gauge Invariance in the Theory of Superconductivity. Phys. Rev. 1960, 117, 648–663. [CrossRef]
- 319. Goldstone, J. Field theories with « Superconductor » solutions. *Il Nuovo C.* **1961**, *19*, 154–164. [CrossRef]
- 320. Goldstone, J.; Salam, A.; Weinberg, S. Broken Symmetries. Phys. Rev. 1962, 127, 965–970. [CrossRef]
- 321. Diessel, O.K.; Diehl, S.; Defenu, N.; Rosch, A.; Chiocchetta, A. Generalized Higgs mechanism in long-range-interacting quantum systems. *Phys. Rev. Res.* 2023, *5*, 033038. [CrossRef]

- 322. Yusuf, E.; Joshi, A.; Yang, K. Spin waves in antiferromagnetic spin chains with long-range interactions. *Phys. Rev. B* 2004, 69, 144412. [CrossRef]
- 323. Lange, R.V. Nonrelativistic Theorem Analogous to the Goldstone Theorem. Phys. Rev. 1966, 146, 301–303. [CrossRef]
- 324. Lange, R.V. Interaction Range, The Goldstone Theorem, and Long-Range Order in the Heisenberg Ferromagnet. *Phys. Rev.* **1967**, 156, 630–631. [CrossRef]
- 325. Wegner, F.J. Critical Exponents in Isotropic Spin Systems. Phys. Rev. B 1972, 6, 1891–1893. [CrossRef]
- 326. Defenu, N.; Trombettoni, A.; Codello, A. Fixed-point structure and effective fractional dimensionality for O(*N*) models with long-range interactions. *Phys. Rev. E* 2015, *92*, 052113. [CrossRef]
- 327. Romen, A.; Birnkammer, S.; Knap, M. Deconfined quantum criticality in the long-range, anisotropic Heisenberg chain. *SciPost Phys. Core.* **2024**, *7*, 8. [CrossRef]
- Feng, L.; Katz, O.; Haack, C.; Maghrebi, M.; Gorshkov, A.; Gong, Z.; Cetina, M. Continuous symmetry breaking in a trapped-ion spin chain. *Nature* 2023, 623, 713–717. [CrossRef]
- 329. Chen, C.; Bornet, G.; Bintz, M.; Emperauger, G.; Leclerc, L.; Liu, V.; Scholl, P.; Barredo, D.; Hauschild, J.; Chatterjee, S.; et al. Continuous symmetry breaking in a two-dimensional Rydberg array. *Nature* **2023**, *616*, 691–695. [CrossRef] [PubMed]
- Haldane, F.D.M. Exact Jastrow-Gutzwiller resonating-valence-bond ground state of the spin-¹/₂ antiferromagnetic Heisenberg chain with 1/r² exchange. *Phys. Rev. Lett.* **1988**, *60*, 635–638. [CrossRef]
- 331. Shastry, B.S. Exact solution of an S=1/2 Heisenberg antiferromagnetic chain with long-ranged interactions. *Phys. Rev. Lett.* **1988**, 60, 639–642. [CrossRef] [PubMed]
- 332. Eßler, F.H.L. Dressed S matrices in models with long-range interactions. Phys. Rev. B 1995, 51, 13357–13366. [CrossRef] [PubMed]
- 333. Wang, L.; Sandvik, A.W. Critical Level Crossings and Gapless Spin Liquid in the Square-Lattice Spin-1/2 J₁ − J₂ Heisenberg Antiferromagnet. *Phys. Rev. Lett.* 2018, 121, 107202. [CrossRef] [PubMed]
- Senthil, T.; Vishwanath, A.; Balents, L.; Sachdev, S.; Fisher, M.P.A. Deconfined Quantum Critical Points. Science 2004, 303, 1490– 1494. [CrossRef] [PubMed]
- Senthil, T.; Balents, L.; Sachdev, S.; Vishwanath, A.; Fisher, M.P.A. Quantum criticality beyond the Landau-Ginzburg-Wilson paradigm. *Phys. Rev. B* 2004, 70, 144407. [CrossRef]
- 336. Vishwanath, A.; Balents, L.; Senthil, T. Quantum criticality and deconfinement in phase transitions between valence bond solids. *Phys. Rev. B* **2004**, *69*, 224416. [CrossRef]
- Senthil, T.; Balents, L.; Sachdev, S.; Vishwanath, A.; Fisher, M.P.A. Deconfined Criticality Critically Defined. J. Phys. Soc. Jpn. 2005, 74, 1–9. [CrossRef]
- Yang, S.; Yao, D.X.; Sandvik, A.W. Deconfined quantum criticality in spin-1/2 chains with long-range interactions. arXiv 2020, arXiv:2001.02821. [CrossRef]
- Adelhardt, P.; Schmidt, K.P. Raw data from the pCUT+MC approach for the antiferromagnetic Heisenberg square lattice bilayer model with (non-frustrating) long-range interactions. Zenodo 2024. [CrossRef]
- Hida, K. Low Temperature Properties of the Double Layer Quantum Heisenberg Antiferromagnet -Modified Spin Wave Method-. J. Phys. Soc. Jpn. 1990, 59, 2230–2236. [CrossRef]
- 341. Hida, K. Quantum Disordered State without Frustration in the Double Layer Heisenberg Antiferromagnet–Dimer Expansion and Projector Monte Carlo Study–. *J. Phys. Soc. Jpn.* **1992**, *61*, 1013–1018. [CrossRef]
- 342. Sandvik, A.W.; Scalapino, D.J. Order-disorder transition in a two-layer quantum antiferromagnet. *Phys. Rev. Lett.* **1994**, 72, 2777–2780. [CrossRef] [PubMed]
- 343. Sandvik, A.W.; Chubukov, A.V.; Sachdev, S. Quantum critical behavior in a two-layer antiferromagnet. *Phys. Rev. B* 1995, 51, 16483–16486. [CrossRef] [PubMed]
- 344. Wang, L.; Beach, K.S.D.; Sandvik, A.W. High-precision finite-size scaling analysis of the quantum-critical point of S = 1/2Heisenberg antiferromagnetic bilayers. *Phys. Rev. B* 2006, 73, 014431. [CrossRef]
- 345. Wenzel, S.; Bogacz, L.; Janke, W. Evidence for an Unconventional Universality Class from a Two-Dimensional Dimerized Quantum Heisenberg Model. *Phys. Rev. Lett.* **2008**, *101*, 127202. [CrossRef] [PubMed]
- 346. Lohöfer, M.; Coletta, T.; Joshi, D.G.; Assaad, F.F.; Vojta, M.; Wessel, S.; Mila, F. Dynamical structure factors and excitation modes of the bilayer Heisenberg model. *Phys. Rev. B* 2015, *92*, 245137. [CrossRef]
- 347. Coester, K. Series Expansions for Dimerized Quantum Spin Systems. Ph.D. Thesis, TU Dortmund, Dortmund, Germany, 2011.
- 348. Hasenbusch, M.; Vicari, E. Anisotropic perturbations in three-dimensional O(*N*)-symmetric vector models. *Phys. Rev. B* 2011, 84, 125136. [CrossRef]
- Chester, S.M.; Landry, W.; Liu, J.; Poland, D.; Simmons-Duffin, D.; Su, N.; Vichi, A. Bootstrapping Heisenberg magnets and their cubic instability. *Phys. Rev. D* 2021, 104, 105013. [CrossRef]
- 350. Schmidt, K.P.; Monien, H.; Uhrig, G.S. Rung-singlet phase of the $S = \frac{1}{2}$ two-leg spin-ladder with four-spin cyclic exchange. *Phys. Rev. B* 2003, *67*, 184413. [CrossRef]
- Barnes, T.; Dagotto, E.; Riera, J.; Swanson, E.S. Excitation spectrum of Heisenberg spin ladders. *Phys. Rev. B* 1993, 47, 3196–3203. [CrossRef] [PubMed]
- 352. Shelton, D.G.; Nersesyan, A.A.; Tsvelik, A.M. Antiferromagnetic spin ladders: Crossover between spin S=1/2 and S=1 chains. *Phys. Rev. B* **1996**, *53*, 8521–8532. [CrossRef] [PubMed]

- Adelhardt, P.; Schmidt, K.P. Raw data to "Continuously varying critical exponents in long-range quantum spin ladders". Zenodo 2023. [CrossRef]
- 354. Ma, N.; Sun, G.Y.; You, Y.Z.; Xu, C.; Vishwanath, A.; Sandvik, A.W.; Meng, Z.Y. Dynamical signature of fractionalization at a deconfined quantum critical point. *Phys. Rev. B* 2018, *98*, 174421. [CrossRef]
- 355. Sandvik, A.W. Evidence for Deconfined Quantum Criticality in a Two-Dimensional Heisenberg Model with Four-Spin Interactions. *Phys. Rev. Lett.* **2007**, *98*, 227202. [CrossRef] [PubMed]
- 356. Melko, R.G.; Kaul, R.K. Scaling in the Fan of an Unconventional Quantum Critical Point. *Phys. Rev. Lett.* 2008, 100, 017203. [CrossRef] [PubMed]
- 357. Jiang, F.J.; Nyfeler, M.; Chandrasekharan, S.; Wiese, U.J. From an antiferromagnet to a valence bond solid: Evidence for a first-order phase transition. *J. Stat. Mech. Theory Exp.* **2008**, 2008, P02009. [CrossRef]
- 358. Lou, J.; Sandvik, A.W.; Kawashima, N. Antiferromagnetic to valence-bond-solid transitions in two-dimensional SU(*N*) Heisenberg models with multispin interactions. *Phys. Rev. B* 2009, *80*, 180414. [CrossRef]
- 359. Sandvik, A.W. Continuous Quantum Phase Transition between an Antiferromagnet and a Valence-Bond Solid in Two Dimensions: Evidence for Logarithmic Corrections to Scaling. *Phys. Rev. Lett.* **2010**, *104*, 177201. [CrossRef] [PubMed]
- Harada, K.; Suzuki, T.; Okubo, T.; Matsuo, H.; Lou, J.; Watanabe, H.; Todo, S.; Kawashima, N. Possibility of deconfined criticality in SU(N) Heisenberg models at small N. Phys. Rev. B 2013, 88, 220408. [CrossRef]
- 361. Nakamura, M. Mechanism of CDW-SDW Transition in One Dimension. J. Phys. Soc. Jpn. 1999, 68, 3123–3126. [CrossRef]
- 362. Nakamura, M. Tricritical behavior in the extended Hubbard chains. Phys. Rev. B 2000, 61, 16377–16392. [CrossRef]
- 363. Sengupta, P.; Sandvik, A.W.; Campbell, D.K. Bond-order-wave phase and quantum phase transitions in the one-dimensional extended Hubbard model. *Phys. Rev. B* 2002, 65, 155113. [CrossRef]
- 364. Sandvik, A.W.; Balents, L.; Campbell, D.K. Ground State Phases of the Half-Filled One-Dimensional Extended Hubbard Model. Phys. Rev. Lett. 2004, 92, 236401. [CrossRef] [PubMed]
- 365. Tsuchiizu, M.; Furusaki, A. Ground-state phase diagram of the one-dimensional half-filled extended Hubbard model. *Phys. Rev.* B 2004, 69, 035103. [CrossRef]
- Jiang, S.; Motrunich, O. Ising ferromagnet to valence bond solid transition in a one-dimensional spin chain: Analogies to deconfined quantum critical points. *Phys. Rev. B* 2019, 99, 075103. [CrossRef]
- 367. Roberts, B.; Jiang, S.; Motrunich, O.I. Deconfined quantum critical point in one dimension. *Phys. Rev. B* 2019, *99*, 165143. [CrossRef]
- 368. Huang, R.Z.; Lu, D.C.; You, Y.Z.; Meng, Z.Y.; Xiang, T. Emergent symmetry and conserved current at a one-dimensional incarnation of deconfined quantum critical point. *Phys. Rev. B* 2019, *100*, 125137. [CrossRef]
- Pollmann, F.; Berg, E.; Turner, A.M.; Oshikawa, M. Symmetry protection of topological phases in one-dimensional quantum spin systems. *Phys. Rev. B* 2012, *85*, 075125. [CrossRef]
- Affleck, I. Critical Behavior of Two-Dimensional Systems with Continuous Symmetries. *Phys. Rev. Lett.* 1985, 55, 1355–1358. [CrossRef] [PubMed]
- 371. Affleck, I.; Haldane, F.D.M. Critical theory of quantum spin chains. Phys. Rev. B 1987, 36, 5291–5300. [CrossRef] [PubMed]
- Tarabunga, P.S.; Surace, F.M.; Andreoni, R.; Angelone, A.; Dalmonte, M. Gauge-Theoretic Origin of Rydberg Quantum Spin Liquids. *Phys. Rev. Lett.* 2022, 129, 195301. [CrossRef] [PubMed]
- 373. Emonts, P.; Wessel, S. Monte Carlo study of the discontinuous quantum phase transition in the transverse-field Ising model on the pyrochlore lattice. *Phys. Rev. B* 2018, *98*, 174433. [CrossRef]
- 374. Patil, P. Quantum Monte Carlo simulations in the restricted Hilbert space of Rydberg atom arrays. arXiv 2023, arXiv:2309.00482. [CrossRef]
- 375. Shankar, R. Renormalization-group approach to interacting fermions. Rev. Mod. Phys. 1994, 66, 129–192. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.